Gwénaël Massuyeau Vladimir Turaev

BRACKETS IN THE PONTRYAGIN ALGEBRAS OF MANIFOLDS

MÉMOIRES DE LA SMF 154

Société Mathématique de France 2017

Comité de rédaction

Christine BACHOC Emmanuel BREUILLARD Yann BUGEAUD Jean-François DAT Marc HERZLICH O' Grady KIERAN Raphaël KRIKORIAN Julien MARCHÉ Laurent MANIVEL Emmanuel RUSS Christophe SABOT Wilhelm SCHLAG

Pascal HUBERT (dir.)

Diffusion

Maison de la SMF B.P. 67 13274 Marseille Cedex 9 France christian.munusami@smf.emath.fr AMS P.O. Box 6248 Providence RI 02940 USA www.ams.org

Tarifs 2017

Vente au numéro : 35 € (\$52) Abonnement électronique : 113 € (\$170) Abonnement avec supplément papier : 162 €, hors Europe : 186 € (\$279) Des conditions spéciales sont accordées aux membres de la SMF.

Secrétariat : Nathalie Christiaën

Mémoires de la SMF Société Mathématique de France Institut Henri Poincaré, 11, rue Pierre et Marie Curie 75231 Paris Cedex 05, France Tél : (33) 01 44 27 67 99 • Fax : (33) 01 40 46 90 96 nathalie.christiaen@smf.emath.fr • http://smf.emath.fr/

© Société Mathématique de France 2017

Tous droits réservés (article L 122–4 du Code de la propriété intellectuelle). Toute représentation ou reproduction intégrale ou partielle faite sans le consentement de l'éditeur est illicite. Cette représentation ou reproduction par quelque procédé que ce soit constituerait une contrefaçon sanctionnée par les articles L 335–2 et suivants du CPI.

> ISSN 0249-633X (print) 2275-3230 (electronic) ISBN 978-2-85629-876-3

> > Stéphane SEURET Directeur de la publication

MÉMOIRES DE LA SMF 154

BRACKETS IN THE PONTRYAGIN ALGEBRAS OF MANIFOLDS

Gwénaël Massuyeau Vladimir Turaev

Société Mathématique de France 2017 Publié avec le concours du Centre National de la Recherche Scientifique

$Gwéna \ddot{e}l\ Massuyeau$

IRMA, Université de Strasbourg & CNRS, 67084 Strasbourg (France), IMB, Université Bourgogne Franche-Comté & CNRS, 21000 Dijon (France).
 E-mail: gwenael.massuyeau@u-bourgogne.fr

 $V ladimir \ Turaev$

Department of Mathematics, Indiana University, Bloomington IN47405 (USA). E-mail:vturaev@yahoo.com

2010 Mathematics Subject Classification. — 17B63, 55N33, 55P50, 57R19.

Key words and phrases. — Smooth manifold, intersection pairing, string topology, loop space, Pontryagin algebra, Poisson bracket, Gerstenhaber bracket, noncommutative Poisson structure, double bracket.

DOI. — 10.24033/msmf.459

BRACKETS IN THE PONTRYAGIN ALGEBRAS OF MANIFOLDS

Gwénaël Massuyeau, Vladimir Turaev

Abstract. — Given a smooth oriented manifold M with non-empty boundary, we study the Pontryagin algebra $A = H_*(\Omega)$ where Ω is the space of loops in M based at a distinguished point of ∂M . Using the ideas of string topology of Chas-Sullivan, we define a linear map

 $\{\{-,-\}\}: A \otimes A \longrightarrow A \otimes A$

which is a double bracket in the sense of Van den Bergh satisfying a version of the Jacobi identity. For dim $(M) \geq 3$, the double bracket $\{\{-, -\}\}$ induces Gerstenhaber brackets in the representation algebras associated with A. This extends our previous work on the case dim(M) = 2 where $A = H_0(\Omega)$ is the group algebra of the fundamental group $\pi_1(M)$ and the double bracket $\{\{-, -\}\}$ induces the standard Poisson brackets on the moduli spaces of representations of $\pi_1(M)$.

Résumé (Crochets dans les algèbres de Pontryagin des variétés)

Étant donnée une variété M, lisse, orientée et à bord non-vide, nous étudions l'algèbre de Pontryagin $A = H_*(\Omega)$ où Ω désigne l'espace des lacets dans M basés en un point distingué de ∂M . En utilisant les idées de la topologie des cordes de Chas et Sullivan, nous définissons une application linéaire

$$\{\{-,-\}\}: A \otimes A \longrightarrow A \otimes A$$

qui est un crochet double au sens de Van den Bergh et satisfait une version de l'identité de Jacobi. Lorsque dim $(M) \ge 3$, le crochet double $\{\{-, -\}\}$ induit des crochets de Gerstenhaber sur les algèbres de représentations associées à A. Ceci étend notre précédent travail sur le cas dim(M) = 2 où $A = H_0(\Omega)$ est l'algèbre de groupe du groupe fondamental $\pi_1(M)$ et le crochet double $\{\{-, -\}\}$ induit les crochets de Poisson habituels sur les espaces de modules de représentations de $\pi_1(M)$.

© Mémoires de la Société Mathématique de France 154, SMF 2017

CONTENTS

Introduction	7
1. Algebras, brackets, and bibrackets	13
1.1. Algebras and brackets	13
1.2. Bibrackets	17
1.3. Equivariance	22
1.4. The associated pairing and the trace	24
2. Bibrackets in unital algebras and in categories	27
2.1. Bibrackets in unital algebras	27
2.2. Bibrackets in categories	29
2.3. Bibrackets in Hopf categories	31
2.4. Hamiltonian reduction of bibrackets	39
3. Face homology	43
3.1. Manifolds with faces and partitions	43
3.2. Polychains, polycycles, and face homology	46
3.3. Face homology versus singular homology	53
3.4. Smooth polychains	63
4. Operations on polychains	60
A.1. Transversality in path spaces	60
4.2. Intersection of polychains	74
4.2 The operation $\widetilde{\Upsilon}$	78
4.4. The operation Υ	70 87
++	-7
5. The intersection bibracket	93
5.1. Construction of the intersection bibracket	93
5.2. The Jacobi identity	97
5.3. Computations and examples	110

CONTENTS

6. Properties of the intersection bibracket11
6.1. The scalar intersection form11
6.2. The reducibility
6.3. The string bracket
6.4. Moment maps and Hamiltonian reduction12
Bibliography13
Index

MÉMOIRES DE LA SMF 154

6

INTRODUCTION

A remarkable feature of an oriented surface Σ discovered by Goldman [20], [21] is a natural Lie bracket in the vector space generated by the free homotopy classes of loops in Σ . If Σ is connected and closed, then Goldman's Lie bracket arises from a symplectic structure on the moduli space of representations of the fundamental group $\pi = \pi_1(\Sigma)$ in a Lie group *G*. This space Hom $(\pi, G)/G$ consists of the conjugacy classes of homomorphisms $\pi \to G$. The resulting symplectic structure incorporates the classical Kähler forms on the Teichmüller space ($G = PSL(2, \mathbb{R})$), on the Jacobi variety (G = U(1)), and on the Narasimhan–Seshadri moduli spaces of semistable vector bundles (G = U(N) with $N \ge 1$). Goldman's construction also yields the Atiyah-Bott symplectic structure determined by a compact Lie group and a nondegenerate ad-invariant symmetric bilinear form on its Lie algebra. If Σ is connected and $\partial \Sigma \neq \emptyset$, then similar methods yield a weaker structure, namely, a Poisson bracket in the algebra of conjugation-invariant smooth functions on $Hom(\pi, G)$, see [17], [23]. This bracket extends to a quasi-Poisson bracket in the algebra of all smooth functions on Hom(π , G), see [2]. Analogous results hold for the general linear group $G = GL_N$ over any commutative ring provided $Hom(\pi, GL_N)$ is treated as an affine algebraic set and smooth functions are traded for regular functions, see [36].

Goldman's Lie bracket for surfaces was generalized by Chas and Sullivan [9], [10] to manifolds of arbitrary dimensions. Chas and Sullivan call this area of study the "string topology". The present memoir exhibits new phenomena in string topology. We consider the Pontryagin algebras of manifolds with boundary and construct a bracket in the associated representation algebras. For surfaces, our bracket is the quasi-Poisson bracket on $\text{Hom}(\pi, \text{GL}_N)$ mentioned above. In dimension ≥ 3 , the representation algebras are graded, and our bracket is a Gerstenhaber bracket, *i.e.*, it satisfies the axioms of a Poisson bracket with appropriate signs. In the rest of the Introduction we focus on manifolds of dimension ≥ 3 .

We recall the concept of a representation algebra following [39], [34], [13]. Fix an integer $N \ge 1$ and a field \mathbb{F} which will be the ground field of the algebras. Given an algebra A and a commutative algebra B, consider the set S = S(A, N, B) of all algebra homomorphisms from A to the algebra $Mat_N(B)$ of $(N \times N)$ -matrices over B.

INTRODUCTION

Each $a \in A$ and each pair of indices $i, j \in \{1, ..., N\}$ determine a mapping $a_{ij}: S \to B$ which evaluates a homomorphism $A \to \operatorname{Mat}_N(B)$ at a and takes the (i, j)-th entry of the resulting matrix. These mappings are the "coordinates" on S, generating an algebra of "polynomial" B-valued functions on S. These coordinates satisfy various polynomial relations some of which are universal, *i.e.*, hold for all B. By definition, the N-th *representation algebra* A_N of A is generated by the symbols $\{a_{ij} | a \in A, 1 \le i, j \le N\}$ subject to those universal relations. One of the universal relations says that the generators commute, so that A_N is a commutative algebra. For every commutative algebra B, the algebra A_N projects onto the algebra of polynomial B-valued functions on S(A, N, B) described above. We view A_N as a universal form of these polynomial algebras. If A is graded, then so is A_N .

Our construction of brackets in the representation algebras $\{A_N\}_{N\geq 1}$ is based on the technique of Van den Bergh [42]. He showed how to construct such brackets from a linear map

$$\{\!\{-,-\}\!\}:A\otimes A\longrightarrow A\otimes A$$

satisfying certain conditions. Van den Bergh calls such maps *double Poisson brackets*. We use the term *bibracket* for the version of double brackets used here. Also, we work in the graded setting and rather consider *Gerstenhaber bibrackets* satisfying a graded version of the Jacobi identity. We show that a Gerstenhaber bibracket $\{\{-, -\}\}$ in a graded algebra *A* induces a Gerstenhaber bracket $\{-, -\}$ in A_N for all $N \ge 1$. In terms of the generators, the bracket $\{-, -\}$ is defined as follows: for any $a, b \in A$, $i, j, u, v \in \{1, ..., N\}$, and any finite expansion

$$\{\!\{a,b\}\!\} = \sum_{\alpha} x_{\alpha} \otimes y_{\alpha} \in A \otimes A,$$

we set

$$\{a_{ij}, b_{uv}\} = \sum_{\alpha} (x_{\alpha})_{uj} (y_{\alpha})_{iv}.$$

The bracket $\{-,-\}$ is invariant under the natural actions of the group $\operatorname{GL}_N(\mathbb{F})$ and the Lie algebra $\operatorname{Mat}_N(\mathbb{F})$ on A_N .

Consider now a smooth oriented manifold *M* of dimension ≥ 3 with base point

$$\star\in\partial M\neq \varnothing.$$

Let $\Omega = \Omega_{\star}$ be the space of loops in *M* based at \star . The graded vector space

$$A = H_*(\Omega; \mathbb{F})$$

carries an associative multiplication induced by concatenation of loops. This turns *A* into a graded algebra, the *Pontryagin algebra* of *M*. We define a so-called *intersection bibracket* in *A* as follows. Pick an embedded path $\varsigma : I = [0, 1] \hookrightarrow \partial M$ connecting the point \star to another point \star' . Consider any singular cycles $\kappa : K \to \Omega = \Omega_{\star}$ and $\lambda : L \to \Omega' = \Omega_{\star'}$. Let *D* be the set of all tuples $(k \in K, s \in I, l \in L, t \in I)$ such that $\kappa(k)(s) = \lambda(l)(t)$. Each tuple $(k, s, l, t) \in D$ determines two loops in *M* based at \star . The first loop goes along ς from \star to \star' , then along the path $\lambda(l)$ from $\star' = \lambda(l)(0)$ to $\lambda(l)(t) = \kappa(k)(s)$ and then along the path $\kappa(k)$ back to $\kappa(k)(1) = \star$. The second loop goes along the path $\kappa(k)$ from $\star = \kappa(k)(0)$ to $\kappa(k)(s) = \lambda(l)(t)$, then along $\lambda(l)$

to $\lambda(l)(1) = \star'$ and finally along ς^{-1} back to \star . Under appropriate transversality assumptions on κ and λ , the resulting map $D \to \Omega \times \Omega$ is a singular cycle of dimension

$$\dim(K) + \dim(L) + 2 - \dim(M).$$

Passing to homology classes and using the isomorphism $A = H_*(\Omega; \mathbb{F}) \simeq H_*(\Omega'; \mathbb{F})$ determined by ς , we obtain the intersection bibracket in A. Our main result is the following theorem.

THEOREM. — The intersection bibracket in the Pontryagin algebra is a well-defined Gerstenhaber bibracket. It is natural with respect to diffeomorphisms of manifolds preserving the orientation and the base point.

The intersection bibracket generalizes to higher dimensions the bibracket of a surface defined in [36]. By the general theory, the intersection bibracket in the Pontryagin algebra *A* induces a Gerstenhaber bracket in A_N for all $N \ge 1$. If the manifold *M* is simply connected and \mathbb{F} is a field of characteristic zero, then the Milnor–Moore theorem identifies *A* with the universal enveloping algebra of the graded Lie algebra $\pi_*(M) = \bigoplus_{p\ge 2} \pi_p(M)$ (with the degree shifted by 1 and the Whitehead bracket in the role of the Lie bracket). In this case, the algebras A_N can be viewed as the representation algebras of $\pi_*(M)$.

Despite the simplicity of the underlying idea, a precise definition of the intersection bibracket requires considerable efforts. First of all, we introduce a version of singular homology using manifolds with corners instead of simplices. Homology theories based on manifolds with corners were implicit already in [9] and were since considered by several authors, see, for example, [7] and [11]. These theories are insufficient for our aims and we develop our own approach. For any topological space *X*, we define *polychains* in *X* as oriented manifolds with corners endowed with additional structure including an identification of some faces, a map to X compatible with this identification, and F-valued weights assigned to the connected components (these weights play the role of the coefficients of singular simplices in singular chains). We define a *reduction* of polychains which eliminates redundant connected components (like, for example, components of weight zero). Each polychain in X has a well-defined reduced boundary. If it is void, then the polychain is a polycycle. The polycycles in X considered up to disjoint unions with reduced boundaries form a graded vector space $\widetilde{H}_*(X)$, the *face homology* of *X*. The key theorem enabling our construction of bibrackets says that the usual singular homology $H_*(X) = H_*(X; \mathbb{F})$ embeds in $\widetilde{H}_*(X)$ as a direct summand.

Given a manifold *M* and a point $\star \in \partial M$ as above, we define smooth polychains in the loop space $\Omega = \Omega_{\star}$ of *M* and show that any pair of face homology classes of Ω can be represented by transversal smooth polycycles. This allows us to carry out the intersection construction outlined above and to obtain a linear map

$$\widetilde{\Upsilon}:\widetilde{H}_*(\Omega)\otimes\widetilde{H}_*(\Omega)\longrightarrow\widetilde{H}_*(\Omega\times\Omega).$$

INTRODUCTION

This map induces a linear map in singular homology

 $\Upsilon: A \otimes A \longrightarrow H_*(\Omega \times \Omega)$

where $A = H_*(\Omega)$. The Künneth theorem allows us to rewrite Υ as a map

 $\{\!\{-,-\}\!\}: A \otimes A \longrightarrow A \otimes A$

which turns out to be a Gerstenhaber bibracket. The assumption that the ground ring is a field is used only in the Künneth theorem; most of the exposition is therefore given over an arbitrary commutative ring. Moreover, our constructions can be generalized by replacing loops based at \star with paths in M having both endpoints in ∂M . This leads us to a notion of a *path homology category* of M and an extension of the intersection bibracket to this category.

Given a smooth oriented manifold W with $\partial W = \emptyset$, we can remove a small open ball from W and obtain thus a manifold with boundary. The intersection bibracket in its Pontryagin algebra and the induced Gerstenhaber brackets are invariants of W. Under further assumptions on W, we obtain an H_0 -Poisson structure [13] on the Pontryagin algebra of W itself.

This work suggests a number of questions. So far, we do not have a general method allowing to compute the face homology, and we do not know whether the face homology carries more information than the singular homology. Other questions concern the intersection bibracket. Is it sensitive to the smooth structure of the manifold? Can it be generalized to PL-manifolds or to topological manifolds? Is it homotopy invariant and can it be defined in homotopy-theoretic terms (*cf.* **[12**])? Note that the technique of face homology allows one to define all the Chas-Sullivan operations **[9]**. It would be useful to formally identify the resulting geometric operations with those in **[12**]. Also, it would be interesting to provide algebraic models for the intersection bibracket. For instance, we do not know how our geometric constructions are related to the cobar constructions of **[3]** applied to the Poincaré duality model of **[31]**, see **[3**, Section 5.5].

Organization of the memoir. — Chapters 1 and 2 are purely algebraic.

In Chapter 1 we define representation algebras and discuss brackets and bibrackets. In Chapter 2 we discuss bibrackets in unital algebras and categories, and we also consider Hamiltonian reduction in this context.

Chapter 3 introduces the face homology.

In Chapter 4 we study transversality of polychains and define intersection operations in the homology of path spaces.

In Chapters 5 and 6 we construct the intersection bibracket and discuss its properties.

Acknowledgements. — Part of this work was done while G. Massuyeau visited Bloomington, Indiana in spring 2013; he would like to thank Indiana University for hospitality and support. The work of V. Turaev on this memoir was partially supported by the NSF grants DMS-1202335 and DMS-1664358. The authors would like to thank F. Eshmatov for an explanation of the paper [3].

INTRODUCTION

Conventions. — Throughout the memoir, the letter \mathbb{K} denotes a commutative ring which serves as the ground ring of all modules and algebras. Thus, by a module (respectively, an algebra, a linear map) we mean a \mathbb{K} -module (respectively, a \mathbb{K} -algebra, a \mathbb{K} -linear map). By the singular homology of a topological space we mean singular homology with coefficients in \mathbb{K} .

Given a smooth oriented manifold M and a smooth orientable submanifold $N \subset M$, an orientation of the normal bundle of N in M determines an orientation of N, and vice versa, via the following rule: a positive frame in the normal bundle of N followed by a positive frame in the tangent bundle of N is a positive frame in the tangent bundle of M. If $\partial M \neq \emptyset$, then the orientation of M induces an orientation of ∂M using the "outward vector first" rule.

11

CHAPTER 1

ALGEBRAS, BRACKETS, AND BIBRACKETS

1.1. Algebras and brackets

We start by recalling some standard terminology.

1.1.1. *Graded modules and graded algebras.* — By a *graded module* we mean a \mathbb{Z} graded module over \mathbb{K} $A = \bigoplus A^p.$

$$\mathbf{A} = \bigoplus_{p \in \mathbb{Z}} A^p.$$

An element *a* of *A* is *homogeneous* if $a \in A^p$ for some *p*; we write then |a| = p and call |a| the *degree* of *a*. By definition, the degree of $0 \in A$ is an arbitrary integer. For any $d \in \mathbb{Z}$, the *d*-degree $|a|_d$ of a homogeneous element $a \in A$ is

$$|a|_d = |a| + d.$$

A *graded algebra* is a graded module *A* endowed with an associative bilinear multiplication such that

$$A^p A^q \subset A^{p+q}$$
 for all $p, q \in \mathbb{Z}$.

Note that if the product of $k \ge 1$ homogeneous elements a_1, \ldots, a_k of A is non-zero, then the degree of this product is equal to

$$|a_1|+\cdots+|a_k|.$$

If $a_1 \cdots a_k = 0$, then we set $|a_1 \cdots a_k| = |a_1| + \cdots + |a_k|$. Similarly, for $d \in \mathbb{Z}$, we write

$$|a_1 \cdots a_k|_d = |a_1| + \cdots + |a_k| + d_1$$

We do not require a graded algebra *A* to have a unit element.

If $ab = (-1)^{|a| \cdot |b|} ba$ for some homogeneous $a, b \in A$, then one says that a and b *commute*.

For a graded algebra *A*, we denote by [*A*, *A*] the graded submodule of *A* spanned by the vectors

$$ab - (-1)^{|a| \cdot |b|} ba$$

where *a*, *b* run over all homogeneous elements of *A*. The graded algebra *A* is *commutative* if [A, A] = 0.

Factoring any graded algebra A by the 2-sided ideal generated by [A, A] we obtain a commutative graded algebra Com(A).

Given graded algebras *A* and *B*, a graded algebra homomorphism $A \rightarrow B$ is a degree-preserving algebra homomorphism from *A* to *B*.

We will consider any $\mathbb{Z}_{\geq 0}$ -graded module

$$A = \bigoplus_{p \ge 0} A^p$$

as a \mathbb{Z} -graded module by setting $A^p = 0$ for all p < 0.

1.1.2. Representation algebras. — Each graded algebra A determines an infinite sequence of graded algebras $\tilde{A}_1, \tilde{A}_2, \ldots$ as follows, *cf.* [34], [13], [42]. The graded algebra

$$\widetilde{A}_N$$
 with $N \ge 1$

is defined by the generators a_{ij} , where *a* runs over all elements of $A, i, j \in \{1, 2, ..., N\}$, and the following relations: for all $a, b \in A, k \in \mathbb{K}$, and $i, j \in \{1, 2, ..., N\}$,

$$(1.1.1) (ka)_{ij} = ka_{ij}, (a+b)_{ij} = a_{ij} + b_{ij}, (ab)_{ij} = a_{il}b_{lj}$$

In the latter formula and in the sequel we always sum up over repeating indices and drop the summation sign. A typical element of \widetilde{A}_N is represented by a noncommutative polynomial in the generators with zero free term. The grading in \widetilde{A}_N is defined by $|a_{ij}| = p$ for all $a \in A^p$.

The construction of A_N is functorial: a graded algebra homomorphism

$$f: A \longrightarrow A'$$

induces a graded algebra homomorphism

$$\tilde{f}_N: \widetilde{A}_N \longrightarrow \widetilde{A}'_N$$
 by $\tilde{f}_N(a_{ij}) = (f(a))_{ij}$ for all $a \in A, i, j \in \{1, \dots, N\}$.

For N = 1 we have $\widetilde{A}_1 = A$ and $\widetilde{f}_1 = f$.

The importance of A_N is due to the following fact. For any graded algebra B, let $Mat_N(B)$ be the graded algebra of $(N \times N)$ -matrices with entries in B. (A matrix has a grading $p \in \mathbb{Z}$ whenever all its entries belong to B^p .) Then there is a canonical bijection

(1.1.2)
$$\operatorname{Hom}_{\mathcal{G}\mathcal{A}}(A_N, B) \xrightarrow{\simeq} \operatorname{Hom}_{\mathcal{G}\mathcal{A}}(A, \operatorname{Mat}_N(B))$$

which is natural in *A* and *B*. Here \mathcal{GA} stands for the category of graded algebras and graded algebra homomorphisms. The bijection (1.1.2) carries a graded algebra homomorphism $r : \widetilde{A}_N \to B$ to the map $A \to \operatorname{Mat}_N(B)$ sending any $a \in A$ to the $(N \times N)$ -matrix $(r(a_{ij}))_{i,j}$. The inverse bijection carries a graded algebra homomorphism $s : A \to \operatorname{Mat}_N(B)$ to the graded algebra homomorphism $\widetilde{A}_N \to B$ sending a generator a_{ij} to the (i, j)-th term of the matrix s(a) for all $a \in A$. Consequently, the endofunctor

$$\mathcal{GA} \longrightarrow \mathcal{GA}, \quad A \longmapsto \widetilde{A}_N$$

is left adjoint to the endofunctor

 $\mathcal{GA} \longrightarrow \mathcal{GA}, \quad B \longmapsto \operatorname{Mat}_N(B).$

The commutative graded algebra

$$A_N = \operatorname{Com}(\widetilde{A}_N)$$

is obtained from \widetilde{A}_N by adding the relations

$$a_{ij}b_{kl} = (-1)^{|a| \cdot |b|} b_{kl}a_{ij}$$

for any homogeneous $a, b \in A$ and any $i, j, k, l \in \{1, ..., N\}$. We call A_N the *N*-th representation algebra of A. The construction of A_N is functorial: a morphism $f : A \to A'$ in \mathcal{GA} induces a morphism $\tilde{f}_N : \tilde{A}_N \to \tilde{A}'_N$ in \mathcal{GA} , which in its turn induces a morphism $f_N : A_N \to A'_N$ in the category of commutative graded algebras \mathcal{CGA} . For any commutative graded algebra B,

$$\operatorname{Hom}_{\mathcal{CGA}}(A_N, B) \simeq \operatorname{Hom}_{\mathcal{GA}}(A_N, B) \simeq \operatorname{Hom}_{\mathcal{GA}}(A, \operatorname{Mat}_N(B)).$$

Consequently, the functor

$$\mathcal{GA} \longrightarrow \mathcal{CGA}, \quad A \longmapsto A_N$$

is left adjoint to the functor

$$CGA \longrightarrow GA, \quad B \longmapsto Mat_N(B).$$

1.1.3. *Brackets.* — Let *A* be a graded module and $d \in \mathbb{Z}$. By a *bracket* in *A* we mean a linear map

 $\{-,-\}:A\otimes A\longrightarrow A.$

A bracket $\{-, -\}$ in A:

▶ *has degree* d if $\{A^p, A^q\} \subset A^{p+q+d}$ for all $p, q \in \mathbb{Z}$;

▷ is *d*-antisymmetric if for all homogeneous $a, b \in A$,

(1.1.3)
$$\{a,b\} = -(-1)^{|a|_d \cdot |b|_d} \{b,a\};$$

▷ satisfies the *d*-graded Jacobi identity if, for all homogeneous $a, b, c \in A$,

 $(\mathbf{1.1.4}) \ \ (-1)^{|a|_d \cdot |c|_d} \left\{ a, \{b,c\} \right\} + (-1)^{|b|_d \cdot |a|_d} \left\{ b, \{c,a\} \right\} + (-1)^{|c|_d \cdot |b|_d} \left\{ c, \{a,b\} \right\} = 0.$

A degree *d* bracket $\{-, -\}$ in *A* satisfying (1.1.3) and (1.1.4) is called a *d*-graded Lie bracket, and the pair $(A, \{-, -\})$ is called then a *d*-graded Lie algebra.

For example, any graded algebra *A* gives rise to a 0-graded Lie algebra of derivations in *A*. Recall that a *derivation in A of degree* $k \in \mathbb{Z}$ is a linear map $\delta : A \to A$ such that

$$\delta(A^p) \subset A^{p+k}$$
 for any $p \in \mathbb{Z}$ and

 $\delta(ab) = \delta(a)b + (-1)^{k|a|}a\delta(b)$ for any homogeneous $a \in A$ and any $b \in A$.

Derivations of *A* of degree *k* form a module $\text{Der}^{k}(A)$. The graded module

$$\operatorname{Der}(A) = \bigoplus_{k \in \mathbb{Z}} \operatorname{Der}^k(A)$$

carries a 0-graded Lie bracket defined by

$$[\delta_1, \delta_2] = \delta_1 \delta_2 - (-1)^{k_1 k_2} \delta_2 \delta_1$$

for any derivations δ_1 and δ_2 of *A* of degrees k_1 and k_2 respectively.

A bracket $\{-, -\}$ in a graded algebra *A* satisfies the *d*-graded Leibniz rules if for all homogeneous $a, b, c \in A$,

(1.1.5)
$$\{a, bc\} = \{a, b\}c + (-1)^{|a|_d \cdot |b|} b\{a, c\},$$

(1.1.6)
$$\{ab,c\} = a\{b,c\} + (-1)^{|b| \cdot |c|_d} \{a,c\}b.$$

A *Gerstenhaber bracket of degree* $d \in \mathbb{Z}$ in a graded algebra A is a d-graded Lie bracket $\{-, -\}$ in A which satisfies the d-graded Leibniz rules. The pair $(A, \{-, -\})$ is called then a *Gerstenhaber algebra of degree* d.

For example, any graded algebra *A* is a Gerstenhaber algebra of degree 0 with respect to the bracket (called the *commutator*) defined by

$$\{a, b\} = ab - (-1)^{|a| \cdot |b|} ba$$

for homogeneous $a, b \in A$ and extended to all $a, b \in A$ by linearity.

1.2. Bibrackets

The rest of this chapter presents an extension of Van den Bergh's [42] theory of double brackets in algebras to graded algebras. Such an extension is outlined in [42, Section 2.7] in the case of degree -1.

Fix throughout this section a graded algebra *A* and an integer *d*.

1.2.1. *Conventions.* — Any $x \in A^{\otimes 2} = A \otimes A$ expands as a sum

$$x = \sum_{\alpha} x'_{\alpha} \otimes x''_{\alpha}$$

where x'_{α} , x''_{α} are homogeneous elements of *A* and the index α runs over a finite set. To simplify notation, we will drop the summation sign and the index and write simply

$$x = x' \otimes x''$$

Similarly, an element *x* of $A^{\otimes 3} = A \otimes A \otimes A$ will be written as

$$x' \otimes x'' \otimes x'''$$

with homogeneous $x', x'', x''' \in A$.

Unless explicitly stated otherwise, we endow $A^{\otimes 2}$ with the "outer" *A*-bimodule structure defined by

$$axb = ax' \otimes x''b$$

for any $a, b \in A$ and $x \in A^{\otimes 2}$. We shall also use the "inner" *A*-bimodule structure on $A^{\otimes 2}$ defined by

(1.2.1)
$$a * x * b = (-1)^{|a| \cdot |b| + |a| \cdot |x'| + |b| \cdot |x''|} x' b \otimes a x''$$

for homogeneous $a, b \in A$ and any $x \in A^{\otimes 2}$.

Given a permutation (i_1, \ldots, i_n) of $(1, \ldots, n)$ with $n \ge 1$, we denote by

$$P_{i_1\cdots i_n}$$

the graded permutation $A^{\otimes n} \to A^{\otimes n}$ carrying any $a_1 \otimes \cdots \otimes a_n$ with homogeneous $a_1, \ldots, a_n \in A$ to

$$(-1)^t a_{i_1} \otimes a_{i_2} \otimes \cdots \otimes a_{i_n}$$

where $t \in \mathbb{Z}$ is the sum of the products $|a_{i_k}| \cdot |a_{i_l}|$ over all pairs of indices k < l such that $i_k > i_l$.

For any $d \in \mathbb{Z}$, we similarly define the *d*-graded permutation

$$\mathcal{P}_{i_1\cdots i_n,d}:A^{\otimes n}\longrightarrow A^{\otimes n}$$

using the *d*-degree $|-|_d = |-| + d$ instead of |-|.

1.2.2. *Bibrackets in A.* — A *bibracket* in *A* is a linear map

$$\{\!\{-,-\}\!\}:A\otimes A\longrightarrow A\otimes A$$

A bibracket $\{\{-, -\}\}$ in *A* has *degree d* if for any integers *p*, *q*,

$$\{\!\{A^p, A^q\}\!\} \subset \bigoplus_{i+j=p+q+d} A^i \otimes A^j.$$

A *d*-graded bibracket in A is a bibracket $\{\{-, -\}\}$ in A of degree d satisfying the following *d*-graded Leibniz rules: for all homogeneous $a, b, c \in A$,

$$(1.2.2) \qquad \{\{a, bc\}\} = \{\{a, b\}\}c + (-1)^{|a|_d \cdot |b|}b\{\{a, c\}\},\$$

 $(1.2.3) \quad \{\{ab,c\}\} = a * \{\{b,c\}\} + (-1)^{|b| \cdot |c|_d} \{\{a,c\}\} * b.$

The following key lemma shows that a *d*-graded bibracket in *A* induces brackets of degree *d* in all representation algebras $\{A_N\}_N$.

LEMMA 1.2.1. — Given a d-graded bibracket $\{\{-,-\}\}$ in A and an integer $N \ge 1$, there is a unique bracket $\{-,-\}$ in A_N satisfying the d-graded Leibniz rules (1.1.5), (1.1.6) and such that

$$(1.2.4) \qquad \qquad \{a_{ij}, b_{uv}\} = \{\!\{a, b\}\!\}'_{ui}\{\!\{a, b\}\!\}'_{iv}$$

for all $a, b \in A$ and $i, j, u, v \in \{1, ..., N\}$. The bracket $\{-, -\}$ has degree d.

Proof. — We extend (1.2.4) to a bilinear form

 $\{-,-\}:A_N\times A_N\longrightarrow A_N$

satisfying (1.1.5) and (1.1.6). To see that this form is well-defined, we need to verify the compatibility with the defining relations of A_N . That the right-hand side of (1.2.4) is linear in *a* and *b* follows from the linearity of $\{\{-, -\}\}$. We now verify the compatibility with the third relation in (1.1.1). Pick any homogeneous *a*, *b*, *c* \in *A* and set *x* = $\{\{a, b\}\}$ and *y* = $\{\{a, c\}\}$. Then

$$\{\!\!\{a, bc\}\!\!\} = xc + (-1)^{|a|_d \cdot |b|} by = x' \otimes x''c + (-1)^{|a|_d \cdot |b|} by' \otimes y''.$$

Therefore, for any $i, j, u, v \in \{1, 2, ..., N\}$,

$$\{a_{ij}, (bc)_{uv}\} = \{\{a, bc\}\}'_{uj}\{\{a, bc\}\}'_{iv}$$

= $x'_{uj}(x''c)_{iv} + (-1)^{|a|_d \cdot |b|}(by')_{uj}y''_{iv}$
= $x'_{uj}x''_{il}c_{lv} + (-1)^{|a|_d \cdot |b|}b_{ul}y'_{lj}y''_{iv}$
= $\{a_{ij}, b_{ul}\}c_{lv} + (-1)^{|a|_d \cdot |b|}b_{ul}\{a_{ij}, c_{lv}\} = \{a_{ij}, b_{ul}c_{lv}\}.$

To check that $\{(ab)_{ij}, c_{uv}\} = \{a_{il}b_{lj}, c_{uv}\}$, set $z = \{\{a, c\}\}$ and $t = \{\{b, c\}\}$. Then

$$\{\!\!\{ab,c\}\!\!\} = a * t + (-1)^{|b| \cdot |c|_d} z * b = (-1)^{|t'| \cdot |a|} t' \otimes at'' + (-1)^{|b| \cdot |cz''|_d} z' b \otimes z''.$$

Therefore

$$\begin{split} \{(ab)_{ij}, c_{uv}\} &= \{\!\{ab, c\}\!\}'_{uj} \{\!\{ab, c\}\!\}''_{iv} \\ &= (-1)^{|t'| \cdot |a|} t'_{uj} (at'')_{iv} + (-1)^{|b| \cdot |cz''|_d} (z'b)_{uj} z''_{iv} \\ &= (-1)^{|t'| \cdot |a|} t'_{uj} a_{il} t''_{lv} + (-1)^{|b| \cdot |cz''|_d} z'_{ul} b_{lj} z''_{iv} \\ &= a_{il} t'_{uj} t''_{lv} + (-1)^{|b| \cdot |c|_d} z'_{ul} z''_{iv} b_{lj} \\ &= a_{il} \{b_{lj}, c_{uv}\} + (-1)^{|b| \cdot |c|_d} \{a_{il}, c_{uv}\} b_{lj} = \{a_{il} b_{lj}, c_{uv}\}. \end{split}$$

The last claim of the lemma follows from the definitions.

1.2.3. *Antisymmetric bibrackets.* — Consider the linear involutions P_{21} and $P_{21,d}$ of $A^{\otimes 2}$ determined by the permutation (21) as in Section 1.2.1: for homogeneous $a, b \in A$, we have

$$\mathbf{P}_{21}(a\otimes b) = (-1)^{|a|\cdot|b|}b\otimes a \quad \text{and} \quad \mathbf{P}_{21,d}(a\otimes b) = (-1)^{|a|_d\cdot|b|_d}b\otimes a.$$

Given $f \in \text{End}(A^{\otimes 2})$, the *d*-transpose of *f* is

$$f_d = \mathcal{P}_{21} f \mathcal{P}_{21,d} \in \operatorname{End}(A^{\otimes 2}).$$

LEMMA 1.2.2. — A bibracket $\{\{-, -\}\}$ satisfies (1.2.2) if and only if its d-transpose $\{\{-, -\}\}_d$ satisfies (1.2.3).

Proof. — Assume that a bibracket $\{\{-,-\}\}$ in *A* satifies (1.2.2). Pick homogeneous $a, b, c \in A$ and set $x = \{\{c, a\}\}, y = \{\{c, b\}\}$. Then

$$\begin{split} \{\!\{ab,c\}\!\}_d &= (-1)^{|ab|_d \cdot |c|_d} \mathcal{P}_{21}(\{\!\{c,ab\}\!\}) \\ &= (-1)^{|ab|_d \cdot |c|_d} \mathcal{P}_{21}(\{\!\{c,a\}\!\}b + (-1)^{|c|_d \cdot |a|}a\{\!\{c,b\}\!\}) \\ &= (-1)^{|ab|_d \cdot |c|_d} \mathcal{P}_{21}(x' \otimes x''b + (-1)^{|c|_d \cdot |a|}ay' \otimes y'') \\ &= (-1)^{|ab|_d \cdot |c|_d + |x'| \cdot |x''b|}x''b \otimes x' + (-1)^{|b|_d \cdot |c|_d + |ay'| \cdot |y''|}y'' \otimes ay' \\ &= (-1)^{|ab|_d \cdot |c|_d} \mathcal{P}_{21}(\{\!\{c,a\}\!\}) * b + (-1)^{|b|_d \cdot |c|_d}a * \mathcal{P}_{21}(\{\!\{c,b\}\!\}) \\ &= (-1)^{|b| \cdot |c|_d}\{\!\{a,c\}\!\}_d * b + a * \{\!\{b,c\}\!\}_d. \end{split}$$

So, $\{\{-,-\}\}_d$ satifies (1.2.3). The converse is shown by a similar computation.

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

A bibracket {{-, -}} in *A* is *d*-antisymmetric if

 $\{\!\{-,-\}\!\}_d = -\{\!\{-,-\}\!\}.$

By Lemma 1.2.2, a *d*-antisymmetric bibracket satisfies (1.2.2) if and only if it satisfies (1.2.3). Note for the record, that given a *d*-antisymmetric bibracket $\{\{-, -\}\}$ in *A*, we have for any homogeneous $a, b \in A$,

LEMMA 1.2.3. — If in Lemma 1.2.1 the bibracket $\{\{-, -\}\}$ is d-antisymmetric, then the induced bracket $\{-, -\}$ in A_N is d-antisymmetric, i.e., satisfies (1.1.3).

Proof. — Pick any homogeneous $a, b \in A$ and set $x = \{\{a, b\}\}$. Then

$$\{b_{uv}, a_{ij}\} \stackrel{(1.2.5)}{=} - (-1)^{|a|_d \cdot |b|_d + |x'| \cdot |x''|} x_{iv}'' x_{uj}' = - (-1)^{|a|_d \cdot |b|_d} x_{uj}' x_{iv}'' = - (-1)^{|a|_d \cdot |b|_d} \{a_{ij}, b_{uv}\}.$$

1.2.4. *The Jacobi identity.* — The bracket in A_N constructed in Lemma 1.2.1 may not satisfy the *d*-graded Jacobi identity (1.1.4). To compute the deviation from this identity, we observe that any bibracket $\{\{-, -\}\}$ in *A* induces a linear endomorphism $\{\{-, -, -\}\}$ of $A^{\otimes 3}$, called the *induced tribracket*, by

(1.2.6)
$$\{\{-,-,-\}\} = \sum_{i=0}^{2} \mathrm{P}_{312}^{i}(\{\{-,-\}\} \otimes \mathrm{id}_{A})(\mathrm{id}_{A} \otimes \{\{-,-\}\}) \mathrm{P}_{312,d}^{-i}$$

where P_{312} , $P_{312,d} \in End(A^{\otimes 3})$ are as defined in Section 1.2.1.

LEMMA 1.2.4. — Let $N \ge 1$. If $\{\{-, -\}\}$ is a d-antisymmetric d-graded bibracket in A, then the associated bracket $\{-, -\}$ in A_N satisfies, for any homogeneous $a, b, c \in A$ and any $p, q, r, s, u, v \in \{1, ..., N\}$,

Proof. — It follows from the definitions that

$$\begin{split} \{\!\!\{a, b, c\}\!\!\} &= \{\!\!\{a, \{\!\!\{b, c\}\!\}'\}\!\} \otimes \{\!\!\{b, c\}\!\}'' + (-1)^{|a|_d \cdot |bc|} \mathbf{P}_{312} \left(\{\!\!\{b, \{\!\!\{c, a\}\!\}'\}\!\} \otimes \{\!\!\{c, a\}\!\}''\right) \\ &+ (-1)^{|ab| \cdot |c|_d} \mathbf{P}_{312}^2 \left(\{\!\!\{c, \{\!\!\{a, b\}\!\}'\}\!\} \otimes \{\!\!\{a, b\}\!\}''\right) \\ &= \{\!\!\{a, \{\!\!\{b, c\}\!\}'\}\!\}' \otimes \{\!\!\{a, \{\!\!\{b, c\}\!\}'\}\!\}'' \otimes \{\!\!\{a, \{\!\!\{b, c\}\!\}'\}\!\}'' \otimes \{\!\!\{a, \{\!\!\{b, c\}\!\}'\}\!\}'' \otimes \{\!\!\{c, a\}\!\}'\}\!\}' \otimes \{\!\!\{c, a\}\!\}'\}'' \\ &+ (-1)^{|a|_d \cdot |bc|} \mathbf{P}_{312} \left(\{\!\!\{b, \{\!\!\{c, a\}\!\}'\}\!\}' \otimes \{\!\!\{b, \{\!\!\{c, a\}\!\}'\}\!\}' \otimes \{\!\!\{c, a\}\!\}'\}\!\}'' \otimes \{\!\!\{a, b\}\!\}'\}'' \\ &+ (-1)^{|ab| \cdot |c|_d} \mathbf{P}_{312}^2 \left(\{\!\!\{c, \{\!\!\{a, b\}\!\}'\}\!\}' \otimes \{\!\!\{c, \{\!\!\{a, b\}\!\}'\}\!\}'' \otimes \{\!\!\{a, b\}\!\}'\} '' \otimes \{\!\!\{a, b\}\!\}''\}'' \otimes \{\!\!\{a, b\}\!\}''\}''' \otimes \{\!\!\{a, b\}\!\}''\}''' \otimes \{\!\!\{a, b\}\!\}''\}''' \otimes \{\!\!\{a, b\}\!\}''\}'' \otimes \{\!\!\{a, b\}\!\}''\}''' \otimes \{\!\!\{a, b\}\!\}''''' \otimes \{\!\!\{a, b\}\!\}''''' \otimes \{\!\!\{a, b\}\!\}''\}''' \otimes \{\!\!\{a, b\}\!\}''\}''' \otimes \{\!\!\{a, b\}\!\}''''' \otimes \{\!\!\{a, b\}\!\}'''''' \otimes \{\!\!\{a, b\}\!\}''\}''' \otimes \{\!\!\{a, b\}\!\}''''''' \otimes$$

20

Using the commutativity of A_N , we deduce that

$$\begin{aligned} (1.2.7) \quad & \{\{a, b, c\}\}'_{uq}\{\{a, b, c\}\}''_{ps}\{\{a, b, c\}\}''_{rv} \\ &= \{\{a, \{\{b, c\}\}'\}\}'_{uq}\{\{a, \{\{b, c\}\}'\}\}''_{ps}\{\{b, c\}\}''_{rv} \\ &+ (-1)^{|a|_d \cdot |bc|}\{\{b, \{\{c, a\}\}'\}\}'_{ps}\{\{b, \{\{c, a\}\}'\}\}''_{rv}\{\{c, a\}\}''_{uq} \\ &+ (-1)^{|ab| \cdot |c|_d}\{\{c, \{\{a, b\}\}'\}\}''_{rv}\{\{c, \{\{a, b\}\}'\}\}''_{uq}\{\{a, b\}\}''_{ps}. \end{aligned}$$

Applying the transpositions $b \leftrightarrow c, r \leftrightarrow u$, and $s \leftrightarrow v$, we obtain

$$\begin{aligned} (1.2.8) \quad & \{\!\{a,c,b\}\!\}_{rq}'\!\{\!\{a,c,b\}\!\}_{pv}''\!\{\!\{a,c,b\}\!\}_{us}'''\\ &= \{\!\{a,\{\!\{c,b\}\!\}'\}\!\}_{rq}'\!\{\!\{a,\{\!\{c,b\}\!\}'\}\!\}_{pv}''\!\{\!\{c,b\}\!\}_{us}''\\ &+ (-1)^{|a|_d \cdot |cb|} \{\!\{c,\{\!\{b,a\}\!\}'\}\!\}_{pv}'\!\{\!\{c,\{\!\{b,a\}\!\}'\}\!\}_{us}''\!\{\!\{b,a\}\!\}'_{rq}\\ &+ (-1)^{|ac| \cdot |b|_d} \{\!\{b,\{\!\{a,c\}\!\}'\}\!\}_{us}'\!\{\!\{b,\{\!\{a,c\}\!\}'\}\!\}_{rq}''\!\{\!\{a,c\}\!\}_{pv}''\} \end{aligned}$$

Equalities (1.2.7) and (1.2.8) allow us to expand the right-hand side of the formula claimed in the lemma. We next expand the left-hand side of this formula. Set $x = \{\!\{b, c\}\!\} \in A^{\otimes 2}$ and observe that

$$\begin{aligned} \left\{ a_{pq}, \left\{ b_{rs}, c_{uv} \right\} \right\} &= \left\{ a_{pq}, x'_{us} x''_{rv} \right\} \\ &= \left\{ a_{pq}, x'_{us} \right\} x''_{rv} + (-1)^{|a|_d \cdot |x'|} x'_{us} \left\{ a_{pq}, x''_{rv} \right\} \\ &= \left\{ \left\{ a, x' \right\} \right\}'_{uq} \left\{ \left\{ a, x' \right\} \right\}''_{ps} x''_{rv} + (-1)^{|a|_d \cdot |x'|} x'_{us} \left\{ \left\{ a, x'' \right\} \right\}''_{rq} \left\{ \left\{ a, x'' \right\} \right\}''_{pv}. \end{aligned}$$

We rewrite the second summand as follows. Since $\{\{-, -\}\}$ has degree d,

$$\left|\{\{a, x''\}\}_{rq}'\{\{a, x''\}\}_{pv}'\right| = \left|\{\{a, x''\}\}''\{\{a, x''\}\}''\right| = |a| + |x''| + d = |a|_d + |x''|.$$

The commutativity of A_N implies that

$$(-1)^{|a|_{d}\cdot|x'|}x'_{us}\{\!\{a,x''\}\!\}'_{rq}\{\!\{a,x''\}\!\}'_{pv} = (-1)^{|x'|\cdot|x''|}\{\!\{a,x''\}\!\}'_{rq}\{\!\{a,x''\}\!\}''_{pv}x'_{us}.$$

The *d*-antisymmetry of $\{\{-, -\}\}$ allows us to compute $x = \{\{b, c\}\}$ from $y = \{\{c, b\}\}$: by (1.2.5), we have $x' \otimes x'' = -(-1)^{|b|_d \cdot |c|_d + |y'| \cdot |y''|} y'' \otimes y'$. Hence,

$$(-1)^{|x'| \cdot |x''|} \{\!\{a, x''\}\!\}_{rq}' \{\!\{a, x''\}\!\}_{pv}'' x_{us}' = -(-1)^{|b|_d \cdot |c|_d} \{\!\{a, y'\}\!\}_{rq}' \{\!\{a, y'\}\!\}_{pv}'' y_{us}''.$$

As a result, we obtain that

$$(1.2.9) \quad \left\{ a_{pq}, \{b_{rs}, c_{uv}\} \right\} = \left\{ \left\{ a, \left\{ \{b, c\} \right\}' \right\}'_{uq} \left\{ \left\{ a, \left\{ \{b, c\} \right\}' \right\}'_{ps} \left\{ \{b, c\} \right\}''_{rv} - (-1)^{|b|_d \cdot |c|_d} \left\{ \left\{ a, \left\{ \{c, b\} \right\}' \right\}'_{rq} \left\{ \left\{ a, \left\{ \{c, b\} \right\}' \right\}''_{pv} \left\{ \{c, b\} \right\}''_{us} \right\} \right\}''_{us} \right\} \right\}$$

Cyclically permuting *a*, *b*, *c* and the indices, we obtain

$$(1.2.10) \quad \left\{ b_{rs}, \left\{ c_{uv}, a_{pq} \right\} \right\} = \left\{ \left\{ b, \left\{ \left\{ c, a \right\} \right\}' \right\}'_{ps} \left\{ \left\{ b, \left\{ \left\{ c, a \right\} \right\}' \right\}'_{rv} \left\{ \left\{ c, a \right\} \right\}'_{uq} - (-1)^{|c|_d \cdot |a|_d} \left\{ \left\{ b, \left\{ \left\{ a, c \right\} \right\}' \right\}'_{us} \left\{ \left\{ b, \left\{ \left\{ a, c \right\} \right\}' \right\}'_{rq} \left\{ \left\{ a, c \right\} \right\}''_{pv} \right\} \right\}'_{us} \left\{ \left\{ b, \left\{ \left\{ a, c \right\} \right\}' \right\}'_{pv} \right\} \right\}'_{us} \left\{ \left\{ b, \left\{ \left\{ a, c \right\} \right\}' \right\}'_{pv} \left\{ \left\{ a, c \right\} \right\}''_{pv} \right\} \right\}'_{us} \left\{ \left\{ b, \left\{ \left\{ a, c \right\} \right\}' \right\}''_{pv} \left\{ \left\{ a, c \right\} \right\}''_{pv} \right\} \right\}''_{us} \left\{ \left\{ b, \left\{ \left\{ a, c \right\} \right\}' \right\}''_{pv} \left\{ \left\{ a, c \right\} \right\}''_{pv} \right\} \right\}''_{us} \left\{ \left\{ b, \left\{ \left\{ a, c \right\} \right\}' \right\}''_{pv} \left\{ \left\{ a, c \right\} \right\}''_{pv} \left\{ a, c \right\}''_$$

$$(1.2.11) \quad \{c_{uv}, \{a_{pq}, b_{rs}\}\} = \{\!\!\{c, \{\!\{a, b\}\!\}'\}\!\}_{rv}^{\prime}\{\!\{c, \{\!\{a, b\}\!\}'\}\!\}_{uq}^{\prime\prime}\{\!\{a, b\}\!\}_{ps}^{\prime\prime} \\ - (-1)^{|a|_d \cdot |b|_d}\{\!\{c, \{\!\{b, a\}\!\}'\}\!\}_{pv}^{\prime}\{\!\{c, \{\!\{b, a\}\!\}'\}\!\}_{us}^{\prime\prime}\{\!\{b, a\}\!\}_{rq}^{\prime\prime}.$$

The required formula directly follows from the equalities (1.2.7)–(1.2.11).

1.2.5. *Gerstenhaber bibrackets.* — A *Gerstenhaber bibracket of degree* d in A is a d-antisymmetric d-graded bibracket {{-, -}} in A such that the induced tribracket (1.2.6) is equal to zero. The pair $(A, \{\{-, -\}\})$ is called then a *double Gerstenhaber algebra of degree* d. This structure was first introduced by Van den Bergh [**42**, Section 2.7] for d = -1; see also [**3**] in the setting of differential graded algebras.

LEMMA 1.2.5. — For any Gerstenhaber bibracket of degree d in A and $N \ge 1$, the bracket $\{-, -\}$ in A_N given by Lemma 1.2.1 is a Gerstenhaber bracket of degree d.

Proof. — This follows from Lemmas 1.2.1, 1.2.3, and 1.2.4. The equality

$$\left\{a_{pq}, \{b_{rs}, c_{uv}\}\right\} + (-1)^{|a|_d \cdot |bc|} \left\{b_{rs}, \{c_{uv}, a_{pq}\}\right\} + (-1)^{|ab| \cdot |c|_d} \left\{c_{uv}, \{a_{pq}, b_{rs}\}\right\} = 0$$

provided by Lemma 1.2.4 implies the *d*-graded Jacobi identity (1.1.4) in which *a*, *b*, *c* are replaced with a_{pq} , b_{rs} , c_{uv} , respectively.

1.3. Equivariance

We show that the bracket constructed in Lemma 1.2.1 is equivariant under the natural actions of the general linear group and the Lie algebra of matrices on the representation algebra. We begin with terminology.

1.3.1. *Lie pairs.* — By a *Lie pair* we mean a pair (G, \mathfrak{g}) where *G* is a group and \mathfrak{g} is a (non-graded) Lie algebra endowed with a (left) action of *G* on \mathfrak{g} by Lie algebra automorphisms. The action is denoted by $w \mapsto {}^{g}w$ for $w \in \mathfrak{g}$ and $g \in G$.

Given a Lie pair (G, \mathfrak{g}) , by a (G, \mathfrak{g}) -*algebra* we mean a graded algebra A endowed with an action of G and an action of \mathfrak{g} such that ${}^{g}wa = gw(g^{-1}a)$ for all $g \in G$, $w \in \mathfrak{g}$, $a \in A$. Here an action of G on A is a group homomorphism from G to the group of graded algebra automorphisms of A, and an action of \mathfrak{g} on A is a Lie algebra homomorphism from \mathfrak{g} to the Lie algebra of derivations of A of degree zero, *cf*. Section 1.1.3.

1.3.2. Action on the representation algebras. — Fix an integer $N \ge 1$. Let

$$G_N = \operatorname{GL}_N(\mathbb{K})$$

be the *N*-th general linear group over \mathbb{K} and let $\mathfrak{g}_N = \operatorname{Mat}_N(\mathbb{K})$ be the Lie algebra of $(N \times N)$ -matrices with Lie bracket

$$[u,v] = uv - vu.$$

The pair (G_N, \mathfrak{g}_N) is a Lie pair where G_N acts on \mathfrak{g}_N by ${}^{g}w = gwg^{-1}$ for any $g \in G_N$ and $w \in \mathfrak{g}_N$. The representation algebra \widetilde{A}_N associated with a graded algebra Ain Section 1.1.2 is a (G_N, \mathfrak{g}_N) -algebra. Here G_N acts on \widetilde{A}_N as follows: for a matrix $g = (g_{k,l})_{k,l=1}^N \in G_N$ and a generator $a_{ij} \in \widetilde{A}_N$, set

(1.3.1)
$$ga_{ij} = (g^{-1})_{i,k} g_{l,j} a_{kl}.$$

In this formula, the numerical coefficients appear to the left of the generator a_{kl} . It is easier to remember (1.3.1) in the equivalent form $ga_{ij} = (g^{-1})_{i,k} a_{kl}g_{l,j}$, and we will use the latter form. Direct computations show that these formulas are compatible with

the relations in \widetilde{A}_N and define an action of G_N on \widetilde{A}_N . We verify the compatibility with the relation $(ab)_{ij} = a_{il}b_{lj}$:

$$g(ab)_{ij} = (g^{-1})_{i,k} (ab)_{kl} g_{l,j} = (g^{-1})_{i,k} a_{kp} b_{pl} g_{l,j}$$

= $(g^{-1})_{i,k} a_{kp} \delta_{pq} b_{ql} g_{l,j} = (g^{-1})_{i,k} a_{kp} g_{p,r} (g^{-1})_{r,q} b_{ql} g_{l,j} = (ga_{ir})(gb_{rj}).$

The Lie algebra \mathfrak{g}_N acts on \widetilde{A}_N as follows: for a matrix $w = (w_{k,l})_{k,l=1}^N \in \mathfrak{g}_N$ and a generator $a_{ij} \in \widetilde{A}_N$, set

$$(1.3.2) wa_{ij} = a_{ik}w_{k,j} - w_{i,k}a_{kj}$$

This formula is compatible with the relations in \widetilde{A}_N and defines an action of \mathfrak{g}_N on \widetilde{A}_N . We verify the compatibility with the relation $(ab)_{ij} = a_{il}b_{lj}$:

$$w(a_{il}b_{lj}) = w(a_{il})b_{lj} + a_{il}w(b_{lj})$$

= $a_{ik}w_{k,l}b_{lj} - w_{i,k}a_{kl}b_{lj} + a_{il}b_{lk}w_{k,j} - a_{il}w_{l,k}b_{kj}$
= $a_{il}b_{lk}w_{k,j} - w_{i,k}a_{kl}b_{lj} = (ab)_{ik}w_{k,j} - w_{i,k}(ab)_{kj} = w(ab)_{ij}.$

It is easy to check that these actions turn \widetilde{A}_N into a (G_N, \mathfrak{g}_N) -algebra. Moreover, these actions descend to the commutative graded algebra $A_N = \text{Com}(\widetilde{A}_N)$ and turn it into a (G_N, \mathfrak{g}_N) -algebra.

The next lemma shows that the bracket in A_N provided by Lemma 1.2.1 is equivariant under the actions of G_N and g_N .

LEMMA 1.3.1. — Let $\{\{-, -\}\}$ be a d-graded bibracket in a graded algebra A. For any $N \ge 1$, the bracket $\{-, -\}$ in A_N defined in Lemma 1.2.1 satisfies

(1.3.3)
$$g\{a,b\} = \{ga,gb\}$$
 and $w\{a,b\} = \{wa,b\} + \{a,wb\}$

for all $g \in G_N$, $w \in \mathfrak{g}_N$ and $a, b \in A_N$.

Proof. — Pick $g = (g_{k,l})_{k,l} \in G_N$. It is easy to see that if the identity $g\{x, y\} = \{gx, gy\}$ holds for all the generators of A_N , then it holds for any $x, y \in A_N$. Given $a, b \in A$ and $i, j, u, v \in \{1, ..., N\}$,

$$\{ga_{ij}, gb_{uv}\} = \{(g^{-1})_{i,k}a_{kl}g_{l,j}, (g^{-1})_{u,s}b_{st}g_{t,v}\}$$

$$= (g^{-1})_{i,k}g_{l,j}(g^{-1})_{u,s}g_{t,v}\{a_{kl}, b_{st}\}$$

$$= (g^{-1})_{i,k}g_{l,j}(g^{-1})_{u,s}g_{t,v}\{\{a,b\}\}'_{sl}\{\{a,b\}\}''_{kt}$$

$$= (g^{-1})_{u,s}\{\{a,b\}\}'_{sl}g_{l,j}(g^{-1})_{i,k}\{\{a,b\}\}''_{ut}g_{t,v}$$

$$= (g\{\{a,b\}\}'_{uj})(g\{\{a,b\}\}''_{iv}) = g\{\{a_{ij},b_{uv}\}\}$$

Similarly, given $w = (w_{k,l})_{k,l} \in g_N$, it is enough to check the identity

$$w\{x, y\} = \{wx, y\} + \{x, wy\}$$

for the generators of A_N . For $a, b \in A$ and $i, j, u, v \in \{1, ..., N\}$,

$$w\{a_{ij}, b_{uv}\} = w(\{\!\{a, b\}\!\}'_{uj}\{\!\{a, b\}\!\}''_{iv})$$

= w(\{\!\{a, b\}\!\}'_{uj})\{\!\{a, b\}\!\}''_{iv} + \{\!\{a, b\}\!\}'_{uj}w(\{\!\{a, b\}\!\}''_{iv})

$$= \{\{a, b\}\}'_{uk} w_{k,j} \{\{a, b\}\}''_{i\nu} - w_{u,k} \{\{a, b\}\}'_{kj} \{\{a, b\}\}''_{i\nu} + \{\{a, b\}\}'_{uj} \{\{a, b\}\}''_{ik} w_{k,\nu} - \{\{a, b\}\}'_{uj} w_{i,k} \{\{a, b\}\}''_{k\nu} = w_{k,j} \{a_{ik}, b_{u\nu}\} - w_{u,k} \{a_{ij}, b_{k\nu}\} + w_{k,\nu} \{a_{ij}, b_{uk}\} - w_{i,k} \{a_{kj}, b_{u\nu}\} = \{a_{ik} w_{k,j} - w_{i,k} a_{kj}, b_{u\nu}\} + \{a_{ij}, b_{uk} w_{k,\nu} - w_{u,k} b_{k\nu}\} = \{wa_{ij}, b_{u\nu}\} + \{a_{ij}, wb_{u\nu}\}.$$

1.4. The associated pairing and the trace

We study the pairing $A \otimes A \rightarrow A$ induced by a bibracket in a graded algebra A and, in particular, discuss its behavior under the trace maps.

1.4.1. *The pairing* $\langle -, - \rangle$. — A bibracket $\{\{-, -\}\}$ in a graded algebra *A* induces an *associated pairing* $\langle -, - \rangle : A \otimes A \rightarrow A$ by

 $\langle a, b \rangle = \{\!\{a, b\}\!\}' \{\!\{a, b\}\!\}'' \in A \text{ for } a, b \in A.$

LEMMA 1.4.1. — Let $\{\{-, -\}\}$ be a *d*-antisymmetric *d*-graded bibracket in *A*. Then the associated pairing $\langle -, - \rangle$ has the following properties:

- (i) $\langle -, \rangle$ has degree d and satisfies the d-graded Leibniz rule (1.1.5),
- (ii) $\langle a, b \rangle \equiv -(-1)^{|a|_d \cdot |b|_d} \langle b, a \rangle \pmod{[A, A]}$ for all homogeneous $a, b \in A$,
- (iii) $\langle [A, A], A \rangle = 0$ and $\langle A, [A, A] \rangle \subset [A, A]$,
- (iv) for any homogeneous $a, b, c \in A$,

$$\left\langle \langle a, b \rangle, c \rangle - \langle a, \langle b, c \rangle \right\rangle + (-1)^{|a|_d \cdot |b|_d} \left\langle b, \langle a, c \rangle \right\rangle$$

= $m((-1)^{|a|_d \cdot |b|_d} \{\{b, a, c\}\} - \{\{a, b, c\}\})$

where $m \in \text{Hom}(A^{\otimes 3}, A)$ carries $x \otimes y \otimes z$ to xyz for all $x, y, z \in A$.

Proof. — Claim (i) is straightforward. To check (ii), set $z = \{\{a, b\}\}$. Then

$$\{\!\!\{b,a\}\!\!\} = -(-1)^{|a|_d \cdot |b|_d + |z'| \cdot |z''|} z'' \otimes z'$$

by (1.2.5) and, modulo [*A*, *A*],

$$\langle b,a\rangle = -(-1)^{|a|_d \cdot |b|_d + |z'| \cdot |z''|} z''z' \equiv -(-1)^{|a|_d \cdot |b|_d} z'z'' = -(-1)^{|a|_d \cdot |b|_d} \langle a,b\rangle.$$

To check (iii), pick any homogeneous $a, b, c \in A$ and set $x = \{\{a, c\}\}, y = \{\{b, c\}\}$. We have

$$\begin{aligned} \{\!\{ab,c\}\!\} &= a * \{\!\{b,c\}\!\} + (-1)^{|b| \cdot |c|_d} \{\!\{a,c\}\!\} * b \\ &= (-1)^{|a| \cdot |y'|} y' \otimes a y'' + (-1)^{|b| \cdot |cx''|_d} x' b \otimes x'' \end{aligned}$$

so that

$$\langle ab, c \rangle = \{\!\!\{ab, c\}\!\}' \{\!\!\{ab, c\}\!\}'' = (-1)^{|a| \cdot |y'|} y' a y'' + (-1)^{|b| \cdot |cx''|_d} x' b x''.$$

Transposing *a* and *b*, we also obtain

$$\langle ba, c \rangle = (-1)^{|b| \cdot |x'|} x' b x'' + (-1)^{|a| \cdot |cy''|_d} y' a y''.$$

Since $\{\{-, -\}\}$ has degree *d*, we have $|cx''|_d \equiv |ax'| \pmod{2}$ and $|cy''|_d \equiv |by'| \pmod{2}$. Therefore

$$\langle ab, c \rangle = (-1)^{|a| \cdot |b|} \langle ba, c \rangle$$

Hence $\langle [A, A], A \rangle = 0$. This equality together with (ii) imply the inclusion

$$\langle A, [A, A] \rangle \subset [A, A].$$

We now prove (iv). Set $x = \{\!\{b, c\}\!\}, y = \{\!\{a, c\}\!\}, \tilde{y} = \{\!\{c, a\}\!\}, z = \{\!\{a, b\}\!\}, \tilde{z} = \{\!\{b, a\}\!\}$. Then

$$\begin{split} \{\!\!\{z'z'',c\}\!\!\} &= z' * \{\!\!\{z'',c\}\!\!\} + (-1)^{|z''| \cdot |c|_d} \{\!\!\{z',c\}\!\!\} * z'' \\ &= (-1)^{|z'| \cdot |\{\!\{z'',c\}\!\}'|} \{\!\!\{z'',c\}\!\!\}' \otimes z' \{\!\!\{z'',c\}\!\!\}'' \\ &+ (-1)^{|z''| \cdot |c\{\!\{z',c\}\!\}''|_d} \{\!\!\{z',c\}\!\!\}'z'' \otimes \{\!\!\{z',c\}\!\!\}''. \end{split}$$

We deduce that

$$\begin{aligned} \textbf{(1.4.1)} \qquad \left\langle \langle a, b \rangle, c \right\rangle &= \langle z'z'', c \rangle = (-1)^{|z'| \cdot |\{\{z'', c\}\}'|} \{\!\{z'', c\}\!\}'z''\{\!\{z'', c\}\!\}'' \\ &+ (-1)^{|z''| \cdot |c\{\{z', c\}\}''|_d} \{\!\{z', c\}\!\}'z''\{\!\{z', c\}\!\}''. \end{aligned}$$

By (i), we have

(1.4.2)
$$\langle a, \langle b, c \rangle \rangle = \langle a, x' x'' \rangle = \langle a, x' \rangle x'' + (-1)^{|a|_d \cdot |x'|} x' \langle a, x'' \rangle,$$

(1.4.3)
$$\langle b, \langle a, c \rangle \rangle = \langle b, y'y'' \rangle = \langle b, y' \rangle y'' + (-1)^{|b|_d \cdot |y'|} y' \langle b, y'' \rangle.$$

By the definition of the tribracket (1.2.6),

$$\begin{split} \{\!\{a, b, c\}\!\} &= \{\!\{a, x'\}\!\} \otimes x'' + (-1)^{|a|_d \cdot |bc|} \mathcal{P}_{312}(\{\!\{b, \tilde{y}'\}\!\} \otimes \tilde{y}'') \\ &+ (-1)^{|ab| \cdot |c|_d} \mathcal{P}_{312}^2(\{\!\{c, z'\}\!\} \otimes z'') \\ &= \{\!\{a, x'\}\!\} \otimes x'' + (-1)^{|a|_d \cdot |bc|} \mathcal{P}_{312}(\{\!\{b, \tilde{y}'\}\!\}' \otimes \{\!\{b, \tilde{y}'\}\!\}' \otimes \tilde{y}'') \\ &+ (-1)^{|ab| \cdot |c|_d} \mathcal{P}_{312}^2(\{\!\{c, z'\}\!\}' \otimes \{\!\{c, z'\}\!\}'' \otimes z'') \\ &= \{\!\{a, x'\}\!\}' \otimes \{\!\{a, x'\}\!\}'' \otimes x'' \\ &+ (-1)^{|a|_d \cdot |bc| + |b\tilde{y}'|_d \cdot |\tilde{y}''|} \tilde{y}'' \otimes \{\!\{b, \tilde{y}'\}\!\}' \otimes \{\!\{b, \tilde{y}'\}\!\}' \otimes z'' \otimes \{\!\{c, z'\}\!\}'' \\ &+ (-1)^{|ab| \cdot |c|_d + |\{\!\{c, z'\}\!\}' \cdot |\{\!\{c, z'\}\!\}'' z''|} \{\!\{c, z'\}\!\}'' \otimes z'' \otimes \{\!\{c, z'\}\!\}' \\ &\stackrel{(\mathbf{1.2.5})}{=} \{\!\{a, x'\}\!\}' \otimes \{\!\{a, x'\}\!\}'' \otimes x'' \\ &- (-1)^{|a|_d \cdot |bc| + |b|_d \cdot |y'| + |a|_d \cdot |c|_d} y' \otimes \{\!\{b, y''\}\!\}' \otimes \{\!\{b, y''\}\!\}'' \\ &- (-1)^{|ab| \cdot |c|_d + |\{\!\{z', c\}\!\}'' |\cdot |z''| + |c|_d \cdot |z'|_d} \{\!\{z', c\}\!\}' \otimes z'' \otimes \{\!\{z', c\}\!\}''. \end{split}$$

Therefore

$$(1.4.4) \qquad m\{\{a, b, c\}\} = \langle a, x' \rangle x'' - (-1)^{|a|_d \cdot |b|_d + |b|_d \cdot |y'|} y' \langle b, y'' \rangle - (-1)^{|c\{\{z', c\}\}''|_d \cdot |z''|} \{\{z', c\}\}' z'''\{\{z', c\}\}''.$$

Transposing $a \leftrightarrow b$, we obtain

$$(1.4.5) \qquad m\{\{b, a, c\}\} = \langle b, y' \rangle y'' - (-1)^{|a|_d \cdot |b|_d + |a|_d \cdot |x'|} x' \langle a, x'' \rangle - (-1)^{|c\{\{\tilde{z}', c\}\}''|_d \cdot |\tilde{z}''|} \{\{\tilde{z}', c\}\}' \tilde{z}'' \{\{\tilde{z}', c\}\}'' \stackrel{(1.2.5)}{=} \langle b, y' \rangle y'' - (-1)^{|a|_d \cdot |b|_d + |a|_d \cdot |x'|} x' \langle a, x'' \rangle + (-1)^{|\{\{z'', c\}\}'| \cdot |z'| + |a|_d \cdot |b|_d} \{\{z'', c\}\}' z' \{\{z'', c\}\}''.$$

Then (iv) follows from (1.4.1)–(1.4.5).

1.4.2. *The trace.* — For a graded algebra *A*, consider the module

$$\check{A} = A/[A, A]$$

with the grading induced by that of *A*. Lemma 1.4.1 implies that the pairing $\langle -, - \rangle : A \otimes A \rightarrow A$ associated with $\{\{-, -\}\}$ induces a pairing

$$\check{A} \otimes \check{A} \longrightarrow \check{A}$$
.

The latter pairing is also denoted by $\langle -, - \rangle$. It has degree *d* and is *d*-antisymmetric. If the induced tribracket of $\{\{-, -\}\}$ is zero, then $\langle -, - \rangle$ is a *d*-graded Lie bracket.

Note that for any $N \ge 1$, the formula $\operatorname{tr}(a) = \sum_{i=1}^{N} a_{ii}$ defines a linear map $\operatorname{tr} : A \longrightarrow A_N$.

Clearly, tr([A, A]) = 0 so that tr induces a linear map $A \to A_N$. This map is also denoted by tr and is called the *trace*. The graded subalgebra of A_N generated by $tr(\check{A}) \subset A_N$ is denoted A_N^t and is called the *N*-th trace algebra of A. We have

$$A_N^t \subset A_N^{G_N}$$

where $A_N^{G_N}$ is the subalgebra of A_N consisting of the elements invariant under the action of $G_N = \operatorname{GL}_N(\mathbb{K})$. When *A* is finitely generated as an algebra and \mathbb{K} is a field of characteristic zero, $A_N^t = A_N^{G_N}$, see [33].

LEMMA 1.4.2. — Under the conditions of Lemma 1.4.1, the map $\text{tr} : \check{A} \to A_N$ carries the pairing $\langle -, - \rangle$ in \check{A} into the bracket $\{-, -\}$ in A_N induced by $\{\{-, -\}\}$. As a consequence,

$$\{A_N^t, A_N^t\} \subset A_N^t$$
 for all $N \ge 1$.

Proof. — Pick any $a, b \in A$ and let \check{a}, \check{b} be their projections to \check{A} . We have

$$\{ \operatorname{tr}(\check{a}), \operatorname{tr}(\check{b}) \} = \left\{ \sum_{i} a_{ii}, \sum_{j} b_{jj} \right\} = \sum_{i,j} \{ a_{ii}, b_{jj} \}$$

$$\stackrel{(1.2.4)}{=} \sum_{i,j} \{ \{a, b\} \}'_{ji} \{ \{a, b\} \}'_{ij} = \sum_{j} \left(\{ \{a, b\} \}' \{ \{a, b\} \}''_{jj} \right)$$

$$= \operatorname{tr}\left(\{ \{a, b\} \}' \{ \{a, b\} \}''_{j} = \operatorname{tr}\left(\langle a, b \rangle \right) = \operatorname{tr}\left(\langle \check{a}, \check{b} \rangle \right).$$

Note that for N = 1, the trace tr : $A \rightarrow A_1 = \text{Com}(A)$ is the canonical projection and $A_1^t = A_1$.

CHAPTER 2

BIBRACKETS IN UNITAL ALGEBRAS AND IN CATEGORIES

2.1. Bibrackets in unital algebras

We define a version of representation algebras in the unital setting.

2.1.1. *Unital algebras.* — A graded algebra *A* is *unital* if it has a two-sided unit $\mathbf{1}_A \in A^0$. Unital graded algebras and graded algebra homomorphisms carrying 1 to 1 form a category \mathcal{GA}^+ . Given a unital graded algebra *A*, we define a sequence of unital graded algebras $\widetilde{A}_1^+, \widetilde{A}_2^+, \ldots$ For $N \ge 1$, \widetilde{A}_N^+ is obtained from the algebra \widetilde{A}_N defined in Section 1.1.2 as follows. First, we adjoin a unit to \widetilde{A}_N , that is consider the unital graded algebra $\mathbb{K} e \oplus \widetilde{A}_N$ with two-sided unit *e*. By definition, \widetilde{A}_N^+ is the quotient of $\mathbb{K} e \oplus \widetilde{A}_N$ by the relations $(\mathbf{1}_A)_{ij} = \delta_{ij}e$ where δ_{ij} is the Kronecker delta and *i*, *j* run over $1, \ldots, N$. For any $B \in Ob(\mathcal{GA}^+)$, the bijection (1.1.2) induces a natural bijection

(2.1.1)
$$\operatorname{Hom}_{\mathcal{G}\mathcal{A}^+}(\widehat{A}_N^+, B) \simeq \operatorname{Hom}_{\mathcal{G}\mathcal{A}^+}(A, \operatorname{Mat}_N(B)).$$

Similarly, let CGA^+ be the category of commutative unital graded algebras and graded algebra homomorphisms carrying 1 to 1. Set

$$A_N^+ = \operatorname{Com}(\widetilde{A}_N^+) \in \operatorname{Ob}(\mathfrak{CGA}^+).$$

Then for any $B \in Ob(CGA^+)$, we have a natural bijection

(2.1.2)
$$\operatorname{Hom}_{\mathcal{CG},\mathcal{A}^+}(A_N^+,B) \simeq \operatorname{Hom}_{\mathcal{G},\mathcal{A}^+}(A,\operatorname{Mat}_N(B)).$$

We call A_N^+ the *N*-th unital representation algebra of *A*. From the viewpoint of algebraic geometry, A_N^+ is the "coordinate algebra" of the "affine scheme" whose set of *B*-points is the set of algebra homomorphisms $A \to \operatorname{Mat}_N(B)$ for any $B \in \operatorname{Ob}(\mathbb{CGA}^+)$. Here an "affine scheme" is a representable functor from \mathbb{CGA}^+ to the category of sets. The same graded algebra A_N^+ can be obtained from A_N by adjoining a two-sided unit *e* and quotienting by the relations

$$(\mathbf{1}_A)_{ij} = \delta_{ij}e$$

where *i*, *j* run over $1, \ldots, N$. For N = 1, we have

 $\widetilde{A}_1^+ = A$ and $A_1^+ = \operatorname{Com}(A)$.

LEMMA 2.1.1. — Let $\{\{-,-\}\}$ be a d-graded bibracket in a unital graded algebra A and let $\{-,-\}$ be the induced bracket in A_N , see Lemma 1.2.1. Then there is a unique bracket $\{-,-\}^+$ in A_N^+ such that the projection $A_N \to A_N^+$ is bracket-preserving. If $\{\{-,-\}\}$ is a Gerstenhaber bibracket of degree d, then $\{-,-\}^+$ is a Gerstenhaber bracket of degree d in A_N^+ .

Proof. — Denote the projection $A_N \to A_N^+$ by *p*. Clearly, *p* is onto which implies the uniqueness of $\{-, -\}^+$. To prove the existence, we extend $\{-, -\}$ to a bracket $\{-, -\}'$ in the algebra

$$A_N' = \mathbb{K} e \oplus A_N \quad \text{by} \quad \{e, A_N'\}' = \{A_N', e\}' = 0.$$

The latter bracket is easily checked to satisfy the *d*-graded Leibniz rules (1.1.5), (1.1.6). Therefore, it suffices to verify that $(\mathbf{1}_A)_{ij} - \delta_{ij}e$ annhibites $\{-, -\}'$ both on the left and on the right for all *i*, *j*. The Leibniz rule (1.2.3) for $\{\{-, -\}\}$ implies that $\{\{\mathbf{1}_A, A\}\} = 0$. Therefore for any $b \in A$ and $u, v \in \{1, ..., N\}$,

$$\{(\mathbf{1}_A)_{ij} - \delta_{ij}e, b_{uv}\}' = \{(\mathbf{1}_A)_{ij}, b_{uv}\} - \delta_{ij}\{e, b_{uv}\}' = 0.$$

Since A'_N is generated by the set $\{b_{uv} | b, u, v\}$, the *d*-graded Leibniz rules (1.1.5), (1.1.6) imply that $\{(\mathbf{1}_A)_{ij} - \delta_{ij}e, A'_N\}' = 0$. Similarly, $\{A'_N, (\mathbf{1}_A)_{ij} - \delta_{ij}e\}' = 0$. The last claim of the lemma follows from Lemma 1.2.5.

The constructions and results given for A_N and A_N in Sections 1.3.2 and 1.4.2 easily extend to \tilde{A}_N^+ and A_N^+ .

2.1.2. *The case of universal enveloping algebras.* — A rich source of unital algebras is the theory of Lie algebras since their universal enveloping algebras are unital. In the graded setting one starts with a 0-graded Lie algebra $L = (L, \{-, -\})$ as in Section 1.1.3.

The *universal enveloping algebra* U(L) of L is the quotient of the graded tensor algebra $\bigoplus_{n>0} L^{\otimes n}$ by the 2-sided ideal generated by the vectors

$$a \otimes b - (-1)^{|a| \cdot |b|} b \otimes a - \{a, b\}$$

where *a*, *b* run over all homogeneous elements of *L*. The graded tensor algebra is unital and so is U(L).

For any unital graded algebra *V*, the composition with the natural linear map $L \rightarrow U(L)$ determines a bijection

(2.1.3)
$$\operatorname{Hom}_{\mathcal{GA}^+}(U(L),V) \simeq \operatorname{Hom}_{\mathcal{L}ie}(L,V)$$

where $\mathcal{L}ie$ is the category of 0-graded Lie algebras and, on the right hand-side, *V* is viewed as a graded Lie algebra with the commutator bracket. Section 2.1.1 yields for each $N \ge 1$, a commutative unital graded algebra $L_N = (U(L))_N^+$. By (2.1.2) and (2.1.3), for any $B \in Ob(\mathcal{CGA}^+)$, we have a natural bijection

(2.1.4)
$$\operatorname{Hom}_{\mathcal{C}G\mathcal{A}^+}(L_N, B) \simeq \operatorname{Hom}_{\mathcal{L}ie}(L, \operatorname{Mat}_N(B)).$$

Note that L_N is generated by the commuting symbols a_{ij} where a runs over homogeneous elements of L and i, j run over 1, ..., N, subject to the first two of the relations (1.1.1) and the relation

$$\{a, b\}_{ij} = a_{il}b_{lj} - (-1)^{|a| \cdot |b|}b_{il}a_{lj}$$

for all homogeneous $a, b \in L$ and all i, j. Lemma 2.1.1 shows how to obtain a bracket in L_N from a bibracket in U(L).

2.2. Bibrackets in categories

We define representation algebras and bibrackets for graded categories. We follow Van den Bergh [42, Section 7] who did it for non-graded categories with finite sets of objects.

2.2.1. *Graded categories and associated algebras.* — A *graded category* is a small category \mathbb{C} such that for any objects X, Y of \mathbb{C} , the set $\operatorname{Hom}_{\mathbb{C}}(X, Y)$ is a graded module, the identity morphisms of all objects are homogeneous of degree zero, and the composition of morphisms is bilinear and degree-additive. The latter condition means that for any homogeneous $f \in \operatorname{Hom}_{\mathbb{C}}(X, Y)$ and $g \in \operatorname{Hom}_{\mathbb{C}}(Y, Z)$, the morphism $g \circ f : X \to Z$ is homogeneous of degree |f| + |g|.

With a graded category C we associate a graded algebra

$$A = A(\mathcal{C}) = \bigoplus_{X,Y \in Ob(\mathcal{C})} \operatorname{Hom}_{\mathcal{C}}(X,Y)$$

where \oplus is the direct sum of graded modules. The product $fg \in A$ of $f \in \text{Hom}_{\mathbb{C}}(X, Y)$ and $g \in \text{Hom}_{\mathbb{C}}(U, Z)$ is equal to $g \circ f$ if Y = U and to zero otherwise. For $X \in \text{Ob}(\mathbb{C})$, the identity morphism of X represents an element of A denoted e_X . Clearly, $e_X e_X = e_X$ and $e_X e_Y = 0$ for $X \neq Y$. If the set $\text{Ob}(\mathbb{C})$ is finite, then $\mathbf{1}_A = \sum_{X \in \text{Ob}(\mathbb{C})} e_X$ is a two-sided unit of A; if the set $\text{Ob}(\mathbb{C})$ is infinite, then A is not unital.

For each integer $N \ge 1$, we introduce a unital graded algebra $\widehat{\mathbb{C}}_N^+$. Consider the unital graded algebra $\mathbb{K} e \oplus \widetilde{A}_N$ obtained by adjoining the two-sided unit e to the graded algebra \widetilde{A}_N associated with $A = A(\mathbb{C})$ in Section 1.1.2. Let $\widetilde{\mathbb{C}}_N^+$ be the quotient of $\mathbb{K} e \oplus \widetilde{A}_N$ by the 2-sided ideal generated by the set $\{(e_X)_{ij} - \delta_{ij}e\}_{X,i,j}$ where X runs over all objects of \mathbb{C} and $i, j \in \{1, \ldots, N\}$. The algebra $\widetilde{\mathbb{C}}_N^+$ has the following universal property. For each unital graded algebra B, we consider the algebra $\operatorname{Mat}_N(B)$ of $(N \times N)$ -matrices over B as a category with a single object. This category is graded: a matrix is homogeneous of degree p if all its entries belong to $B^p \subset B$. There is a natural bijection

$$\operatorname{Hom}_{\mathcal{GA}^+}(\mathcal{C}_N^+, B) \xrightarrow{\simeq} \operatorname{Fun}(\mathcal{C}, \operatorname{Mat}_N(B))$$

where \mathcal{GA}^+ is the category of unital graded algebras and $\operatorname{Fun}(\mathcal{C}, \operatorname{Mat}_N(B))$ is the set of degree-preserving linear functors $\mathcal{C} \to \operatorname{Mat}_N(B)$. Note that such functors can be interpreted as *N*-dimensional *B*-representations of \mathcal{C} .

29

The commutative unital graded algebra $\mathcal{C}_N^+ = \operatorname{Com}(\widetilde{\mathcal{C}}_N^+)$ plays a similar role in the category \mathcal{CGA}^+ of commutative unital graded algebras: for any $B \in Ob(\mathcal{CGA}^+)$, there is a natural bijection

(2.2.1)
$$\operatorname{Hom}_{\mathcal{C}\mathcal{G}\mathcal{A}^+}(\mathcal{C}_N^+, B) \xrightarrow{\simeq} \operatorname{Fun}\left(\mathcal{C}, \operatorname{Mat}_N(B)\right).$$

2.2.2. Double Gerstenhaber categories. — Let *d* be an integer. A *d*-graded bibracket in a graded category C is a *d*-graded bibracket $\{\{-, -\}\}$ in the graded algebra A = A(C) such that

$$\{\!\{A, e_X\}\!\} = \{\!\{e_X, A\}\!\} = 0 \text{ for all } X \in Ob(\mathcal{C}).$$

If such a bibracket in *A* is a Gerstenhaber bibracket of degree *d*, then the pair $(\mathcal{C}, \{\{-,-\}\})$ is called a *double Gerstenhaber category of degree d*.

- LEMMA 2.2.1. Let $\{\{-,-\}\}$ be a d-graded bibracket in a graded category C. Then
 - ▷ for any $X, Y, U, V \in Ob(\mathcal{C})$,

 $\{\!\!\{\operatorname{Hom}_{\mathfrak{C}}(X,Y),\operatorname{Hom}_{\mathfrak{C}}(U,V)\}\!\!\} \subset \operatorname{Hom}_{\mathfrak{C}}(U,Y) \otimes \operatorname{Hom}_{\mathfrak{C}}(X,V);$

- ▶ for any integer $N \ge 1$, the bracket in A_N determined by Lemma 1.2.1 induces a bracket $\{-,-\}$ in \mathbb{C}_N^+ satisfying the Leibniz rules (1.1.5), (1.1.6);
- ▶ *if* $(\mathbb{C}, \{\{-, -\}\})$ *is a double Gerstenhaber category of degree d, then the pair* $(\mathbb{C}_N^+, \{-, -\})$ *is a unital Gerstenhaber algebra of degree d for all* $N \ge 1$.

Proof. — Using the identity $\{\!\{A, e_X\}\!\} = \{\!\{e_X, A\}\!\} = 0$ and the Leibniz rules for $\{\!\{-, -\}\!\}$, we obtain that for any $f \in \operatorname{Hom}_{\mathbb{C}}(X, Y)$, $g \in \operatorname{Hom}_{\mathbb{C}}(U, V)$,

$$\begin{split} \{\!\{f,g\}\!\} &= \{\!\{e_X f e_Y, e_U g e_V\}\!\} \\ &= e_U \{\!\{e_X f e_Y, g\}\!\} e_V \\ &= e_U \left(\!e_X * \{\!\{f,g\}\!\} * e_Y\!\right) e_V \\ &= e_U \left(\!e_X * \left(\!\{f,g\}\!\}' \otimes \{\!\{f,g\}\!\}'') * e_Y\!\right) e_V \\ &= e_U \{\!\{f,g\}\!\}' e_Y \otimes e_X \{\!\{f,g\}\!\}'' e_V \in \operatorname{Hom}_{\mathbb{C}}(U,Y) \otimes \operatorname{Hom}_{\mathbb{C}}(X,V). \end{split}$$

П

Other claims of the lemma follow from the definitions and Lemma 1.2.5.

We conclude that a double Gerstenhaber category $(\mathcal{C}, \{\{-, -\}\})$ of degree d gives rise to a system of unital Gerstenhaber algebras $\{\mathcal{C}_N^+\}_{N\geq 1}$ of degree d. Moreover, for any full subcategory \mathcal{C}' of \mathcal{C} , the algebra $A' = A(\mathcal{C}')$ may be viewed as a subalgebra of $A = A(\mathcal{C})$ in the obvious way. The first claim of Lemma 2.2.1 implies that the bibracket $\{\{-, -\}\}$ in A restricts to a bibracket in A'. In this way, \mathcal{C}' becomes a double Gerstenhaber category of degree d. In particular, any object X of \mathcal{C} determines a full subcategory \mathcal{C}_X of \mathcal{C} consisting of X and all its endomorphisms. Then the restriction of $\{\{-, -\}\}$ to the unital graded algebra $A_X = A(\mathcal{C}_X) = \operatorname{End}_{\mathcal{C}}(X)$ is a Gerstenhaber bibracket of degree d, and we have $(\mathcal{C}_X)_N^+ = (A_X)_N^+$.

2.2.3. *Remark.* — In analogy with non-unital algebras, one can consider "categories without identity morphisms". However, such generalized categories do not appear in our geometric context and we do not study them.

2.3. Bibrackets in Hopf categories

We define Hopf categories and we introduce a class of bibrackets in Hopf categories called reducible bibrackets.

2.3.1. *Hopf categories.* — Consider a graded category \mathcal{C} and the associated graded algebra $A = A(\mathcal{C})$, see Section 2.2.1. For $X \in Ob(\mathcal{C})$, we let $e_X \in A^0 \subset A$ be the element represented by the identity morphism of *X*. We view $A \otimes A$ as an algebra with multiplication defined by

$$(a_1 \otimes a_2)(b_1 \otimes b_2) = (-1)^{|a_2| \cdot |b_1|} a_1 b_1 \otimes a_2 b_2$$

for any homogeneous $a_1, a_2, b_1, b_2 \in A$. A *comultiplication* in \mathbb{C} is a degree-preserving algebra homomorphism $\Delta : A \to A \otimes A$ such that

$$(\Delta \otimes \mathrm{id}_A)\Delta = (\mathrm{id}_A \otimes \Delta)\Delta$$
 and $\Delta(e_X) = e_X \otimes e_X$ for all $X \in \mathrm{Ob}(\mathbb{C})$.

As a consequence, Δ must carry $H = \text{Hom}_{\mathbb{C}}(X, Y) \subset A$ to $H \otimes H$ for any objects X, Y of \mathbb{C} . The image of any $a \in H$ under Δ expands (non-uniquely) as a sum

$$\sum_i a_i^{(1)} \otimes a_i^{(2)}$$

where *i* runs over a finite set and $a_i^{(1)}, a_i^{(2)}$ are homogeneous elements of *H*. We use Sweedler's notation, *i.e.*, drop the index *i* and the summation sign and write simply

$$\Delta(a) = a^{(1)} \otimes a^{(2)}.$$

The condition that Δ is degree-preserving means that $|a^{(1)}| + |a^{(2)}| = |a|$ for any homogeneous $a \in A$. That Δ is an algebra homomorphism means the identity

$$\Delta(ab) = (-1)^{|a^{(2)}| \cdot |b^{(1)}|} a^{(1)} b^{(1)} \otimes a^{(2)} b^{(2)}$$

for any homogeneous $a, b \in A$.

An *augmentation* of \mathcal{C} is a linear map $\varepsilon : A \to \mathbb{K}$ carrying the identity morphisms of all objects to 1, carrying A^p to 0 for all $p \neq 0$, and satisfying $\varepsilon(fg) = \varepsilon(f)\varepsilon(g)$ for any morphisms f, g in \mathcal{C} with target(f) = source(g). A *counit* for a comultiplication $\Delta : A \to A \otimes A$ is an augmentation

$$\varepsilon: A \longrightarrow \mathbb{K}$$

of C such that

$$(\mathrm{id}_A \otimes \varepsilon)\Delta = \mathrm{id}_A = (\varepsilon \otimes \mathrm{id}_A)\Delta \colon A \longrightarrow A$$

Clearly, if ε is a counit of Δ , then Δ is a split injection with left inverses $id_A \otimes \varepsilon$ and $\varepsilon \otimes id_A$. Also, ε induces linear maps ε_{in} , $\varepsilon_{out} : A \to A$ such that

$$\varepsilon_{\text{in}}(a) = \varepsilon(a)e_X$$
 and $\varepsilon_{\text{out}}(a) = \varepsilon(a)e_Y$,

for all $X, Y \in Ob(\mathcal{C})$ and $a \in Hom_{\mathcal{C}}(X, Y)$. An *antipode* in \mathcal{C} is a degree-preserving linear map $s : A \to A$ carrying $Hom_{\mathcal{C}}(X, Y)$ to $Hom_{\mathcal{C}}(Y, X)$ for any $X, Y \in Ob(\mathcal{C})$ and satisfying

$$a^{(1)}s(a^{(2)}) = \varepsilon_{\mathrm{in}}(a)$$
 and $a^{(2)} = \varepsilon_{\mathrm{out}}(a)$,

for all $a \in A$. It follows immediately that $s(e_X) = e_X$ for any $X \in Ob(\mathcal{C})$.

A graded category \mathcal{C} endowed with a comultiplication Δ , a counit ε , and an antipode *s* is called a *Hopf category*. When \mathcal{C} has a single object, we recover the usual notion of a *graded Hopf algebra*. A Hopf category $(\mathcal{C}, \Delta, \varepsilon, s)$ is *cocommutative* if $\Delta = P_{21}\Delta$ and is *involutive* if *s* is an involution.

Basic properties of Hopf algebras (see, for instance, [27, Theorem III.3.4]) generalize to Hopf categories. We state the properties used in the sequel.

LEMMA 2.3.1. — The antipode *s* of a Hopf category \mathbb{C} is an antiendomorphism of the underlying algebra of $A = A(\mathbb{C})$ in the sense that, for any homogeneous $a, b \in A$,

$$s(ab) = (-1)^{|a| \cdot |b|} s(b) s(a).$$

Also, s is an anti-endomorphism of the underlying coalgebra of A in the sense that, for any $a \in A$ *,*

$$\varepsilon(s(a)) = \varepsilon(a)$$
 and $(s(a))^{(1)} \otimes (s(a))^{(2)} = (-1)^{|a^{(1)}| \cdot |a^{(2)}|} s(a^{(2)}) \otimes s(a^{(1)})$

Finally, the cocommutativity of C *implies its involutivity, and the latter is equivalent to any of the following two properties:*

- (i) for all $a \in A$, $(-1)^{|a^{(1)}| \cdot |a^{(2)}|} s(a^{(2)})a^{(1)} = \varepsilon_{\text{out}}(a);$
- (ii) for all $a \in A$, $(-1)^{|a^{(1)}| \cdot |a^{(2)}|} a^{(2)} s(a^{(1)}) = \varepsilon_{in}(a)$.

Proof. — In the proof we will use the following notation. Recall that the algebra A is linearly generated by morphisms in C. Given two expressions linearly depending on one or several elements a, b, \ldots of A, we relate these expressions by the symbol

=

if they are equal for all a, b, ... and this equality follows from the axioms of a Hopf category whenever a, b, ... are morphisms in \mathcal{C} .

Let *C* be the module of degree-preserving linear maps $A \otimes A \rightarrow A$. Note that the comultiplication in *A* induces a degree-preserving coassociative comultiplication in $A \otimes A$ carrying $a \otimes b$ with $a, b \in A$ to

$$(-1)^{|b^{(1)}| \cdot |a^{(2)}|} (a^{(1)} \otimes b^{(1)}) \otimes (a^{(2)} \otimes b^{(2)}).$$

This comultiplication induces the *convolution product* * in *C* by

$$(f * g)(a \otimes b) = (-1)^{|b^{(1)}| \cdot |a^{(2)}|} f(a^{(1)} \otimes b^{(1)}) g(a^{(2)} \otimes b^{(2)})$$

for any $f, g \in C$ and any $a, b \in A$. We define elements l, r of C by

$$l(a \otimes b) = s(ab)$$
 and $r(a \otimes b) = (-1)^{|a| \cdot |b|} s(b)s(a)$

for any homogeneous $a, b \in A$. To prove the first claim of the lemma we must show that l = r. To this end we define $m, u, v \in C$ by

$$m(a \otimes b) = ab, \quad u(a \otimes b) = \varepsilon_{out}(ab), \quad v(a \otimes b) = \varepsilon_{in}(ab)$$

for any $a, b \in A$. Observe that

$$(u * r)(a \otimes b) = (-1)^{|b^{(1)}| \cdot |a^{(2)}| + |b^{(2)}| \cdot |a^{(2)}|} \varepsilon_{\text{out}}(a^{(1)}b^{(1)})(s(b^{(2)})s(a^{(2)}))$$

$$\begin{split} &\stackrel{\cdots}{=} (-1)^{|b| \cdot |a^{(2)}|} \varepsilon(a^{(1)}b^{(1)}) s(b^{(2)}) s(a^{(2)}) \\ &\stackrel{\cdots}{=} (-1)^{|b| \cdot |a^{(2)}|} \varepsilon(a^{(1)}) \varepsilon(b^{(1)}) s(b^{(2)}) s(a^{(2)}) \\ &= (-1)^{|b| \cdot |a|} s(\varepsilon(b^{(1)})b^{(2)}) s(\varepsilon(a^{(1)})a^{(2)}) = r(a \otimes b) \end{split}$$

and

$$\begin{aligned} (l*m)(a\otimes b) &= (-1)^{|b^{(1)}|\cdot|a^{(2)}|} s(a^{(1)}b^{(1)})(a^{(2)}b^{(2)}) \\ &= s((ab)^{(1)})((ab)^{(2)}) = \varepsilon_{\text{out}}(ab) = u(a\otimes b). \end{aligned}$$

Furthermore,

$$\begin{split} (m*r)(a\otimes b) &= (-1)^{|b^{(1)}|\cdot|a^{(2)}|+|b^{(2)}|\cdot|a^{(2)}|} (a^{(1)}b^{(1)}) (s(b^{(2)})s(a^{(2)})) \\ &= (-1)^{|b|\cdot|a^{(2)}|}a^{(1)} (b^{(1)}s(b^{(2)}))s(a^{(2)}) \\ &= (-1)^{|b|\cdot|a^{(2)}|}a^{(1)}\varepsilon_{\mathrm{in}}(b)s(a^{(2)}) \\ &= a^{(1)}\varepsilon_{\mathrm{in}}(b)s(a^{(2)}) \\ &= a^{(1)}\varepsilon_{\mathrm{in}}(b)s(a^{(2)}) \\ &= a^{(1)}\varepsilon_{\mathrm{in}}(b)s(a^{(2)})\varepsilon(a^{(3)}) \\ &\stackrel{\cdots}{=} a^{(1)}s(a^{(2)})\varepsilon(a^{(3)}b) \\ &= \varepsilon_{\mathrm{in}}(a^{(1)})\varepsilon(a^{(2)}b) \stackrel{\cdots}{=} \varepsilon_{\mathrm{in}}(ab) = v(a\otimes b) \end{split}$$

and

$$\begin{aligned} (l*v)(a\otimes b) &= (-1)^{|b^{(1)}|\cdot|a^{(2)}|} s(a^{(1)}b^{(1)})\varepsilon_{\mathrm{in}}(a^{(2)}b^{(2)}) \\ &\stackrel{\cdots}{=} (-1)^{|b^{(1)}|\cdot|a^{(2)}|} s(a^{(1)}b^{(1)})\varepsilon(a^{(2)}b^{(2)}) \\ &\stackrel{\cdots}{=} (-1)^{|b^{(1)}|\cdot|a^{(2)}|} s(a^{(1)}b^{(1)})\varepsilon(a^{(2)})\varepsilon(b^{(2)}) \\ &= s(a^{(1)}\varepsilon(a^{(2)})b^{(1)}\varepsilon(b^{(2)})) = l(a\otimes b). \end{aligned}$$

Since * is an associative operation, we deduce that

l = l * v = l * m * r = u * r = r.

We now verify that *s* is an antiendomorphism of the unital coalgebra (A, Δ, ε) . For this, we consider the module *D* of degree-preserving linear maps $A \to A \otimes A$, and we equip it with the *convolution product* defined by

$$(f * g)(a) = f(a^{(1)})g(a^{(2)})$$

for any $f, g \in D$ and any $a \in A$. Let $l, r, u, v \in D$ be defined by

$$l = \Delta s$$
, $r = (s \otimes s) P_{21} \Delta$, $u = \Delta \varepsilon_{out}$, $v = \Delta \varepsilon_{in}$.

We must prove that l = r. Observe that, for any $a \in A$,

$$(u * r)(a) = \Delta \varepsilon_{\text{out}}(a^{(1)}) ((s \otimes s) P_{21} \Delta(a^{(2)}))$$

$$\stackrel{\cdots}{=} (-1)^{|a^{(3)}| \cdot |a^{(4)}|} (\varepsilon_{\text{out}}(a^{(1)}) \otimes \varepsilon_{\text{out}}(a^{(2)})) (s(a^{(4)}) \otimes s(a^{(3)}))$$

$$= (-1)^{|a^{(3)}| \cdot |a^{(4)}|} \varepsilon_{\text{out}}(a^{(1)}) s(a^{(4)}) \otimes \varepsilon_{\text{out}}(a^{(2)}) s(a^{(3)})$$

$$\stackrel{\cdots}{=} (-1)^{|a^{(2)}a^{(3)}| \cdot |a^{(4)}|} \varepsilon(a^{(1)}) s(a^{(4)}) \otimes \varepsilon(a^{(2)}) s(a^{(3)})$$

$$= (-1)^{|a^{(2)}| \cdot |a^{(3)}|} \varepsilon(a^{(1)}) s(a^{(3)}) \otimes s(a^{(2)})$$

= $(-1)^{|a^{(1)}a^{(2)}| \cdot |a^{(3)}|} s(a^{(3)}) \otimes s(\varepsilon(a^{(1)})a^{(2)})$
= $(-1)^{|a^{(1)}| \cdot |a^{(2)}|} s(a^{(2)}) \otimes s(a^{(1)}) = r(a)$

and

$$(l * \Delta)(a) = \Delta(s(a^{(1)}))\Delta(a^{(2)}) = \Delta(s(a^{(1)})a^{(2)}) = u(a).$$

Furthermore,

$$\begin{aligned} (\Delta * r)(a) &= \Delta(a^{(1)}) \left((s \otimes s) \mathcal{P}_{21} \Delta(a^{(2)}) \right) \\ &= (-1)^{|a^{(3)}| \cdot |a^{(4)}|} (a^{(1)} \otimes a^{(2)}) \left(s(a^{(4)}) \otimes s(a^{(3)}) \right) \\ &= (-1)^{|a^{(2)}a^{(3)}| \cdot |a^{(4)}|} a^{(1)} s(a^{(4)}) \otimes a^{(2)} s(a^{(3)}) \\ &= (-1)^{|a^{(2)}| \cdot |a^{(3)}|} a^{(1)} s(a^{(3)}) \otimes \varepsilon_{\mathrm{in}}(a^{(2)}) \\ &\stackrel{\cdots}{=} a^{(1)} s(a^{(3)}) \otimes \varepsilon_{\mathrm{in}}(a^{(2)}) \stackrel{\cdots}{=} v(a) \end{aligned}$$

and

$$(l * v)(a) = \Delta s(a^{(1)}) \Delta \varepsilon_{in}(a^{(2)})$$

= $\Delta (s(a^{(1)})\varepsilon_{in}(a^{(2)}))$
 $\stackrel{\cdots}{=} \Delta (s(a^{(1)})\varepsilon(a^{(2)})) = \Delta (s(a^{(1)}\varepsilon(a^{(2)}))) = l(a).$

Using the associativity of *, we deduce that

$$l = l * v = l * \Delta * r = u * r = r.$$

Also, *s* preverves the counit: for any $a \in A$, we have

$$\varepsilon(a) \stackrel{\dots}{=} \varepsilon(\varepsilon_{\mathrm{in}}(a)) = \varepsilon(a^{(1)}s(a^{(2)}))$$
$$\stackrel{\dots}{=} \varepsilon(a^{(1)})\varepsilon(s(a^{(2)})) = \varepsilon(s(\varepsilon(a^{(1)})a^{(2)})) = \varepsilon(s(a)).$$

We now prove the part of the lemma concerning the involutivity. If $s^2 = id_A$, then the condition (i) is satisfied:

$$(-1)^{|a^{(1)}| \cdot |a^{(2)}|} s(a^{(2)}) a^{(1)} = s(s(a^{(1)})a^{(2)}) = s(\varepsilon_{\text{out}}(a)) = \varepsilon_{\text{out}}(a).$$

Assume now that condition (i) is met and consider the convolution product * in the module, *E*, of degree-preserving linear maps $A \rightarrow A$. For any $a \in A$,

$$(s * s^{2})(a) = s(a^{(1)})s(s(a^{(2)})) = (-1)^{|a^{(1)}| \cdot |a^{(2)}|}s(s(a^{(2)})a^{(1)})$$

= $s(\varepsilon_{\text{out}}(a)) = \varepsilon_{\text{out}}(a).$

Thus, $s * s^2 = \varepsilon_{out}$. It follows from the axioms of a Hopf category that

$$\operatorname{id}_A * s = \varepsilon_{\operatorname{in}}$$
 and $\varepsilon_{\operatorname{in}} * f = f = f * \varepsilon_{\operatorname{out}}$

for each $f \in E$ carrying the set $\text{Hom}_{\mathfrak{C}}(X, Y)$ into itself for all $X, Y \in \text{Ob}(\mathfrak{C})$. Applying this to $f = s^2$ and to $f = \text{id}_A$ and using the associativity of *, we obtain

$$s^2 = \varepsilon_{\mathrm{in}} * s^2 = \mathrm{id}_A * s * s^2 = \mathrm{id}_A * \varepsilon_{\mathrm{out}} = \mathrm{id}_A$$
 .
This shows the equivalence between the involutivity and (i); the equivalence with (ii) is proved similarly. Finally, if \mathcal{C} is cocommutative, then the identity $s * id_A = \varepsilon_{out}$ implies (i), so that *A* is involutive.

2.3.2. *Bibrackets re-examined.* — Bibrackets in a Hopf category $(\mathcal{C}, \Delta, \varepsilon, s)$ have a useful reformulation which we now describe. Consider the associated graded algebra $A = A(\mathcal{C})$ and a *d*-graded bibracket

$$\{\!\{-,-\}\!\}: A \otimes A \longrightarrow A \otimes A$$

with $d \in \mathbb{Z}$. We define a linear map $\Lambda = \Lambda(\{\{-, -\}\}) : A \otimes A \to A \otimes A$ by

(2.3.1)
$$\Lambda(a,b) = a^{(1)}s(\{\{a^{(2)}, b^{(1)}\}\}') \otimes \{\{a^{(2)}, b^{(1)}\}\}''s(b^{(2)})$$

for any $a, b \in A$. Note that, if $a \in \text{Hom}_{\mathbb{C}}(X, Y)$ and $b \in \text{Hom}_{\mathbb{C}}(U, V)$ with X, Y, U, Vin $Ob(\mathbb{C})$, then $\Lambda(a, b) \in \text{Hom}_{\mathbb{C}}(X, U) \otimes \text{Hom}_{\mathbb{C}}(X, U)$.

LEMMA 2.3.2. — For any $a \in A$ and any homogeneous $b, c \in A$, we have

$$\begin{split} \Lambda(a,bc) &= \Lambda(a,b^{(1)}) \varepsilon(b^{(2)}c) + (-1)^{|b| \cdot |c|} \Lambda(a,c) (s \otimes s) (\Delta(b)), \\ \Lambda(ab,c) &= \Lambda(a^{(1)},c) \varepsilon(a^{(2)}b) + \Delta(a) \Lambda(b,c). \end{split}$$

Proof. — Since both sides of the first identity are linear in *b* and *c*, it suffices to consider the case where $b \in \operatorname{Hom}_{\mathbb{C}}(U, V)$ and $c \in \operatorname{Hom}_{\mathbb{C}}(W, Z)$ for some objects U, V, W, Z of \mathbb{C} . If $V \neq W$, then both sides of the identity are equal to zero. If V = W, then $\Lambda(a, b^{(1)}) \varepsilon(b^{(2)}c) = \Lambda(a, b)\varepsilon(c)$ and

$$\begin{split} \Lambda(a,bc) &= a^{(1)}s\big(\{\!\{a^{(2)},(bc)^{(1)}\}\!\}'\big) \otimes \{\!\{a^{(2)},(bc)^{(1)}\}\!\}''s((bc)^{(2)}) \\ &= (-1)^{|b^{(2)}|\cdot|c|}a^{(1)}s\big(\{\!\{a^{(2)},b^{(1)}c^{(1)}\}\!\}'\big) \otimes \{\!\{a^{(2)},b^{(1)}c^{(1)}\}\!\}''s(c^{(2)})s(b^{(2)}) \\ &= (-1)^{|b^{(2)}|\cdot|c|}a^{(1)}s\big(\{\!\{a^{(2)},b^{(1)}\}\!\}'\big) \otimes \{\!\{a^{(2)},b^{(1)}\}\!\}''c^{(1)}s(c^{(2)})s(b^{(2)}) \\ &\quad + \eta_1a^{(1)}s\big(\{\!\{a^{(2)},c^{(1)}\}\!\}'\big)s(b^{(1)}\big) \otimes \{\!\{a^{(2)},c^{(1)}\}\!\}''s(c^{(2)})s(b^{(2)}) \\ &= \varepsilon(c)\Lambda(a,b) + \eta_2\Lambda(a,c)\big(s(b^{(1)})\otimes s(b^{(2)})\big) \end{split}$$

where the signs $\eta_1, \eta_2 = \pm 1$ are computed by

$$\begin{aligned} \eta_1 &= (-1)^{|b^{(2)}| \cdot |c| + |b^{(1)}| \cdot |\{\{a^{(2)}, c^{(1)}\}\}'| + |b^{(1)}| \cdot |a^{(2)}|_d}, \\ \eta_2 &= \eta_1 \cdot (-1)^{|b^{(1)}| \cdot |\{\{a^{(2)}, c^{(1)}\}\}'' c^{(2)}|} = (-1)^{|b| \cdot |c|}. \end{aligned}$$

The second identity is proved similarly with the key case being the one where *a*, *b* are morphisms in C and the target object of *a* coincides with the source object of *b*. Then $\Lambda(a^{(1)}, c)\varepsilon(a^{(2)}b) = \Lambda(a, c)\varepsilon(b)$ and

$$\begin{split} \Lambda(ab,c) &= (ab)^{(1)} s\bigl(\{\!\{(ab)^{(2)},c^{(1)}\}\!\}'\bigr) \otimes \{\!\{(ab)^{(2)},c^{(1)}\}\!\}'' s(c^{(2)}) \\ &= (-1)^{|a^{(2)}|\cdot|b^{(1)}|} a^{(1)} b^{(1)} s\bigl(\{\!\{a^{(2)}b^{(2)},c^{(1)}\}\!\}'\bigr) \otimes \{\!\{a^{(2)}b^{(2)},c^{(1)}\}\!\}'' s(c^{(2)}) \\ &= \theta_1 a^{(1)} b^{(1)} s\bigl(\{\!\{b^{(2)},c^{(1)}\}\!\}'\bigr) \otimes a^{(2)} \{\!\{b^{(2)},c^{(1)}\}\!\}'' s(c^{(2)}) \\ &\quad + \theta_2 a^{(1)} b^{(1)} s\bigl(\{\!\{a^{(2)},c^{(1)}\}\!\}'b^{(2)}\bigr) \otimes \{\!\{a^{(2)},c^{(1)}\}\!\}'' s(c^{(2)}) \end{split}$$

$$= (a^{(1)} \otimes a^{(2)}) \Lambda(b, c) + \theta_3 a^{(1)} b^{(1)} s(b^{(2)}) s(\{\!\{a^{(2)}, c^{(1)}\}\!\}') \otimes \{\!\{a^{(2)}, c^{(1)}\}\!\}'' s(c^{(2)}) = \Delta(a) \Lambda(b, c) + (-1)^{|a^{(2)}| \cdot |b|} \varepsilon(b) \Lambda(a, c) = \Delta(a) \Lambda(b, c) + \varepsilon(b) \Lambda(a, c)$$

where the signs $\theta_1, \theta_2, \theta_3 = \pm 1$ are computed by

$$\begin{aligned} \theta_1 &= (-1)^{|a^{(2)}| \cdot |b^{(1)}| + |a^{(2)}| \cdot |\{b^{(2)}, c^{(1)}\}'|}, \\ \theta_2 &= (-1)^{|a^{(2)}| \cdot |b^{(1)}| + |b^{(2)}| \cdot |c^{(1)}|_d + |b^{(2)}| \cdot |\{a^{(2)}, c^{(1)}\}''|}, \\ \theta_3 &= \theta_2 \cdot (-1)^{|b^{(2)}| \cdot |\{\{a^{(2)}, c^{(1)}\}''|} = (-1)^{|a^{(2)}| \cdot |b|}. \end{aligned}$$

The bibracket $\{\{-, -\}\}\$ may be recovered from the map Λ at least in the case where the antipode *s* in C is invertible. Indeed, for any $a, b \in A$,

$$\begin{aligned} (2.3.2) \qquad s(a^{(1)})\Lambda(a^{(2)},b^{(1)})b^{(2)} \\ &= s(a^{(1)})a^{(2)}s(\{\!\{a^{(3)},b^{(1)}\}\!\}')\otimes\{\!\{a^{(3)},b^{(1)}\}\!\}''s(b^{(2)})b^{(3)} \\ &= \varepsilon_{\text{out}}(a^{(1)})s(\{\!\{a^{(2)},b^{(1)}\}\!\}')\otimes\{\!\{a^{(2)},b^{(1)}\}\!\}''\varepsilon_{\text{out}}(b^{(2)}) \\ &= \varepsilon(a^{(1)})s(\{\!\{a^{(2)},b^{(1)}\}\!\}')\otimes\{\!\{a^{(2)},b^{(1)}\}\!\}''\varepsilon(b^{(2)}) \\ &= (s\otimes \text{id}_A)(\{\!\{a,b\}\!\}). \end{aligned}$$

If follows that, if the antipode *s* is invertible, then

$$\begin{aligned} \{\!\{a,b\}\!\} &= (s^{-1} \otimes \mathrm{id}_A) \big(s(a^{(1)}) \Lambda(a^{(2)}, b^{(1)}) b^{(2)} \big) \\ &= (-1)^{|a^{(1)}| \cdot |a^{(2)}b^{(1)}|_d} (s^{-1} \otimes \mathrm{id}_A) \big(\Lambda(a^{(2)}, b^{(1)}) \big) \big(a^{(1)} \otimes b^{(2)} \big). \end{aligned}$$

2.3.3. *Reducible bibrackets.* — Let $\{\{-, -\}\}$ be a bibracket in a Hopf category $\mathcal{C} = (\mathcal{C}, \Delta, \varepsilon, s)$.

It induces, in the notation of the previous subsection, a bilinear pairing

 $\lambda = \lambda(\{\{-,-\}\}) : A \times A \longrightarrow A \quad \text{by} \quad \lambda = (\varepsilon \otimes \text{id}_A)\Lambda.$

Explicitly, for any $a, b \in A$ we have

$$(2.3.3) \qquad \lambda(a,b) = \varepsilon(\{\{a,b^{(1)}\}\}')\{\{a,b^{(1)}\}\}''s(b^{(2)})$$

It follows from Lemma 2.3.2 that, for any $a \in A$ and any homogeneous $b, c \in A$,

$$\lambda(a,bc) = \lambda(a,b^{(1)})\varepsilon(b^{(2)}c) + (-1)^{|b| \cdot |c|}\lambda(a,c)s(b),$$

$$\lambda(ab,c) = \lambda(a^{(1)},c)\varepsilon(a^{(2)}b) + a\lambda(b,c).$$

We call a bibracket $\{\!\{-,-\}\!\}$ in \mathcal{C} *reducible* if $\Lambda(A \otimes A) \subset \Delta(A)$. Then

$$\lambda = (\varepsilon \otimes \mathrm{id}_A)\Lambda = (\mathrm{id}_A \otimes \varepsilon)\Lambda \colon A \times A \longrightarrow A \quad \text{and} \quad \Lambda = \Delta \circ \lambda.$$

As a consequence, a reducible bibracket in a Hopf category with invertible antipode is fully determined by the associated pairing λ .

Lемма 2.3.3. — Suppose that the Hopf category C is cocommutative.

(i) If $\{\{-,-\}\}$ is reducible, then, for any $a, b \in A$,

$$\{\!\{s(a), s(b)\}\!\} = (s \otimes s) P_{21}\{\!\{a, b\}\!\};$$

- (ii) If $\{\{-,-\}\}$ is d-antisymmetric, then $(s \otimes s)\Lambda = -P_{21}\Lambda P_{21,d}$;
- (iii) If $\{\{-,-\}\}$ is reducible and d-antisymmetric, then $s\lambda = -\lambda P_{21,d}$.

Proof. — In the proof we will often use that $s^2 = id_A$. We begin with (i). It easily follows from Lemma 2.3.2 that $\Lambda(a, e_X) = 0 = \Lambda(e_X, a)$ for any $a \in A$ and $X \in Ob(\mathbb{C})$. Hence, for any $x, y \in A$,

$$\begin{split} 0 &= \Lambda(x, \varepsilon_{\text{out}}(y)) = \Lambda(x, s(y^{(1)})y^{(2)}) \\ &= \Lambda(x, s(y^{(1)}))\varepsilon(y^{(2)}) + (-1)^{|y^{(1)}| \cdot |y^{(2)}|} \Lambda(x, y^{(2))}) (s \otimes s) \Delta(s(y^{(1)})) \\ &= \Lambda(x, s(y)) + (-1)^{|y^{(1)}y^{(2)}| \cdot |y^{(3)}| + |y^{(1)}| \cdot |y^{(2)}|} \Lambda(x, y^{(3)}) (y^{(2)} \otimes y^{(1)}). \end{split}$$

Therefore

(2.3.4) $\Lambda(x, s(y)) = -(-1)^{|y^{(1)}y^{(2)}| \cdot |y^{(3)}| + |y^{(1)}| \cdot |y^{(2)}|} \Lambda(x, y^{(3)}) (y^{(2)} \otimes y^{(1)}).$ Similarly, for any $x, y \in A$,

$$\begin{split} 0 &= \Lambda(\varepsilon_{\text{out}}(x), y) = (-1)^{|x^{(1)}| \cdot |x^{(2)}|} \Lambda(s(x^{(2)})x^{(1)}, y) \\ &= (-1)^{|x^{(1)}| \cdot |x^{(2)}|} \Lambda(s(x^{(2)}), y) \varepsilon(x^{(1)}) + (-1)^{|x^{(1)}| \cdot |x^{(2)}|} \Delta(s(x^{(2)})) \Lambda(x^{(1)}, y) \\ &= \Lambda(s(x), y) + (-1)^{|x^{(1)}| \cdot |x^{(2)}x^{(3)}| + |x^{(2)}| \cdot |x^{(3)}|} (s(x^{(3)}) \otimes s(x^{(2)})) \Lambda(x^{(1)}, y), \end{split}$$

and therefore

(2.3.5)
$$\Lambda(s(x), y) = -(-1)^{|x^{(1)}| \cdot |x^{(2)}x^{(3)}| + |x^{(2)}| \cdot |x^{(3)}|} (s(x^{(3)}) \otimes s(x^{(2)})) \Lambda(x^{(1)}, y).$$

We have

$$\begin{split} (s \otimes \mathrm{id}_{A}) \{\!\!\{s(a), s(b)\}\!\} &\stackrel{(2,3,2)}{=} s((s(a))^{(1)}) \Lambda((s(a))^{(2)}, (s(b))^{(1)})(s(b))^{(2)} \\ &= (-1)^{|a^{(1)}| \cdot |a^{(2)}| + |b^{(1)}| \cdot |b^{(2)}|} a^{(2)} \Lambda(s(a^{(1)}), s(b^{(2)})) s(b^{(1)}) \\ &\stackrel{(2,3,4)}{=} \theta_{1} a^{(2)} (\Lambda(s(a^{(1)}), b^{(4)}) * b^{(3)}) b^{(2)} s(b^{(1)}) \\ &= \theta_{2} a^{(2)} (\Lambda(s(a^{(1)}), b^{(3)}) * b^{(2)}) \varepsilon_{\mathrm{in}}(b^{(1)}) \\ &= \theta_{3} a^{(2)} (\Lambda(s(a^{(1)}), b^{(3)}) * b^{(2)}) \varepsilon_{\mathrm{in}}(b^{(1)}) \\ &= \theta_{4} a^{(2)} (\Lambda(s(a^{(1)}), b^{(3)}) * b^{(2)}) \varepsilon(b^{(1)}) \\ &= \theta_{4} a^{(2)} (\Lambda(s(a^{(1)}), b^{(2)}) * b^{(1)}) \\ &\stackrel{(2,3,5)}{=} \theta_{5} a^{(4)} s(a^{(3)}) (s(a^{(2)}) * \Lambda(a^{(1)}, b^{(2)}) * b^{(1)}) \\ &= \theta_{6} \varepsilon_{\mathrm{in}}(a^{(3)}) (s(a^{(2)}) * \Lambda(a^{(1)}, b^{(2)}) * b^{(1)}) \\ &= \theta_{7} \varepsilon(a^{(3)}) (s(a^{(2)}) * \Lambda(a^{(1)}, b^{(2)}) * b^{(1)}) \\ &= \theta_{8} (s(a^{(2)}) * \Lambda(a^{(1)}, b^{(2)}) * b^{(1)}) \end{split}$$

where the signs $\theta_1, \theta_2, \ldots$ are computed by

$$\begin{split} \theta_{1} &= -(-1)^{|a^{(1)}| \cdot |a^{(2)}| + |b^{(1)}| \cdot |b^{(2)}b^{(3)}b^{(4)}| + |b^{(2)}b^{(3)}| \cdot |b^{(4)}| + |b^{(2)}| \cdot |b^{(3)}|} \\ &= -(-1)^{|a^{(1)}| \cdot |a^{(2)}| + |b^{(1)}| \cdot |b^{(2)}| + |b^{(1)}b^{(2)}| \cdot |b^{(3)}b^{(4)}| + |b^{(3)}| \cdot |b^{(4)}|} ,\\ \theta_{2} &= -(-1)^{|a^{(1)}| \cdot |a^{(2)}| + |b^{(1)}| \cdot |b^{(2)}b^{(3)}| + |b^{(2)}| \cdot |b^{(3)}|} ,\\ \theta_{3} &= -(-1)^{|a^{(1)}| \cdot |a^{(2)}| + |b^{(1)}| \cdot |b^{(2)}b^{(3)}| = -(-1)^{|a^{(1)}| \cdot |a^{(2)}| + |b^{(1)}b^{(2)}| \cdot |b^{(3)}|} ,\\ \theta_{4} &= -(-1)^{|a^{(1)}| \cdot |a^{(2)}| + |b^{(1)}| \cdot |b^{(2)}|} ,\\ \theta_{5} &= (-1)^{|a^{(1)}a^{(2)}a^{(3)}| \cdot |a^{(4)}| + |a^{(1)}| \cdot |a^{(2)}a^{(3)}| + |a^{(2)}| \cdot |a^{(3)}| + |b^{(1)}| \cdot |b^{(2)}|} ,\\ \theta_{5} &= (-1)^{|a^{(3)}| \cdot |a^{(4)}| + |a^{(1)}a^{(2)}| \cdot |a^{(3)}a^{(4)}| + |a^{(1)}| \cdot |a^{(2)}| + |b^{(1)}| \cdot |b^{(2)}|} ,\\ \theta_{6} &= (-1)^{|a^{(1)}a^{(2)}| \cdot |a^{(3)}| + |a^{(1)}| \cdot |a^{(2)}| + |b^{(1)}| \cdot |b^{(2)}|} ,\\ \theta_{7} &= (-1)^{|a^{(1)}| \cdot |a^{(2)}| + |b^{(1)}| \cdot |b^{(2)}|} = (-1)^{|a^{(1)}| \cdot |a^{(2)}a^{(3)}| + |b^{(1)}| \cdot |b^{(2)}|} ,\\ \theta_{8} &= (-1)^{|a^{(1)}| \cdot |a^{(2)}| + |b^{(1)}| \cdot |b^{(2)}|} . \end{split}$$

Therefore, using the cocommutativity of C, we obtain

$$(s \otimes \mathrm{id}_A) \{\!\!\{s(a), s(b)\}\!\!\} = s(a^{(1)}) * \Lambda(a^{(2)}, b^{(1)}) * b^{(2)}.$$

Besides,

$$(s \otimes id_A)(s \otimes s)P_{21}\{\!\{a, b\}\!\} = P_{21}(s \otimes id_A)\{\!\{a, b\}\!\}$$
$$\stackrel{(2.3.2)}{=} P_{21}(s(a^{(1)})\Lambda(a^{(2)}, b^{(1)})b^{(2)})$$
$$= s(a^{(1)}) * (P_{21}\Lambda(a^{(2)}, b^{(1)})) * b^{(2)}$$
$$= s(a^{(1)}) * \Lambda(a^{(2)}, b^{(1)}) * b^{(2)}$$

where the last equality uses the formula $P_{21}\Lambda = \Lambda$ which follows from the reducibility of $\{\{-, -\}\}$. We conclude that $\{\{s(a), s(b)\}\} = (s \otimes s)P_{21}\{\{a, b\}\}$.

We now prove (ii). If the bibracket $\{\{-, -\}\}$ is *d*-antisymmetric, then for any homogeneous $a, b \in A$,

$$\begin{split} \Lambda \mathcal{P}_{21,d}(a \otimes b) &= (-1)^{|a|_d \cdot |b|_d} \Lambda(b, a) \\ &= (-1)^{|a|_d \cdot |b|_d} b^{(1)} s\big(\{\!\{b^{(2)}, a^{(1)}\}\!\}'\big) \otimes \{\!\{b^{(2)}, a^{(1)}\}\!\}'' s(a^{(2)}) \\ &= \theta_1 b^{(1)} s\big(\{\!\{a^{(1)}, b^{(2)}\}\!\}''\big) \otimes \{\!\{a^{(1)}, b^{(2)}\}\!\}' s(a^{(2)}) \\ &= \theta_2 \mathcal{P}_{21}\big(\{\!\{a^{(1)}, b^{(2)}\}\!\}' s(a^{(2)}) \otimes b^{(1)} s\big(\{\!\{a^{(1)}, b^{(2)}\}\!\}''\big) \big) \\ &= \theta_3 \mathcal{P}_{21}(s \otimes s) \big(a^{(2)}(\{\!\{a^{(1)}, b^{(2)}\}\!\}') \otimes \{\!\{a^{(1)}, b^{(2)}\}\!\}'' s(b^{(1)})\big) \\ &= -\mathcal{P}_{21}(s \otimes s) \Lambda(a, b) \end{split}$$

where the last equality is a consequence of the cocommutativity of $\ensuremath{\mathbb{C}}$ and

$$\begin{split} \theta_{1} &= -(-1)^{d|b^{(1)}|+d|a^{(2)}|+|a^{(1)}|\cdot|b^{(1)}|+|a^{(2)}|\cdot|b^{(2)}|+|a^{(2)}|\cdot|b^{(1)}|+|\{\{a^{(1)},b^{(2)}\}\}'\{\{a^{(1)},b^{(2)}\}\}''|,\\ \theta_{2} &= -(-1)^{d|b^{(1)}|+d|a^{(2)}|+|a^{(1)}|\cdot|b^{(1)}|+|a^{(2)}|\cdot|b^{(2)}|+|a^{(2)}|\cdot|\{\{a^{(1)},b^{(2)}\}\}''|+|\{\{a^{(1)},b^{(2)}\}\}'|\cdot|b^{(1)}|,\\ \theta_{3} &= -(-1)^{|a^{(2)}|\cdot|a^{(1)}|+|b^{(2)}|\cdot|b^{(1)}|. \end{split}$$

Finally, we deduce (iii) from (ii):

$$s\lambda = s(\varepsilon \otimes \mathrm{id}_A)\Lambda = (\varepsilon \otimes \mathrm{id}_A)(s \otimes s)\Lambda$$
$$= -(\varepsilon \otimes \mathrm{id}_A)\mathrm{P}_{21}\Lambda\mathrm{P}_{21,d}$$
$$= -(\mathrm{id}_A \otimes \varepsilon)\Lambda\mathrm{P}_{21,d} = -\lambda\mathrm{P}_{21,d}.$$

2.3.4. *Remark.* — Reducible bibrackets are interesting from the algebraic viewpoint because they induce brackets in more general representation algebras associated with algebraic groups. This class of algebras includes the representation algebras considered here and associated with the general linear groups. For more on this, see [37]. The bibrackets arising below in the geometric context are reducible.

2.4. Hamiltonian reduction of bibrackets

We formulate Hamiltonian reduction for Gerstenhaber bibrackets based on a notion of an H_0 -Poisson structure. In the non-graded case, the content of this section is due to Crawley-Boevey [13] and Van den Bergh [42].

2.4.1. *H*₀-*Poisson structures.* — An *H*₀-*Poisson structure of degree* $d \in \mathbb{Z}$ on a graded algebra *A* is a *d*-graded Lie bracket $\langle -, - \rangle$ in the graded module

$$\check{A} = A/[A, A]$$

such that, for all homogeneous $x \in A$, the map

 $\langle x, - \rangle : \check{A} \longrightarrow \check{A}$

lifts to a derivation $A \rightarrow A$ of degree $|x|_d = |x| + d$. If A is a commutative graded algebra, then an H_0 -Poisson structure of degree d in A is nothing but a Gerstenhaber bracket of degree d in A.

LEMMA 2.4.1. — Given a Gerstenhaber bibracket of degree *d* in a graded algebra *A*, the induced bracket $\langle -, - \rangle$ in \check{A} is an H_0 -Poisson structure of degree *d* on *A*.

Proof. — That $\langle -, - \rangle$ is a *d*-graded Lie bracket in \check{A} follows from Lemma 1.4.1. The same lemma shows that the formula *x* $\mapsto \langle x, - \rangle$ defines a linear map $\check{A} \to \text{Der}(A)$ which preserves the Lie bracket and carries \check{A}^p to $\text{Der}^{p+d}(A)$ for all *p* ∈ \mathbb{Z} . This implies the claim of the lemma.

THEOREM 2.4.2. — Let $\langle -, - \rangle$ be an H_0 -Poisson structure of degree d on a graded algebra A and let $N \ge 1$. Then there is a unique Gerstenhaber bracket $\{-, -\}$ of degree d in the trace algebra $A_N^t \subset A_N$ such that, for any $\check{a}, \check{b} \in \check{A}$,

$$\{\operatorname{tr}(\check{a}), \operatorname{tr}(\check{b})\} = \operatorname{tr}\langle\check{a}, \check{b}\rangle.$$

Proof. — The proof follows the same lines as in the non-graded case, see [13, Theorem 4.5]. The uniqueness of $\{-, -\}$ is obvious because the image of the trace map tr : $\check{A} \to A_N$ generates A_N^t . To prove the existence, consider the commutative graded algebra $S = S(\check{A})$ freely generated by the graded module \check{A} (the *symmetric algebra* of \check{A}). The *d*-graded Lie bracket $\langle -, - \rangle$ in \check{A} uniquely extends to a Gerstenhaber

bracket $\langle -, -\rangle_S$ of degree d in S. The map tr : $A \to A_N$ uniquely extends to a graded algebra homomorphism $T : S \to A_N^t$, which is surjective. Therefore, it suffices to prove the existence of a map $\{-, -\} : A_N^t \times A_N^t \to A_N^t$ such that the following diagram commutes:



In other words, we need to show that the pairing $T\langle -, -\rangle_S : S \times S \to A_N^t$ annihilates Ker(T). Since the bracket $\langle -, -\rangle_S$ is *d*-antisymmetric, it suffices to show that

$$T\langle r, \operatorname{Ker}(T) \rangle_{s} = 0$$
 for any $r \in S$.

Since the bracket $\langle -, - \rangle_S$ satisfies the *d*-graded Leibniz rule in the first variable and *T* is an algebra homomorphism, it suffices to consider the case $r \in \check{A}$. By the definition of an H_0 -Poisson structure, the map $\langle r, - \rangle : \check{A} \to \check{A}$ lifts to a derivation $\delta : A \to A$. There is a unique derivation $\delta_N : A_N \to A_N$ such that $\delta_N(a_{ij}) = (\delta(a))_{ij}$ for any $a \in A$ and $i, j \in \{1, ..., N\}$. Then, for any $a \in A$,

$$\delta_N(\operatorname{tr}(a)) = \delta_N\left(\sum_i a_{ii}\right) = \sum_i \left(\delta(a)\right)_{ii} = \operatorname{tr} \delta(a) = \operatorname{tr} \langle r, \check{a} \rangle_S.$$

It follows that the maps $\delta_N T : S \to A_N$ and $T \langle r, -\rangle_S : S \to A_N^t \subset A_N$ are equal on $\check{A} \subset S$. Since \check{A} generates the algebra S and both these maps are derivations, they must be equal. As a consequence, $T \langle r, \text{Ker}(T) \rangle_S = 0$.

Combining Lemma 2.4.1 and Theorem 2.4.2, we obtain that any Gerstenhaber bibracket of degree d in A induces a Gerstenhaber bracket of degree d in A_N^t . Clearly this bracket is the restriction of the Gerstenhaber bracket in A_N provided by Lemma 1.2.5.

2.4.2. *Moment maps.* — Let *A* be a unital graded algebra equipped with a Gerstenhaber bibracket $\{\{-, -\}\}$ of degree *d*.

A *moment map* for $\{\{-, -\}\}$ is an element $\mu \in A^{-d}$ such that

$$\{\!\{\mu,a\}\!\} = a \otimes \mathbf{1}_A - \mathbf{1}_A \otimes a$$

for all $a \in A$ or, equivalently,

$$\{\!\{a,\mu\}\!\} = a \otimes \mathbf{1}_A - \mathbf{1}_A \otimes a$$

for all $a \in A$. If $d \neq 0$, then there is at most one moment map. If d = 0, then for any moment map $\mu \in A^0$ and any $k \in \mathbb{K}$, the sum $\mu + k\mathbf{1}_A$ is a moment map.

LEMMA 2.4.3. — Let $\mu \in A^{-d}$ be a moment map. The bracket $\langle -, - \rangle$ in A associated with $\{\{-, -\}\}$ induces an H_0 -Poisson structure of degree d on $B = A/A\mu A$.

Proof. — Let $p : A \to B$ and $h : B \to \check{B} = B/[B, B]$ be the canonical projections. Clearly, p carries [A, A] to [B, B] and induces a linear map $\check{p} : \check{A} \to \check{B}$. Lemma 1.4.1(iii) shows that the bracket $\langle -, - \rangle$ in A induces a pairing $\langle -, - \rangle : \check{A} \otimes A \to A$. We claim that there are linear maps u, v such that the following diagram commutes:

Such maps u, v are necessarily unique because p, \check{p} , h are onto. As a consequence, the following diagram commutes:

$$\begin{array}{c} \check{A} \otimes \check{A} & \xrightarrow{\check{p} \otimes \check{p}} & \check{B} \otimes \check{B} \\ \langle \neg, \neg \rangle & & & \downarrow \nu \\ \check{A} & \xrightarrow{\check{p}} & & \check{B}. \end{array}$$

Therefore *v* is a *d*-graded Lie bracket in \mathring{B} . Since $\langle x, - \rangle : A \to A$ is a derivation for all $x \in \check{A}$ and \check{p} is onto, the Lie bracket *v* is an *H*₀-Poisson structure on *B*.

It remains to verify the claim above. The definitions of the moment map μ and the bracket $\langle -, - \rangle$ in *A* imply that $\langle A, \mu \rangle = 0$. Hence,

$$\langle A, A\mu A \rangle \subset A\mu A = \operatorname{Ker} p.$$

This inclusion implies the existence of *u*. By Lemma 1.4.1 (ii),

$$\langle A\mu A, A \rangle \subset A\mu A + [A, A] = \operatorname{Ker} hp.$$

This implies the existence of *v*.

By Theorem 2.4.2, we obtain that under the assumptions of Lemma 2.4.3, the bibracket in *A* induces Gerstenhaber brackets of degree *d* on the trace algebras of $B = A/A\mu A$. As an exercise, the reader may extend Lemma 2.4.3 to the setting of graded categories discussed in Section 2.2.

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

CHAPTER 3

FACE HOMOLOGY

3.1. Manifolds with faces and partitions

We recall manifolds with faces and discuss partitions on such manifolds.

3.1.1 *Manifolds with faces.* — We start with a bigger class of *manifolds with corners*, see **[8]**, **[14]**, **[25]**, **[35]**, and **[26]**. An *n*-dimensional manifold with corners with $n \ge 0$, or, shorter, an *n*-manifold with corners, is a paracompact Hausdorff topological space locally differentiably (C^{∞}) modelled on open subsets of $[0, \infty)^n$. For a definition in terms of local coordinate systems and for further details, see **[26]**. The underlying topological space of an *n*-manifold with corners *K* is an *n*-dimensional topological manifold with boundary. The topological boundary of *K* is denoted by ∂K (the symbol ∂K has a different meaning in **[26]**). The dimension function $d_K : K \to \mathbb{Z}$ carries a point of *K* represented by a tuple (x_1, \ldots, x_n) in a local coordinate system to the number of non-zero terms in this tuple (this number does not depend on the choice of the local coordinate system). For $r \ge 0$, the set

$$K_r = \left\{ x \in K : d_K(x) \le r \right\}$$

is a closed subset of K. It is clear that

$$K_0 \subset K_1 \subset \cdots \subset K_{n-1} = \partial K \subset K_n = K.$$

Also, $K_0 = d_X^{-1}(0)$ is a discrete set, and $K_r \setminus K_{r-1}$ is a smooth *r*-dimensional manifold for all $r \ge 1$.

The set $P(K) = \partial K \setminus K_{n-2}$ is an open subset of ∂K and any $x \in \partial K$ belongs to the closure of at most $n - d_K(x)$ connected components of P(K). We call K a manifold with faces if K is compact and every $x \in \partial K$ belongs to the closure of precisely $n - d_K(x)$ different components of P(K). This condition implies that the closure in K of any component of P(K) is an (n - 1)-dimensional manifold with faces whose dimension function is the restriction of d_K . We call the closure of a component of P(K) a principal face of K. We can now define recursively on $n = \dim K$ the notion of a face of K. By definition, a face of K is a connected component of K, or a principal face of K, or a face of a principal face of K. Clearly, K has only a finite number of faces, and each face

of *K* is a connected manifold with faces. The union of faces of *K* of dimension $\leq r$ is equal to K_r for all $r \geq 0$. The faces of *K* contained in ∂K are said to be *proper*.

Every point *x* of *K* lies in the interior of a unique face F_x of *K*. If $d_K(x) \ge 1$, then F_x is the closure of the component of $K_r \setminus K_{r-1}$ containing *x* for $r = d_K(X)$. If $d_K(x) = 0$, then $F_x = \{x\}$. Note that F_x is the smallest face of *K* containing *x*: any face of *K* containing *x* contains F_x as a face.

For example, any compact smooth manifold M is a manifold with faces, and its faces are the components of M and of ∂M . For any $n \ge 0$, an n-dimensional simplex is a manifold with faces and its faces are the usual combinatorial faces. Finite disjoint unions and finite products of manifolds with faces are manifolds with faces in the obvious way. The empty set is considered as an n-manifold with faces for any $n \ge 0$.

Following [35], we call a map f from an n-manifold with faces K to an m-manifold with faces L smooth if, restricting f to any local coordinate systems in these manifolds, we obtain a map that extends to a C^{∞} -map from an open subset of \mathbb{R}^n to \mathbb{R}^m . (Such a map f is said to be "weakly smooth" in [26].) A smooth map $f : K \to L$ is continuous and its restriction to any face F of K is a smooth map $F \to L$. A map $f : K \to L$ is a *diffeomorphism* if it is a bijection and both f and f^{-1} are smooth. Diffeomorphisms of manifolds with faces preserve the dimension function and carry faces onto faces.

We can define smooth (C^{∞}) triangulations of a manifold with faces repeating word for word the standard definition of a smooth triangulation of an ordinary manifold [**38**, Section 8.3] and requiring all faces to be subcomplexes. (The latter condition is probably satisfied automatically but we prefer to spell it out.) The standard methods of the theory of smooth triangulations [**38**, Section 10.6] apply in this setting and show that all manifolds with faces have smooth triangulations.

A manifold with faces *K* is *oriented* if its underlying topological manifold is oriented. The oriented manifold with faces obtained from *K* by inverting the orientation is denoted by -K.

3.1.2. *Partitions.* — By a *partition* φ on a manifold with faces K we mean a partition of the set of faces of K into disjoint subsets, called *types*, and a family of diffeomorphisms $\{\varphi_{F,G} : F \to G\}_{(F,G)}$ numerated by ordered pairs (F,G) of faces of K of the same type such that

- (a) $\varphi_{F,F} = id_F$ for any face *F* of *K* and $\varphi_{G,H}\varphi_{F,G} = \varphi_{F,H}$ for any faces *F*, *G*, *H* of *K* of the same type;
- (b) if *F*, *G* are faces of *K* of the same type, then φ_{F,G} : *F* → *G* carries any face *F'* of *F* onto a face *G'* of *G* so that *F'*, *G'* have the same type as faces of *K* and φ_{F',G'} = φ_{F,G|F'} : *F'* → *G'*.

The diffeomorphisms $\{\varphi_{F,G}\}_{(F,G)}$ will be called *identification maps*. For example, every manifold with faces *K* has a *trivial partition* such that two faces have the same type if and only if they coincide.

Given a partition φ on *K*, we write

 $x \sim_{\varphi} y$

for points $x, y \in K$ if there are faces F, G of K of the same type such that $x \in F, y \in G$, and $\varphi_{F,G}(x) = y$. Clearly, $x \sim_{\varphi} y$ if and only if the faces F_x, F_y have the same type and $\varphi_{F_x,F_y}(x) = y$. Then \sim_{φ} is an equivalence relation on *K*. The quotient topological space

$$K_{\varphi} = K / \sim_{\varphi}$$

may not be a manifold. For any set $L \subset K$, we denote by L_{φ} the image of L under the projection $K \to K_{\varphi}$.

A smooth triangulation *T* of *K* fits a partition φ on *K* if the identification map $\varphi_{F,G} : F \to G$ is a simplicial isomorphism for any faces *F*, *G* of the same type.

LEMMA 3.1.1. — For any partition φ on K, there exists a smooth triangulation T of K which fits φ and projects to a triangulation, T_{φ} , of the quotient space K_{φ} .

Proof. — We construct by induction on $r \ge 0$ a smooth triangulation T^r of K_r satisfying the following condition: all the identification maps between faces of K of dimension $\le r$ are simplicial isomorphisms. The case r = 0 is obvious: we just take $T^0 = K_0$. Given T^{r-1} , we construct T^r as follows: pick one r-dimensional face of K in each type and extend T^{r-1} to the union of K_{r-1} with these faces using the theory of smooth triangulations [**38**, Section 10.6]. The resulting triangulation of this union uniquely extends to a triangulation T^r of K_r satisfying the condition above. Set $n = \dim K$. Clearly, $T = T^n$ is a smooth triangulation of K that fits φ .

Let *T*' and *T*'' be the first and second barycentric subdivisions of *T*, respectively. Both *T*' and *T*'' fit φ . We claim that:

- (i) the projection $\pi : K \to K_{\varphi}$ is injective on each simplex of *T*' and
- (ii) the images under π of any two simplices of T'' (which by (i) are simplices) meet along a common face.

Thus, the triangulation T'' of K projects to a triangulation of K_{φ} and satisfies the conditions of the lemma.

To prove (i), consider a simplex τ of T'. Since all simplices of T' are faces of *n*-simplices, it suffices to consider the case where $\dim(\tau) = n$. Note that the restriction of $\pi : K \to K_{\varphi}$ to the interior of any face of K is injective. Moreover, for any faces $F \subset G$ of K, the restriction of π to $\operatorname{Int}(F) \cup \operatorname{Int}(G)$ is injective. Therefore, to prove the injectivity of $\pi_{|\tau}$, it is enough to find a sequence of faces $F_0 \subset F_1 \subset \cdots \subset \sigma_n$ be the simplices of T whose barycenters are the vertices of τ where $\dim(\sigma_i) = i$ for all i. Let F_i be the smallest face of K containing σ_i . The inclusions $\sigma_{i-1} \subset \partial \sigma_i \subset F_i$ imply that $F_{i-1} \subset F_i$ for all i. Note that $\operatorname{Int}(\sigma_i) \subset \operatorname{Int}(F_i)$ since ∂F_i is a subcomplex of T. Thus, $\tau \subset \bigcup_i \operatorname{Int}(\sigma_i) \subset \bigcup_i \operatorname{Int}(F_i)$.

To prove (ii), observe first that for any simplex Δ of T', the set $\pi^{-1}(\pi(\Delta))$ is a subcomplex of T'. Indeed, this set is equal to $\cup_{F,G} \varphi_{F,G}(\Delta \cap F)$ where F, G run over all faces of K of the same type. Since T' fits φ and both Δ and F are subcomplexes of T', so are the sets $\Delta \cap F$, $\varphi_{F,G}(\Delta \cap F)$, and $\pi^{-1}(\pi(\Delta))$.

Consider any simplices τ_1, τ_2 of the triangulation T''. Let Δ_1 and Δ_2 be simplices of T' containing τ_1 and τ_2 respectively. Set $R = \pi(\Delta_1) \cap \pi(\Delta_2)$. Clearly, R is the image of the set $\Delta_1 \cap \pi^{-1}(\pi(\Delta_2))$ under π . By the above, the latter set is a subcomplex of Δ_1 . Therefore, R is a subcomplex of the simplex $\pi(\Delta_1)$. We claim that $R \cap \pi(\tau_1)$ is a face of the simplex $\pi(\tau_1)$. This claim would imply that $R \cap \pi(\tau_1) = \pi(\tau_0)$ for

a simplex τ_0 of T''. Since $R \subset \pi(\Delta_2)$, we can assume (replacing if necessary τ_0 by some $\varphi_{F,G}(\tau_0)$) that $\tau_0 \subset \Delta'_2$ where Δ'_2 is the barycentric subdivision of Δ_2 . Then

$$\pi(\tau_1) \cap \pi(\tau_2) = R \cap \pi(\tau_1) \cap \pi(\tau_2) = \pi(\tau_0) \cap \pi(\tau_2)$$

is an intersection of two simplices of $\pi(\Delta'_2)$. Hence, it is a simplex of $\pi(\Delta'_2)$ and a face of $\pi(\tau_2)$. By symmetry between τ_1 and τ_2 , the intersection $\pi(\tau_1) \cap \pi(\tau_2)$ is also a face of $\pi(\tau_1)$.

To prove the claim above, we need only to show that any subcomplex *R* of an arbitrary simplex Δ meets any simplex τ of the first barycentric subdivision Δ' along a face of τ . Clearly, $R \cap \tau$ is an intersection of two subcomplexes of Δ' and therefore a subcomplex of τ . Set $k = \dim \tau$ and let $\sigma_0 \subset \cdots \subset \sigma_k$ be the faces of Δ whose barycenters v_0, \ldots, v_k are the vertices of τ . Let *i* be the largest integer such that $v_i \in R$. Since *R* contains an interior point of σ_i and *R* is a subcomplex of Δ , we have $\sigma_i \subset R$. Then *R* contains the face $\langle v_0, \ldots, v_i \rangle$ of τ . Since $R \cap \tau$ is a subcomplex of τ not containing v_{i+1}, \ldots, v_k , we have $R \cap \tau = \langle v_0, \ldots, v_i \rangle$.

3.2. Polychains, polycycles, and face homology

We introduce the face homology of a topological space *X*.

3.2.1. *Polychains.* — Given a partition φ on a manifold with faces *K*, we say that a continuous map $\kappa : K \to X$ is *compatible* with φ if

$$\kappa \circ \varphi_{F,G} = \kappa|_F : F \longrightarrow X$$

for any faces *F*, *G* of *K* of the same type. Every such κ is obtained by composing the projection $K \to K_{\varphi}$ with a continuous map $K_{\varphi} \to X$.

An *n*-dimensional polychain or, shorter, an *n*-polychain in X with $n \ge 0$ is a quadruplet

$$\mathcal{K} = (K, \varphi, u, \kappa)$$

where *K* is an oriented *n*-manifold with faces, φ is a partition on *K*, *u* is a map $\pi_0(K) \to \mathbb{K}$ called the *weight*, and $\kappa : K \to X$ is a continuous map compatible with φ . By convention, for every $n \ge 0$, there is an *empty n-polychain* \emptyset whose underlying *n*-manifold is the empty set.

A *diffeomorphism* of *n*-polychains $\mathcal{K} = (K, \varphi, u, \kappa)$ and $\mathcal{K}' = (K', \varphi', u', \kappa')$ in *X* is a diffeomorphism $f : K \to K'$ such that

- (1) $\kappa = \kappa' \circ f;$
- (2) faces *F*, *G* of *K* have the same type if and only the faces f(F), f(G) of *K'* have the same type and then $f_{|G} \circ \varphi_{F,G} = \varphi'_{f(F),f(G)} \circ f_{|F} : F \to f(G)$;
- (3) $u'(f(C)) = \deg (f_{|C} : C \to f(C))u(C)$ for any connected component *C* of *K* where deg denotes the degree of a diffeomorphism.

We say that *n*-polychains \mathcal{K} and \mathcal{K}' in X are *diffeomorphic* and we write $\mathcal{K} \cong \mathcal{K}'$ if there exists a diffeomorphism of \mathcal{K} onto \mathcal{K}' . It is clear that \cong is an equivalence relation.

By definition, the diffeomorphism class of a polychain $\mathcal{K} = (K, \varphi, u, \kappa)$ is preserved if one simultaneously inverts the orientation of a component of *K* and multiplies the corresponding weight by -1. Therefore the *opposite* polychain $-\mathcal{K} = (-K, \varphi, u, \kappa)$ is diffeomorphic to $(K, \varphi, -u, \kappa)$.

Examples of polychains are provided by *singular manifolds* in *X*, that is pairs (an oriented smooth compact manifold *M*, a continuous map $\kappa : M \to X$). Such a pair determines a polychain (M, φ, u, κ) where *M* is viewed as a manifold with faces as in Section 3.1.1, φ is the trivial partition, and $u = 1 \in \mathbb{K}$ is the constant function on $\pi_0(M)$. As explained below, polychains in *X* may be also extracted from singular chains in *X*. Thus, we can view polychains as common generalisations of singular manifolds and singular chains in which the role of source spaces is played by manifolds with faces.

3.2.2. *Reduced polychains.* — A polychain $\mathcal{K} = (K, \varphi, u, \kappa)$ in *X* is *reduced* if any distinct connected components of *K* have different types with respect to φ and $u(C) \neq 0$ for any connected component *C* of *K*. We define two transformations of an arbitrary polychain $\mathcal{K} = (K, \varphi, u, \kappa)$ in *X* whose composition turns \mathcal{K} into a reduced polychain.

To define the first transformation, pick a representative in each type of connected components of K, and let $K_+ \subset K$ be the union of these representatives. Clearly, K_+ is a manifold with faces which we endow with orientation induced from that of K. Restricting φ and κ to K_+ we obtain a partition φ_+ on K_+ and a map $\kappa_+ : K_+ \to X$ compatible with φ_+ . We define a weight u_+ on K_+ by

$$u_+(C) = \sum_{C'} \deg(\varphi_{C,C'}) u(C')$$

where *C* is a component of *K* lying in K_+ and *C'* runs over all components of *K* of the same type as *C*. It is clear that $(K_+, \varphi_+, u_+, \kappa_+)$ is a polychain in *X* whose distinct components have different types. This polychain, denoted red₊(\mathcal{K}), is determined by \mathcal{K} uniquely up to diffeomorphism.

The second transformation of a polychain $\mathcal{K} = (K, \varphi, u, \kappa)$ removes from *K* all connected components with zero weight and restricts φ , *u*, κ to the remaining manifold with faces. The resulting polychain is denoted red₀(\mathcal{K}).

The two-step operation $\operatorname{red} = \operatorname{red}_0 \operatorname{red}_+$ transforms an arbitrary polychain into a reduced polychain defined uniquely up to diffeomorphism. It is clear that a polychain \mathcal{K} is reduced if and only if $\operatorname{red}(\mathcal{K}) \cong \mathcal{K}$.

3.2.3. *Operations.* — The *boundary* of an *n*-polychain $\mathcal{K} = (K, \varphi, u, \kappa)$ in X is the (n-1)-polychain

$$\partial \mathcal{K} = (K^{\partial}, \varphi^{\partial}, u^{\partial}, \kappa^{\partial})$$

in *X* defined as follows:

▶ The manifold with faces K^{∂} is the disjoint union of all principal faces of *K* endowed with orientation induced from that of *K* (see the Introduction for our orientation conventions).

▶ Let $\iota : K^{\partial} \to K$ be the natural map identifying each component of K^{∂} with its copy in *K*. Two faces *F*, *G* of K^{∂} have the same type if the faces $\iota(F), \iota(G)$ of *K* have

the same type and

$$\varphi_{F,G}^{\partial} = (\iota_{|G})^{-1} \varphi_{\iota(F),\iota(G)} \iota : F \to G$$

▶ For any connected component *P* of K^{∂} , we set $u^{\partial}(P) = u(K^{P})$ where K^{P} is the connected component of *K* containing the principal face $\iota(P)$.

 $\blacktriangleright \text{ We set } \kappa^{\partial} = \kappa \iota : K^{\partial} \to X.$

The boundary of a polychain is well defined up to diffeomorphism, and diffeomorphic polychains have diffeomorphic boundaries. The *reduced boundary* $\partial^r \mathcal{K}$ of a polychain \mathcal{K} is defined by

$$\partial^r \mathcal{K} = \operatorname{red}(\partial \mathcal{K}).$$

LEMMA 3.2.1. — For any polychain \mathcal{K} in X,

$$\partial^r \operatorname{red}(\mathcal{K}) = \partial^r \mathcal{K} \quad and \quad \partial^r \partial^r \mathcal{K} = \emptyset.$$

Proof. — The first identity is clear. The second identity follows from the first:

$$\partial^r \partial^r \mathcal{K} = \partial^r \operatorname{red}(\partial \mathcal{K}) = \partial^r \partial \mathcal{K} = \operatorname{red}_0 \operatorname{red}_+(\partial \partial \mathcal{K}) = \emptyset.$$

The *disjoint union* of two *n*-polychains $\mathcal{K}_1, \mathcal{K}_2$ in *X* is defined in the obvious way and is denoted $\mathcal{K}_1 \sqcup \mathcal{K}_2$. Clearly,

$$\operatorname{red}(\mathfrak{K}_1 \sqcup \mathfrak{K}_2) = \operatorname{red}(\mathfrak{K}_1) \sqcup \operatorname{red}(\mathfrak{K}_2) \quad \text{and} \quad \partial(\mathfrak{K}_1 \sqcup \mathfrak{K}_2) = \partial \mathfrak{K}_1 \sqcup \partial \mathfrak{K}_2$$

so that $\partial^r(\mathcal{K}_1 \sqcup \mathcal{K}_2) = \partial^r(\mathcal{K}_1) \sqcup \partial^r(\mathcal{K}_2).$

For $k \in \mathbb{K}$ and a polychain $\mathcal{K} = (K, \varphi, u, \kappa)$ in X, set $k\mathcal{K} = (K, \varphi, ku, \kappa)$. Clearly,

$$\operatorname{red}(k\mathcal{K}) = \operatorname{red}(k\operatorname{red}(\mathcal{K}))$$
 and $\partial(k\mathcal{K}) = k\partial\mathcal{K}$

so that $\partial^r(k\mathcal{K}) = \operatorname{red}(k\partial^r\mathcal{K})$. Note that the polychain $(-1)\mathcal{K}$ is diffeomorphic to the polychain $-\mathcal{K}$ opposite to \mathcal{K} .

3.2.4. *Face homology.* — The diffeomorphism classes of *n*-polychains in *X* may be added and multiplied by elements of \mathbb{K} , but do not form a module because the distributivity relation $(k+l)\mathcal{K} \cong k\mathcal{K} \sqcup l\mathcal{K}$ fails. Also, it is natural to throw in relations identifying \mathcal{K} with red(\mathcal{K}) for all \mathcal{K} . Quotienting the set of diffeomorphism classes of *n*-polychains in *X* by the relations of these two types, we obtain the \mathbb{K} -*module of n*-polychains in *X*. These modules together with the boundary maps induced by ∂ form the *face chain complex* of *X* whose homology is the *face homology* of *X*. However, we prefer the following more direct definition of face homology.

We say that *n*-polychains \mathcal{K}_1 and \mathcal{K}_2 in *X* are *homologous*, and write

$$\mathcal{K}_1 \simeq \mathcal{K}_2,$$

if there exist (n + 1)-polychains $\mathcal{L}_1, \mathcal{L}_2$ in *X* such that

$$\operatorname{red}(\mathcal{K}_1) \sqcup \partial^r \mathcal{L}_1 \cong \operatorname{red}(\mathcal{K}_2) \sqcup \partial^r \mathcal{L}_2.$$

Clearly, the homology relation \simeq is an equivalence relation (weaker than the diffeomorphism relation \cong). The homology class of an *n*-polychain \mathcal{K} in X is denoted by $\langle \mathcal{K} \rangle$. Note that $\langle \mathcal{K} \rangle = \langle \operatorname{red}(\mathcal{K}) \rangle$. If \mathcal{K}_1 and \mathcal{K}_2 are homologous, then $\partial^r \mathcal{K}_1 \cong \partial^r \mathcal{K}_2$.

A polychain $\mathcal{K} = (K, \varphi, u, \kappa)$ is a *polycycle* if $\partial^r \mathcal{K} = \emptyset$. A polychain homologous to a polycycle is itself a polycycle. In particular, if \mathcal{K} is a polycycle, then so is $red(\mathcal{K})$ and vice versa. Let

$$\widetilde{H}_n(X) = \{n \text{-polycycles in } X\} / \simeq$$

be the set of homology classes of *n*-polycycles in *X*. Note that the disjoint union of polycycles is a polycycle, and multiplication of polycycles by elements of \mathbb{K} yield polycycles.

LEMMA 3.2.2. — Disjoint union of polycycles together with multiplication of polycycles by elements of \mathbb{K} turn $\widetilde{H}_n(X)$ into a module (over \mathbb{K}).

Proof. — Clearly, the disjoint union of polychains is compatible with \simeq and induces a binary operation in $\widetilde{H}_n(X)$. This operation is associative and commutative with \varnothing representing the zero element. Thus, $\widetilde{H}_n(X)$ is an abelian monoid.

To prove that $\widetilde{H}_n(X)$ is a group, we use the cylinder construction on polychains. Consider an *n*-polychain

$$\mathcal{K} = (K, \varphi, u, \kappa)$$
 in X .

We define the cylinder polychain

$$\overline{\mathcal{K}} = (\overline{K}, \overline{\varphi}, \overline{u}, \overline{\kappa})$$

as follows. Set $\overline{K} = K \times I$ where I = [0, 1] is viewed as a manifold with faces I, $\{0\}$, $\{1\}$ and endow \overline{K} with the product orientation. Two faces $F \times J$, $G \times J'$ of $K \times I$ are of the same type if F, G are faces of K of the same type and J = J' is any face of I; then

$$\overline{\varphi}_{F \times J, G \times J} = \varphi_{F, G} \times \mathrm{id}_J.$$

By definition, $\overline{u}(C \times I) = u(C)$ for any connected component *C* of *K*, and $\overline{\kappa} : \overline{K} \to X$ is the composition of the cartesian projection $\overline{K} \to K$ with $\kappa : K \to X$. It follows from the definitions that

$$\operatorname{red}(\overline{\mathcal{K}}) \cong \operatorname{red}(\mathcal{K}) \quad \text{and} \quad \partial \overline{\mathcal{K}} \cong \mathcal{K} \sqcup (-\mathcal{K}) \sqcup \partial \overline{\mathcal{K}}.$$

Therefore

$$\partial^r \overline{\mathcal{K}} = \operatorname{red}(\partial \overline{\mathcal{K}}) \cong \operatorname{red}(\mathcal{K}) \sqcup \operatorname{red}(-\mathcal{K}) \sqcup \operatorname{red}(\partial \overline{\mathcal{K}}) \cong \operatorname{red}(\mathcal{K}) \sqcup \operatorname{red}(-\mathcal{K}) \sqcup \partial^r (\mathcal{K}).$$

If \mathcal{K} is a polycycle, this gives $\partial^r \overline{\mathcal{K}} \cong \operatorname{red}(\mathcal{K}) \sqcup \operatorname{red}(-\mathcal{K})$. Therefore $\mathcal{K} \sqcup (-\mathcal{K}) \simeq \emptyset$. We conclude that $\widetilde{H}_n(X)$ is an abelian group.

Given two homologous *n*-polycycles \mathcal{K}_1 and \mathcal{K}_2 in *X*, pick (n+1)-polychains $\mathcal{L}_1, \mathcal{L}_2$ in *X* such that

$$\operatorname{red}(\mathcal{K}_1) \sqcup \partial^r \mathcal{L}_1 \cong \operatorname{red}(\mathcal{K}_2) \sqcup \partial^r \mathcal{L}_2$$

Then, for any $k \in \mathbb{K}$,

$$\operatorname{red}\left(k(\operatorname{red}(\mathcal{K}_1)\sqcup\partial^r\mathcal{L}_1)\right)\cong\operatorname{red}\left(k(\operatorname{red}(\mathcal{K}_2)\sqcup\partial^r\mathcal{L}_2)\right).$$

For each $i \in \{1, 2\}$,

$$\operatorname{red}\left(k(\operatorname{red}(\mathcal{K}_i)\sqcup\partial^r\mathcal{L}_i)\right) = \operatorname{red}\left(k\operatorname{red}(\mathcal{K}_i)\right)\sqcup\operatorname{red}(k\partial^r\mathcal{L}_i) = \operatorname{red}(k\mathcal{K}_i)\sqcup\partial^r(k\mathcal{L}_i).$$

We deduce that $k\mathcal{K}_1 \simeq k\mathcal{K}_2$. Thus, the multiplication by $k \in \mathbb{K}$ induces a well defined map

$$\widetilde{H}_n(X) \longrightarrow \widetilde{H}_n(X).$$

The axioms of a \mathbb{K} -module are straightforward except the linearity in *k*. The latter is a consequence of the following fact: if

$$\mathfrak{K}_1 = (K, \varphi, u_1, \kappa)$$
 and $\mathfrak{K}_2 = (K, \varphi, u_2, \kappa)$

are *n*-polycycles in *X* (with the same *K*, φ , κ), then the *n*-polychain

$$\mathfrak{K} = (K, \varphi, u_1 + u_2, \kappa)$$

is a polycycle homologous to $\mathcal{K}_1 \sqcup \mathcal{K}_2$. To see this, consider the cylinder polychain

$$\overline{\mathcal{K}_1 \sqcup \mathcal{K}_2} \cong \overline{\mathcal{K}}_1 \sqcup \overline{\mathcal{K}}_2$$

(as defined above) and modify its partition by additionally declaring that, for any face *F* of *K*, the faces $(F \times \{0\})_1$ and $(F \times \{0\})_2$ of $(K \times I)_1 \sqcup (K \times I)_2$ have the same type and the corresponding identification map is the identity map. This gives an (n + 1)-polychain \mathcal{L} such that

 $\operatorname{red}_+(\partial \mathcal{L}) \cong \operatorname{red}_+(\mathcal{K}_1) \sqcup \operatorname{red}_+(\mathcal{K}_2) \sqcup \operatorname{red}_+(-\mathcal{K}) \sqcup (a \text{ polychain with zero weight}).$

Therefore

 $\partial^r \mathcal{L} \cong \operatorname{red}(\mathcal{K}_1) \sqcup \operatorname{red}(\mathcal{K}_2) \sqcup \operatorname{red}(-\mathcal{K}).$

Hence, \mathcal{K} is a polycycle homologous to $\mathcal{K}_1 \sqcup \mathcal{K}_2$.

We call $\widetilde{H}_n(X)$ the *n*-th face homology of X (with coefficients in K).

The face homology extends to a functor from the category of topological spaces to the category of modules: a continuous map $f : X \rightarrow Y$ induces a linear map

$$f_*: \widetilde{H}_n(X) \longrightarrow \widetilde{H}_n(Y)$$

carrying the homology class of a polycycle $\mathcal{K} = (K, \varphi, u, \kappa)$ in *X* to the homology class of the polycycle $f_*(\mathcal{K}) = (K, \varphi, u, f\kappa)$ in *Y*.

3.2.5. *Deformations.* — A *deformation* of a polychain $\mathcal{K} = (K, \varphi, u, \kappa)$ in X is a family of polychains $\{\mathcal{K}^t = (K, \varphi, u, \kappa^t)\}_{t \in I}$ with the same K, φ, u such that $\{\kappa^t : K \to X\}_{t \in I}$ is a (continuous) homotopy of $\kappa^0 = \kappa$. By the definition of a polychain, the map κ^t is compatible with φ for all $t \in I$.

LEMMA 3.2.3. — If $\{\mathcal{K}^t\}_{t \in I}$ is a deformation of a polycycle \mathcal{K} , then \mathcal{K}^1 is a polycycle homologous to $\mathcal{K} = \mathcal{K}^0$.

Proof. — Equality $\partial^r \mathcal{K}^1 = \emptyset$ is a direct consequence of the assumption $\partial^r \mathcal{K} = \emptyset$. Consider the cylinder polychain $\overline{\mathcal{K}} = (\overline{K}, \overline{\varphi}, \overline{u}, \overline{\kappa})$ associated with $\mathcal{K} = (K, \varphi, u, \kappa)$ in the proof of Lemma 3.2.2. Let $\widehat{\kappa} : \overline{K} = K \times I \to X$ be the map determined by the homotopy $\{\kappa^t\}_{t \in I}$ of κ . Then $\mathcal{R} = (\overline{K}, \overline{\varphi}, \overline{u}, \widehat{\kappa})$ is a polychain such that

$$\partial^r \mathcal{R} \cong \operatorname{red}(\mathcal{K}^1) \sqcup \operatorname{red}(-\mathcal{K}^0).$$

This implies that $\mathcal{K}^0 \simeq \mathcal{K}^1$.

MÉMOIRES DE LA SMF 154

LEMMA 3.2.4. — Let X, Y be topological spaces. If maps $f, g: X \to Y$ are homotopic, then $f_* = g_* : \widetilde{H}_*(X) \longrightarrow \widetilde{H}_*(Y).$

Proof. — Pick a homotopy $\{f^t\}_{t \in I}$ between $f^0 = f$ and $f^1 = g$. For any polycycle $\mathcal{K} = (K, \varphi, u, \kappa)$ in X, we have a deformation $\{(K, \varphi, u, f^t \kappa)\}_{t \in I}$ relating the polycycles $f_*(\mathcal{K}) = (K, \varphi, u, f \kappa)$ and $g_*(\mathcal{K}) = (K, \varphi, u, g \kappa)$. Lemma 3.2.3 implies that these polycycles are homologous. Hence, $f_* = g_*$.

Lemma 3.2.4 implies that a homotopy equivalence between topological spaces induces an isomorphism of their face homology.

3.2.6. *Cross product.* — The cartesian product $K \times L$ of two manifolds with faces K and L can be viewed as a manifold with faces in the obvious way. The faces of $K \times L$ are the products $F \times G$ where F runs over faces of K and G runs over faces of L. When K and L are oriented, we always provide $K \times L$ with the product orientation. This construction leads to a cross product in face homology as follows.

Let *X* and *Y* be topological spaces. The *cross product* of a *p*-polychain $\mathcal{K} = (K, \varphi, u, \kappa)$ in *X* and a *q*-polychain $\mathcal{L} = (L, \psi, v, \lambda)$ in *Y* is the (p + q)-polychain

$$\mathcal{K} \times \mathcal{L} = (K \times L, \varphi \times \psi, u \times v, \kappa \times \lambda)$$

in $X \times Y$. Here $\varphi \times \psi$ is the following partition on $K \times L$: for faces F, F' of K and G, G' of L, the faces $F \times G$ and $F' \times G'$ of $K \times L$ have the same type if, and only if, F has the same type as F' and G has the same type as G', and then

$$(\varphi \times \psi)_{F \times G, F' \times G'} = \varphi_{F, F'} \times \psi_{G, G'}.$$

The weight $u \times v$ carries $C \times D$ to u(C)v(D) for any connected components *C* of *K* and *D* of *L*. We also define the *reduced cross product* of \mathcal{K} and \mathcal{L} by

$$\mathcal{K} \times^{r} \mathcal{L} = \operatorname{red} \left(\mathcal{K} \times \mathcal{L} \right).$$

Note that

$$(3.2.1) \qquad \qquad \mathcal{K} \times^{r} \mathcal{L} \cong \mathcal{K} \times^{r} \operatorname{red}(\mathcal{L}) \cong \operatorname{red}(\mathcal{K}) \times^{r} \mathcal{L} \cong \operatorname{red}(\mathcal{K}) \times^{r} \operatorname{red}(\mathcal{L}).$$

LEMMA 3.2.5

(i) For any *p*-polychains $\mathcal{K}_1, \mathcal{K}_2$ in *X* and *q*-polychain \mathcal{L} in *Y*,

$$(\mathcal{K}_1 \sqcup \mathcal{K}_2) \times^r \mathcal{L} \cong (\mathcal{K}_1 \times^r \mathcal{L}) \sqcup (\mathcal{K}_2 \times^r \mathcal{L}).$$

(ii) For any *p*-polychain \mathcal{K} in X and for any *q*-polychain \mathcal{L} in Y,

 $\partial^r (\mathcal{K} \times^r \mathcal{L}) \cong (\partial^r \mathcal{K} \times^r \mathcal{L}) \sqcup (-1)^p (\mathcal{K} \times^r \partial^r \mathcal{L}).$

Proof. — Clearly, $(\mathcal{K}_1 \sqcup \mathcal{K}_2) \times \mathcal{L} = (\mathcal{K}_1 \times \mathcal{L}) \sqcup (\mathcal{K}_2 \times \mathcal{L})$ so that

$$(\mathcal{K}_1 \sqcup \mathcal{K}_2) \times^r \mathcal{L} = \operatorname{red} \left((\mathcal{K}_1 \sqcup \mathcal{K}_2) \times \mathcal{L} \right) \cong \operatorname{red}(\mathcal{K}_1 \times \mathcal{L}) \sqcup \operatorname{red}(\mathcal{K}_2 \times \mathcal{L})$$

which proves (i). We now prove (ii). Let *K* and *L* be the oriented manifolds with faces underlying \mathcal{K} and \mathcal{L} respectively. A principal face of $K \times L$ has either the form $P \times D$, where *P* is a principal face of *K* and *D* is a component of *L*, or the form $C \times Q$, where *C* is a component of *K* and *Q* is a principal face of *L*. The orientation of $P \times D \subset \partial(K \times L)$

inherited from $K \times L$ coincides with the product orientation of $P \times D$ where $P \subset \partial K$ inherits orientation from K. The orientation of $C \times Q \subset \partial(K \times L)$ inherited from $K \times L$ differs from the product orientation of $C \times Q$, where $Q \subset \partial L$ inherits orientation from L, by the sign $(-1)^p$. So

$$\partial(\mathcal{K} \times \mathcal{L}) \cong (\partial \mathcal{K} \times \mathcal{L}) \sqcup (-1)^p (\mathcal{K} \times \partial \mathcal{L}).$$

Therefore

$$\begin{aligned} \partial^r (\mathcal{K} \times^r \mathcal{L}) \ &= \ \partial^r \operatorname{red}(\mathcal{K} \times \mathcal{L}) = \partial^r (\mathcal{K} \times \mathcal{L}) \\ &= \operatorname{red} \partial (K \times L) \\ &\cong (\partial \mathcal{K} \times^r \mathcal{L}) \sqcup (-1)^p (\mathcal{K} \times^r \partial \mathcal{L}) \,. \end{aligned}$$

We conclude thanks to (3.2.1).

LEMMA 3.2.6. — The cross product of polychains induces a bilinear map

$$(3.2.2) \qquad \qquad \times : H_*(X) \times H_*(Y) \longrightarrow H_*(X \times Y).$$

Proof. — Let \mathcal{K} be a polycycle in X and \mathcal{L} be a polycycle in Y. Lemma 3.2.5 (ii) implies that $\mathcal{K} \times r^r \mathcal{L}$ is a polycycle. This polycycle is the reduction of $\mathcal{K} \times \mathcal{L}$, and therefore $\mathcal{K} \times \mathcal{L}$ also is a polycycle. We claim that assigning to $(\mathcal{K}, \mathcal{L})$ the homology class $\langle \mathcal{K} \times r^r \mathcal{L} \rangle = \langle \mathcal{K} \times \mathcal{L} \rangle$ one obtains a well defined pairing (3.2.2). Let us prove the independence of the choice of \mathcal{K} in its homology class (the second variable is treated similarly). Consider two homologous polycycles \mathcal{K}_1 and \mathcal{K}_2 in X, and let $\mathcal{P}_1, \mathcal{P}_2$ be polychains in X such that

$$\operatorname{red}(\mathcal{K}_1) \sqcup \partial^r \mathcal{P}_1 \cong \operatorname{red}(\mathcal{K}_2) \sqcup \partial^r \mathcal{P}_2.$$

Lemma 3.2.5 (i) implies that

$$(\operatorname{red}(\mathcal{K}_1) \times^r \mathcal{L}) \sqcup (\partial^r \mathcal{P}_1 \times^r \mathcal{L}) \cong (\operatorname{red}(\mathcal{K}_2) \times^r \mathcal{L}) \sqcup (\partial^r \mathcal{P}_2 \times^r \mathcal{L}).$$

For $i \in \{1, 2\}$, formula (3.2.1) gives $\operatorname{red}(\mathcal{K}_i) \times^r \mathcal{L} \cong \operatorname{red}(\mathcal{K}_i \times^r \mathcal{L})$. Since $\partial^r \mathcal{L} = \emptyset$, Lemma 3.2.5 (ii) gives $\partial^r \mathcal{P}_i \times^r \mathcal{L} = \partial^r (\mathcal{P}_i \times^r \mathcal{L})$. Therefore

$$\mathcal{K}_1 \times^r \mathcal{L} \simeq \mathcal{K}_2 \times^r \mathcal{L}.$$

The linearity of (3.2.2) in the first variable follows from Lemma 3.2.5 (i) and the equality $(k\mathcal{K}) \times \mathcal{L} = k(\mathcal{K} \times \mathcal{L})$ for all $k \in \mathbb{K}$. The linearity in the second variable is proved similarly.

3.2.7. Remarks

(1) A polychain derived from a singular manifold $\kappa : M \to X$ (see Section 3.2.1) is a polycycle if and only if $\partial M = \emptyset$. The oriented bordism classes of *n*-dimensional singular manifolds

$$\kappa: M \longrightarrow X$$
 with $\partial M = \emptyset$

form an abelian group $\Omega_n(X)$, called the *n*-dimensional oriented bordism group of X. Treating singular manifolds as polychains, we obtain an additive map

$$\Omega_n(X) \longrightarrow H_n(X).$$

MÉMOIRES DE LA SMF 154

By Remark 3.3.5.2 below, this map is not surjective for some *n*, *X*, and $\mathbb{K} = \mathbb{Z}$. Thus, some face homology classes over \mathbb{Z} are not representable by singular manifolds.

(2) For a topological pair (X, Y) and an integer $n \ge 0$, we define the *n*-th *relative face* homology $\widetilde{H}_n(X, Y)$ as follows. Given *n*-polychains $\mathcal{K}_1, \mathcal{K}_2$ in *X*, we write

$$\mathcal{K}_1 \simeq_Y \mathcal{K}_2$$

if there exist (n + 1)-polychains $\mathcal{L}_1, \mathcal{L}_2$ in *X* and *n*-polychains $\mathcal{N}_1, \mathcal{N}_2$ in *Y* such that

$$\operatorname{red}(\mathcal{K}_1) \sqcup \partial^r \mathcal{L}_1 \sqcup \iota_*(\mathcal{N}_1) \cong \operatorname{red}(\mathcal{K}_2) \sqcup \partial^r \mathcal{L}_2 \sqcup \iota_*(\mathcal{N}_2)$$

where $\iota : Y \hookrightarrow X$ is the inclusion map. An *n*-polychain \mathcal{K} in *X* is a *polycycle relative to Y* if $\partial^r \mathcal{K}$ is the image of an (n - 1)-polychain in *Y* under ι . Set

$$\widetilde{H}_n(X,Y) = \{n \text{-polycycles in } X \text{ relative to } Y\} / \simeq_Y$$
.

The properties of the face homology of topological spaces stated above directly extend to the face homology of topological pairs.

3.3. Face homology versus singular homology

In this section, we construct two natural transformations

$$[-]: \widetilde{H}_* \longrightarrow H_*$$
 and $\langle - \rangle: H_* \longrightarrow \widetilde{H}_*$

relating face homology to singular homology.

3.3.1. *Preliminaries.* — For an integer $n \ge 0$, the symbol Δ^n denotes the standard *n*-simplex that is the convex hull of the standard basis (e_0, \ldots, e_n) of \mathbb{R}^{n+1} . We endow Δ^n with orientation induced by the order of its vertices, *i.e.* the orientation represented by the basis $(\overrightarrow{e_0e_1}, \overrightarrow{e_1e_2}, \ldots, \overrightarrow{e_{n-1}e_n})$ in the tangent space of Δ^n at any point. Each subset $A = \{i_0, i_1, \ldots, i_r\}$ of $\{0, \ldots, n\}$ with $i_0 < i_1 < \cdots < i_r$ and $0 \le r \le n$ determines an affine map $e_A : \Delta^r \to \Delta^n$ carrying the vertices e_0, e_1, \ldots, e_r of Δ^r to the vertices $e_{i_0}, e_{i_1}, \ldots, e_{i_r}$ of Δ^n , respectively; the image of the map e_A is the *combinatorial face* of Δ^n corresponding to A.

A *singular n-simplex* in a topological space *X* is a continuous map

$$\Delta^n \longrightarrow X.$$

A *singular n-chain* in *X* is a finite formal linear combination of singular *n*-simplices with coefficients in \mathbb{K} . The *boundary* of a singular *n*-simplex $\sigma : \Delta^n \to X$ is the singular (n-1)-chain

(3.3.1)
$$\partial \sigma = \sum_{a=0}^{n} (-1)^{a} \cdot \sigma e_{\hat{a}} \quad \text{where } \hat{a} = \{0, 1, \dots, n\} \setminus \{a\}.$$

The boundary of singular simplices extends to singular chains by linearity. The modules of singular chains together with the boundary homomorphisms form the *singular chain complex* $C_*(X)$ of X. Its homology is the *singular homology* $H_*(X)$ of X (with coefficients in \mathbb{K}).

3.3.2. *The transformation* [-]. — Consider an *n*-dimensional oriented manifold with faces *K*. Each weight $u : \pi_0(K) \to \mathbb{K}$ determines a homology class

$$[K, u] = \sum_{C} u(C) [C] \in \bigoplus_{C} H_n(C, \partial C) = H_n(K, \partial K)$$

where *C* runs over all connected components of *K* and $[C] \in H_n(C, \partial C)$ is the fundamental class of *C*. We say that a partition φ on *K* is *compatible* with *u* if for any principal face *P* of *K*,

(3.3.2)
$$\sum_{Q} \deg(\varphi_{P,Q}) u(K^{Q}) = 0$$

where Q runs over all (principal) faces of K of the same type as P and K^Q is the connected component of K containing Q.

LEMMA 3.3.1. — Let φ be a partition on K compatible with a weight $u : \pi_0(K) \to \mathbb{K}$. Then there is a unique homology class

$$[K_{\varphi}, u] \in H_n(K_{\varphi})$$

whose image in $H_n(K_{\varphi}, (\partial K)_{\varphi})$ is equal to the image of [K, u] under the map

$$H_n(K,\partial K) \longrightarrow H_n(K_{\varphi},(\partial K)_{\varphi})$$

induced by the projection $K \to K_{\varphi}$.

Proof. — Consider the commutative diagram

$$\begin{array}{cccc} H_n(\partial K) & \longrightarrow & H_n(K) & \longrightarrow & H_n(K, \partial K) & \xrightarrow{\partial_*} & H_{n-1}(\partial K) \\ & & & & \downarrow \pi_* & & \downarrow \pi_* & & \downarrow \pi_* \\ H_n\left((\partial K)_{\varphi}\right) & \longrightarrow & H_n\left(K_{\varphi}\right) & \longrightarrow & H_n\left(K_{\varphi}, (\partial K)_{\varphi}\right) & \xrightarrow{\partial_*} & H_{n-1}\left((\partial K)_{\varphi}\right), \end{array}$$

where the vertical maps are induced by the projection $\pi : K \to K_{\varphi}$ and each row is a part of the long exact sequence of a topological pair. We have

$$H_n((\partial K)_{\varphi}) = 0$$

since $(\partial K)_{\varphi}$ is an (n-1)-dimensional polyhedron. Hence, it is enough to show that

$$\pi_*\partial_*([K,u]) = 0 \in H_{n-1}\left((\partial K)_\varphi\right)$$

Consider the commutative square

$$\partial_*([K,u]) \in H_{n-1}(\partial K) \xrightarrow{j} H_{n-1}(\partial K, K_{n-2})$$

$$\downarrow^{\pi_*} \qquad \qquad \downarrow^{\pi_*} \\ H_{n-1}\left((\partial K)_{\varphi}\right) \xrightarrow{j_{\varphi}} H_{n-1}\left((\partial K)_{\varphi}, (K_{n-2})_{\varphi}\right)$$

where j and j_{φ} are the inclusion homomorphisms. Since $(K_{n-2})_{\varphi}$ is an (n-2)-dimensional polyhedron, Ker $j_{\varphi} = 0$ and it suffices to prove that

$$j_{\varphi}\pi_*\partial_*([K,u])=0$$

or, equivalently, that $\pi_* j \partial_* ([K, u]) = 0$. We have

$$\partial_*([K,u]) = \sum_C u(C)\partial_*([C]) = \sum_C u(C)[\partial C]$$

where the sum runs over the connected components C of K. Then

$$j\partial_*([K,u]) = \sum_C u(C) \sum_{P \subset C} [P] = \sum_P u(K^P)[P]$$

where *P* runs over all principal faces of *K* and K^P is the connected component of *K* containing *P*. Pick a face $P_i \in i$ in each type *i* of principal faces of *K*. Then

$$\pi_* j\partial_*([K, u]) = \sum_P u(K^P)\pi_*([P])$$

= $\sum_i \sum_{Q \in i} u(K^Q)\pi_*([Q])$
= $\sum_i \sum_{Q \in i} u(K^Q)\pi_*(\deg(\varphi_{P_i,Q}) \cdot (\varphi_{P_i,Q})_*([P_i]))$
= $\sum_i \left(\sum_{Q \in i} u(K^Q)\deg(\varphi_{P_i,Q})\right)\pi_*([P_i]) = 0$

where at the last step we use the compatibility condition (3.3.2).

It follows from the definitions that a polychain (K, φ, u, κ) in a topological space *X* is a polycycle if and only if *u* and φ are compatible in the sense of (3.3.2). Therefore, given an *n*-polycycle $\mathcal{K} = (K, \varphi, u, \kappa)$ in *X*, Lemma 3.3.1 gives the homology class

$$[K_{\varphi}, u] \in H_n(K_{\varphi}).$$

Since the map $\kappa : K \to X$ is compatible with φ , it induces a continuous map $K_{\varphi} \to X$ denoted by κ_{φ} . We define

$$[\mathcal{K}] = (\kappa_{\varphi})_* ([K_{\varphi}, u]) \in H_n(X).$$

The homology class $[\mathcal{K}]$ can be represented by explicit singular cycles which are best described in terms of locally ordered triangulations. A *local order* on a triangulation *T* of a topological space is a binary relation on the set of vertices of *T* which restricts to a total order on the set of vertices of any simplex of *T*. For example, any total order on the set of vertices of *T* is a local order on *T*. A triangulation endowed with a local order is *locally ordered*. We say that a locally ordered smooth triangulation *T* of *K fits the partition* φ if, for any faces *F*, *G* of *K* of the same type, the identification map $\varphi_{F,G} : F \to G$ is a simplicial isomorphism preserving the local order on the vertices. To construct such a locally ordered triangulation one can take a triangulation *T* of *K* provided by Lemma 3.1.1 and lift an arbitrary total order \leq on the set of vertices of T_{φ} to *T*. More precisely, denote by $\pi : K \to K_{\varphi}$ the canonical projection and, for any vertices $a, b \in T$, declare that $a \leq b$ if $\pi(a) \leq \pi(b)$. Since any simplex of *T* projects isomorphically onto a simplex of T_{φ} , this gives a local order on *T* which, obviously, fits φ .

Pick a locally ordered smooth triangulation *T* of *K* which fits φ . Each *r*-simplex Δ of *T* with $r \ge 0$ determines a singular simplex in *K* denoted by σ_{Δ} and obtained as the composition of the affine isomorphism $\Delta^r \to \Delta$ preserving the order of the vertices with the inclusion $\Delta \hookrightarrow K$. We define the *fundamental n-chain*

(3.3.3)
$$\sigma = \sigma(T, u) = \sum_{\Delta} \varepsilon_{\Delta} u(K^{\Delta}) \sigma_{\Delta} \in C_n(K)$$

where Δ runs over all *n*-simplices of *T*, K^{Δ} is the connected component of *K* containing Δ , $\varepsilon_{\Delta} = +1$ if the orientation of Δ induced by that of *K* is compatible with the order of the vertices of Δ and $\varepsilon_{\Delta} = -1$ otherwise. Clearly, the image of σ in $C_n(K, \partial K)$ is a relative *n*-cycle representing [K, u]. Projecting σ to K_{φ} , we obtain a singular *n*-chain $\sigma_{\varphi} \in C_n(K_{\varphi})$. The compatibility of φ and *u* implies that σ_{φ} is an *n*-cycle. Therefore $[\sigma_{\varphi}] \in H_n(K_{\varphi})$ satisfies the requirements of Lemma 3.3.1 so that

$$[K_{\varphi}, u] = [\sigma_{\varphi}].$$

It follows that $[\mathcal{K}]$ is represented by the singular *n*-cycle

$$(\kappa_{\varphi})_*(\sigma_{\varphi}) = \kappa_*(\sigma) = \sum_{\Delta} \varepsilon_{\Delta} u(K^{\Delta}) \kappa \sigma_{\Delta} \in C_n(X).$$

LEMMA 3.3.2. — The formula $\langle \mathfrak{K} \rangle \mapsto [\mathfrak{K}]$ defines a linear map

$$[-]: \widetilde{H}_n(X) \longrightarrow H_n(X).$$

Moreover, [-] *is a natural transformation from* \widetilde{H}_n *to* H_n *.*

Proof. — It follows from the definitions that $[\mathcal{K}] \in H_n(X)$ depends only on the diffeomorphism class of \mathcal{K} , that $[\mathcal{K}] = [red(\mathcal{K})]$, and that $[k\mathcal{K}] = k[\mathcal{K}]$ for any $k \in \mathbb{K}$. Moreover, $[\mathcal{K}_1 \sqcup \mathcal{K}_2] = [\mathcal{K}_1] + [\mathcal{K}_2]$ for any *n*-polycycles $\mathcal{K}_1, \mathcal{K}_2$ in *X*. Therefore, to prove the first claim of the lemma, it is enough to show that $[\partial \mathcal{L}] = 0$ for any (n + 1)-polychain $\mathcal{L} = (L, \psi, \nu, \lambda)$ in *X*. For this, pick a locally ordered smooth triangulation *T* of *L* that fits ψ and consider the singular chain

$$\sigma = \sigma(T, v) = \sum_{\Delta} \varepsilon_{\Delta} v(L^{\Delta}) \sigma_{\Delta} \in C_{n+1}(L).$$

Here Δ runs over all (n + 1)-dimensional simplices of T, ε_{Δ} is the sign determined by the orientation of L and the order of the vertices of Δ , and L^{Δ} is the component of L containing Δ . Projecting σ to L_{ψ} we obtain a singular chain σ_{ψ} in L_{ψ} . Next we consider the *n*-polycycle $\partial \mathcal{L} = (L^{\partial}, \psi^{\partial}, v^{\partial}, \lambda^{\partial})$. The triangulation T of L induces a triangulation T^{∂} of L^{∂} . The local order on the set of vertices of T restricts to a local order on the set of vertices of T^{∂} . Consider the fundamental *n*-chain $\tau = \sigma(T^{\partial}, v^{\partial})$ in L^{∂} as defined before the statement of the lemma. Projecting τ to the quotient space $(L^{\partial})_{\psi^{\partial}}$ we obtain a singular *n*-cycle $\tau_{\psi^{\partial}}$ representing

$$[(L^{\partial})_{\psi^{\partial}}, u^{\partial}] \in H_n((L^{\partial})_{\psi^{\partial}}).$$

The natural map $\iota : L^{\partial} \to L$ induces a map $\iota_{\psi} : (L^{\partial})_{\psi^{\partial}} \to L_{\psi}$ carrying $\tau_{\psi^{\partial}}$ to $\partial \sigma_{\psi}$. By definition, $\lambda^{\partial} = \lambda \iota : L^{\partial} \to X$. Therefore

$$(\lambda^{\partial})_{\psi^{\partial}} = \lambda_{\psi}\iota_{\psi} : (L^{\partial})_{\psi^{\partial}} \longrightarrow X$$

where $\lambda_{\psi} : L_{\psi} \to X$ is the map induced by $\lambda : L \to X$. Hence,

$$[\partial \mathcal{L}] = \left((\lambda^{\partial})_{\psi^{\partial}} \right)_{*} \left([\tau_{\psi^{\partial}}] \right) = (\lambda_{\psi} \iota_{\psi})_{*} \left([\tau_{\psi^{\partial}}] \right) = (\lambda_{\psi})_{*} \left([\partial \sigma_{\psi}] \right) = 0.$$

To prove the second claim of the lemma, consider a continuous map $f : X \to Y$. For any *n*-polycycle $\mathcal{K} = (K, \varphi, u, \kappa)$ in *X*, we have

$$f_*([\mathcal{K}]) = f_*(\kappa_{\varphi})_*([K_{\varphi}, u]) = (f \kappa_{\varphi})_*([K_{\varphi}, u]) = ((f \kappa)_{\varphi})_*([K_{\varphi}, u]) = [f_*(\mathcal{K})]$$

since $f_*(\mathcal{K}) = (K, \varphi, u, f \kappa)$ by definition.

3.3.3. *The transformation* $\langle - \rangle$. — Let *X* be a topological space and let $n \ge 0$ be an integer. We associate with each singular *n*-chain σ in *X* an *n*-polychain $\mathcal{P}(\sigma)$ in *X*. Pick an expansion

$$\sigma = \sum_i k_i \sigma_i,$$

where *i* runs over a finite set of indices, $k_i \in \mathbb{K}$, and $\{\sigma_i\}_i$ are singular *n*-simplices in *X*. Let *K* be the manifold with faces obtained as a disjoint union of copies $(\Delta^n)_i$ of Δ^n numerated by all *i*. We define a partition φ on *K* as follows: a face *F* of $(\Delta^n)_i$ corresponding to a set $A \subset \{0, ..., n\}$ and a face *F'* of $(\Delta^n)_{i'}$ corresponding to a set $A' \subset \{0, ..., n\}$ are declared to be of the same type if *A* and *A'* have the same cardinality $r \leq n + 1$, and $\sigma_i e_A = \sigma_{i'} e_{A'} : \Delta^{r-1} \to X$ (where $e_A, e_{A'}$ are the maps defined in Section 3.3.1). Then we set

$$F_{F,F'} = e_{A'}e_A^{-1} : F \longrightarrow F'.$$

Clearly, the map $\kappa = \coprod_i \sigma_i : K \to X$ is compatible with φ . We define a weight

$$u:\pi_0(K)\longrightarrow\mathbb{K}$$

by $u((\Delta^n)_i) = k_i$ for all *i*. The tuple (K, φ, u, κ) is an *n*-polychain in *X* depending on the choice of the expansion $\sigma = \sum_i k_i \sigma_i$. However, the polychain

$$\mathcal{P}(\sigma) = \operatorname{red}(K, \varphi, u, \kappa)$$

does not depend on this choice. Indeed, any two expansions of σ may be related by the following operations: replacement of $k\sigma_{\bullet} + l\sigma_{\bullet}$ by $(k + l)\sigma_{\bullet}$ for any $k, l \in \mathbb{K}$ and any singular *n*-simplex σ_{\bullet} in *X*; addition of a term $0\sigma_{\bullet}$ for an arbitrary singular *n*-simplex σ_{\bullet} in *X*; the inverse operations. It is easy to see that $\mathcal{P}(\sigma)$ is preserved under these transformations. By definition, if $\sigma = 0$, then $\mathcal{P}(\sigma) = \emptyset$.

The face homology class $\langle \mathfrak{P}(\sigma) \rangle$ of the polychain $\mathfrak{P}(\sigma)$ will be denoted by $\langle \sigma \rangle$.

LEMMA 3.3.3. — If σ is a cycle, then $\mathfrak{P}(\sigma)$ is a polycycle. The formula $[\sigma] \mapsto \langle \sigma \rangle$, applied to singular *n*-cycles in *X*, defines a linear map

$$\langle - \rangle : H_n(X) \longrightarrow \widetilde{H}_n(X).$$

Moreover, $\langle - \rangle$ *is a natural transformation from* H_n *to* H_n *.*

Proof. — We check first that for any singular *n*-chain σ in *X*,

(3.3.4) $\partial^r \mathfrak{P}(\sigma) \cong \mathfrak{P}(\partial \sigma).$

Pick an expansion $\sigma = \sum_i k_i \sigma_i$ such that the simplices $\{\sigma_i\}_i$ are pairwise distinct and $k_i \neq 0$ for all *i*. Then the associated polychain (K, φ, u, κ) is reduced and

$$\mathcal{P}(\sigma) = (K, \varphi, u, \kappa).$$

A connected component *P* of $\partial \mathcal{P}(\sigma) = (K^{\partial}, \varphi^{\partial}, u^{\partial}, \kappa^{\partial})$ is nothing but a principal face of $(\Delta^{n})_{i} \subset K$ for some i = i(P) corresponding to the complement of a singleton $a_{P} \in \{0, ..., n\}$. By the definition of u^{∂} , we have

$$u^{o}(P) = k_{i(P)}.$$

We compute red₊($\partial \mathcal{P}(\sigma)$) as described in Section 3.2.1. Pick a representative *P* for each type of connected components of K^{∂} , and let $K^{\partial}_{+} \subset K^{\partial}$ be the union of these representatives. Restricting φ^{∂} and κ^{∂} to K^{∂}_{+} we obtain a partition φ^{∂}_{+} on K^{∂}_{+} and a compatible map $\kappa^{\partial}_{+} : K^{\partial}_{+} \to X$. The weight u^{∂}_{+} on K^{∂}_{+} is evaluated on each component *P* of K^{∂}_{+} by

$$u_{+}^{\partial}(P) = \sum_{Q} \deg(\varphi_{P,Q}) u^{\partial}(Q) = \sum_{Q} (-1)^{a_{P} + a_{Q}} k_{i(Q)} = (-1)^{a_{P}} \sum_{Q} (-1)^{a_{Q}} k_{i(Q)}$$

where *Q* runs over all components of K^{∂} of the same type as *P*. Note that the total coefficient of the singular simplex $\sigma_{i(P)}e_{\widehat{a_P}} : \Delta^{n-1} \to X$ in $\partial \sigma$ is

$$\sum_Q (-1)^{a_Q} k_{i(Q)}$$

Also, $(-1)^{a_P}$ is the degree of $e_{\widehat{a_P}} : \Delta^{n-1} \to P$ (recall that Δ^{n-1} is oriented as in Section 3.3.1 while $P \subset \partial(\Delta^n)_i$ inherits orientation from $(\Delta^n)_i$ where i = i(P)). We conclude that the polychain red₊ $(\partial \mathcal{P}(\sigma))$ consists of $\mathcal{P}(\partial \sigma)$ and eventually several connected components of weight zero. Hence

$$\partial^r \mathfrak{P}(\sigma) = \operatorname{red}(\partial \mathfrak{P}(\sigma)) = \operatorname{red}_0 \operatorname{red}_+ (\partial \mathfrak{P}(\sigma)) \cong \mathfrak{P}(\partial \sigma).$$

This proves (3.3.4). The first assertion of the lemma follows.

Next we claim that, for any singular *n*-cycles
$$\sigma$$
, τ in *X*,

(3.3.5) $\mathcal{P}(\sigma + \tau) \simeq \mathcal{P}(\sigma) \sqcup \mathcal{P}(\tau).$

To see this, pick expansions $\sigma = \sum_i k_i \sigma_i$, $\tau = \sum_j l_j \tau_j$ and let

$$\mathcal{K} = (K, \varphi, u, \kappa) \quad \mathcal{L} = (L, \psi, v, \lambda)$$

be the associated polychains, respectively. Consider the cylinder polychain

$$\overline{\mathcal{K} \sqcup \mathcal{L}} \cong \overline{\mathcal{K}} \sqcup \overline{\mathcal{L}}$$

(as defined in the proof of Lemma 3.2.2) and modify its partition by additionally declaring that for any face *F* of $(\Delta^n)_i \subset K$ corresponding to $A \subset \{0, \ldots, n\}$ and for any face *G* of $(\Delta^n)_j \subset L$ corresponding to $B \subset \{0, \ldots, n\}$ such that $\sigma_i e_A = \tau_j e_B$, the faces $F \times \{0\}$ and $G \times \{0\}$ of $\overline{\mathcal{K} \sqcup \mathcal{L}}$ are of the same type, and the corresponding identification map is $e_B e_A^{-1} \times \operatorname{id}_{\{0\}} : F \times \{0\} \to G \times \{0\}$. The resulting (n + 1)-polychain, \mathcal{M} , in *X* satisfies

 $\operatorname{red}_+ \partial \mathcal{M} \cong \operatorname{red}_+(\mathcal{K}) \sqcup \operatorname{red}_+(\mathcal{L}) \sqcup \operatorname{red}_+(-\mathcal{R}) \sqcup (a \text{ polychain with zero weight})$

where \Re is the polychain associated with the expansion $\sum_i k_i \sigma_i + \sum_j l_j \tau_j$ of $\sigma + \tau$. Hence,

$$\partial^r \mathcal{M} \cong \mathcal{P}(\sigma) \sqcup \mathcal{P}(\tau) \sqcup (-\mathcal{P}(\sigma + \tau))$$

and our claim follows.

If \mathbb{K} has no zero-divisors, then $\mathcal{P}(k\sigma) \cong k\mathcal{P}(\sigma)$ for any singular *n*-chain σ in *X* and any non-zero $k \in \mathbb{K}$. For an arbitrary \mathbb{K} and all $k \in \mathbb{K}$, we have

(3.3.6)
$$\mathcal{P}(k\sigma) \cong \operatorname{red}(k\mathcal{P}(\sigma)) \simeq k\mathcal{P}(\sigma).$$

Equalities (3.3.4)–(3.3.6) imply that the formula $[\sigma] \mapsto \langle \mathcal{P}(\sigma) \rangle$ defines a linear map

$$\langle - \rangle : H_n(X) \longrightarrow \widetilde{H}_n(X).$$

To prove the last claim of the lemma, consider a continuous map $f : X \to Y$. Let σ be a singular *n*-cycle in X, and let $\mathcal{K} = (K, \varphi, u, \kappa)$ be the *n*-polycycle associated to an expansion $\sum_i k_i \sigma_i$ of σ . The *n*-polycycle associated to the expansion $\sum_i k_i (f \sigma_i)$ of $f_*(\sigma)$ has the form $\mathcal{K}' = (K, \varphi', u, f \kappa)$ and differs from $f_*(\mathcal{K}) = (K, \varphi, u, f \kappa)$ only in the partition. Modifying appropriately the partition of the cylinder polychain $\overline{f_*(\mathcal{K})}$, we obtain an (n + 1)-polychain \mathcal{M} in X such that

 $\operatorname{red}_+ \partial \mathcal{M} = \operatorname{red}_+ f_*(\mathcal{K}) \sqcup \operatorname{red}_+(-\mathcal{K}') \sqcup (a \text{ polychain with zero weight}).$

We deduce that $\partial^r \mathcal{M} = \operatorname{red} f_*(\mathcal{K}) \sqcup \operatorname{red}(-\mathcal{K}')$ and

$$f_*(\langle \mathcal{P}(\sigma) \rangle) = f_*(\langle \mathcal{K} \rangle) = \langle f_*(\mathcal{K}) \rangle = \langle \mathcal{K}' \rangle = \langle \mathcal{P}(f_*(\sigma)) \rangle.$$

The next theorem implies that $H_n(X)$ is canonically isomorphic to a direct summand of $\widetilde{H}_n(X)$.

Тнеокем 3.3.4. — *We have*

$$[-] \circ \langle - \rangle = \mathrm{id} : H_n(X) \longrightarrow H_n(X).$$

Proof. — Let $\sigma = \sum_i k_i \sigma_i$ be a singular *n*-cycle in *X* and let $\mathcal{P}(\sigma) = (K, \varphi, u, \kappa)$ be the corresponding reduced *n*-polycycle. Then

$$\left[\langle [\sigma] \rangle\right] = \left[\mathcal{P}(\sigma)\right] = (\kappa_{\varphi})_* \left([K_{\varphi}, u]\right) = \left[\sum_i k_i \sigma_i\right] = [\sigma] \in H_n(X).$$

Here the third equality is obtained by considering the tautological locally ordered smooth triangulation *T* of *K* and the corresponding fundamental *n*-chain $\sigma(T, u)$ (see the paragraph preceding Lemma 3.3.2).

3.3.4. *Cross product re-examined.* — The following lemma shows that the transformation $[-] : \widetilde{H}_* \to H_*$ carries the cross product × in face homology to the standard cross product × in singular homology.

LEMMA 3.3.5. — For any topological spaces *X*, *Y*, the following diagram commutes:

 $(3.3.7) \qquad \begin{array}{c} \widetilde{H}_*(X) \times \widetilde{H}_*(Y) & \xrightarrow{\times} & \widetilde{H}_*(X \times Y) \\ & & & & \downarrow^{[-]\times[-]} \\ & & & \downarrow^{[-]} \\ & & & H_*(X) \times H_*(Y) & \xrightarrow{\times} & H_*(X \times Y). \end{array}$

We first recall the definition of the map $\times : H_*(X) \times H_*(Y) \to H_*(X \times Y)$ and then prove Lemma 3.3.5. Fix integers $p, q \ge 0$. Any *p*-element subset *S* of $\{1, \ldots, p + q\}$ determines non-decreasing maps

 $\alpha = \alpha_S : \{0, \dots, p+q\} \rightarrow \{0, \dots, p\}$ and $\beta = \beta_S : \{0, \dots, p+q\} \rightarrow \{0, \dots, q\}$ such that $\alpha(0) = \beta(0) = 0$ and for any $i = 1, \dots, p+q$,

$$(\alpha(i),\beta(i)) = \begin{cases} (\alpha(i-1)+1,\beta(i-1)) & \text{if } i \in S, \\ (\alpha(i-1),\beta(i-1)+1) & \text{if } i \notin S. \end{cases}$$

Let $\omega_S \subset \Delta^p \times \Delta^q$ be the convex hull of the set $\{(e_{\alpha(0)}, e_{\beta(0)}), \dots, (e_{\alpha(p+q)}, e_{\beta(p+q)})\}$.

LEMMA 3.3.6. — The set ω_S is an embedded (p+q)-simplex in $\Delta^p \times \Delta^q$ with vertices

$$\left\{\left(e_{\alpha(i)},e_{\beta(i)}\right)\right\}_{i=0}^{p+q}.$$

The simplices $\{\omega_S\}_S$ *and their faces form a triangulation of* $\Delta^p \times \Delta^q$ *.*

Proof. — This lemma is well known but we give a proof for completeness. For $n \ge 0$, denote by \mathbb{A}^n the affine space formed by the points of \mathbb{R}^{n+1} with sum of coordinates 1. The basis (e_0, \ldots, e_n) of \mathbb{R}^{n+1} is an affine basis of \mathbb{A}^n and $\Delta^n \subset \mathbb{A}^n$. Consider the basis $(v_1, \ldots, v_p) = (\overrightarrow{e_0e_1}, \overrightarrow{e_1e_2}, \ldots, \overrightarrow{e_{p-1}e_p})$ of the vector space underlying \mathbb{A}^p and the basis $(v_{p+1}, \ldots, v_{p+q}) = (\overrightarrow{e_0e_1}, \overrightarrow{e_1e_2}, \ldots, \overrightarrow{e_{q-1}e_q})$ of the vector space underlying \mathbb{A}^q . Then (v_1, \ldots, v_{p+q}) is a basis of the vector space underlying the product affine space $\mathbb{A}^p \times \mathbb{A}^q$.

Recall that a (p, q)-shuffle is a permutation *s* of $\{1, \ldots, p+q\}$ such that

$$s(1) < s(2) < \dots < s(p)$$
 and $s(p+1) < \dots < s(p+q)$.

Any *p*-element subset *S* of $\{1, ..., p+q\}$ determines a unique (p, q)-shuffle *s* such that

$$S = s(\{1,\ldots,p\}),$$

The first claim of the lemma follows from the fact that the basis $(v_{s^{-1}(1)}, \ldots, v_{s^{-1}(p+q)})$ underlies the set of vertices of $\omega_S \subset \mathbb{A}^p \times \mathbb{A}^q$:

$$(3.3.8) \quad (e_{\alpha(0)}, e_{\beta(0)}) \xrightarrow{\nu_{s^{-1}(1)}} (e_{\alpha(1)}, e_{\beta(1)}) \xrightarrow{\nu_{s^{-1}(2)}} \cdots \xrightarrow{\nu_{s^{-1}(p+q)}} (e_{\alpha(p+q)}, e_{\beta(p+q)}).$$

To prove the second claim, observe that given n + 1 affinely independent points f_0, \ldots, f_n in an *n*-dimensional affine space, an arbitrary point $f_0 + \sum_{i=1}^n t_i \overrightarrow{f_{i-1}f_i}$ of this space (with $t_1, \ldots, t_n \in \mathbb{R}$) belongs to the affine simplex spanned by f_0, \ldots, f_n if and only if $1 \ge t_1 \ge \cdots \ge t_n \ge 0$. Therefore any point $z \in \Delta^p \times \Delta^q$ expands uniquely as

$$z = (e_0, e_0) + z_1 v_1 + \dots + z_{p+q} v_{p+q}$$

with

$$1 \ge z_1 \ge \cdots \ge z_p \ge 0$$
 and $1 \ge z_{p+1} \ge \cdots \ge z_{p+q} \ge 0$.

By the same observation and (3.3.8), the inclusion $z \in \omega_S$ for a *p*-element subset *S* of $\{1, \ldots, p+q\}$ holds if and only if

$$z_{s^{-1}(1)} \ge z_{s^{-1}(2)} \ge \cdots \ge z_{s^{-1}(p+q)}$$

where *s* is the (p,q)-shuffle determined by *S*. Therefore $\Delta^p \times \Delta^q$ is the union of the simplices $\{\omega_S\}_S$, and any two of these simplices meet along a common face.

For each *p*-element subset $S \subset \{1, ..., p + q\}$, we turn ω_S into a singular simplex in $\Delta^p \times \Delta^q$ by sending the ordered vertices $e_0 < e_1 < \cdots < e_{p+q}$ of Δ^{p+q} to

$$(3.3.9) (e_{\alpha(0)}, e_{\beta(0)}) < (e_{\alpha(1)}, e_{\beta(1)}) < \dots < (e_{\alpha(p+q)}, e_{\beta(p+q)})$$

respectively. Summing up over all such S we obtain a singular chain

(3.3.10)
$$\omega_{p,q} = \sum_{S} \varepsilon_{S} \omega_{S} \in C_{p+q}(\Delta^{p} \times \Delta^{q})$$

where ε_S is the sign comparing the orientation in ω_S determined by the order of its vertices (3.3.9) with the product orientation in $\Delta^p \times \Delta^q$. The *Eilenberg-Zilber chain map*

$$(3.3.11) \qquad \qquad EZ: C_*(X) \otimes C_*(Y) \longrightarrow C_*(X \times Y)$$

is defined by

$$EZ(\sigma \otimes \tau) = (\sigma \times \tau)_*(\omega_{p,q})$$

for any singular simplices $\sigma : \Delta^p \to X$ and $\tau : \Delta^q \to Y$. Here

$$(\sigma \times \tau)_* \colon C_*(\Delta^p \times \Delta^q) \longrightarrow C_*(X \times Y)$$

is the chain map induced by $\sigma \times \tau : \Delta^p \times \Delta^q \to X \times Y$.

The *cross product* of singular homology classes $x \in H_p(X)$ and $y \in H_q(Y)$ is defined by taking any cycles $\sigma \in C_p(X)$ and $\tau \in C_q(Y)$ representing x and y respectively, and letting $x \times y \in H_{p+q}(X \times Y)$ be the homology class of $EZ(\sigma \otimes \tau)$.

Proof of Lemma 3.3.5. — Let $\mathcal{K} = (K, \varphi, u, \kappa)$ be a *p*-polycycle in *X* and $\mathcal{L} = (L, \psi, v, \lambda)$ be a *q*-polycycle in *Y*. We must prove that

$$(3.3.12) \qquad \qquad [\mathcal{K} \times \mathcal{L}] = [\mathcal{K}] \times [\mathcal{L}] \in H_{p+q}(X \times Y).$$

Fix a locally ordered smooth triangulation *T* of *K* which fits φ and consider the fundamental *p*-chain $\sum_i \varepsilon_i u(K^i) \sigma_i \in C_p(K)$ where *i* runs over *p*-simplices of *T*, $\sigma_i : \Delta^p \to K$ is the smooth singular simplex determined by *i*, ε_i is the sign comparing the orientation induced by the order of the vertices of *i* to the orientation of *K*, and K^i is the connected component of *K* containing *i*. Then

$$[\mathcal{K}] = \left[\sum_{i} \varepsilon_{i} u(K^{i}) (\kappa \circ \sigma_{i})\right] \in H_{p}(X)$$

where the square brackets on the right-hand side stand for the homology class of a singular cycle. Similarly, fixing a locally ordered smooth triangulation W of L which fits ψ , we obtain

$$[\mathcal{L}] = \left[\sum_{j} \varepsilon_{j} v(L^{j}) (\lambda \circ \tau_{j})\right] \in H_{q}(Y)$$

where *j* runs over *q*-simplices of *W*, $\tau_j : \Delta^q \to L$ is the smooth singular simplex determined by *j*, ε_j is the sign comparing the orientation induced by the order of the vertices of *j* to the orientation of *L*, and L^j is the component of *L* containing *j*. By the definition of the cross product in singular homology,

(3.3.13)
$$[\mathcal{K}] \times [\mathcal{L}] = \left[\sum_{i,j} \varepsilon_i \varepsilon_j u(K^i) v(L^j) EZ(\kappa \sigma_i \otimes \lambda \tau_j) \right]$$
$$= \left[\sum_{i,j,S} \varepsilon_i \varepsilon_j \varepsilon_S u(K^i) v(L^j) (\kappa \sigma_i \times \lambda \tau_j) \omega_S \right]$$

where *S* runs over *p*-element subsets of $\{1, \ldots, p+q\}$.

For any simplices *i*, *j* as above, we push forward via $\sigma_i \times \tau_j$ the triangulation of $\Delta^p \times \Delta^q$ provided by Lemma 3.3.6 to a triangulation of $i \times j \subset K \times L$. This gives a smooth triangulation *Z* of $K \times L$. The set of vertices Z^0 of *Z* is the cartesian product of the sets of vertices T^0 and W^0 of *T* and *W*, respectively; we endow Z^0 with the product of the binary relations on T^0 and W^0 determined by the local orders on *T* and *W*. This defines a local order on *Z* which fits the partition $\varphi \times \psi$.

The simplices of *Z* can be identified with the triples (i, j, S) as above, and the corresponding singular simplices in $K \times L$ are the maps $(\sigma_i \times \tau_j)\omega_S : \Delta^{p+q} \to K \times L$. (Here we use the fact that the order (3.3.9) of the vertices of ω_S is given by the product binary relation on the set of vertices of $\Delta^p \times \Delta^q$ determined by the natural orders on the sets of vertices of Δ^p and Δ^q .) Let $\varepsilon_{i,j,S}$ be the sign comparing the orientation of the simplex (i, j, S) induced by the order of the vertices to the product orientation of $K \times L$. Let $(K \times L)^{i,j,S}$ be the component of $K \times L$ containing the simplex (i, j, S). Then

$$(3.3.14) \qquad [\mathcal{K} \times \mathcal{L}] = \Big[\sum_{i,j,S} \varepsilon_{i,j,S} \cdot (u \times v) \big((K \times L)^{i,j,S} \big) \cdot \big((\kappa \times \lambda) \circ (\sigma_i \times \tau_j) \omega_S \big) \Big].$$

Clearly,

$$(u \times v)((K \times L)^{i,j,S}) = u(K^i)v(L^j).$$

Note that $\varepsilon_{i,j,S} = \varepsilon_i \varepsilon_j \varepsilon_S$ since $\varepsilon_i \varepsilon_j$ is the degree of the diffeomorphism

$$\sigma_i \times \tau_i : \Delta^p \times \Delta^q \longrightarrow i \times j$$

with respect to the product orientation in $\Delta^p \times \Delta^q$ and the product orientation in $K \times L$ restricted to $i \times j$. Comparing (3.3.13) to (3.3.14), we obtain (3.3.12).

3.3.5. Remarks

(1) Though we shall not need it in the sequel, note that the sign ε_S in (3.3.10) can be computed explicitly:

$$\varepsilon_S = (-1)^{n_S}$$

where n_S is the number of pairs i < j with $i \in \{1, ..., p+q\} \setminus S$ and $j \in S$. Indeed, in the notation introduced in the proof of Lemma 3.3.6, the orientation of ω_S determined by the sequence (3.3.9) is represented by the (p+q)-vector

$$v_{s^{-1}(1)} \wedge \dots \wedge v_{s^{-1}(p+q)} = (-1)^m v_1 \wedge \dots \wedge v_{p+q}$$

where *s* is the (p, q)-shuffle associated with *S* and *m* is the number of inversions in *s*. Therefore $\varepsilon_S = (-1)^m$ and it remains to observe that $m = n_S$.

The definition of n_S may be also reformulated in terms of the maps $\alpha = \alpha_S$ and $\beta = \beta_S$. Namely, n_S is the number of pairs i < j such that

$$\beta(i) = \beta(i-1) + 1$$
 and $\alpha(j) = \alpha(j-1) + 1$.

This implies the following formula for n_S used, for example, in [15, Section 4 (b)]:

$$n_{S} = \sum_{1 \leq i < j \leq p+q} \left(\beta(i) - \beta(i-1) \right) \left(\alpha(j) - \alpha(j-1) \right).$$

(2) By a celebrated result of Thom, there are topological spaces *X* and integers n > 0 such that some *n*-dimensional singular homology classes of *X* with coefficients in $\mathbb{K} = \mathbb{Z}$ are not realizable by closed singular manifolds. For such *X* and *n*, the natural map from the *n*-dimensional oriented bordism group $\Omega_n(X)$ to $H_n(X)$, carrying a closed singular manifold

$$\kappa: M \longrightarrow X$$

to $\kappa_*([M])$, is not surjective. This map splits as a composition of the map

$$\Omega_n(X) \longrightarrow \widetilde{H}_n(X)$$

described in Remark 3.2.7 with the surjective map $[-] : \widetilde{H}_n(X) \to H_n(X)$. Therefore, for such *X* and *n*, the map $\Omega_n(X) \to \widetilde{H}_n(X)$ is not surjective.

(3) The face homology seems to be difficult to compute. As a consequence, the authors do not know whether the transformation $[-] : \widetilde{H}_* \to H_*$ is injective, and, equivalently, whether $\langle - \rangle : H_* \to \widetilde{H}_*$ is surjective. In fact, the authors even do not know whether the face homology of a point is trivial in positive degrees.

(4) The constructions and results of this section easily extend to the face homology of topological pairs (*cf.* Remark 3.2.7).

3.4. Smooth polychains

We reformulate face homology of the path spaces of manifolds in terms of smooth polychains. We start by studying polychains in manifolds.

3.4.1. *Polychains in manifolds.* — Recall from Section 3.1.1 that a map κ from an *n*-dimensional manifold with faces *K* to a smooth *m*-dimensional manifold *M* (possibly, with boundary) is said to be smooth if restricting κ to any local coordinate systems in *K* and *M* we obtain a map that extends to a C^{∞} -map from an open subset of \mathbb{R}^n to \mathbb{R}^m . If $N \subset K$ is a union of (some) faces of *K*, then we call a map $N \to M$ smooth whenever

its restrictions to all faces of *K* contained in *N* are smooth. Such a map $N \rightarrow M$ is necessarily continuous. This terminology applies in particular to $N = \partial K$.

LEMMA 3.4.1. — Let *K* be a manifold with faces. Any smooth map $\partial K \to M$ extends to a smooth map from a neighborhood of ∂K in *K* to *M*.

Proof. — Using a partition of unity on *K* and local coordinates on *M*, we easily reduce the lemma to the case where $K = \mathbb{R}^n_+ = [0, \infty)^n$ with $n \ge 0$ and $M = \mathbb{R}$. We need to prove that every function

$$f:\partial\mathbb{R}^n_+\longrightarrow\mathbb{R}$$

whose restrictions to all proper faces of \mathbb{R}^n_+ are smooth extends to a smooth function on \mathbb{R}^n_+ . We exhibit one such extension explicitly. For a subset *S* of the set $\{1, ..., n\}$ and a point $x = (x_1, ..., x_n) \in \mathbb{R}^n_+$ denote by x_S the point of \mathbb{R}^n_+ whose *i*-th coordinate is x_i if $i \in S$ and zero otherwise. If $S \neq \{1, ..., n\}$, *i.e.* if *S* is a proper subset of $\{1, ..., n\}$, then $x_S \in \partial \mathbb{R}^n_+$. Set

(3.4.1)
$$\bar{f}(x) = \sum_{S \subseteq \{1,...,n\}} (-1)^{\operatorname{card}(S)+n+1} f(x_S).$$

Each function $x \mapsto f(x_S)$ is smooth because it is a composition of f with the projection of \mathbb{R}^n_+ onto its proper face. Therefore the function $\overline{f} : \mathbb{R}^n_+ \to \mathbb{R}$ is smooth. Moreover, it satisfies

$$\overline{f}(x) = f(x)$$
 for all $x \in \partial \mathbb{R}^n_+$.

Indeed, pick $i \in \{1, ..., n\}$ such that $x_i = 0$ and observe that each term in (3.4.1) corresponding to S with $i \notin S$ cancels with the term corresponding to $S \cup \{i\}$ provided the latter set is proper. This leaves only the term $f(x_S) = f(x)$ determined by $S = \{1, ..., n\} \setminus \{i\}$.

LEMMA 3.4.2. — Let $\kappa : K \to M$ be a continuous map from a manifold with faces K to a smooth manifold M. Then any homotopy of

$$\kappa_{\mid \partial K}: \partial K \longrightarrow M$$

to a smooth map extends to a homotopy of κ to a smooth map $K \to M$.

Proof. — Observe first that if

$$\kappa_{\mid \partial K} : \partial K \longrightarrow M$$

is smooth, then there is a homotopy of κ rel ∂K to a smooth map $K \to M$. Indeed, Lemma 3.4.1 yields an extension of $\kappa|_{\partial K} : \partial K \to M$ to a smooth map $U \to M$ where U is a collar of ∂K in K. The latter map obviously extends to a continuous map $\kappa' : K \to M$ homotopic to κ rel ∂K . Since $\kappa'|_U$ is smooth and $\overline{K \setminus U}$ is a compact subset of the smooth manifold $K \setminus \partial K$, there is a homotopy of κ' to a smooth map $K \to M$, and this homotopy may be chosen to be constant in a neighborhood of ∂K in U. The resulting smooth map $K \to M$ is homotopic to κ rel ∂K .

To prove the lemma, take an arbitrary (continuous) extension of the given homotopy of $\kappa_{|\partial K}$ to a homotopy of κ , and compose it with a homotopy rel ∂K of the resulting map $K \to M$ to a smooth map as in the previous paragraph. Next, using a

collar of ∂K in *K*, deform the composed homotopy of κ into a homotopy satisfying the conditions of the lemma.

As an exercise, the reader may deduce from Lemma 3.4.2 (by an inductive construction on the faces of *K*) that any continuous map $K \rightarrow M$ is homotopic to a smooth map.

A polychain (K, φ, u, κ) in M is *smooth* if the map $\kappa : K \to M$ is smooth. We explain now how to deform arbitrary polychains in M into smooth polychains. For the notion of a deformation of a polychain in M, see Section 3.2.5. We explain first how to extend deformations. Let $N \subset K$ consist of some faces of K. We say that a homotopy

$$\left\{ (\kappa_{|N})^t : N \longrightarrow M \right\}_{t \in I}$$

of $\kappa_{|N}$ is *compatible* with the partition φ if

$$(\kappa_{|N})^t|_G \circ \varphi_{F,G} = (\kappa_{|N})^t|_F$$

for any $t \in I$ and any faces of the same type $F, G \subset N$.

LEMMA 3.4.3. — Let $\mathcal{K} = (K, \varphi, u, \kappa)$ be a polychain in M and let N be a union of faces of K. Let

$$\left\{ (\kappa_{|N})^t : N \to M \right\}_{t \in I}$$

be a homotopy of $\kappa_{|N}$ *compatible with* φ *such that* $(\kappa_{|N})^1 : N \to M$ *is smooth. Then there is a deformation*

$$\left\{ \mathcal{K}^t = (K, \varphi, u, \kappa^t) \right\}_{t \in I}$$

of $\mathcal{K}^0 = \mathcal{K}$ such that \mathcal{K}^1 is smooth and for all $t \in I$,

$$\kappa^t|_N = (\kappa|_N)^t : N \longrightarrow M.$$

Proof. — For any integer *r*, denote (as in Section 3.1.1) by K_r the union of all faces of *K* of dimension ≤ *r*. Recursively in r = -1, 0, ..., we construct a homotopy $(\kappa_{|K_r})^t$ of $\kappa_{|K_r}$ to a smooth map $(\kappa_{|K_r})^1 : K_r \to M$. For r = -1, there is nothing to do since $K_{-1} = \emptyset$. The induction step goes as follows. For each type of *r*-dimensional faces of *K*, select a representative face *F* so that if at least one face of the given type lies in *N*, then $F \subset N$. If $F \subset N$, then set $(\kappa_{|F})^t = (\kappa_{|N})^t|_F$ for all *t*. If $F \nsubseteq N$, then by Lemma 3.4.2 there is a homotopy of $\kappa_{|F}$ to a smooth map extending the homotopy of κ on $\partial F \subset K_{r-1}$ obtained at the previous step. The homotopy on the selected *r*-dimensional faces uniquely extends to a homotopy of κ on K_r compatible with φ . For $r = \dim(K)$, we obtain the required deformation of κ .

LEMMA 3.4.4. — For any polychain $\mathcal{K} = (K, \varphi, u, \kappa)$ in M, there is a deformation

$$\left\{ \mathcal{K}^t = (K, \varphi, u, \kappa^t) \right\}_{t \in I}$$

of $\mathfrak{K}^0 = \mathfrak{K}$ such that \mathfrak{K}^1 is smooth and $\kappa^t|_F = \kappa|_F$ for all $t \in I$ and all faces F of K on which κ is smooth.

Proof. — This is a special case of Lemma 3.4.3 where *N* is the union of all faces of *K* on which κ is smooth and $\{(\kappa_{|N})^t\}_{t \in I}$ is the constant homotopy.

We can now reformulate the face homology of M in terms of smooth polychains. Note that if a polychain \mathcal{K} in M is smooth, then so are the polychains red \mathcal{K} , $\partial \mathcal{K}$, and $\partial^r \mathcal{K}$. Disjoint unions of smooth polychains are smooth. Applying the definitions of Section 3.2.4 to X = M but considering only smooth polycycles and smooth polychains we obtain *smooth face homology* $\widetilde{H}^s_*(M)$.

THEOREM 3.4.5. — The natural linear map $\widetilde{H}^s_*(M) \to \widetilde{H}_*(M)$ is an isomorphism.

Proof. — Lemmas 3.2.3 and 3.4.4 imply that any polycycle in *M* is homologous to a smooth polycycle. This proves the surjectivity of the map in the statement of the theorem. To prove the injectivity it suffices to show that, for any homologous reduced smooth *n*-polychains \mathcal{K}_1 , \mathcal{K}_2 in *M* there are smooth (n + 1)-polychains \mathcal{R}'_1 , \mathcal{R}'_2 in *M* such that

$$\mathcal{K}_1 \sqcup \partial^r \mathcal{R}'_1 \cong \mathcal{K}_2 \sqcup \partial^r \mathcal{R}'_2$$

By assumption, there are (n + 1)-polychains \mathcal{R}_1 , \mathcal{R}_2 in *M* and a diffeomorphism

$$f: \mathcal{K}_1 \sqcup \partial^r \mathcal{R}_1 \longrightarrow \mathcal{K}_2 \sqcup \partial^r \mathcal{R}_2.$$

For i = 1, 2, set

$$(P_i, \varphi_i, u_i, \kappa_i) = \mathcal{K}_i \sqcup \partial^r \mathcal{R}_i$$

and let $K_i \subset P_i$ be the union of the components of P_i underlying \mathcal{K}_i . Lemma 3.4.4 yields a homotopy

$$\{\kappa_1^t: P_1 \to M\}_t$$

of $\kappa_1^0 = \kappa_1$ to a smooth map κ_1^1 in the class of maps compatible with the partition φ_1 , which is constant on all faces of P_1 on which κ_1 is smooth. In particular, this homotopy is constant on K_1 . Consider the homotopy

$$\{\kappa_2^t = \kappa_1^t f^{-1} : P_2 \to M\}_t$$

of $\kappa_2^0 = \kappa_2$. This homotopy is compatible with the partition φ_2 and is constant on K_2 (because $\kappa_1 = \kappa_2 f$ is smooth on $f^{-1}(K_2)$). Clearly, $f : P_1 \to P_2$ is a diffeomorphism of the smooth polychains

$$(P_1, \varphi_1, u_1, \kappa_1^1)$$
 and $(P_2, \varphi_2, u_2, \kappa_2^1 = \kappa_1^1 f^{-1}).$

For i = 1, 2, the polychain $(P_i, \varphi_i, u_i, \kappa_i^1)$ is obtained from $\mathcal{K}_i \sqcup \partial^r \mathcal{R}_i$ by a deformation which is constant on \mathcal{K}_i and transforms $\partial^r \mathcal{R}_i$ into a smooth polychain.

To finish the proof, we need only to show that this deformation of $\partial^r \mathfrak{R}_i$ extends to a deformation of \mathfrak{R}_i into a smooth polychain. This is done in three steps. First of all, applying Lemma 3.4.3 to $\mathfrak{K} = \operatorname{red}_+ \partial \mathfrak{R}_i$ and taking for *N* the union of all connected components of non-zero weight we obtain that our deformation of $\partial^r \mathfrak{R}_i = \operatorname{red}_0 \operatorname{red}_+ \partial \mathfrak{R}_i$ extends to a deformation of $\operatorname{red}_+ \partial \mathfrak{R}_i$ into a smooth polychain. The latter deformation induces a deformation of $\partial \mathfrak{R}_i$ into a smooth polychain. One more application of Lemma 3.4.3 allows us to extend the latter deformation to a deformation of \mathfrak{R}_i into a smooth polychain \mathfrak{R}_i . Then $\mathfrak{K}_1 \sqcup \partial^r \mathfrak{R}_1' \cong \mathfrak{K}_2 \sqcup \partial^r \mathfrak{R}_2'$.

3.4.2. *Polychains in path spaces.* — Pick two points \star, \star' in a smooth manifold *M* (possibly, $\star = \star'$ and $\partial M \neq \emptyset$). A *path* in *M* from \star to \star' is a continuous map $I = [0, 1] \rightarrow M$ carrying 0 to \star and 1 to \star' . Let

$$\Omega = \Omega(M, \star, \star')$$

be the space of such paths with compact-open topology; we call Ω the path space of M. Note that a map σ from a topological space K to Ω is continuous if and only if the adjoint map $\tilde{\sigma} : K \times I \to M$, carrying any pair $(k \in K, s \in I)$ to $\sigma(k)(s) \in M$ is continuous; see, for example, [18, Section 1.2.7].

Given a subspace *X* of Ω , we call a map from a manifold with faces *K* to *X* smooth if the adjoint map $K \times I \to M$ is smooth in the sense of Section 3.4.1. A polychain $\mathcal{K} = (K, \varphi, u, \kappa)$ in *X* is smooth if $\kappa : K \to X$ is smooth. The definitions of Section 3.2.4 restricted to smooth polycycles and smooth polychains in *X*, yield the smooth face homology $\widetilde{H}^s_*(X)$ of *X*. In the next theorem, $X = \Omega$.

THEOREM 3.4.6. — The natural linear map $\widetilde{H}^s_*(\Omega) \to \widetilde{H}_*(\Omega)$ is an isomorphism.

Proof. — We follow the lines of Section 3.4.1 with *M* replaced by Ω . First, we show that given a manifold with faces *K*, any smooth map

$$f:\partial K\longrightarrow \Omega$$

extends to a smooth map from a neighborhood of ∂K in K to Ω . Indeed, the adjoint map $\tilde{f} : \partial K \times I \to M$ extends to a smooth map $\partial (K \times I) \to M$ by sending $K \times \{0\}$ to \star and $K \times \{1\}$ to \star' . By Lemma 3.4.1, the latter map extends to a smooth map

$$\bar{f}: U \longrightarrow M$$

for some neighborhood *U* of $\partial(K \times I)$ in $K \times I$. Clearly, $U \supset V \times I$ for a neighborhood *V* of ∂K in *K*. The map $\overline{f}_{|V \times I}$ is adjoint to a smooth map $V \to \Omega$ extending *f*.

Lemmas 3.4.2–3.4.4 remain true with *M* replaced by Ω . The proofs above apply with the only difference that Lemma 3.4.1 should be replaced by the result of the previous paragraph. The proof of Theorem 3.4.5 also works with *M* replaced by Ω . This gives the desired result.

In the case where $\star, \star' \in \partial M$, the path space $\Omega = \Omega(M, \star, \star')$ is homotopy equivalent to a smaller space. Let

$$\Omega^{\circ} = \Omega^{\circ}(M, \star, \star')$$

be the subspace of Ω consisting of all paths $\alpha : I \to M$ from \star to \star' such that $\alpha^{-1}(\partial M) = \partial I$. We call Ω° *the proper path space* of (M, \star, \star') .

LEMMA 3.4.7. — The inclusion map $\Omega^{\circ} \hookrightarrow \Omega$ is a homotopy equivalence.

Proof. — We begin with an observation in set-theoretic topology. Consider a topological pair $Y \subset X$ and suppose that there is a homotopy

$${f_t : X \to X}_{t \in I}$$

of the identity map $f_0 = id_X$ such that $f_t(X) \subset Y$ for all t > 0. Then the inclusion $\iota : Y \hookrightarrow X$ is a homotopy equivalence and its homotopy inverse $g : X \to Y$ is obtained

from f_1 by reducing the image to Y. Indeed, the family $\{f_t : X \to X\}_t$ is a homotopy between $f_0 = id_X$ and $f_1 = \iota g$. Since $f_t \iota(Y) \subset Y$ for all $t \in I$, we have the family of maps $\{f_t \iota : Y \to Y\}_t$. This is a homotopy between id_Y and $g\iota$.

Using a tubular neighborhood of ∂M in M, we can easily construct a (smooth) family of embeddings

 $\{F_{s,t}: M \hookrightarrow M\}_{s,t \in I}$

such that $F_{s,t} = id_M$ if $s \in \{0, 1\}$ or t = 0, and $F_{s,t}(M) \subset Int M = M \setminus \partial M$ for all other pairs (s, t). Given $t \in I$ and a path $\alpha : I \to M$ from \star to \star' , we define a path $\alpha_t : I \to M$ by $\alpha_t(s) = F_{s,t}(\alpha(s))$ for all $s \in I$. This gives a family of paths $\{\alpha_t\}_{t \in I}$ such that $\alpha_0 = \alpha$ and $\alpha_t \in \Omega^\circ$ for all t > 0. The formula $f_t(\alpha) = \alpha_t$ defines a homotopy $\{f_t : \Omega \to \Omega\}_{t \in I}$ of $f_0 = id_\Omega$ such that $f_t(\Omega) \subset \Omega^\circ$ for all t > 0. Now, the result of the previous paragraph implies that the inclusion $\Omega^\circ \hookrightarrow \Omega$ is a homotopy equivalence.

Lemma 3.4.7 implies that $\widetilde{H}_*(\Omega^\circ) \simeq \widetilde{H}_*(\Omega)$. The following theorem computes the face homology of Ω° and Ω in terms of smooth polychains in Ω° .

THEOREM 3.4.8. — The natural linear map $\widetilde{H}^{s}_{*}(\Omega^{\circ}) \to \widetilde{H}_{*}(\Omega^{\circ})$ is an isomorphism.

Proof. — Consider the homotopy $\{f_t\}_{t \in I}$ of $f_0 = id_\Omega$ introduced in the proof of Lemma 3.4.7 and set $f = f_1 : \Omega \to \Omega^\circ \subset \Omega$. For any manifold with faces *L* and for any map $\lambda : L \to \Omega$, the adjoint map $\tilde{f_t}\lambda : L \times I \to M$ of $f_t\lambda$ is given by

$$\widetilde{f_t\lambda}(l,s) = f_t(\lambda(l))(s) = \lambda(l)_t(s) = F_{s,t}(\lambda(l)(s)) = F_{s,t}(\widetilde{\lambda}(l,s)).$$

Consequently, if λ is smooth, then $f_t \lambda : L \to \Omega$ is smooth for all $t \in I$. We conclude that for any smooth polychain \mathcal{L} in Ω , the polychain $f_*(\mathcal{L})$ in Ω° is smooth.

By the surjectivity of the map in Theorem 3.4.6, any polycycle \mathcal{K} in Ω° is homologous in Ω to a smooth polycycle \mathcal{K}' in Ω . Applying f, we obtain that $f_*(\mathcal{K})$ is homologous in Ω° to the smooth polycycle $f_*(\mathcal{K}')$. The homotopy $\{f_t\}_t$ induces a deformation of \mathcal{K} into $f_*(\mathcal{K})$ in Ω° . Therefore \mathcal{K} is homologous to $f_*(\mathcal{K})$ in Ω° . Thus, \mathcal{K} is homologous to $f_*(\mathcal{K}')$ in Ω° . This proves the surjectivity of the natural map $\widetilde{H}^s_*(\Omega^{\circ}) \to \widetilde{H}_*(\Omega^{\circ})$. To prove the injectivity, consider two reduced smooth *n*-polycycles $\mathcal{K}_1, \mathcal{K}_2$ in Ω° that are homologous in Ω° . Then they are homologous in Ω . By the injectivity of the map in Theorem 3.4.6, there are smooth (n + 1)-polychains $\mathcal{L}_1, \mathcal{L}_2$ in Ω such that $\mathcal{K}_1 \sqcup \partial^r \mathcal{L}_1 \cong \mathcal{K}_2 \sqcup \partial^r \mathcal{L}_2$. Applying f we obtain

$$f_*(\mathcal{K}_1) \sqcup \partial^r (f_*(\mathcal{L}_1)) \cong f_*(\mathcal{K}_2) \sqcup \partial^r (f_*(\mathcal{L}_2))$$

where $f_*(\mathcal{K}_1), f_*(\mathcal{L}_1), f_*(\mathcal{K}_2), f_*(\mathcal{L}_2)$ are smooth polychains in Ω° . So,

$$\langle f_*(\mathcal{K}_1) \rangle = \langle f_*(\mathcal{K}_2) \rangle \in H^s_n(\Omega^\circ).$$

The homotopy $\{f_t\}_t$ induces a smooth deformation of \mathcal{K}_i into $f_*(\mathcal{K}_i)$ and therefore $\langle \mathcal{K}_i \rangle = \langle f_*(\mathcal{K}_i) \rangle \in \widetilde{H}_n^s(\Omega^\circ)$ for i = 1, 2. Hence $\langle \mathcal{K}_1 \rangle = \langle \mathcal{K}_2 \rangle \in \widetilde{H}_n^s(\Omega^\circ)$. This completes the proof of the injectivity of the natural map $\widetilde{H}_*^s(\Omega^\circ) \to \widetilde{H}_*(\Omega^\circ)$ and of the theorem.

CHAPTER 4

OPERATIONS ON POLYCHAINS

Throughout this chapter, *M* is an oriented smooth *n*-dimensional manifold with boundary, where $n \ge 2$. We fix points $\star_1, \star_2, \star_3, \star_4 \in \partial M$ and assume, unless explicitly stated to the contrary, that $\{\star_1, \star_2\} \cap \{\star_3, \star_4\} = \emptyset$ (possibly, $\star_1 = \star_2$ and/or $\star_3 = \star_4$). For $i, j \in \{1, 2, 3, 4\}$, let

- ▶ $\Omega_{ij} = \Omega(M, \star_i, \star_j)$ be the path space and
- ▶ $\Omega_{ij}^{\circ} = \Omega^{\circ}(M, \star_i, \star_j) \subset \Omega_{ij}$ be the proper path space of (M, \star_i, \star_j) .

4.1. Transversality in path spaces

In this section, we study transversality of polychains in the proper path spaces Ω_{12}° and Ω_{34}° .

4.1.1. Transversal maps. — The diagonal of M

$$\operatorname{diag}_{M} = \{(x, x) | x \in M\} \subset M \times M$$

is a smooth manifold diffeomorphic to M. We say that a smooth map g from a manifold with faces N to $M \times M$ is *weakly transversal to* diag_M if g(N) does not meet ∂ (diag_M) and the restriction of g to Int(N) = $N \setminus \partial N$ is transversal to Int(diag_M) in the usual sense of differential topology. (The interiors of N and diag_M are smooth manifolds so this condition makes sense.) The map g is *transversal to* diag_M if its restriction to any face of N is weakly transversal to diag_M.

Fix manifolds with faces *K* and *L*. Consider smooth maps

$$\kappa: K \longrightarrow \Omega_{12}^{\circ}, \quad \lambda: L \longrightarrow \Omega_{34}^{\circ}$$

and let $\tilde{\kappa} : K \times I \to M$, $\tilde{\lambda} : L \times I \to M$ be the adjoint maps. The product map

$$\tilde{\kappa} \times \lambda : K \times I \times L \times I \longrightarrow M \times M$$

carries a tuple $(k \in K, s \in I, l \in L, t \in I)$ to the point $(\kappa(k)(s), \lambda(l)(t))$. The latter point can lie on diag_{*M*} only when $s, t \in Int(I) = (0, 1)$ and never lies in $\partial(\text{diag}_M)$. We say that κ and λ are *transversal* if the map $\tilde{\kappa} \times \tilde{\lambda}$ is transversal to diag_{*M*} in the sense above. Note that the maps κ and λ are transversal in our sense if and only if they are transversal in the sense of [35], see Proposition 7.2.2 therein. (Our notion of transversality is stronger than the one in [26], see Remark 6.3 therein.) If κ and λ are transversal, then their restrictions to arbitrary faces of *K*, *L* are transversal. Clearly, smooth homotopies of κ and λ that are sufficiently *C*¹-small preserve transversality.

LEMMA 4.1.1. — For any smooth maps $\kappa : K \to \Omega_{12}^{\circ}$ and $\lambda : L \to \Omega_{34}^{\circ}$, there is an arbitrarily C^{∞} -small smooth homotopy $\{\kappa^t\}_{t \in I}$ of $\kappa^0 = \kappa$ such that κ^1 and λ are transversal.

Proof. — Proceeding by induction on $\dim(L) \ge -1$, we can assume that κ is transversal to the restrictions of λ to all proper faces of L. All subsequent homotopies of κ are chosen to be small enough to preserve this property. Fix a smooth triangulation T of $K \times I$. A map from a simplex e of T to M is *smooth* if it is smooth as a singular simplex in M. A smooth map $f : e \to M$ is λ -transversal if the map $f \times \tilde{\lambda} : e \times L \times I \to M \times M$ is weakly transversal to diag_M.

We call a map $g : K \times I \to M$ good if the adjoint map from K to the space of paths in M takes values in $\Omega_{12}^{\circ} = \Omega^{\circ}(M, \star_1, \star_2)$. The map g is *T*-good if it is good and the image of any simplex of T under g lies in a closed ball in M.

Consider the map $\tilde{\kappa} : K \times I \to M$ adjoint to κ . Clearly, $\tilde{\kappa}$ is good. Since

$$\{\star_1, \star_2\} \cap \{\star_3, \star_4\} = \emptyset,$$

the set $\tilde{\kappa}(K \times \partial I)$ is disjoint from $\tilde{\lambda}(L \times I)$. By continuity, there is a small $\delta > 0$ such that the sets $\tilde{\kappa}(K \times [0, \delta])$ and $\tilde{\kappa}(K \times [1 - \delta, 1])$ are disjoint from $\tilde{\lambda}(L \times I)$. Subdividing *T*, we can assume that $\tilde{\kappa}$ is *T*-good and $T_{\delta} = K \times ([0, \delta] \cup [1 - \delta, 1])$ is a subcomplex of *T*. For any simplex *e* of T_{δ} , the map $\tilde{\kappa}_{|e}$ is λ -transversal because $(\tilde{\kappa} \times \tilde{\lambda})(e \times L \times I) \cap \text{diag}_M = \emptyset$.

Set $p = \dim(K)$. We shall construct p + 2 homotopies

$$\kappa = \kappa_0 \rightsquigarrow \kappa_1 \rightsquigarrow \cdots \rightsquigarrow \kappa_{p+2}$$

in the class of *T*-good maps $K \times I \to M$ such that for all $r \ge 0$ and any simplex *e* of *T* of dimension $\le r - 1$, the map $\kappa_{r|e}$ is λ -transversal. Then, the restriction of κ_{p+2} to any simplex of *T* is λ -transversal. For each face *E* of *K*, the product $E \times I$ is a subcomplex of *T*. Therefore, the restriction of $\kappa_{p+2} \times \tilde{\lambda}$ to $E \times I \times L \times I$ is weakly transversal to diag_{*M*}. By the beginning of the proof, the same holds when *L* is replaced with any of its proper faces. This shows that the smooth map $K \to \Omega_{12}^{\circ}$ determined by κ_{p+2} is transversal to λ .

The homotopies $\tilde{\kappa} = \kappa_0 \rightsquigarrow \kappa_1 \rightsquigarrow \cdots$ are constructed recursively. For r = 0, the condition on κ_r is void, and we can take $\kappa_0 = \tilde{\kappa}$. Assume that we have required homotopies $\kappa_0 \rightsquigarrow \kappa_1 \rightsquigarrow \cdots \rightsquigarrow \kappa_r$ for some $r \ge 0$. Consider an *r*-dimensional simplex $e \in T \setminus T_{\delta}$. Clearly, $e \subset K \times \text{Int}(I)$. For $\varepsilon > 0$, let U_{ε} denote the (closed) metric ε -neighborhood of ∂e in $K \times \text{Int}(I)$. The inductive assumption implies that

(*) $\begin{cases} \text{for a sufficiently small } \varepsilon > 0, \text{ the restriction of the} \\ \max \kappa_r \times \tilde{\lambda} \text{ to } U_{\varepsilon} \times L \times I \text{ is weakly transversal to } \operatorname{diag}_M. \end{cases}$

Since κ_r is good and $e \subset K \times \text{Int}(I)$, we have $\kappa_r(e) \subset \text{Int}(M)$. Since κ_r is *T*-good, one has $\kappa_r(e) \subset B$ for a closed ball $B \subset M$. We can choose *B* so that $\kappa_r(e) \subset \text{Int}(B)$. We identify *B* with the closed unit ball in Euclidean space with center 0. Pick a small neighborhood $S \subset B$ of $0 \in B$ so that $\kappa_r(e) + s \subset \text{Int}(B)$ for all $s \in S$. Consider the
smooth maps $\{f_s : e \to B \subset M\}_{s \in S}$ where f_s carries any $u \in e$ to $\kappa_r(u) + s$. It is obvious that the family of maps

$$\left\{f_s \times \tilde{\lambda} : e \times L \times I \to M \times M\right\}_{s \in S}$$

is weakly transversal to diag_M in the sense that the adjoint map

$$e \times L \times I \times S \longrightarrow M \times M$$

is weakly transversal to diag_M . By the classical transversality theorem (see [22, Section 2.3]),

(**) $\begin{cases} \text{the set } S_{\lambda} = \{s \in S : \text{the map } f_s \times \tilde{\lambda} \text{ is weakly transversal to } \text{diag}_M \} \\ \text{is dense (and open) in } S. \end{cases}$

(This argument is adapted from that of Laudenbach [**32**, Proof of Lemma 2.6].) In our terminology, f_s is λ -transversal for any $s \in S_{\lambda}$. For each $s \in S_{\lambda}$, we define a map

$$g_s: e \longrightarrow B \subset M$$

by $g_s(u) = \kappa_r(u) + h(u)s$ for all $u \in e$ where $h : e \to I$ is a smooth function carrying $e \cap U_{\varepsilon/2}$ to 0 and $e \setminus U_{\varepsilon}$ to 1. Then $g_s = \kappa_r$ on $e \cap U_{\varepsilon/2}$ and $g_s = f_s$ on $e \setminus U_{\varepsilon}$. We deduce from (*) and (**) that, for all sufficiently small $s \in S_{\lambda}$, the map g_s is λ -transversal. Pick such an s and consider the obvious linear homotopy $\kappa_{r|e} \to g_s$ constant on $e \cap U_{\varepsilon/2}$. Combining such homotopies corresponding to all r-dimensional simplices $e \in T \setminus T_{\delta}$ and extending by the constant homotopy on T_{δ} we obtain a homotopy of κ_r on the union of T_{δ} with the r-skeleton of T. The latter homotopy extends to a homotopy $\kappa_r \to \kappa_{r+1}$ in the class of good maps $K \times I \to M$. Taking the vectors sin this construction small enough, we can always choose the homotopy is constant on T_{δ} and on the (r - 1)-th skeleton of T, the λ -transversality of κ_r on the simplices of Tof dimension < r acquired at the previous steps is preserved during the homotopy. By construction, κ_{r+1} is λ -transversal on all r-dimensional simplices of T.

LEMMA 4.1.2. — The homotopy in Lemma 4.1.1 may be chosen to be constant on the union of all faces *E* of *K* such that $\kappa_{|E}$ is transversal to λ .

Proof. — The proof proceeds by induction on $\dim(K)$. If $\dim(K) = 0$, then we take the constant homotopy on all connected components of *K* on which κ is transversal to λ and we take the homotopy provided by Lemma 4.1.1 on all other components of *K*. Let *p* be a positive integer such that the lemma holds for all *K* of dimension < *p*. We prove the lemma for an arbitrary *p*-dimensional manifold with faces *K*. As above, if κ is transversal to λ on some connected components of *K*, then we take the constant homotopy on that components. Thus we can assume without loss of generality that κ may be transversal to λ only on proper faces of *K*.

Let Σ be the set of all faces E of K such that $\kappa_{|E}$ is transversal to λ . By the definition of transversality, if $E \in \Sigma$, then all faces of E also belong to Σ . Set

$$|\Sigma| = \bigcup_{E \in \Sigma} E$$

and note that $|\Sigma| \subset \partial K$ by our assumption. All homotopies of $\kappa : K \to \Omega_{12}^{\circ}$ in the following construction are arbitrarily C^{∞} -small smooth homotopies constant on $|\Sigma|$. We recursively construct p homotopies $\kappa = \kappa_{-1} \to \kappa_0 \to \cdots \to \kappa_{p-1}$ such that the restriction of κ_r to any face of K of dimension $\leq r$ is transversal to λ for all r. Assume that we already have homotopies $\kappa = \kappa_{-1} \to \kappa_0 \to \cdots \to \kappa_{r-1}$ with the required properties where $0 \leq r < p$. Consider an r-dimensional face E of K not belonging to Σ . By the assumptions on κ_{r-1} , the restriction of κ_{r-1} to any proper face of E is transversal to λ . Since dim(E) = r < p, the inductive assumption guarantees that there is an arbitrarily C^{∞} -small, smooth, constant on ∂E homotopy of $\kappa_{r-1}|_E$ into a map $E \to \Omega_{12}^{\circ}$ transversal to λ . Combining these homotopies over all E as above together with the constant homotopy on $|\Sigma|$ and extending to a small smooth homotopy of κ_{r-1} on the rest of K, we obtain a homotopy $\kappa_{r-1} \to \kappa_r$ with the required properties.

Next pick a collar $U \cong \partial K \times I \subset K$ of $\partial K \cong \partial K \times \{0\}$ in K. Set

$$V = \overline{K \setminus U} \subset K.$$

Then *V* is a manifold with faces and $\partial V = U \cap V \cong \partial K$. By the above,

$$\kappa' = \kappa_{p-1} : K \longrightarrow \Omega_{12}^{\circ}$$

is a smooth map whose restriction to all proper faces of *K* is transversal to λ . Therefore, choosing the collar *U* sufficiently narrow, we can ensure that the map $\kappa'|_U : U \to \Omega_{12}^\circ$ is transversal to λ . By Lemma 4.1.1, there is an arbitrarily C^∞ -small homotopy $\{\kappa^t|_V\}_{t \in I}$ of $\kappa^0|_V = \kappa'|_V$ such that $\kappa^1|_V$ is transversal to λ . This homotopy extends to a small homotopy $\{\kappa^t|_V\}_{t \in I}$ of $\kappa^0 = \kappa'$ constant on a neighborhood of ∂K in *U*. If the homotopy $\{\kappa^t|_V\}_{t \in I}$ is sufficiently small, then the extension may be chosen so that $\kappa^t|_U$ is transversal to λ for all $t \in I$. Then $\kappa^1|_U$ is transversal to λ , and so, $\kappa^1 : K \to \Omega_{12}^\circ$ is transversal to λ . The composite homotopy

$$\kappa = \kappa_{-1} \rightsquigarrow \cdots \rightsquigarrow \kappa_{p-1} = \kappa' = \kappa^0 \rightsquigarrow \kappa^1$$

satisfies all the conditions of the lemma.

4.1.2. *Transversal polychains.* — We call smooth polychains $\mathcal{K} = (K, \varphi, u, \kappa)$ in Ω_{12}° and $\mathcal{L} = (L, \psi, v, \lambda)$ in Ω_{34}° *transversal* if the maps $\kappa : K \to \Omega_{12}^{\circ}$ and $\lambda : L \to \Omega_{34}^{\circ}$ are transversal. The following two lemmas show that any smooth polychain can be made transversal to a given smooth polychain by a small deformation.

LEMMA 4.1.3. — Let $\mathcal{K} = (K, \varphi, u, \kappa)$ and $\mathcal{L} = (L, \psi, v, \lambda)$ be smooth polychains in Ω_{12}° and $\Omega_{34'}^{\circ}$ respectively. Let N be a union of faces of K. Let

$$\left\{ (\kappa_{|N})^t : N \to \Omega_{12}^\circ \right\}_{t \in I}$$

be a smooth homotopy of $\kappa_{|N}$ compatible with φ such that $(\kappa_{|N})^1 : N \to \Omega_{12}^\circ$ is transversal to \mathcal{L} . Then there is a smooth deformation $\{\mathcal{K}^t = (K, \varphi, u, \kappa^t)\}_{t \in I}$ of $\mathcal{K}^0 = \mathcal{K}$ such that \mathcal{K}^1 is transversal to \mathcal{L} and for all $t \in I$,

$$\kappa^t{}_{|N} = (\kappa_{|N})^t : N \longrightarrow \Omega_{12}^\circ.$$

Proof. — We apply the same recursive method as in the proof of Lemma 3.4.3 with *M* replaced by Ω_{12}° . The homotopy of κ on a representative face *F* is obtained in two steps. First, we take an arbitrary smooth homotopy of $\kappa_{|F}$ extending the homotopy of $\kappa_{|\partial F}$ obtained at the previous step. Then we compose with an additional smooth homotopy rel ∂F to a map $F \to \Omega_{12}^{\circ}$ transversal to λ . The latter homotopy is provided by Lemma 4.1.2.

LEMMA 4.1.4. — Let $\mathcal{K} = (K, \varphi, u, \kappa)$ and $\mathcal{L} = (L, \psi, v, \lambda)$ be smooth polychains in Ω_{12}° and $\Omega_{34'}^{\circ}$ respectively. There exists an arbitrarily C^{∞} -small smooth deformation

$$\left\{\mathcal{K}^t = (K, \varphi, u, \kappa^t)\right\}_{t \in I}$$

of \mathcal{K} such that the polychain \mathcal{K}^1 is transversal to \mathcal{L} and $\kappa^t|_F = \kappa|_F$ for all $t \in I$ and all faces F of K on which κ is transversal to λ .

Proof. — This is a special case of Lemma 4.1.3 where *N* is the union of all faces of *K* on which κ is transversal to λ and $\{(\kappa_{|N})^t\}_t$ is the constant homotopy. That the deformation $\{\mathcal{K}^t\}_t$ may be chosen arbitrarily C^{∞} -small follows from Lemmas 4.1.1 and 4.1.2.

We say that a pair of face homology classes $(a \in \tilde{H}_*(\Omega_{12}^\circ), b \in \tilde{H}_*(\Omega_{34}^\circ))$ is *transversely represented* by a pair $(\mathcal{K}, \mathcal{L})$ if \mathcal{K} is a smooth reduced polycycle in Ω_{12}° representing *a*, \mathcal{L} is a smooth reduced polycycle in Ω_{34}° representing *b*, and \mathcal{K} is transversal to \mathcal{L} . The following lemma will play a key role in the sequel.

LEMMA 4.1.5. — Every pair $(a \in \hat{H}_*(\Omega_{12}^\circ), b \in \hat{H}_*(\Omega_{34}^\circ))$ can be transversely represented by a pair of polycycles. Any two pairs of polycycles transversely representing (a, b) can be related by a finite sequence of transformations $(\mathfrak{K}, \mathcal{L}) \mapsto (\check{\mathfrak{K}}, \check{\mathcal{L}})$ of the following types:

- (i) L ≃ L̃ and K̃ ≃ K ⊔ ∂^rM or K ≃ K̃ ⊔ ∂^rM where M is a smooth polychain in Ω^o₁₂ transversal to L;
- (ii) $\mathcal{K} \cong \check{\mathcal{K}}$ and $\check{\mathcal{L}} \cong \mathcal{L} \sqcup \partial^r \mathcal{N}$ or $\mathcal{L} \cong \check{\mathcal{L}} \sqcup \partial^r \mathcal{N}$ where \mathcal{N} is a smooth polychain in Ω_{34}° transversal to \mathcal{K} .

Proof. — The first claim follows from Lemma 4.1.4 and the surjectivity in Theorem 3.4.8. That we need only reduced polycycles follows from the fact that the reduction of a (smooth) polycycle gives a homologous (smooth) polycycle.

We prove the second claim of the lemma. Consider pairs of reduced polycycles $(\mathcal{K}_1, \mathcal{L})$ and $(\mathcal{K}_2, \mathcal{L})$ transversely representing (a, b). Since \mathcal{K}_1 is homologous to \mathcal{K}_2 , we have

$$\mathcal{K}_1 \sqcup \partial^r \mathcal{R}_1 \cong \mathcal{K}_2 \sqcup \partial^r \mathcal{R}_2$$

for some (n + 1)-polychains $\mathcal{R}_1, \mathcal{R}_2$ in Ω_{12}° . The injectivity in Theorem 3.4.8 ensures that $\mathcal{R}_1, \mathcal{R}_2$ can be chosen to be smooth. Then there are smooth polychains $\mathcal{R}'_1, \mathcal{R}'_2$ in Ω_{12}° transversal to \mathcal{L} such that $\mathcal{K}_1 \sqcup \partial^r \mathcal{R}'_1 \cong \mathcal{K}_2 \sqcup \partial^r \mathcal{R}'_2$. These polychains are obtained from $\mathcal{R}_1, \mathcal{R}_2$ using the same method as in the proof of Theorem 3.4.5 with the following replacements:

 $M \rightsquigarrow \Omega_{12}^{\circ}$, "smooth" \rightsquigarrow "transversal to \mathcal{L} ", "homotopy" \rightsquigarrow "smooth homotopy", Lemma 3.4.3 \rightsquigarrow Lemma 4.1.3 and Lemma 3.4.4 \rightsquigarrow Lemma 4.1.4.

The move $(\mathcal{K}_1, \mathcal{L}) \mapsto (\mathcal{K}_2, \mathcal{L})$ expands as the composition of the following type (i) moves:

$$(\mathcal{K}_1, \mathcal{L}) \longmapsto (\mathcal{K}_1 \sqcup \partial^r \mathcal{R}'_1, \mathcal{L}) \longmapsto (\mathcal{K}_2 \sqcup \partial^r \mathcal{R}'_2, \mathcal{L}) \longmapsto (\mathcal{K}_2, \mathcal{L}).$$

(The middle move is a type (i) move corresponding to $\mathcal{M} = \emptyset$.) A similar argument shows that if two pairs of polycycles $(\mathcal{K}, \mathcal{L}_1)$ and $(\mathcal{K}, \mathcal{L}_2)$ transversely represent (a, b), then the move $(\mathcal{K}, \mathcal{L}_1) \mapsto (\mathcal{K}, \mathcal{L}_2)$ is a composition of type (ii) moves.

Consider now any pairs of polycycles $(\mathcal{K}_1, \mathcal{L}_1)$ and $(\mathcal{K}_2, \mathcal{L}_2)$ transversely representing (a, b). By Lemma 4.1.4, there is an arbitrarily C^{∞} -small smooth deformation of \mathcal{K}_1 into a polycycle $\mathcal K$ transversal to $\mathcal L_2$. We assume the deformation to be so small that $\mathcal K$ is transversal to \mathcal{L}_1 as well. By Lemma 3.2.3, $\mathcal{K} \simeq \mathcal{K}_1$ represents *a*. By the previous paragraph, each of the moves

$$(\mathfrak{K}_1, \mathfrak{L}_1) \longmapsto (\mathfrak{K}, \mathfrak{L}_1) \longmapsto (\mathfrak{K}, \mathfrak{L}_2) \longmapsto (\mathfrak{K}_2, \mathfrak{L}_2)$$

expands as a composition of moves of types (i) and (ii).

4.2. Intersection of polychains

We define "intersection" for transversal polychains in Ω_{12}° and Ω_{34}° .

4.2.1. *The intersection polychain.* — Let $\mathcal{K} = (K, \varphi, u, \kappa)$ be a smooth polychain of dimension p in Ω_{12}° and let $\mathcal{L} = (L, \psi, v, \lambda)$ be a smooth polychain of dimension q in Ω_{34}° . Assume that \mathcal{K} and \mathcal{L} are transversal in the sense of Section 4.1.2. We derive from \mathcal{K} and \mathcal{L} an "intersection polychain" in $\Omega_{32} \times \Omega_{14}$.

Let $\tilde{\kappa} : K \times I \to M$ and $\tilde{\lambda} : L \times I \to M$ be the adjoint maps of κ and λ respectively. Set $N = K \times I \times L \times I$ and consider the map $\tilde{\kappa} \times \tilde{\lambda} : N \to M \times M$. Since K, L and I are manifolds with faces of dimensions p, q and 1, respectively, N is a manifold with faces of dimension p + q + 2. The transversality of κ and λ implies that the set

$$D = \left(\tilde{\kappa} \times \tilde{\lambda}\right)^{-1} (\operatorname{diag}_M) \subset N$$

is empty if p + q + 2 < n and is a (p + q + 2 - n)-dimensional manifold with corners if $p+q+2 \ge n$. In the latter case each point of *D* has a neighborhood *V* in *N* such that $V \cap D$ is homeomorphic to $\mathbb{R}^{u} \times [0, \infty)^{v}$ for some integers $u, v \ge 0$ with u + v = p + q + 2 - nand V is homeomorphic to $\mathbb{R}^n \times (V \cap D)$. These claims follow from the general theorems about transversality and about submanifolds of manifolds with corners, see [35, Propositions 3.1.14 and 7.2.7]. Consequently, $P(D) \subset P(N)$ so that we can consider the commutative diagram of inclusion maps



The structure of V described above implies that i is a bijection between v-element sets. Since N is a manifold with faces, j is an injection. Therefore j' is injective which

implies that *D* is a manifold with faces. The faces of *D* are the connected components of the intersections of *D* with faces of *N*.

We now upgrade *D* to a polychain in $\Omega_{32} \times \Omega_{14}$. First of all, we orient *D* as follows. We use the orientation of $\operatorname{diag}_M \approx M$ and the product orientation of $M \times M$ to orient the normal vector bundle of diag_M in $M \times M$ (see the Introduction for our orientation conventions). Next, we pull-back this orientation of the normal vector bundle along $\tilde{\kappa} \times \tilde{\lambda}$ to obtain an orientation of the normal vector bundle of *D* in *N*. The latter orientation together with the product orientation in $N = K \times I \times L \times I$ induces an orientation of *D*. We also equip *D* with the weight $w : \pi_0(D) \to \mathbb{K}$ which, for any connected components *X* of *K* and *Y* of *L*, carries all connected components of *D* contained in $X \times I \times Y \times I$ to $u(X)v(Y) \in \mathbb{K}$.

Next, we define a continuous map $\kappa \cong \lambda : D \times I \to M \times M$ by

$$(\kappa \stackrel{\sim}{\bullet} \lambda)(x, s, y, t, u) = \begin{cases} (\lambda(y)(t \ast u), \kappa(x)(s \ast u)) & \text{if } 0 \le u \le 1/2, \\ (\kappa(x)(s \ast u), \lambda(y)(t \ast u)) & \text{if } 1/2 \le u \le 1 \end{cases}$$

for any $(x, s, y, t) \in D \subset K \times I \times L \times I$ and $u \in I$, where we set

$$\ell * u = \begin{cases} 2\ell u & \text{for } \ell \in I, u \in [0, 1/2], \\ 1 - 2(1 - \ell)(1 - u) & \text{for } \ell \in I, u \in [1/2, 1]. \end{cases}$$

The key property of the operation * is that for any $\ell, u \in I$, we have $0 \le \ell * u \le \ell$ if $u \in [0, 1/2]$ and $\ell \le \ell * u \le 1$ if $u \in [1/2, 1]$. For a fixed $(x, s, y, t) \in D$, the point

$$(\kappa \, \widehat{\triangleleft} \lambda)(x, s, y, t, u) \in M \times M$$

moves along the path $(\tilde{\lambda}(y, t * u), \tilde{\kappa}(x, s * u))$ from (\star_3, \star_1) to the diagonal point $(\tilde{\kappa}(x, s), \tilde{\lambda}(y, t))$ as *u* increases from 0 to 1/2 and, next, it moves from that diagonal point to (\star_2, \star_4) along the path $(\tilde{\kappa}(x, s * u), \tilde{\lambda}(y, t * u))$ as *u* increases from 1/2 to 1: see Figure 4.2.1 next page. Thus the map $\kappa \approx \lambda$ is adjoint to a continuous map

$$\kappa \! \diamond \! \lambda : D \longrightarrow \Omega(M imes M, (\star_3, \star_1), (\star_2, \star_4)) = \Omega_{32} imes \Omega_{14}$$

whose coordinate maps are denoted by $\kappa \triangleleft \lambda : D \rightarrow \Omega_{32}$ and $\kappa \triangleright \lambda : D \rightarrow \Omega_{14}$.

figures-COLOR/diamond.pdf.{ps,eps} not found (or no BBox)

FIGURE 4.2.1. The pair $(\triangleleft, \triangleright) = (\kappa \tilde{\clubsuit} \lambda)(x, s, y, t, u) \in M \times M$ for a fixed $(x, s, y, t) \in D$ and *u* running from 0 to 1.

Finally, we define a partition of *D*. Here we need the assumption that the images of κ and λ lie in $\Omega_{12}^{\circ} \subset \Omega_{12}$ and $\Omega_{34}^{\circ} \subset \Omega_{34}$, respectively. This assumption implies that

$$D \subset K \times \operatorname{Int}(I) \times L \times \operatorname{Int}(I) \subset N.$$

Therefore each face *F* of *D* is contained in a unique smallest face

$$N_F = A_F \times I \times B_F \times I$$

of *N*, where A_F is a face of *K* and B_F is a face of *L*. Note that the codimension of *F* in *D* is equal to the codimension of N_F in *N*. We declare two faces *F* and *F*' of *D* to

have the same type if and only if $A = A_F$ has the same type as $A' = A_{F'}$ in K, $B = B_F$ has the same type as $B' = B_{F'}$ in L, and the diffeomorphism

$$N_F = A \times I \times B \times I \xrightarrow{\varphi_{A,A'} \times \mathrm{id}_I \times \psi_{B,B'} \times \mathrm{id}_I} A' \times I \times B' \times I = N_{F'}$$

carries *F* onto *F'*. By restriction, we obtain a diffeomorphism $\theta_{F,F'} : F \to F'$ for any such *F*, *F'*. This defines a partition, θ , of *D*. Note that, for each face *F* of *D*, the faces of *D* of the same type as *F* are in one-to-one correspondence with the pairs (A', B') where *A'* is a face of *K* of the same type as *A_F* and *B'* is a face of *L* of the same type as *B_F*.

Lемма 4.2.1. — The tuple
$$\mathcal{D}(\mathcal{K},\mathcal{L}) = (D,\theta,w,\kappa \bullet \lambda)$$
 is a polychain in $\Omega_{32} \times \Omega_{14}$

Proof. — We need only to show that the map $\kappa \ll \lambda$ is compatible with the partition θ . Consider two faces *F* and *F*' of the same type in *D* and set

$$A = A_F, B = B_F, A' = A_{F'}, \text{ and } B' = B_{F'}.$$

For any $(x, s, y, t) \in F$ and $u \in [0, 1/2]$, we have

$$(\kappa \rightsquigarrow \lambda) (\theta_{F,F'}(x, s, y, t)) (u) = (\kappa \rightsquigarrow \lambda) (\varphi_{A,A'}(x), s, \psi_{B,B'}(y), t) (u)$$
$$= (\lambda(\psi_{B,B'}(y))(t * u), \kappa(\varphi_{A,A'}(x))(s * u))$$
$$= (\lambda(y)(t * u), \kappa(x)(s * u))$$
$$= (\kappa \rightsquigarrow \lambda) (x, s, y, t) (u)$$

where the third equality follows from the compatibility of κ with φ and of λ with ψ . A similar argument works for $u \in [1/2, 1]$. Thus, $(\kappa \Rightarrow \lambda)\theta_{F,F'} = (\kappa \Rightarrow \lambda)|_F$. \Box

4.2.2. *Properties of* \mathcal{D} . — We study the behavior of the polychain $\mathcal{D}(\mathcal{K}, \mathcal{L})$ under the operations on \mathcal{K} and \mathcal{L} introduced in Sections 3.2.2 and 3.2.3.

LEMMA 4.2.2. — Let $\mathcal{K}, \mathcal{K}'$ be smooth *p*-polychains in Ω_{12}° and $\mathcal{L}, \mathcal{L}'$ be smooth *q*-polychains in Ω_{34}° such that $\mathcal{K}, \mathcal{K}'$ and $\mathcal{L}, \mathcal{L}'$ are pairwise transversal. Then

- (i) $\mathcal{D}(k\mathfrak{K},\mathcal{L}) \cong \mathcal{D}(\mathfrak{K},k\mathcal{L}) \cong k\mathcal{D}(\mathfrak{K},\mathcal{L})$ for any $k \in \mathbb{K}$;
- (ii) red $\mathcal{D}(\operatorname{red} \mathcal{K}, \operatorname{red} \mathcal{L}) = \operatorname{red} \mathcal{D}(\mathcal{K}, \mathcal{L});$
- (iii) $\partial \mathcal{D}(\mathcal{K}, \mathcal{L}) \cong (-1)^n \mathcal{D}(\partial \mathcal{K}, \mathcal{L}) \sqcup (-1)^{n+p+1} \mathcal{D}(\mathcal{K}, \partial \mathcal{L});$
- (iv) $\partial^r \mathcal{D}(\mathcal{K}, \mathcal{L}) = (-1)^n \operatorname{red} \mathcal{D}(\partial^r \mathcal{K}, \operatorname{red} \mathcal{L}) \sqcup (-1)^{n+p+1} \operatorname{red} \mathcal{D}(\operatorname{red} \mathcal{K}, \partial^r \mathcal{L});$
- (v) $\mathcal{D}(\mathcal{K} \sqcup \mathcal{K}', \mathcal{L}) \cong \mathcal{D}(\mathcal{K}, \mathcal{L}) \sqcup \mathcal{D}(\mathcal{K}', \mathcal{L}), \mathcal{D}(\mathcal{K}, \mathcal{L} \sqcup \mathcal{L}') \cong \mathcal{D}(\mathcal{K}, \mathcal{L}) \sqcup \mathcal{D}(\mathcal{K}, \mathcal{L}').$

Proof. — Claims (i) and (v) are obvious. Claim (iv) easily follows from (ii) and (iii). We prove (ii). It is clear that

$$\operatorname{red}_{+} \mathcal{D}(\operatorname{red}_{+}(-), \operatorname{red}_{+}(-)) = \operatorname{red}_{+} \mathcal{D}(-, -),$$

$$\operatorname{red}_{0} \mathcal{D}(\operatorname{red}_{0}(-), \operatorname{red}_{0}(-)) = \operatorname{red}_{0} \mathcal{D}(-, -).$$

Using the identities $red red_0 = red = red red_+$, we conclude that

$$\operatorname{red} \mathcal{D}(\operatorname{red} \mathcal{K}, \operatorname{red} \mathcal{L}) = \operatorname{red} \operatorname{red}_0 \mathcal{D}(\operatorname{red}_0 \operatorname{red}_+ \mathcal{K}, \operatorname{red}_0 \operatorname{red}_+ \mathcal{L})$$

 $= \operatorname{red} \operatorname{red}_0 \mathcal{D}(\operatorname{red}_+ \mathcal{K}, \operatorname{red}_+ \mathcal{L})$

$$= \operatorname{red} \operatorname{red}_{+} \mathcal{D}(\operatorname{red}_{+} \mathcal{K}, \operatorname{red}_{+} \mathcal{L})$$
$$= \operatorname{red} \operatorname{red}_{+} \mathcal{D}(\mathcal{K}, \mathcal{L}) = \operatorname{red} \mathcal{D}(\mathcal{K}, \mathcal{L}).$$

We now prove (iii). Let $\mathcal{K} = (K, \varphi, u, \kappa)$, $\mathcal{L} = (L, \psi, v, \lambda)$ and $\mathcal{D}(\mathcal{K}, \mathcal{L}) = (D, \theta, w, \kappa \diamond \lambda)$ as in Section 4.2.1. Consider the boundary polychain

 $\partial \mathcal{D}(\mathcal{K},\mathcal{L}) = (D^{\partial},\theta^{\partial},w^{\partial},(\kappa \rightsquigarrow \lambda)^{\partial})$

as defined in Section 3.2.3, as well as the polychains

$$\mathcal{D}(\partial \mathcal{K}, \mathcal{L}) = ({}^*D, {}^*\theta, {}^*w, \kappa^{\partial} \rightsquigarrow \lambda) \quad \text{and} \quad \mathcal{D}(\mathcal{K}, \partial \mathcal{L}) = (D^*, \theta^*, w^*, \kappa \rightsquigarrow \lambda^{\partial}).$$

We verify that

(4.2.1)
$$D^{\partial} \cong (-1)^{n*}D \sqcup (-1)^{n+p+1}D^*.$$

Consider a principal face *F* of *D*. Since the codimension of *F* in *D* is equal to the codimension of $N_F = A_F \times I \times B_F \times I$ in $N = K \times I \times L \times I$, the face N_F is a principal face of *N*. Therefore either A_F is a connected component of *K* and B_F is a principal face of *L*, or, A_F is a principal face of *K* and B_F is a connected component of *L*. We first analyze the former case. Set

$$N^* = K \times I \times L^{\partial} \times I.$$

Then $F \subset D \subset N$ corresponds to a connected component F^* of $D^* \subset N^*$ via the map $\operatorname{id}_K \times \operatorname{id}_I \times \iota \times \operatorname{id}_I : N^* \to N$ where $\iota : L^\partial \to L$ is the natural map as in Section 3.2.3. The orientation of F induced by that of $\partial D \subset D$ may differ from the orientation of F^* induced by D^* , and we now compute this difference. Let ε^1 be the trivial 1-dimensional vector bundle equipped with the canonical orientation and let $-\varepsilon^1$ be the same bundle with the opposite orientation. Given a cartesian product of topological spaces, let pr_i denote the projection onto the *i*-th factor. Set

$$N_F^* = K \times I \times B_F \times I,$$

which is a submanifold with faces of N^* of codimension 0 containing F^* . We can also view N_F^* as a submanifold of N of codimension 1, so that $F^* \subset N_F^*$ corresponds to $F \subset N$. Using the orientation conventions of the Introduction and using the letter T for the tangent vector bundle of a manifold, we obtain the following orientation-preserving isomorphisms of oriented vector bundles:

$$TN_{|N_F^*} = \operatorname{pr}_1^*(TK)_{|N_F^*} \oplus \operatorname{pr}_2^*(TI)_{|N_F^*} \oplus \operatorname{pr}_3^*(TL)_{|N_F^*} \oplus \operatorname{pr}_4^*(TI)_{|N_F^*}$$

$$\cong \operatorname{pr}_1^*(TK) \oplus \operatorname{pr}_2^*(TI) \oplus \varepsilon^1 \oplus \operatorname{pr}_3^*(TB_F) \oplus \operatorname{pr}_4^*(TI)$$

$$\cong (-1)^{p+1} \varepsilon^1 \oplus \operatorname{pr}_1^*(TK) \oplus \operatorname{pr}_2^*(TI) \oplus \operatorname{pr}_3^*(TB_F) \oplus \operatorname{pr}_4^*(TI).$$

 $=TN_{F}^{*}$

Therefore

$$TN_{|F^*} \cong (-1)^{p+1} \varepsilon^1 \oplus TN^*_{F|F^*}$$
$$\cong (-1)^{p+1} \varepsilon^1 \oplus \mathcal{V}_{N^*_F} F^* \oplus TF^* = (-1)^{p+1} \varepsilon^1 \oplus \mathcal{V}_{N^*} F^* \oplus TF$$

where the letter V stands for the normal vector bundle of a submanifold in the ambient manifold. On the other hand, restricting the orientation-preserving isomorphism of oriented vector bundles $TN_{|D} \cong V_N D \oplus TD$ to F we obtain that

$$TN_{|F} \cong \mathcal{V}_N D_{|F} \oplus TD_{|F} \cong \mathcal{V}_N D_{|F} \oplus \varepsilon^1 \oplus TF \cong (-1)^n \varepsilon^1 \oplus \mathcal{V}_N D_{|F} \oplus TF.$$

Since the orientations of $\mathcal{V}_{N^*}F^* = \mathcal{V}_{N_F^*}F^*$ and $\mathcal{V}_N D$ are both induced by the orientation of the normal bundle of diag_{*M*} inx $M \times M$, the bundle isomorphism $\mathcal{V}_{N_F^*}F^* \to \mathcal{V}_N D_{|F}$ induced by the inclusion $N_F^* \subset N$ is orientation-preserving. Combining with the computations above, we deduce that

$$TF \cong (-1)^{n+p+1}TF^*.$$

The case where A_F is a principal face of K and B_F is a connected component of L is treated similarly. In this case, F corresponds to a connected component *F of *D, and the orientation of F induced from that of D differs by $(-1)^n$ from the orientation of *F induced by *D. This gives the diffeomorphism (4.2.1) of oriented manifolds with faces, which is easily checked to be a diffeomorphism of polychains as in (iii).

4.3. The operation $\widetilde{\Upsilon}$

We introduce an operation $\widetilde{\Upsilon}$ in the face homology of path spaces.

4.3.1. Definition and properties of $\tilde{\Upsilon}$. — First, we show that the intersection operation defined in Section 4.2 induces an operation in face homology.

LEMMA 4.3.1. — For any integers $p, q \ge 0$, the intersection $(\mathcal{K}, \mathcal{L}) \mapsto \mathcal{D}(\mathcal{K}, \mathcal{L})$ from Section 4.2.1 induces a bilinear map

$$H_p(\Omega_{12}) \times H_q(\Omega_{34}) \longrightarrow H_{p+q+2-n}(\Omega_{32} \times \Omega_{14}).$$

Proof. — Consider any face homology classes $a \in \tilde{H}_p(\Omega_{12})$ and $b \in \tilde{H}_q(\Omega_{34})$. By Lemmas 3.4.7 and 4.1.5, the pair (a, b) can be transversely represented by a smooth reduced *p*-polycycle \mathcal{K} in Ω_{12}° and a smooth reduced *q*-polycycle \mathcal{L} in Ω_{34}° . It follows from Lemma 4.2.2.(iv) that the polychain $\mathcal{D}(\mathcal{K}, \mathcal{L})$ in $\Omega_{32} \times \Omega_{14}$ is a polycycle. Consider another such pair $(\check{\mathcal{K}}, \check{\mathcal{L}})$ transversely representing (a, b). We claim that the polycycles $\mathcal{D}(\mathcal{K}, \mathcal{L})$ and $\mathcal{D}(\check{\mathcal{K}}, \check{\mathcal{L}})$ are homologous. By Lemma 4.1.5, it suffices to prove this claim in the following two cases:

- $\succ \mathcal{L} = \check{\mathcal{L}} \text{ and there exists a smooth } (p+1)\text{-polychain } \mathcal{M} \text{ in } \Omega_{12}^{\circ} \text{ transversal to } \mathcal{L} \text{ such that } \mathcal{K} \cong \check{\mathcal{K}} \sqcup \partial^r \mathcal{M} \text{ or } \check{\mathcal{K}} \cong \mathcal{K} \sqcup \partial^r \mathcal{M};$
- ▶ $\mathcal{K} = \check{\mathcal{K}}$ and there exists a smooth (q+1)-polychain \mathcal{N} in Ω_{34}° transversal to \mathcal{K} such that $\mathcal{L} \cong \check{\mathcal{L}} \sqcup \partial^r \mathcal{N}$ or $\check{\mathcal{L}} \cong \mathcal{L} \sqcup \partial^r \mathcal{N}$.

Assume for concreteness that $\mathcal{L} = \check{\mathcal{L}}$ and $\mathcal{K} \cong \check{\mathcal{K}} \sqcup \partial^r \mathfrak{M}$ (the other cases can be treated similarly). Since \mathcal{L} is a reduced polycycle, Lemma 4.2.2.(iv) implies that

$$\partial^r \mathcal{D}(\mathcal{M},\mathcal{L}) = (-1)^n \operatorname{red} \mathcal{D}(\partial^r \mathcal{M},\mathcal{L}).$$

4.3. THE OPERATION $\widetilde{\Upsilon}$

This and Lemma 4.2.2.(v) imply that

$$\operatorname{red} \mathcal{D}(\mathcal{K},\mathcal{L}) \cong \operatorname{red} \mathcal{D}(\check{\mathcal{K}} \sqcup \partial^r \mathcal{M},\check{\mathcal{L}}) \cong \operatorname{red} \mathcal{D}(\check{\mathcal{K}},\check{\mathcal{L}}) \sqcup \partial^r \mathcal{D}((-1)^n \mathcal{M},\check{\mathcal{L}}).$$

We conclude that $\mathcal{D}(\mathcal{K},\mathcal{L})$ is homologous to $\mathcal{D}(\check{\mathcal{K}},\check{\mathcal{L}})$. Thus, the face homology class

 $\left\langle \mathcal{D}(\mathcal{K},\mathcal{L})\right\rangle \in \widetilde{H}_{p+q+2-n}(\Omega_{32}\times\Omega_{14})$

of $\mathcal{D}(\mathcal{K}, \mathcal{L})$ depends only on $a \in \widetilde{H}_p(\Omega_{12})$ and $b \in \widetilde{H}_q(\Omega_{34})$. This defines the pairing in the statement of the lemma. The bilinearity of this pairing follows from assertions (i) and (v) in Lemma 4.2.2.

The pairing produced by Lemma 4.3.1 induces a linear map

$$\widetilde{H}_p(\Omega_{12}) \otimes \widetilde{H}_q(\Omega_{34}) \longrightarrow \widetilde{H}_{p+q+2-n}(\Omega_{32} \times \Omega_{14}).$$

Taking the direct sum over all $p, q \ge 0$, we obtain a linear map of degree 2 - n

(4.3.1)
$$\widetilde{\Upsilon}: \widetilde{H}_*(\Omega_{12}) \otimes \widetilde{H}_*(\Omega_{34}) \longrightarrow \widetilde{H}_*(\Omega_{32} \times \Omega_{14}).$$

To stress the role of the tuple of base points $(\star_1, \star_2, \star_3, \star_4)$ we will also denote this map by $\widetilde{\Upsilon}_{12,34}$. Any permutation $(\star_i, \star_j, \star_k, \star_l)$ of $(\star_1, \star_2, \star_3, \star_4)$ such that $\{\star_i, \star_j\} \cap \{\star_k, \star_l\} = \emptyset$ yields a map

$$\widetilde{\Upsilon}_{ij,kl}: \widetilde{H}_*(\Omega_{ij}) \otimes \widetilde{H}_*(\Omega_{kl}) \longrightarrow \widetilde{H}_*(\Omega_{kj} \times \Omega_{il}).$$

We now establish the following symmetry for $\widetilde{\Upsilon}$.

LEMMA 4.3.2. — Let $p: \Omega_{32} \times \Omega_{14} \to \Omega_{14} \times \Omega_{32}$ be the map permuting the two factors of the cartesian product. For any $a \in \tilde{H}_p(\Omega_{12})$ and $b \in \tilde{H}_q(\Omega_{34})$ with $p, q \ge 0$,

$$p_* \widetilde{\Upsilon}_{12,34}(a \otimes b) = (-1)^{(p+1)(q+1)+n} \widetilde{\Upsilon}_{34,12}(b \otimes a).$$

Proof. — We assume that (a, b) is transversely represented by a smooth reduced *p*-polycycle $\mathcal{K} = (K, \varphi, u, \kappa)$ in Ω_{12}° and a smooth reduced *q*-polycycle $\mathcal{L} = (L, \psi, v, \lambda)$ in Ω_{34}° . Let

$$\mathcal{D}(\mathcal{K},\mathcal{L}) = (D,\theta,w,\kappa \rightsquigarrow \lambda) \text{ and } \mathcal{D}(\mathcal{L},\mathcal{K}) = (D',\theta',w',\lambda \rightsquigarrow \kappa).$$

Let q be the permutation map $M \times M \to M \times M, (m_1, m_2) \mapsto (m_2, m_1)$. This map preserves diag_M pointwise and preserves (respectively, inverts) the orientation of the normal bundle of diag_M in $M \times M$ if *n* is even (respectively, odd). Let

$$\mathbf{h}: K \times I \times L \times I \rightarrow L \times I \times K \times I$$

be the permutation map defined by $(k, s, l, t) \mapsto (l, t, k, s)$. Clearly,

deg h =
$$(-1)^{(p+1)(q+1)}$$
 and $(\tilde{\lambda} \times \tilde{\kappa})$ h = q $(\tilde{\kappa} \times \tilde{\lambda})$.

Thus, h restricts to a diffeomorphism $h_{|D} : D \to D'$ of degree $(-1)^{(p+1)(q+1)+n}$. This diffeomorphism carries the weight *w* to *w'* and the partition θ to θ' . Also,

$$(\lambda \triangleleft \kappa) \circ \mathbf{h}_{|D} = \mathbf{p} \circ (\kappa \triangleleft \lambda).$$

Thus, $h_{|D}$ is a diffeomorphism of the polychains $p_* \mathcal{D}(\mathcal{K}, \mathcal{L})$ and $(-1)^{(p+1)(q+1)+n} \mathcal{D}(\mathcal{L}, \mathcal{K})$. We conclude that

$$\begin{split} \mathbf{p}_* \widetilde{\mathbf{\Upsilon}}_{12,34}(a \otimes b) &= \left\langle \mathbf{p}_* \mathcal{D}(\mathcal{K}, \mathcal{L}) \right\rangle = (-1)^{(p+1)(q+1)+n} \left\langle \mathcal{D}(\mathcal{L}, \mathcal{K}) \right\rangle \\ &= (-1)^{(p+1)(q+1)+n} \widetilde{\mathbf{\Upsilon}}_{34,12}(b \otimes a). \end{split}$$

4.3.2. Computation of $\widetilde{\Upsilon}$. — To evaluate $\widetilde{\Upsilon}$ on a pair of face homology classes in Ω_{12}, Ω_{34} , we represent these classes by smooth reduced transversal polycycles in $\Omega_{12}^{\circ}, \Omega_{34}^{\circ}$ and take the face homology class of the intersection polycycle. We now explain how to compute $\widetilde{\Upsilon}$ from more general polycycles in Ω_{12}, Ω_{34} .

We say that polycycles (possibly non-smooth and non-reduced)

 $\mathcal{K} = (K, \varphi, u, \kappa)$ in Ω_{12} and $\mathcal{L} = (L, \psi, v, \lambda)$ in Ω_{34}

are *admissible* if there exist open sets $U \subset K \times Int(I)$ and $V \subset L \times Int(I)$ such that

- (i) the maps $\tilde{\kappa}_{|U} : U \to M$ and $\tilde{\lambda}_{|V} : V \to M$ are smooth and their images do not meet ∂M ;
- (ii) $(\tilde{\kappa} \times \tilde{\lambda})^{-1}(\operatorname{diag}_M) \subset U \times V$;
- (iii) for any face *E* of *K* and any face *F* of *L*, the restriction of $\tilde{\kappa} \times \tilde{\lambda}$ to

 $(\operatorname{Int}(E \times I) \cap U) \times (\operatorname{Int}(F \times I) \cap V)$

is transversal to $Int(diag_M)$ in the usual sense of differential topology.

If \mathcal{K} and \mathcal{L} are admissible, then we can define the intersection polycycle $\mathcal{D}(\mathcal{K}, \mathcal{L})$ in $\Omega_{32} \times \Omega_{14}$ repeating word for word the definitions of Section 4.2. The polycycle $\mathcal{D}(\mathcal{K}, \mathcal{L})$ depends only on \mathcal{K}, \mathcal{L} and does not depend on the choice of U, V.

LEMMA 4.3.3. — Let \mathcal{K} and \mathcal{L} be admissible polycycles in Ω_{12} and Ω_{34} representing, respectively, $a \in \widetilde{H}_*(\Omega_{12})$ and $b \in \widetilde{H}_*(\Omega_{34})$. Then

$$\widetilde{\Upsilon}(a,b) = \langle \mathcal{D}(\mathcal{K},\mathcal{L}) \rangle.$$

Proof. — Let $\mathcal{K} = (K, \varphi, u, \kappa)$ and $\mathcal{L} = (L, \psi, v, \lambda)$. The set

$$(\tilde{\lambda})^{-1}(\operatorname{diag}_{M}) \subset K \times I \times L \times I$$

is closed and, hence, compact. By (ii), there are compact sets $A \subset U$ and $B \subset V$ such that $(\tilde{\kappa} \times \tilde{\lambda})^{-1}(\operatorname{diag}_M) \subset A \times B$. Pick a small deformation of κ and λ into smooth maps (in the class of maps compatible with the partitions). The deformation may be chosen to be constant on some open neighborhoods $U' \subset U$, $V' \subset V$ of A, B, respectively, and to be so small that the condition (ii) with $U \times V$ replaced by $U' \times V'$ is met during the deformation. The condition (iii) with U, V replaced by U', V' is automatically met during the deformation. By Lemma 3.2.3, the face homology class $\langle \mathcal{D}(\mathcal{K}, \mathcal{L}) \rangle$ is preserved under such a deformation. Thus, without loss of generality we can assume from the very beginning that the maps κ and λ are smooth.

Pick a small neighborhood *W* of ∂M in *M* such that $\tilde{\kappa}(U) \cup \tilde{\lambda}(V) \subset M \setminus W$. The proof of Lemma 3.4.7 and Theorem 3.4.8 provides, for any $i, j \in \{1, 2, 3, 4\}$, a homotopy of the identity map id : $\Omega_{ij} \rightarrow \Omega_{ij}$ into a map $f_{ij} : \Omega_{ij} \rightarrow \Omega_{ij}^{\circ} \subset \Omega_{ij}$ such that smooth polycycles in Ω_{ij} remain smooth throughout the homotopy. The homotopy acts on a

path in *M* from \star_i to \star_j by pushing the interior points of the path inside *M* along a 1-parameter family of embeddings $M \hookrightarrow M$. We can assume that these embeddings are constant on $M \setminus W$ and so, the homotopy fixes all points of the paths lying in $M \setminus W$. For i = 1, j = 2 and i = 3, j = 4, these homotopies induce a smooth deformation of polycycles

$$\{\mathcal{K}^t\}_{t\in I} = \left\{ (K,\varphi,u,\kappa^t) \right\}_{t\in I}, \quad \{\mathcal{L}^t\}_{t\in I} = \left\{ (L,\psi,v,\lambda^t) \right\}_{t\in I}$$

where $\kappa^0 = \kappa$, $\lambda^0 = \lambda$, $\kappa^1 = f_{12}\kappa$, $\lambda^1 = f_{34}\lambda$. Our assumptions ensure that $\tilde{\kappa}^t|_U = \tilde{\kappa}|_U$ and $\tilde{\lambda}^t|_V = \tilde{\lambda}|_V$ for all $t \in I$. Thus the set $(\tilde{\kappa}^t \times \tilde{\lambda}^t)^{-1}(\operatorname{diag}_M)$ does not depend on t, and $\mathcal{K}^t, \mathcal{L}^t$ are admissible for all $t \in I$. Then the polycycle $\mathcal{D}(\mathcal{K}^1, \mathcal{L}^1)$ is obtained from the polycycle $\mathcal{D}(\mathcal{K}^0, \mathcal{L}^0) = \mathcal{D}(\mathcal{K}, \mathcal{L})$ by deformation. Hence, by Lemma 3.2.3,

$$\langle \mathcal{D}(\mathcal{K}^1, \mathcal{L}^1) \rangle = \langle \mathcal{D}(\mathcal{K}, \mathcal{L}) \rangle.$$

The polycycles $\operatorname{red}(\mathcal{K}^1)$ in Ω_{12}° and $\operatorname{red}(\mathcal{L}^1)$ in Ω_{34}° transversely represent the pair (a, b). We conclude that

$$\begin{split} \widetilde{\mathbf{f}}(a,b) &= \left\langle \mathcal{D}(\operatorname{red}\mathcal{K}^{1},\operatorname{red}\mathcal{L}^{1})\right\rangle \\ &= \left\langle \operatorname{red}\mathcal{D}(\operatorname{red}\mathcal{K}^{1},\operatorname{red}\mathcal{L}^{1})\right\rangle \\ &= \left\langle \operatorname{red}\mathcal{D}(\mathcal{K}^{1},\mathcal{L}^{1})\right\rangle = \left\langle \mathcal{D}(\mathcal{K}^{1},\mathcal{L}^{1})\right\rangle = \left\langle \mathcal{D}(\mathcal{K},\mathcal{L})\right\rangle \end{split}$$

where the third equality is given by Lemma 4.2.2 (ii).

4.3.3. The Leibniz rule. — We formulate for $\widetilde{\Upsilon}$ a Leibniz-type rule in the second variable. (Since $\widetilde{\Upsilon}$ is symmetric in the sense of Lemma 4.3.2, a Leibniz-type rule in the first variable easily follows.) Pick a fifth base point $\star_5 \in \partial M$. For any *i*, *j*, $k \in \{1, \ldots, 5\}$, the concatenation of paths $c : \Omega_{ij} \times \Omega_{jk} \to \Omega_{ik}$ induces a bilinear *concatenation pairing*

$$(4.3.2) \qquad \qquad \widetilde{H}_*(\Omega_{ij}) \times \widetilde{H}_*(\Omega_{jk}) \longrightarrow \widetilde{H}_*(\Omega_{ik}), \ (a,b) \longmapsto ab = c_*(a \times b).$$

Similarly, for any $i, j, k, l, m \in \{1, ..., 5\}$, the map $c : \Omega_{ij} \times \Omega_{jk} \to \Omega_{ik}$ induces bilinear pairings

$$\begin{split} H_*(\Omega_{lm} \times \Omega_{ij}) \times H_*(\Omega_{jk}) &\longrightarrow H_*(\Omega_{lm} \times \Omega_{ik}), \quad (x, a) \longmapsto xa = (\mathrm{id} \times \mathrm{c})_*(x \times a), \\ \widetilde{H}_*(\Omega_{ij}) \times \widetilde{H}_*(\Omega_{jk} \times \Omega_{lm}) &\longrightarrow \widetilde{H}_*(\Omega_{ik} \times \Omega_{lm}), \quad (a, x) \longmapsto ax = (\mathrm{c} \times \mathrm{id})_*(a \times x). \end{split}$$

LEMMA 4.3.4. — If $\star_5 \in \partial M \setminus \{\star_1, \star_2\}$, then for any $a \in \widetilde{H}_p(\Omega_{12})$, $b \in \widetilde{H}_q(\Omega_{34})$, and $c \in \widetilde{H}_i(\Omega_{45})$ with $p, q, i \ge 0$,

$$\widetilde{\Upsilon}_{12,35}(a\otimes bc)=(-1)^{i}\widetilde{\Upsilon}_{12,34}(a\otimes b)c+(-1)^{(p+n+1)q}b\widetilde{\Upsilon}_{12,45}(a\otimes c).$$

Proof. — Let $\mathcal{K} = (K, \varphi, u, \kappa)$, $\mathcal{L} = (L, \psi, v, \lambda)$, $\mathcal{R} = (R, \chi, z, \rho)$ be smooth polycycles in $\Omega_{12}^{\circ}, \Omega_{34}^{\circ}, \Omega_{45}^{\circ}$ representing *a*, *b*, *c* respectively. Applying Lemma 4.1.4 twice (and choosing homotopy there sufficiently small), we can assume that \mathcal{K} is transversal to both \mathcal{L} and \mathcal{R} . Then *bc* is represented by the following polycycle in Ω_{35} :

$$\mathcal{N} = c_*(\mathcal{L} \times \mathcal{R}) = (L \times R, \psi \times \chi, v \times z, \eta)$$

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

81

where $\eta = c(\lambda \times \rho)$ and the adjoint map $\tilde{\eta} : L \times R \times I \to M$ is computed by

$$\widetilde{\eta}(l,r,t) = \begin{cases} \widehat{\lambda}(l,2t) & \text{for } l \in L, r \in R, t \in [0,1/2] \\ \widetilde{\rho}(r,2t-1) & \text{for } l \in L, r \in R, t \in [1/2,1] \end{cases}$$

The polycycles \mathcal{K} and \mathcal{N} are admissible in the sense of Section 4.3.2: we can take

 $U = K \times \text{Int}(I)$ and $V = N \times (\text{Int}(I) \setminus \{1/2\}).$

It follows from Lemma 4.3.3 that $\widetilde{\Upsilon}_{12,35}(a \otimes bc) = \langle \mathcal{D}(\mathcal{K}, \mathcal{N}) \rangle$. Thus, to prove the lemma, it is enough to show that

$$\begin{array}{ll} (4.3.3) \qquad \mathcal{D}(\mathcal{K},\mathcal{N}) \simeq (-1)^{i} (\operatorname{id} \times \operatorname{c})_{*} \big(\mathcal{D}(\mathcal{K},\mathcal{L}) \times \mathcal{R} \big) \\ \\ \sqcup \ (-1)^{(p+n+1)q} (\operatorname{c} \times \operatorname{id})_{*} \big(\mathcal{L} \times \mathcal{D}(\mathcal{K},\mathcal{R}) \big). \end{array}$$

To this end, we compare $\mathcal{D}(\mathcal{K}, \mathcal{N}) = (D, \theta, w, \kappa \triangleleft \eta)$ with

$$\mathcal{D}(\mathcal{K},\mathcal{L}) = (D',\theta',w',\kappa \diamond \lambda) \quad \text{and} \quad \mathcal{D}(\mathcal{K},\mathcal{R}) = (D',\theta',w,\kappa \diamond \rho).$$

Consider the embedding

$$P': (K \times I \times L \times I) \times R \longleftrightarrow K \times I \times (L \times R) \times I$$

defined by P'(k, s, l, t, r) = (k, s, l, r, t/2) and the embedding

$$P: L \times (K \times I \times R \times I) \longleftrightarrow K \times I \times (L \times R) \times I$$

defined by P(l, k, s, r, t) = (k, s, l, r, (t + 1)/2). Note that P' has degree $(-1)^i$ while P has degree $(-1)^{(p+1)q}$. Consider also the cartesian projections

$$\begin{aligned} & \mathrm{pr}': (K \times I \times L \times I) \times R \longrightarrow K \times I \times L \times I, \\ & \mathrm{'pr}: L \times (K \times I \times R \times I) \longrightarrow K \times I \times R \times I. \end{aligned}$$

Clearly, $(\tilde{\kappa} \times \tilde{\eta})P' = (\tilde{\kappa} \times \tilde{\lambda})$ pr'. Therefore, the map *P*' restricts to a diffeomorphism

$$D' \times R \longrightarrow P'(D' \times R) \subset D$$

of degree $(-1)^i$. Similarly, since $(\tilde{\kappa} \times \tilde{\eta})'P = (\tilde{\kappa} \times \tilde{\rho})'$ pr, the map '*P* restricts to a diffeomorphism

$$L \times D \longrightarrow P(L \times D) \subset D$$

of degree $(-1)^{(p+1)q+nq}$. Here we use the following general fact involving our orientation conventions stated in the Introduction: if X, Y are oriented manifolds and Sis an oriented submanifold of X, then the bundle map $\mathcal{V}_{X \times Y}(S \times Y) \rightarrow \mathcal{V}_X S$ induced by the cartesian projection $X \times Y \rightarrow X$ is an orientation-preserving isomorphism on each fiber, while the bundle map $\mathcal{V}_{Y \times X}(Y \times S) \rightarrow \mathcal{V}_X S$ induced by the cartesian projection $Y \times X \rightarrow X$ is orientation-preserving if and only if the product $(\dim X - \dim S) \cdot \dim(Y)$ is even.

It is clear from the definition of N and the computations of degrees above that

$$P' \sqcup 'P : (-1)^i (D' \times R) \sqcup (-1)^{(p+n+1)q} (L \times 'D) \longrightarrow D$$

is an orientation-preserving diffeomorphism. We claim that it transports the polychain structures of $(\operatorname{id} \times c)_* (\mathcal{D}(\mathcal{K}, \mathcal{L}) \times \mathcal{R})$ and $(c \times \operatorname{id})_* (\mathcal{L} \times \mathcal{D}(\mathcal{K}, \mathcal{R}))$ into the polychain

4.3. THE OPERATION $\widetilde{\Upsilon}$

structure of $\mathcal{D}(\mathcal{K},\mathbb{N})$ up to deformation of the latter. This will imply (4.3.3) and the lemma.

To prove our claim, we need to verify that $P' \sqcup P$ preserves the face partitions and the weights and commutes with the maps to the path spaces up to deformation. We start with the face partitions. Let F', G' be faces of D' of the same type and let H, J be faces of R of the same type. Then $F' \times H$ and $G' \times J$ are faces of $D' \times R$ of the same type. We claim that the faces $F = P'(F' \times H)$ and $G = P'(G' \times J)$ of D have the same type. By Section 4.2.1,

$$\theta'_{F'C'}:F'\longrightarrow G'$$

is the restriction of the diffeomorphism

$$\varphi_{A_{F'},A_{G'}} \times \operatorname{id} \times \psi_{B_{F'},B_{G'}} \times \operatorname{id} : N_{F'} = A_{F'} \times I \times B_{F'} \times I \longrightarrow A_{G'} \times I \times B_{G'} \times I = N_{G'}$$

to F' where $N_{F'}$ (respectively, $N_{G'}$) is the smallest face of $K \times I \times L \times I$ containing F' (respectively, G'). The smallest faces N_F and N_G of $K \times I \times (L \times R) \times I$ containing F and G respectively are

$$N_F = A_{F'} \times I \times (B_{F'} \times H) \times I$$
 and $N_G = A_{G'} \times I \times (B_{G'} \times J) \times I$.

Clearly, the diagram



commutes, so that the bottom diffeomorphism in that diagram carries *F* onto *G*. We deduce that *F* and *G* have the same type in *D* and the identification map $\theta_{F,G} : F \to G$ (which, by definition, is the restriction of the bottom diffeomorphism to *F*) satisfies

$$\theta_{F,G} \circ P'_{|F' \times H} = P'_{|G' \times J} \circ (\theta'_{F',G'} \times \chi_{H,J}).$$

This proves that P' carries the partition $\theta' \times \chi$ on $D' \times R$ to the partition θ restricted to $P'(D' \times R) \subset D$. A similar argument shows that '*P* carries the partition $\psi \times \theta$ on $L \times D$ to the partition θ restricted to ' $P(L \times D) \subset D$. It remains only to observe that a face of *D* lying in $P'(D' \times R)$ cannot have the same type as a face of *D* lying in ' $P(L \times D)$. To see this, we use the fact that every face *F* of *D* determines a smallest face

$$N_F = A_F \times I \times (B_F \times C_F) \times I$$

of $K \times I \times (L \times R) \times I$ such that $F \subset N_F$ and A_F, B_F, C_F are faces of K, L, R respectively. If F, G are faces of D of the same type, then A_F, B_F, C_F must have the same type as A_G, B_G, C_G respectively, and the diffeomorphism

$$\varphi_{A_F,A_G} \times \operatorname{id} \times (\psi_{B_F,B_G} \times \chi_{C_F,C_G}) \times \operatorname{id} : N_F \longrightarrow N_G$$

carries *F* onto *G*. Since this diffeomorphism preserves the last coordinate and

$$P'(D' \times R) \subset K \times I \times (L \times R) \times [0, 1/2], \quad P(L \times D) \subset K \times I \times (L \times R) \times [1/2, 1]$$

we deduce that *F* and *G* are both contained either in $P'(D' \times R)$ or in $P(L \times D)$.

We next show that the diffeomorphism $P' \sqcup P$ preserves the weights. Let W' be a connected component of D' and let Z be a connected component of R. The weight of the connected component $W' \times Z$ of $D' \times R$ is

$$(w' \times z)(W' \times Z) = w'(W')z(Z) = u(U)v(V)z(Z)$$

where *U* and *V* are connected components of *K* and *L*, respectively, such that $W' \subset U \times I \times V \times I$. Clearly, $P'(W' \times Z) \subset U \times I \times (V \times Z) \times I$ so that

$$w(P'(W' \times Z)) = u(U) \cdot (v \times z)(V \times Z) = u(U)v(V)z(Z) = (w' \times z)(W' \times Z).$$

This proves that P' carries the weight $w' \times z$ on $D' \times R$ to the weight w restricted to $P'(D' \times R)$. A similar argument shows that '*P* carries the weight $v \times w$ on $L \times D$ to the weight w restricted to '*P*($L \times D$).

We now show that $P' \sqcup P$ commutes with the maps to $\Omega_{32} \times \Omega_{15}$ up to deformation. The maps in question are $\kappa \rightsquigarrow \eta : D \to \Omega_{32} \times \Omega_{15}$ and $f \sqcup g$ where

$$(4.3.4) f = (\mathrm{id} \times \mathrm{c})((\kappa \rightsquigarrow \lambda) \times \rho) : D' \times R \longrightarrow \Omega_{32} \times \Omega_{15}$$

$$(4.3.5) g = (c \times id)(\lambda \times (\kappa \Rightarrow \rho)) : L \times D \longrightarrow \Omega_{32} \times \Omega_{15}$$

We first compute $(\kappa \Rightarrow \eta)P'$. Pick any $(k, s, l, t) \in D'$ and $r \in R$. For $x \in [0, 1/2]$,

$$(\kappa \rightsquigarrow \eta)(P'(k, s, l, t, r))(x) = (\kappa \stackrel{\sim}{\Rightarrow} \eta)(k, s, l, r, t/2, x)$$
$$= (\tilde{\eta}(l, r, (t/2) * x), \tilde{\kappa}(k, s * x))$$
$$= (\tilde{\eta}(l, r, tx), \tilde{\kappa}(k, 2sx))$$
$$= (\tilde{\lambda}(l, 2tx), \tilde{\kappa}(k, 2sx))$$
$$= (\tilde{\lambda}(l, t * x), \tilde{\kappa}(k, 2sx)).$$

Similarly, for $x \in [1/2, 1]$,

$$(\kappa \rightsquigarrow \eta) (P'(k, s, l, t, r))(x) = (\kappa \H \gg \eta)(k, s, l, r, t/2, x)$$
$$= (\widetilde{\kappa}(k, s \ast x), \widetilde{\eta}(l, r, (t/2) \ast x))$$
$$= (\widetilde{\kappa}(k, s \ast x), \widetilde{\eta}(l, r, 1 - (2 - t)(1 - x))).$$

We separate two cases depending on whether or not $1 - (2 - t)(1 - x) \le 1/2$ or, equivalently, on whether or not $x \le (3 - 2t)/(4 - 2t)$.

For $x \in [1/2, (3-2t)/(4-2t)]$, we obtain

$$(\kappa \rightsquigarrow \eta)(P'(k,s,l,t,r))(x) = (\widetilde{\kappa}(k,s*x),\lambda(l,2-2(2-t)(1-x))).$$

For $x \in [(3 - 2t)/(4 - 2t), 1]$, we obtain

$$(\kappa \rightsquigarrow \eta)(P'(k, s, l, t, r))(x) = (\widetilde{\kappa}(k, s \ast x), \widetilde{\rho}(r, 1 - 2(2 - t)(1 - x))).$$

These computations show that the first coordinate map $D' \times R \to \Omega_{32}$ of $(\kappa \rightsquigarrow \eta)P'$ is equal to $(\kappa \blacktriangleleft \lambda) \circ \operatorname{pr'}_{|D' \times R}$, which is also the first coordinate of the map f given by (4.3.4). The second coordinate maps $D' \times R \to \Omega_{15}$ of $(\kappa \rightsquigarrow \eta)P'$ and f may differ. Nonetheless, they are homotopic in the following way. For any $s, t, y \in I$, consider the numbers

$$0 < \frac{1}{4} \le a_y = \frac{1+y}{4} \le \frac{1}{2} \le b_{t,y} = \frac{1}{2} + y\frac{1-t}{4-2t} \le \frac{3-2t}{4-2t} < 1$$

4.3. THE OPERATION $\widetilde{\Upsilon}$

and let

$$\alpha_{s,y}:[0,a_y] \longrightarrow [0,s], \quad \beta_{t,y}:[a_y,b_{t,y}] \longrightarrow [t,1], \quad \gamma_{t,y}:[b_{t,y},1] \longrightarrow [0,1]$$

be the affine maps carrying the left/right endpoints of segments to the left/right endpoints respectively. We define a continuous map $e : D' \times R \times I \times I \rightarrow M$ by

(4.3.6)
$$e(k, s, l, t, r, x, y) = \begin{cases} \widetilde{\kappa}(k, \alpha_{s,y}(x)) & \text{if } x \in [0, a_y], \\ \widetilde{\lambda}(l, \beta_{t,y}(x)) & \text{if } x \in [a_y, b_{t,y}], \\ \widetilde{\rho}(r, \gamma_{t,y}(x)) & \text{if } x \in [b_{t,y}, 1]. \end{cases}$$

Observing that $a_0 = 1/4$, $b_{t,0} = 1/2$ and $a_1 = 1/2$, $b_{t,1} = (3-2t)/(4-2t)$, we conclude that *e* determines a homotopy between the second coordinate maps of *f* and $(\kappa \Rightarrow \eta)P'$ in the class of maps $D' \times R \to \Omega_{15}$. It remains to check that this homotopy is compatible with the partition $\theta' \times \chi$ on $D' \times R$. Any faces *F*, *G* of $D' \times R$ of the same type expand as $F = F' \times H$ and $G = G' \times J$ where F', G' are faces of D' of the same type and H, Jare faces of *R* of the same type. Let

$$N_{F'} = A_{F'} \times I \times B_{F'} \times I$$
 and $N_{G'} = A_{G'} \times I \times B_{G'} \times I$

be the smallest faces of $K \times I \times L \times I$ containing F' and G' respectively. The identifying map $(\theta' \times \chi)_{F,G} : F \to G$ is the restriction of the diffeomorphism

$$(\varphi_{A_{F'},A_{G'}} \times \operatorname{id} \times \psi_{B_{F'},B_{G'}} \times \operatorname{id}) \times \chi_{H,J} : N_{F'} \times H \longrightarrow N_{G'} \times J.$$

Since the maps κ , λ , ρ are compatible with the partitions φ , ψ , χ respectively, we deduce from (4.3.6) that for any $(k, s, l, t) \in F'$, $r \in H$, and $x, y \in I$,

$$e((\theta' \times \chi)_{F,G}(k, s, l, t, r), x, y) = e(\varphi_{A_{F'}, A_{G'}}(k), s, \psi_{B_{F'}, B_{G'}}(l), t, \chi_{H,J}(r), x, y) = e(k, s, l, t, r, x, y).$$

Hence for each $y \in I$, the map

$$D' \times R \longrightarrow \Omega_{15}, \quad (k, s, l, t, r) \longmapsto (x \mapsto e(k, s, l, t, r, x, y))$$

is compatible with the partition $\theta' \times \chi$. We conclude that the homotopy of f to $(\kappa \Rightarrow \eta)P'$ determined by e is compatible with the partition $\theta' \times \chi$. One similarly constructs a deformation of the map (4.3.5) into $(\kappa \Rightarrow \eta)'P$ compatible with the partition.

4.3.4. *Change of base points.* — Consider one more tuple $(\star'_1, \star'_2, \star'_3, \star'_4)$ of points of ∂M such that $\{\star'_1, \star'_2\} \cap \{\star'_3, \star'_4\} = \emptyset$ and set $\Omega'_{ij} = \Omega(M, \star'_i, \star'_j)$. Section 4.3.1 yields a linear map

$$\widetilde{\Upsilon}':\widetilde{H}_*(\Omega_{12}')\otimes\widetilde{H}_*(\Omega_{34}')\longrightarrow\widetilde{H}_*(\Omega_{32}'\times\Omega_{14}').$$

We compare $\widetilde{\Upsilon}'$ to the map $\widetilde{\Upsilon}: \widetilde{H}_*(\Omega_{12}) \otimes \widetilde{H}_*(\Omega_{34}) \to \widetilde{H}_*(\Omega_{32} \times \Omega_{14})$ assuming that \star_i and \star'_i belong to the same connected component of ∂M for all $i \in \{1, 2, 3, 4\}$.

Choose a path $\varsigma_i : I \to \partial M$ from \star_i to \star'_i for each *i*. The formula $\gamma \mapsto \varsigma_i^{-1} \gamma \varsigma_j$ defines a continuous map $(\varsigma_i, \varsigma_j)_{\#}$ from Ω_{ij} to $\Omega_{i'j'}$. Homotopic paths yield homotopic maps, and constant paths yield maps homotopic to the identity. Therefore $(\varsigma_i, \varsigma_j)_{\#}$

is a homotopy equivalence with homotopy inverse $(\varsigma_i^{-1}, \varsigma_j^{-1})_{\#}$. The homotopy equivalence $(\varsigma_i, \varsigma_j)_{\#}$ induces an isomorphism in the face homology which we denote by the same symbol:

(4.3.7)
$$(\varsigma_i, \varsigma_j)_{\#} : \widetilde{H}_*(\Omega_{ij}) \xrightarrow{\simeq} \widetilde{H}_*(\Omega'_{ij}).$$

Similarly, the isomorphism $\widetilde{H}_*(\Omega_{ij} \times \Omega_{kl}) \to \widetilde{H}_*(\Omega'_{ij} \times \Omega'_{kl})$ induced by the homotopy equivalence $(\varsigma_i, \varsigma_j)_{\#} \times (\varsigma_k, \varsigma_l)_{\#}$ is also denoted by $(\varsigma_i, \varsigma_j)_{\#} \times (\varsigma_k, \varsigma_l)_{\#}$.

LEMMA 4.3.5. — If $n \ge 3$, then the following diagram commutes:

$$(4.3.8) \qquad \begin{split} \widetilde{H}_{*}(\Omega_{12}) \otimes \widetilde{H}_{*}(\Omega_{34}) & \xrightarrow{\widetilde{\Upsilon}} & \widetilde{H}_{*}(\Omega_{32} \times \Omega_{14}) \\ (\mathfrak{c}_{1,\mathfrak{c}_{2}})_{\#} \otimes (\mathfrak{c}_{3,\mathfrak{c}_{4}})_{\#} & \swarrow & \swarrow \\ \widetilde{H}_{*}(\Omega_{12}') \otimes \widetilde{H}_{*}(\Omega_{34}') & \xrightarrow{\widetilde{\Upsilon}'} & \widetilde{H}_{*}(\Omega_{32}' \times \Omega_{14}'). \end{split}$$

Proof. — Since the isomorphism $(\varsigma_i, \varsigma_j)_{\#}$ depends only on the homotopy classes of the paths ς_i , ς_j , and since composition of the paths leads to composition of the corresponding isomorphisms, it is enough to consider the case where three of the paths ς_i 's are constant. Assume for concreteness that $\star_1 = \star'_1, \star_2 = \star'_2, \star_3 = \star'_3$, and $\varsigma_1, \varsigma_2, \varsigma_3$ are constant paths. The assumption $n \ge 3$ implies that deforming if necessary the path $\varsigma = \varsigma_4$, we can ensure that $\varsigma(I) \subset \partial M \setminus \{\star_1, \star_2\}$.

Let $a \in \tilde{H}_p(\Omega_{12})$ and $b \in \tilde{H}_q(\Omega_{34})$. Consider smooth polycycles $\mathcal{K} = (K, \varphi, u, \kappa)$ in Ω_{12}° and $\mathcal{L} = (L, \psi, v, \lambda)$ in Ω_{34}° transversely representing the pair (a, b). The class $(1, \varsigma)_{\#}(b) \in \tilde{H}(\Omega'_{34})$ is represented by the polycycle $\mathcal{L}' = (1, \varsigma)_{\#}\mathcal{L}$ in Ω'_{34} (but not in Ω_{34}°). The polycycles \mathcal{K} and \mathcal{L}' are admissible in the sense of Section 4.3.2: we can take $U = K \times \text{Int}(I)$ and $V = L \times (0, 1/2)$. Set

$$\mathcal{D}(\mathcal{K},\mathcal{L}) = (D,\theta,w,\kappa \diamond \lambda) \text{ and } \mathcal{D}(\mathcal{K},\mathcal{L}') = (D',\theta',w',\kappa \diamond \lambda')$$

where $\lambda' = (1, \varsigma)_{\sharp} \lambda$. It is easy to construct a diffeomorphism $f : D \to D'$ preserving the orientation, the weight, and the face partition, and such that $(\kappa \gg \lambda') \circ f$ is homotopic to $(\operatorname{id} \times (1, \varsigma)_{\#}) \circ (\kappa \twoheadrightarrow \lambda)$ in the class of maps $D \to \Omega_{32} \times \Omega'_{14}$ compatible with θ . Lemma 4.3.3 implies that

$$\begin{split} \widetilde{\Upsilon}'(a \otimes (1,\varsigma)_{\#}(b)) &= \langle \mathcal{D}(\mathcal{K},\mathcal{L}') \rangle \\ &= \left\langle (\operatorname{id} \times (1,\varsigma)_{\#}) \mathcal{D}(\mathcal{K},\mathcal{L}) \right\rangle \\ &= \left(\operatorname{id} \times (1,\varsigma)_{\#} \right) \left\langle \mathcal{D}(\mathcal{K},\mathcal{L}) \right\rangle = \left(\operatorname{id} \times (1,\varsigma)_{\#} \right) \widetilde{\Upsilon}(a,b). \end{split}$$

This proves the commutativity of the diagram (4.3.8).

4.3.5. *Extension of* $\widetilde{\mathbf{T}}$. — Assuming that $n \ge 3$, we extend the definition of $\widetilde{\mathbf{T}}$ to all 4-tuples of points $\star_1, \star_2, \star_3, \star_4 \in \partial M$. Deforming these points in ∂M , we can obtain points $\star'_1, \star'_2, \star'_3, \star'_4 \in \partial M$ such that $\{\star'_1, \star'_2\} \cap \{\star'_3, \star'_4\} = \emptyset$. For $i = 1, \ldots, 4$, pick a path $\varsigma_i : I \to \partial M$ from \star_i to \star'_i . Section 4.3.1 yields a linear map

$$\Upsilon': \tilde{H}_*(\Omega'_{12}) \otimes \tilde{H}_*(\Omega'_{34}) \longrightarrow \tilde{H}_*(\Omega'_{32} \times \Omega'_{14})$$

where $\Omega'_{ii} = \Omega(M, \star'_i, \star'_i)$ for all *i*, *j*. Then we define

$$\widetilde{\Upsilon} = \widetilde{\Upsilon}_{12,34} : \widetilde{H}_*(\Omega_{12}) \otimes \widetilde{H}_*(\Omega_{34}) \longrightarrow \widetilde{H}_*(\Omega_{32} \times \Omega_{14})$$

to be the unique linear map such that the diagram (4.3.8) commutes. Lemma 4.3.5 implies that this map depends neither on the choice of the paths ς_1 , ς_2 , ς_3 , ς_4 nor on the choice of the points \star'_1 , \star'_2 , \star'_3 , \star'_4 . If $\{\star_1, \star_2\} \cap \{\star_3, \star_4\} = \emptyset$, then we can take $\star'_i = \star_i$ and the constant path ς_i for all *i*, and recover the same map $\widetilde{\Upsilon}$ as before.

The properties of $\widetilde{\Upsilon}$ established under the assumption $\{\star_1, \star_2\} \cap \{\star_3, \star_4\} = \emptyset$ remain true for arbitrary base points in ∂M . This easily follows from the definitions and the fact that the concatenation pairing (4.3.2) is preserved under the change-of-base-points isomorphism (4.3.7).

4.3.6. Renormalization. — We will use a renormalized version

(4.3.9)
$$\check{\Upsilon} = \check{\Upsilon}_{12,34} : \widetilde{H}_*(\Omega_{12}) \otimes \widetilde{H}_*(\Omega_{34}) \longrightarrow \widetilde{H}_*(\Omega_{32} \times \Omega_{14})$$

of $\widetilde{\Upsilon}$ defined by

$$\check{\Upsilon}(a \otimes b) = (-1)^{|b|+n|a|} \widetilde{\Upsilon}(a \otimes b)$$

for any homogeneous $a \in \widetilde{H}_*(\Omega_{12})$ and $b \in \widetilde{H}_*(\Omega_{34})$. The properties of $\widetilde{\Upsilon}$ can be rephrased for Υ . In particular, Lemma 4.3.2 yields the identity

(4.3.10)
$$p_* \check{\Upsilon}_{12,34}(a \otimes b) = -(-1)^{|a|n \cdot |b|n} \check{\Upsilon}_{34,12}(b \otimes a)$$

where $|-|_n = |-| + n$ is the *n*-degree. Also, for any $\star_5 \in \partial M$ (distinct from \star_1 and \star_2 if n = 2) and any homogeneous $a \in \widetilde{H}_*(\Omega_{12})$, $b \in \widetilde{H}_*(\Omega_{34})$, $c \in \widetilde{H}_*(\Omega_{45})$, Lemma 4.3.4 yields the Leibniz rule

(4.3.11)
$$\mathring{\Upsilon}_{12,35}(a \otimes bc) = \mathring{\Upsilon}_{12,34}(a \otimes b)c + (-1)^{|a|_{R} \cdot |b|}b \mathring{\Upsilon}_{12,45}(a \otimes c).$$

Finally, the diagram (4.3.8) remains commutative with $\tilde{\Upsilon}$ replaced by $\check{\Upsilon}$.

4.4. The operation Υ

We derive from $\check{\Upsilon}$ an operation Υ in singular homology. In this section we drop the assumption $\{\star_1, \star_2\} \cap \{\star_3, \star_4\} = \emptyset$ when $n \ge 3$.

4.4.1. *Definition and properties of* Υ . — Consider the linear map

$$(4.4.1) \qquad \Upsilon = \Upsilon_{12,34} : H_*(\Omega_{12}) \otimes H_*(\Omega_{34}) \longrightarrow H_*(\Omega_{32} \times \Omega_{14})$$

defined by the commutative diagram

$$\begin{split} \widetilde{H}_*(\Omega_{12}) \otimes \widetilde{H}_*(\Omega_{34}) & \xrightarrow{\Upsilon} & \widetilde{H}_*(\Omega_{32} \times \Omega_{14}) \\ & \swarrow^{(-) \times \langle - \rangle} & & \downarrow^{[-]} \\ & & \downarrow^{[-]} \\ & H_*(\Omega_{12}) \otimes H_*(\Omega_{34}) - - -\overset{\Upsilon}{-} - \to H_*(\Omega_{32} \times \Omega_{14}). \end{split}$$

Formula (4.3.10) and the naturality of the transformation $[-] : \widetilde{H}_* \to H_*$ imply the following antisymmetry of Υ : for any homogeneous $a \in H_*(\Omega_{12}), b \in H_*(\Omega_{34})$,

(4.4.2)
$$p_*\Upsilon_{12,34}(a\otimes b) = -(-1)^{|a|_n \cdot |b|_n}\Upsilon_{34,12}(b\otimes a)$$

where $p_* : H_*(\Omega_{32} \times \Omega_{14}) \to H_*(\Omega_{14} \times \Omega_{32})$ is the linear map induced by the permutation map $p : \Omega_{32} \times \Omega_{14} \to \Omega_{14} \times \Omega_{32}$.

If $n \ge 3$, then the diagram (4.3.8) with $\widetilde{\Upsilon}$ replaced by $\check{\Upsilon}$ and the naturality of the transformations $\langle - \rangle$ and [-] imply that the following diagram commutes:

$$(4.4.3) \qquad \begin{array}{c} & H_*(\Omega_{12}) \otimes H_*(\Omega_{34}) & \xrightarrow{\Upsilon} & H_*(\Omega_{32} \times \Omega_{14}) \\ (\mathfrak{c}_{1,\mathfrak{c}_2})_{\#} \otimes (\mathfrak{c}_{3,\mathfrak{c}_4})_{\#} & \swarrow & \swarrow & (\mathfrak{c}_{3,\mathfrak{c}_2})_{\#} \times (\mathfrak{c}_{1,\mathfrak{c}_4})_{\#} \\ & H_*(\Omega_{12}') \otimes H_*(\Omega_{34}') & \xrightarrow{\Upsilon'} & H_*(\Omega_{32}' \times \Omega_{14}'). \end{array}$$

Here, for every $i \in \{1, 2, 3, 4\}$, \star'_i is a point of ∂M connected to \star_i by a path $\varsigma_i : I \to \partial M$, Υ' is the map (4.4.1) determined by the base points $\star'_1, \star'_2, \star'_3, \star'_4$, and, for all $i, j \in \{1, 2, 3, 4\}$, $(\varsigma_i, \varsigma_j)_{\#}$ stands for the homotopy equivalence

 $\Omega_{ij} \to \Omega'_{ij} = \Omega(M, \star'_i, \star'_j), \gamma \longmapsto \varsigma_i^{-1} \gamma \varsigma_j$

and for the induced isomorphism in singular homology.

The following crucial lemma will be proved in Section 4.4.3.

LEMMA 4.4.1. — The following diagram commutes:

~

(4.4.4)
$$\begin{array}{c} \widetilde{H}_{*}(\Omega_{12}) \times \widetilde{H}_{*}(\Omega_{34}) & \xrightarrow{\Upsilon} & \widetilde{H}_{*}(\Omega_{32} \times \Omega_{14}) \\ & & & & \\ & & & \\ & & & \\ & & & & \\ & & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\ & & & \\$$

4.4.2. The Leibniz rule for Υ . — As in Section 4.3.3 in the case of face homology, the concatenation of paths induces three kinds of bilinear pairings in singular homology:

$$(4.4.5) \qquad \begin{cases} H_*(\Omega_{ij}) \times H_*(\Omega_{jk}) \longrightarrow H_*(\Omega_{ik}), & (a,b) \longmapsto ab = c_*(a \times b), \\ H_*(\Omega_{lm} \times \Omega_{ij}) \times H_*(\Omega_{jk}) \longrightarrow H_*(\Omega_{lm} \times \Omega_{ik}), \\ (x,a) \longmapsto xa = (\mathrm{id} \times c)_*(x \times a), \\ H_*(\Omega_{ij}) \times H_*(\Omega_{jk} \times \Omega_{lm}) \longrightarrow H_*(\Omega_{ik} \times \Omega_{lm}), \\ (a,x) \longmapsto ax = (c \times \mathrm{id})_*(a \times x) \end{cases}$$

LEMMA 4.4.2. — For any $\star_5 \in \partial M$ (distinct from \star_1 and \star_2 if n = 2) and any homogeneous $a \in H_*(\Omega_{12}), b \in H_*(\Omega_{34}), c \in H_*(\Omega_{45})$,

(4.4.6)
$$\Upsilon_{12,35}(a \otimes bc) = \Upsilon_{12,34}(a \otimes b)c + (-1)^{|a|_{n} \cdot |b|}b\Upsilon_{12,45}(a \otimes c).$$

Proof. — For any
$$x \in H_*(\Omega_{ij})$$
, $y \in H_*(\Omega_{jk})$ with $i, j, k \in \{1, ..., 5\}$, we have
(4.4.7) $[xy] = [c_*(x \times y)] = c_*[x \times y] = c_*([x] \times [y]) = [x][y]$

$$bc = [\langle b \rangle] [\langle c \rangle] = [\langle b \rangle \langle c \rangle].$$

We deduce that

$$\begin{split} \Upsilon_{12,35}(a \otimes bc) &= \Upsilon_{12,35}\big([\langle a \rangle] \otimes [\langle b \rangle \langle c \rangle]\big) \\ &= \big[\check{\Upsilon}_{12,35}(\langle a \rangle, \langle b \rangle \langle c \rangle)\big] \\ &= \big[\check{\Upsilon}_{12,35}(\langle a \rangle, \langle b \rangle) \langle c \rangle + (-1)^{|a|n \cdot |b|} \langle b \rangle \check{\Upsilon}_{12,35}(\langle a \rangle, \langle c \rangle)\big] \\ &= \big[\check{\Upsilon}_{12,35}(\langle a \rangle, \langle b \rangle)\big]c + (-1)^{|a|n \cdot |b|} b\big[\check{\Upsilon}_{12,35}(\langle a \rangle, \langle c \rangle)\big] \\ &= \Upsilon_{12,35}(a,b)c + (-1)^{|a|n \cdot |b|} b\Upsilon_{12,35}(a,c) \end{split}$$

where the second, third, fourth and fifth formulas follow respectively from (4.4.4), (4.3.11), (4.4.7), and the definition of Υ .

4.4.3. *Proof of Lemma* **4.4.1.** — We claim that, for all $a \in \widetilde{H}_*(\Omega_{12})$ and $b \in \widetilde{H}_*(\Omega_{34})$,

(4.4.8)
$$\left[\Upsilon(\langle [a]\rangle, b)\right] = \left[\Upsilon(a, b)\right] = \left[\Upsilon(a, \langle [b]\rangle)\right].$$

This would imply similar equalities with $\widetilde{\Upsilon}$ replaced by $\check{\Upsilon}$. Therefore

$$\Upsilon([a], [b]) = [\check{\Upsilon}(\langle [a] \rangle, \langle [b] \rangle)] = \left[\check{\Upsilon}(a, \langle [b])\right] = \left[\check{\Upsilon}(a, b)\right]$$

which proves the commutativity of the diagram (4.4.4).

We prove the first equality in (4.4.8); the second equality follows by the symmetry of $\tilde{\Upsilon}$ (Lemma 4.3.2). By Section 4.3.5, we can assume that $\{\star_1, \star_2\} \cap \{\star_3, \star_4\} = \emptyset$. We need to prove that, for any smooth polycycle $\mathcal{K} = (K, \varphi, u, \kappa)$ in Ω_{12}° and any smooth polycycle $\mathcal{L} = (L, \psi, v, \lambda)$ in Ω_{34}° transversal to \mathcal{K} ,

(4.4.9)
$$\left[\widetilde{\Upsilon}(\langle [\mathcal{K}] \rangle, \langle \mathcal{L} \rangle)\right] = \left[\mathcal{D}(\mathcal{K}, \mathcal{L})\right].$$

Set $p = \dim(K)$. Pick a locally ordered smooth triangulation *T* of *K* which fits φ . The construction of such a triangulation in Section 3.3.2 (using Lemma 3.1.1) shows that we can further assume that (*) $F \cap \tau$ is a face of τ for any face *F* of *K* and any simplex τ of *T* (*cf.* the last paragraph in the proof of Lemma 3.1.1). Consider the fundamental *p*-chain

$$\sigma = \sigma(T, u) = \sum_{\Delta} \varepsilon_{\Delta} u(K^{\Delta}) \sigma_{\Delta} \in C_p(K)$$

determined by *T* as in Section 3.3.2 (here Δ runs over all *p*-simplices of *T*). Then $\kappa_*(\sigma)$ is a smooth singular *p*-cycle in Ω_{12}° representing the singular homology class

$$[\mathcal{K}] \in H_p(\Omega_{12}^\circ).$$

Next consider the smooth *p*-polycycle

$$\mathcal{K}' = (K', \varphi', u', \kappa')$$

in Ω_{12}° associated with the expansion

$$\kappa_*(\sigma) = \sum_\Delta arepsilon_\Delta u(K^\Delta) \kappa \sigma_\Delta$$

as in Section 3.3.3. By construction, K' is a disjoint union of copies of the standard p-simplex Δ^p indexed by p-simplices Δ of T and $\kappa' = \kappa \zeta$ where $\zeta = \coprod_{\Delta} \sigma_{\Delta} : K' \to K$. By the definition of the transformation $\langle - \rangle : H_* \to \widetilde{H}_{**}$

$$\langle [\mathcal{K}] \rangle = \langle [\kappa_*(\sigma)] \rangle = \langle \mathcal{K}' \rangle \in \widetilde{H}_p(\Omega_{12}),$$

so that (4.4.9) is equivalent to

(4.4.10)
$$\left[\widetilde{\Upsilon}(\langle \mathcal{K}' \rangle, \langle \mathcal{L} \rangle)\right] = \left[\mathcal{D}(\mathcal{K}, \mathcal{L})\right].$$

Lemma 4.1.4 yields a deformation of \mathcal{L} into a polycycle \mathcal{L}^1 transversal to \mathcal{K}' . Such a polycycle \mathcal{L}^1 is also transversal to \mathcal{K} . By Lemma 3.2.3, $\langle \mathcal{L} \rangle = \langle \mathcal{L}^1 \rangle$ so that

$$\left[\widetilde{\Upsilon}(\langle \mathcal{K}' \rangle, \langle \mathcal{L} \rangle)\right] = \left[\widetilde{\Upsilon}(\langle \mathcal{K}' \rangle, \langle \mathcal{L}^1 \rangle)\right]$$

By Lemma 4.3.1, the polycycles $\mathcal{D}(\mathcal{K}, \mathcal{L})$ and $\mathcal{D}(\mathcal{K}, \mathcal{L}^1)$ are homologous so that

$$\left[\mathcal{D}(\mathcal{K},\mathcal{L})\right] = \left[\mathcal{D}(\mathcal{K},\mathcal{L}^1)\right].$$

Thus, in order to prove (4.4.10), we may assume without loss of generality that \mathcal{L} is transversal to \mathcal{K}' . We need to prove that

(4.4.11)
$$\begin{bmatrix} \mathcal{D}(\mathcal{K}',\mathcal{L}) \end{bmatrix} = \begin{bmatrix} \mathcal{D}(\mathcal{K},\mathcal{L}) \end{bmatrix}.$$

Set $\mathcal{D}(\mathcal{K},\mathcal{L}) = (D,\theta,w,\kappa \rightsquigarrow \lambda), \mathcal{D}(\mathcal{K}',\mathcal{L}) = (D',\theta',w',\kappa' \rightsquigarrow \lambda)$ and
 $\bar{\zeta} = \zeta \times \operatorname{id}_I \times \operatorname{id}_L \times \operatorname{id}_I : K' \times I \times L \times I \longrightarrow K \times I \times L \times I.$

It follows from the definition that $D' = \overline{\zeta}^{-1}(D)$. Thus D' is obtained by cutting D into pieces, each "piece" being a connected component of $D \cap (\Delta \times I \times L \times I)$ where Δ is a *p*-simplex of *T*. The map $\overline{\zeta}_{|D'} : D' \to D$ is the obvious gluing map. It is surjective and its restriction to every connected component of D' is injective.

Consider the equivalence relations \sim_{θ} on D and $\sim_{\theta'}$ on D' defined by the partitions (see Section 3.1.2). We claim that if some points $d_1, d_2 \in D'$ satisfy $\overline{\zeta}(d_1) \sim_{\theta} \overline{\zeta}(d_2)$, then $d_1 \sim_{\theta'} d_2$. We now check this claim. Assume that for i = 1, 2,

$$d_i = (k'_i, s, l_i, t) \in D' \subset K' \times I \times L \times I$$
, and set $k_i = \zeta(k'_i) \in K$.

By assumption, there exist faces F_1 , F_2 of D of the same type such that

$$\overline{\zeta}(d_i) = (k_i, s, l_i, t) \in F_i \quad \text{for } i = 1, 2$$

and $\theta_{F_1,F_2} : F_1 \to F_2$ carries $\overline{\zeta}(d_1)$ to $\overline{\zeta}(d_2)$. For i = 1, 2, let A_i and B_i be faces of K and L respectively such that $A_i \times I \times B_i \times I$ is the smallest face of $K \times I \times L \times I$ containing F_i . Then $k_i \in A_i$, $l_i \in B_i$, A_1 has the same type as A_2 , B_1 has the same type as B_2 , and

$$\varphi_{A_1,A_2}(k_1) = k_2, \quad \psi_{B_1,B_2}(l_1) = l_2$$

To proceed, let $\Delta_i \approx \Delta^p$ be the connected component of k'_i in K', and let

$$\sigma_i = \sigma_{\Delta_i} : \Delta^p \longrightarrow K$$

be the corresponding singular simplex (which is a simplicial isomorphism onto a *p*-simplex of the triangulation *T*). Then $k_i \in A_i \cap \sigma_i(\Delta_i)$. By the assumption (*) above, $A_i \cap \sigma_i(\Delta_i)$ is a face of the *p*-simplex $\sigma_i(\Delta_i)$. Since *T* fits φ , the sets

$$au_1=\sigma_1(\Delta_1)\cap arphi_{A_2,A_1}(A_2\cap \sigma_2(\Delta_2)), \qquad au_2=\sigma_2(\Delta_2)\cap arphi_{A_1,A_2}(A_1\cap \sigma_1(\Delta_1))$$

are faces of the *p*-simplices $\sigma_1(\Delta_1), \sigma_2(\Delta_2)$ containing k_1, k_2 respectively.

The map $\varphi_{A_1,A_2} : A_1 \to A_2$ restricts to a simplicial isomorphism $\varphi_{12} : \tau_1 \to \tau_2$ preserving the order of the vertices and carrying k_1 to k_2 . Set $r = \dim(\tau_1) = \dim(\tau_2)$. Then $\tau'_i = \sigma_i^{-1}(\tau_i)$ is an *r*-dimensional face of the *p*-simplex Δ_i containing k'_i for i = 1, 2. The map

$$\varphi_{12}' = \sigma_2^{-1} \varphi_{12} \sigma_1_{|\tau_1'} : \tau_1' \longrightarrow \tau_2'$$

is an order-preserving simplicial isomorphism and $\varphi'_{12}(k'_1) = k'_2$. Since Δ_1, Δ_2 are copies of the standard *p*-simplex Δ^p , their faces τ'_1, τ'_2 correspond to certain (r + 1)-element subsets S_1, S_2 of the set $\{0, \ldots, p\}$. Since $\kappa \varphi_{A_1, A_2} = \kappa_{|A_1|}$ we have

$$(\kappa\sigma_1) \circ e_{S_1} = (\kappa\sigma_2) \circ e_{S_2} : \Delta^r \longrightarrow \Omega_{12}^\circ.$$

By the definition of \mathcal{K}' , the latter equality implies that the faces τ'_1 , τ'_2 of K' have the same type and $\varphi'_{\tau'_1,\tau'_2} = \varphi'_{12} : \tau'_1 \to \tau'_2$. Consider now the map

$$\Theta = \left(\varphi_{\tau_1',\tau_2'}' \times \operatorname{id} \times \psi_{B_1,B_2} \times \operatorname{id}\right) : \tau_1' \times I \times B_1 \times I \to \tau_2' \times I \times B_2 \times I.$$

We have

$$\Theta(d_1) = (\varphi'_{12}(k'_1), s, \psi_{B_1, B_2}(l_1), t) = d_2.$$

Let G_i be the connected component of d_i in $D' \cap (\tau'_i \times I \times B_i \times I)$. The equality $\Theta(d_1) = d_2$ implies that $\Theta(G_1) = G_2$. Thus, G_1 and G_2 are faces of D' of the same type, and the identification map $\theta'_{G_1,G_2} = \Theta_{|G_1|}$ carries d_1 to d_2 . This proves that $d_1 \sim_{\theta'} d_2$ as claimed.

Consider the canonical projections $\pi : D \to D_{\theta}$ and $\pi' : D' \to D'_{\theta'}$. The previous claim implies that there exists a unique map *g* such that the diagram



commutes. The map *g* is continuous because π' is continuous and $\overline{\zeta}, \pi$ are quotient maps. Set

$$S = \bar{\zeta}(\partial D') = D \cap T^{p-1}$$

where T^{p-1} is the (p-1)-skeleton of *T*. Clearly, $\partial D \subset S$. Consider the commutative diagram

$$\begin{array}{c} H_*(D_{\theta}, (\partial D)_{\theta}) \xleftarrow{\pi_*} H_*(D, \partial D) \ni [D, w] \\ \downarrow \\ [D_{\theta}, w] \in H_*(D_{\theta}) \longrightarrow H_*(D_{\theta}, S_{\theta}) \xleftarrow{\pi_*} H_*(D, S) \ni (\bar{\zeta}_{|D'})_*([D', w']) \\ g_* \downarrow \\ g_* \downarrow \\ [D_{\theta'}, w'] \in H_*(D_{\theta'}') \longrightarrow H_*(D_{\theta'}', (\partial D')_{\theta'}) \xleftarrow{\pi_*'} H_*(D', \partial D') \ni [D', w'] \end{array}$$

where the unlabelled arrows are the inclusion maps. The definition of \mathcal{K}' implies that the weight $u' : \pi_0(K') \to \mathbb{K}$ of \mathcal{K}' is the composition of $\zeta_{\#} : \pi_0(K') \to \pi_0(K)$ with the weight $u : \pi_0(K) \to \mathbb{K}$ of \mathcal{K} . Using the definition of the operation \mathcal{D} , we deduce

that the weight $w' : \pi_0(D') \to \mathbb{K}$ is the composition of $(\bar{\zeta}_{|D'})_{\#} : \pi_0(D') \to \pi_0(D)$ with the weight $w : \pi_0(D) \to \mathbb{K}$. This fact and the definition of [D', w'], [D, w] imply that $(\bar{\zeta}_{|D'})_* ([D', w'])$ is the image of [D, w] in $H_*(D, S)$. By Lemma 3.3.1, the image of $[D_{\theta}, w]$ in $H_*(D_{\theta}, (\partial D)_{\theta})$ is equal to $\pi_*([D, w])$. Using this fact, the uniqueness in Lemma 3.3.1, and a simple diagram chasing we obtain that

(4.4.12)
$$g_*([D_{\theta}, w]) = [D'_{\theta'}, w'] \in H_*(D'_{\theta'}).$$

Next, we verify that

 $(4.4.13) \qquad (\kappa' \rightsquigarrow \lambda)_{\theta'}g = (\kappa \rightsquigarrow \lambda)_{\theta} : D_{\theta} \longrightarrow \Omega_{32} \times \Omega_{14}.$ Given $d = (k, s, l, t) \in D$, we have $g\pi(d) = \pi'(k', s, l, t)$ for any $k' \in \zeta^{-1}(k) \subset K'$. Then $(\kappa' \rightsquigarrow \lambda)_{\theta'}g(\pi(d)) = (\kappa' \rightsquigarrow \lambda)(k', s, l, t) = (\kappa \rightsquigarrow \lambda)(k, s, l, t) = (\kappa \rightsquigarrow \lambda)_{\theta}(\pi(d))$

where we use the equality $\kappa'(k') = \kappa(k)$. Since $\pi : D \to D_{\theta}$ is onto, we conclude that (4.4.13) holds. This and (4.4.12) imply (4.4.11):

$$\begin{split} \left[\mathcal{D}(\mathcal{K}, \mathcal{L}) \right] &= \left((\kappa \rightsquigarrow \lambda)_{\theta} \right)_* \left([D_{\theta}, w] \right) \\ &= \left((\kappa' \rightsquigarrow \lambda)_{\theta'} \right)_* g_* \left([D_{\theta}, w] \right) \\ &= \left((\kappa' \rightsquigarrow \lambda)_{\theta'} \right)_* \left([D'_{\theta'}, w'] \right) = \left[\mathcal{D}(\mathcal{K}', \mathcal{L}) \right]. \end{split}$$

CHAPTER 5

THE INTERSECTION BIBRACKET

Throughout this chapter, *M* is an oriented smooth *n*-dimensional manifold with non-void boundary, where $n \ge 2$.

5.1. Construction of the intersection bibracket

We introduce the path homology category of M and define the intersection bibracket in this category.

5.1.1. *The path homology category.* — Let C = C(M) be the graded category whose set of objects is ∂M and whose graded modules of morphisms are defined by

$$\operatorname{Hom}_{\mathcal{C}}(\star, \star') = H_*(\Omega(M, \star, \star'))$$

for any $\star, \star' \in \partial M$. Composition in \mathbb{C} is the pairing (4.4.5) defined via concatenation of paths. For $\star \in \partial M$, the identity morphism of \star in \mathbb{C} is the element of $H_0(\Omega(M, \star, \star))$ represented by the constant path in \star . We call \mathbb{C} the *path homology category* of M. By Section 2.2.1, this category determines a graded algebra

(5.1.1)
$$A = A(\mathcal{C}) = \bigoplus_{\star,\star' \in \partial M} H_*(\Omega(M,\star,\star')).$$

The subcategory \mathbb{C}^0 of \mathbb{C} formed by all objects and morphisms of degree 0 can be formulated in terms of paths in M: for any $\star, \star' \in \partial M$, the module of morphisms $\operatorname{Hom}_{\mathbb{C}^0}(\star, \star')$ is freely generated by the set of homotopy classes of paths from \star to \star' in M. Thus the category \mathbb{C}^0 is the linearization of the fundamental groupoid $\pi_1(M, \partial M)$ of M based at ∂M , and the algebra $A(\mathbb{C}^0)$ is the corresponding groupoid algebra. Clearly, $A(\mathbb{C}^0)$ embeds in A as a subalgebra.

5.1.2. *The intersection bibracket.* — Assume that $n = \dim(M) \ge 3$ and

(5.1.2)
$$\begin{cases} \text{The cross product in the homology } H_*(\Omega_{\star}) \text{ of the loop space } \Omega_{\star} = \Omega(M, \star, \star) \text{ based at } \star \in \partial M \text{ induces an isomorphism } H_*(\Omega_{\star}) \otimes H_*(\Omega_{\star}) \simeq H_*(\Omega_{\star} \times \Omega_{\star}). \end{cases}$$

By the Künneth theorem, the condition (5.1.2) holds if \mathbb{K} is a principal ideal domain and $H_*(\Omega_{\star}) = H_*(\Omega_{\star}; \mathbb{K})$ is a flat \mathbb{K} -module (this occurs, for instance, when \mathbb{K} is a field); it also holds for any \mathbb{K} if $H_*(\Omega_{\star}; \mathbb{Z})$ is a free abelian group. Then the cross product induces an isomorphism

$$\varpi_{32,14}: H_*(\Omega_{32}) \otimes H_*(\Omega_{14}) \longrightarrow H_*(\Omega_{32} \times \Omega_{14})$$

for any choice of base points $\star_1, \star_2, \star_3, \star_4 \in \partial M$. Composing the inverse isomorphism with the map $\Upsilon_{12,34}$ defined in Section 4.4.1, we obtain a linear map

$$(\varpi_{32,14})^{-1}\Upsilon_{12,34}: H_*(\Omega_{12}) \otimes H_*(\Omega_{34}) \longrightarrow H_*(\Omega_{32}) \otimes H_*(\Omega_{14})$$

The direct sum of these maps over all 4-tuples of points in ∂M is a linear map

$$\{\!\{-,-\}\!\}: A \otimes A \longrightarrow A \otimes A$$

called the *intersection bibracket* of *M*. We can now state our main result.

THEOREM 5.1.1. — Under the assumptions above, $(\mathbb{C}, \{\{-, -\}\})$ is a double Gerstenhaber category of degree d = 2 - n.

Proof. — That the bibracket $\{\{-,-\}\}$ has degree *d* follows from the fact that the intersection polychain of a *p*-polycycle and a *q*-polycycle has dimension

$$p+q+2-n=p+q+d$$
 for any p,q .

The *d*-antisymmetry of $\{\{-,-\}\}$ follows from the formula (4.4.2) and the following well-known fact:

For any topological spaces *X* and *Y* such that the cross product induces an isomorphism $H_*(X) \otimes H_*(Y) \rightarrow H_*(X \times Y)$, the isomorphism

$$H_*(X) \otimes H_*(Y) \longrightarrow H_*(Y) \otimes H_*(X)$$

induced by the interchange of factors $X \times Y \to Y \times X$ and the cross product isomorphisms, carries $a \otimes b$ to $(-1)^{|a| \cdot |b|} b \otimes a$ for any homogeneous $a \in H_*(X)$, $b \in H_*(Y)$; see, for example, **[15**, Section 4(b)].

The bibracket $\{\{-, -\}\}$ satisfies the first Leibniz rule (1.2.2) as easily follows from Lemma 4.4.2 using the associativity and the naturality of the cross product in singular homology. By Lemma 1.2.2 and the *d*-antisymmetry, the bibracket $\{\{-, -\}\}$ also satisfies the second Leibniz rule (1.2.3). Therefore $\{\{-, -\}\}$ is a *d*-graded bibracket in *A*.

It is obvious from the definitions that $\{\{-, -\}\}$ annihilates the identity morphisms of all objects. It remains only to prove that the associated tribracket is equal to zero; we postpone the proof to Section 5.2.

Since d = 2 - n < 0, the restriction of the intersection bibracket in C to C⁰ is equal to zero. Moreover, the morphisms in C⁰ represented by paths in ∂M annihilate the bibracket in C both on the right and on the left.

Theorem 5.1.1 and Lemma 2.2.1 imply that for every integer $N \ge 1$, the associated representation algebra \mathcal{C}_N^+ is a unital Gerstenhaber algebra of degree 2 - n.

5.1.3. *The Pontryagin algebra.* — We now fix a base point $\star \in \partial M$. By the *Pontryagin algebra of M*, we mean the unital graded algebra

$$A_{\star} = \operatorname{End}_{\mathcal{C}(M)}(\star) = H_*(\Omega_{\star}) \quad \text{where } \Omega_{\star} = \Omega(M, \star, \star).$$

Multiplication in A_{\star} is the *Pontryagin product* given by $ab = c_*(a \times b)$.

Pontryagin algebras have been extensively studied since Serre's thesis [40]. They can be explicitly computed using the Adams-Hilton model [1] or the techniques of rational homotopy theory (at least, in the simply connected case). We only mention the relation with the homotopy groups, and refer to [15] for a detailed exposition. Consider the boundary homomorphism

$$\partial_i: \pi_i(M) = \pi_i(M, \star) \longrightarrow \pi_{i-1}(\Omega_{\star}) = \pi_{i-1}(\Omega_{\star}, e_{\star})$$

for the path space fibration of M where $i \ge 1$ and $e_{\star} \in \Omega_{\star}$ is the constant path at \star . Since the total space of that fibration is contractible, ∂_i is an isomorphism for all i. Composing ∂_i with the Hurewicz homomorphism $\pi_{i-1}(\Omega_{\star}) \to H_{i-1}(\Omega_{\star})$, we obtain an additive map $\overline{\partial}_i : \pi_i(M) \to A_{\star}^{i-1}$ called the *connecting homomorphism*. For i = 1, this homomorphism extends to a ring isomorphism $\mathbb{K}[\pi_1(M, \star)] \simeq A_{\star}^0$. For i = 2, this homomorphism induces an isomorphism from $\mathbb{K} \otimes_{\mathbb{Z}} \pi_2(M)$ onto $H_1(\Omega_{\star}^{\text{null}}) \subset A_{\star}^1$ where $\Omega_{\star}^{\text{null}}$ is the connected component of Ω_{\star} formed by all null-homotopic loops.

The group $\pi_1(M, \star)$ acts on A_{\star} by graded algebra automorphisms: the action of any $g \in \pi_1(M, \star)$ is the automorphism $a \mapsto a^g = gag^{-1}$ of A_{\star} where g is viewed as an invertible element of A^0_{\star} . The inclusion $\partial M \subset M$ allows us to consider the induced action of $\pi_1(\partial M, \star)$ on A_{\star} .

THEOREM 5.1.2. — If $n = \dim(M) \ge 3$ and the condition (5.1.2) is satisfied, then the restriction of the intersection bibracket $\{\{-, -\}\}$ in the category $\mathcal{C}(M)$ to A_{\star} is a $\pi_1(\partial M, \star)$ -equivariant Gerstenhaber bibracket of degree 2 - n.

Proof. — Clearly, $A_{\star} = A(\mathcal{C}_{\star})$ where \mathcal{C}_{\star} is the full subcategory of \mathcal{C} determined by the object \star . Therefore our claim is a consequence of Theorem 5.1.1 and the results stated at the end of Section 2.2.2. We only need to check the equivariance. Let $a, b \in A_{\star}$ and let ς be a loop in ∂M based at \star representing $g \in \pi_1(\partial M, \star)$. We deduce from the commutativity of the diagram (4.4.3) that

$$\Upsilon(a^g \otimes b^g) = \Upsilon\left((\varsigma^{-1}, \varsigma^{-1})_{\sharp}(a), (\varsigma^{-1}, \varsigma^{-1})_{\sharp}(b)\right) = \left((\varsigma^{-1}, \varsigma^{-1})_{\sharp} \times (\varsigma^{-1}, \varsigma^{-1})_{\sharp}\right) \Upsilon(a, b).$$

Using the naturally of the cross product, we conclude that

$$\{\!\{a^g, b^g\}\!\} = \left(\{\!\{a, b\}\!\}'\right)^g \otimes \left(\{\!\{a, b\}\!\}''\right)^g.$$

By Theorem 5.1.2 and Lemma 2.1.1, the intersection bibracket in A_{\star} induces a natural structure of a Gerstenhaber algebra of degree 2 - n in the commutative unital graded algebra $(A_{\star})_N^+$ for all $N \ge 1$. We call $(A_{\star})_N^+$ the *N*-th representation algebra of M. The action of $\pi_1(\partial M, \star)$ on A_{\star} induces an action of $\pi_1(\partial M, \star)$ on $(A_{\star})_N^+$ by graded algebra automorphisms, and the Gerstenhaber bracket in $(A_{\star})_N^+$ is $\pi_1(\partial M, \star)$ -equivariant. The isomorphism classes of the double Gerstenhaber algebra A_{\star} and the Gerstenhaber algebras $\{(A_{\star})_N^+\}_N$ depend only on the connected component of \star in ∂M .

5.1.4. *The induced Lie bracket.* — We keep notation of Section 5.1.3 and let \dot{A}_{\star} be the quotient of $A_{\star} = H_{*}(\Omega_{\star})$ by the submodule $[A_{\star}, A_{\star}]$ spanned by the vectors $ab - (-1)^{|a| \cdot |b|} ba$ where a, b run over all homogeneous elements of A_{\star} . Under the assumptions of Theorem 5.1.2, the intersection bibracket $\{\{-, -\}\}$ in A_{\star} composed with the multiplication of A_{\star} induces a (2 - n)-graded Lie bracket $\langle -, - \rangle$ in \check{A}_{\star} , see Section 1.4.1.

The Lie bracket $\langle -, - \rangle$ can be computed using the map

$$\mathbf{c}_*: H_*(\Omega_{\star} \times \Omega_{\star}) \longrightarrow H_*(\Omega_{\star})$$

induced by the concatenation of loops. Namely, if $h : A_{\star} \to \check{A}_{\star}$ is the natural projection, then for any homogeneous $a, b \in A_{\star}$,

The resulting expression may be used as the definition of $\langle -, - \rangle$ avoiding the use of $\{\{-, -\}\}$. This gives a (2 - n)-graded Lie bracket in \check{A}_{\star} over an arbitrary commutative ring K. The Jacobi identity for $\langle -, - \rangle$ may be deduced from Lemma 5.2.6 below. Presumably, the Lie bracket $\langle -, - \rangle$ is related to the operation discussed in [28, Remark 3.2.3] using Chas-Sullivan's techniques.

5.1.5. *The simply connected case.* — Suppose that the manifold M is simply connected and the ground ring K is a field of characteristic zero. The classical Milnor-Moore theorem (see [15, Theorem 21.5]) asserts that, the Pontryagin algebra

$$A_{\star} = H_*(\Omega_{\star})$$

is fully determined by $\pi_*(M) = \bigoplus_{p \ge 0} \pi_p(M)$ and the Whitehead bracket $[-, -]_{Wh}$ in $\pi_*(M)$. More precisely, consider the graded module

$$L_{\star} = \bigoplus_{p \ge 0} \mathbb{K} \otimes_{\mathbb{Z}} \pi_p(\Omega_{\star})$$

(obtained from $\pi_*(M)$ by tensorizing with \mathbb{K} and shifting the degree by 1), and equip L_{\star} with the bracket defined by

$$[k \otimes \alpha, l \otimes \beta] = kl \otimes (-1)^{p+1} \partial_{p+q+1} \left([\partial_{p+1}^{-1}(\alpha), \partial_{q+1}^{-1}(\beta)]_{\mathrm{Wh}} \right) \in \mathbb{K} \otimes \pi_{p+q}(\Omega_{\star})$$

for any $k, l \in \mathbb{K}$, $\alpha \in \pi_p(\Omega_{\star})$, $\beta \in \pi_q(\Omega_{\star})$. Then L_{\star} is a 0-graded Lie algebra and the Hurewicz homomorphism $L_{\star} \to A_{\star}$ extends to an isomorphism of the universal enveloping algebra U(L) onto A_{\star} . Moreover, under this isomorphism, the standard comultiplication in U(L) carrying any $\alpha \in L_{\star}$ to $\alpha \otimes 1 + 1 \otimes \alpha$ corresponds to the comultiplication in A_{\star} induced by the diagonal map $\Omega_{\star} \to \Omega_{\star} \times \Omega_{\star}$. Note that, by the Poincaré-Birkhoff-Witt theorem for graded Lie algebras [15, Theorem 21.1], the natural linear map $L_{\star} \to U(L_{\star})$ is injective so that L_{\star} can be treated as a submodule of $U(L_{\star}) \simeq A_{\star}$.

Recall from Section 2.1.2 that the 0-graded Lie algebra L_{\star} gives rise to representation algebras $\{(L_{\star})_N\}_{N\geq 1}$. The Milnor-Moore isomorphism $U(L_{\star}) \simeq A_{\star}$ induces

an isomorphism $(L_{\star})_N \simeq (A_{\star})_N^+$ for all $N \ge 1$. In this way, the algebras $\{(L_{\star})_N\}_{N\ge 1}$ acquire a structure of Gerstenhaber algebras of degree 2 - n.

5.1.6. The *2*-dimensional case. — The case n = 2 (so far ruled out in this section by the assumptions of Section 5.1.2) has been extensively studied by several authors and gave the original impetus to this work. We briefly discuss this case.

A connected oriented surface M with $\partial M \neq \emptyset$ is an Eilenberg-MacLane space $K(\pi, 1)$ where π is the fundamental group of M. For any points $\star_1, \star_2 \in \partial M$, the space $\Omega(M, \star_1, \star_2)$ is homotopy equivalent to the underlying discrete set of π . Therefore, in the notation of Section 5.1.1, we have $\mathcal{C} = \mathcal{C}^0$ and $A = A(\mathcal{C}^0)$ is the groupoid algebra of $\pi_1(M, \partial M)$.

For any points $\star_1, \star_2, \star_3, \star_4 \in \partial M$ such that $\{\star_1, \star_2\} \cap \{\star_3, \star_4\} = \emptyset$, Section 4.4.1 yields a linear map

$$\Upsilon_{12,34}: H_0(\Omega_{12}) \otimes H_0(\Omega_{34}) \longrightarrow H_0(\Omega_{32} \times \Omega_{14}) = H_0(\Omega_{32}) \otimes H_0(\Omega_{14})$$

(the latter equality holds for all \mathbb{K}). This construction extends to arbitrary 4-tuples of points in ∂M by slightly pushing these points in the positive direction along ∂M and proceeding as in Section 4.3.5. After an appropriate normalization, this yields a 0-antisymmetric o-graded bibracket of degree o in the groupoid algebra A. This bibracket is quasi-Poisson in an appropriate sense, *cf.* **[2]**, **[42]**, **[36]**. For $\star \in \partial M$, the restriction of this bibracket to the group algebra $A_{\star} = \mathbb{K}[\pi_1(M, \star)]$ is the double bracket $\{\{-, -\}\}^s$ studied in **[36**, Section 7]. It is closely related to the homotopy intersection form in $\mathbb{K}[\pi_1(M, \star)]$ introduced in **[41**]; see also **[29]** for a similar operation. The associated Lie bracket $\langle -, - \rangle$ in \check{A}_{\star} was first introduced by Goldman **[21**].

Lemma 2.2.1 implies that for every integer $N \ge 1$, the above bibracket in A induces a bracket in the associated representation algebra \mathbb{C}_N^+ . This bracket is quasi-Poisson (and not Poisson), *cf.* [36]. Note that \mathbb{C}_N^+ is the coordinate algebra of the affine scheme (over \mathbb{K}) that associates to any unital commutative algebra B the set of groupoid homomorphisms $\pi_1(M, \partial M) \to \operatorname{GL}_N(B)$. Indeed, through linear extension of groupoid homomorphisms, the latter set may be identified with the set of linear functors from \mathbb{C} to the category $\operatorname{Mat}_N(B)$ considered in Section 2.2.1. We conclude by applying (2.2.1).

5.2. The Jacobi identity

We conclude the proof of Theorem 5.1.1 by proving that the tribracket associated with the intersection bibracket is equal to zero. We resume notation of Chapter 4, *i.e.*, fix points $\star_1, \star_2, \star_3, \star_4 \in \partial M$ such that $\{\star_1, \star_2\} \cap \{\star_3, \star_4\} = \emptyset$ and, for any $i, j \in \{1, 2, 3, 4\}$, let $\Omega_{ij} = \Omega(M, \star_i, \star_j)$ be the path space and $\Omega_{ij}^\circ = \Omega^\circ(M, \star_i, \star_j)$ be the proper path space of (M, \star_i, \star_j) . We start by developing a parametrized version of the theory of polychains in path spaces.

5.2.1. *Parametrized versions of* Υ , Υ , and Υ . — Let *Z* be an arbitrary topological space. Given a polychain $\mathcal{L} = (L, \psi, v, \lambda)$ in $\Omega_{34}^{\circ} \times Z$, we let λ' and λ'' be the compositions of $\lambda : L \to \Omega_{34}^{\circ} \times Z$ with the projections to Ω_{34}° and *Z*, respectively. We call the polychain \mathcal{L} smooth if the map $\lambda' : L \to \Omega_{34}^{\circ}$ is smooth in the sense of Section 3.4.2. Applying

the definitions of Section 3.2.4 but considering only smooth polychains in $\Omega_{34}^{\circ} \times Z$, we obtain *smooth face homology* $\widetilde{H}_{*}^{s}(\Omega_{34}^{\circ} \times Z)$. The proof of Theorem 3.4.8 easily adapts to this setting and yields that the natural linear map

$$\widetilde{H}^s_*(\Omega_{34}^\circ \times Z) \longrightarrow \widetilde{H}_*(\Omega_{34}^\circ \times Z) \simeq \widetilde{H}_*(\Omega_{34} \times Z)$$

is an isomorphism. This computes the face homology of $\Omega_{34} \times Z$ in terms of smooth polychains in $\Omega_{34}^{\circ} \times Z$.

We say that smooth polychains $\mathcal{K} = (K, \varphi, u, \kappa)$ in Ω_{12}° and $\mathcal{L} = (L, \psi, v, \lambda)$ in $\Omega_{34}^{\circ} \times Z$ are *transversal* if the maps

$$\kappa: K \longrightarrow \Omega_{12}^{\circ} \quad \text{and} \quad \lambda': L \longrightarrow \Omega_{34}^{\circ}$$

are transversal in the sense of Section 4.1.1. A pair $(a, b) \in H_p(\Omega_{12}) \times H_q(\Omega_{34} \times Z)$ with $p, q \ge 0$ is *transversely represented* by a pair $(\mathcal{K}, \mathcal{L})$ if \mathcal{K} is a smooth reduced p-polycycle in Ω_{12}° and \mathcal{L} is a smooth reduced q-polycycle in $\Omega_{34}^{\circ} \times Z$ transversal to \mathcal{K} . Adapting the proof of Lemma 4.1.5, we obtain that any pair (a, b) as above can be transversely represented by a pair of polycycles, and, furthermore, any two such pairs of polycycles can be related by a finite sequence of transformations $(\mathcal{K}, \mathcal{L}) \mapsto (\check{\mathcal{K}}, \check{\mathcal{L}})$ of the following types:

- (i) $\mathcal{L} \cong \mathring{\mathcal{L}}$ and $\mathring{\mathcal{K}} \cong \mathscr{K} \sqcup \partial^r \mathscr{M}$ or $\mathscr{K} \cong \mathring{\mathcal{K}} \sqcup \partial^r \mathscr{M}$ where \mathscr{M} is a smooth (p+1)-polychain in Ω_{12}° transversal to \mathcal{L} ;
- (ii) K ≅ K and L ≅ L ⊔ ∂^rN or L ≅ L ⊔ ∂^rN where N is a smooth (q + 1)-polychain in Ω^o₃₄ × Z transversal to K.

We next adapt the construction of the intersection polychain. Consider smooth transversal polychains $\mathcal{K} = (K, \varphi, u, \kappa)$ in Ω_{12}° and $\mathcal{L} = (L, \psi, v, \lambda)$ in $\Omega_{34}^{\circ} \times Z$. Since the polychain $\mathcal{L}' = (L, \psi, v, \lambda')$ in Ω_{34}° is smooth and transversal to \mathcal{K} , Section 4.2.1 yields an intersection polychain $\mathcal{D}(\mathcal{K}, \mathcal{L}') = (D, \theta, w, \kappa * \lambda')$ in $\Omega_{32} \times \Omega_{14}$. We lift $\mathcal{D}(\mathcal{K}, \mathcal{L}')$ to a polychain in $\Omega_{32} \times \Omega_{14} \times Z$ as follows.

LEMMA 5.2.1. — Let $pr: K \times I \times L \times I \rightarrow L$ be the cartesian projection. The tuple

 $\mathcal{D}^{Z}(\mathcal{K},\mathcal{L}) = (D,\theta,w,\delta)$

with $\delta = (\kappa \rightsquigarrow \lambda', \lambda'' \circ \operatorname{pr}_{|D})$ is a polychain in $\Omega_{32} \times \Omega_{14} \times Z$.

Proof. — We need only to check that the map $\lambda'' \circ \text{pr}_{|D} : D \to Z$ is compatible with the partition θ . Let *F*, *G* be two faces of *D* of the same type and let

$$N_F = A_F \times I \times B_F \times I, \quad N_G = A_G \times I \times B_G \times I$$

be the smallest faces of $K \times I \times L \times I$ containing F, G, respectively. Since λ'' is compatible with the partition ψ of L, we have for any $(k, s, l, t) \in F$

$$\lambda^{\prime\prime} \operatorname{pr} \left(\theta_{F,G}(k, s, l, t) \right) = \lambda^{\prime\prime} \operatorname{pr} \left(\varphi_{A_{F},A_{G}}(k), s, \psi_{B_{F},B_{G}}(l), t \right)$$
$$= \lambda^{\prime\prime} \left(\psi_{B_{F},B_{G}}(l) \right)$$
$$= \lambda^{\prime\prime} (l) = \lambda^{\prime\prime} \operatorname{pr}(k, s, l, t).$$

The next claim is a parametrized version of Lemma 4.3.1 and is proved similarly.

LEMMA 5.2.2. — For any integers $p, q \ge 0$, the intersection $(\mathfrak{K}, \mathcal{L}) \mapsto \mathcal{D}^{Z}(\mathfrak{K}, \mathcal{L})$ induces a bilinear map $\widetilde{H}_{p}(\Omega_{12}) \times \widetilde{H}_{q}(\Omega_{34} \times Z) \to \widetilde{H}_{p+q+2-n}(\Omega_{32} \times \Omega_{14} \times Z)$.

The direct sum over all integers $p, q \ge 0$ of the pairings produced by Lemma 5.2.2 is a linear map of degree 2 - n

$$\widetilde{\Upsilon}_{12,34Z}:\widetilde{H}_*(\Omega_{12})\otimes\widetilde{H}_*(\Omega_{34}\times Z)\longrightarrow\widetilde{H}_*(\Omega_{32}\times\Omega_{14}\times Z).$$

As in Section 4.3.6, a normalized version of this map

$$\check{\Upsilon}_{12,34Z}: \widetilde{H}_*(\Omega_{12}) \otimes \widetilde{H}_*(\Omega_{34} \times Z) \longrightarrow \widetilde{H}_*(\Omega_{32} \times \Omega_{14} \times Z)$$

is defined by

$$\check{\Upsilon}_{12,34Z}(a\otimes b) = (-1)^{|b|+n|a|} \widetilde{\Upsilon}_{12,34Z}(a\otimes b)$$

for any homogeneous $a \in \widetilde{H}_*(\Omega_{12})$ and $b \in \widetilde{H}_*(\Omega_{34} \times Z)$. We also define an operation $\Upsilon_{12,34Z}$ in singular homology by the commutative diagram

(5.2.1)
$$\begin{split} \widetilde{H}_{*}(\Omega_{12}) \otimes \widetilde{H}_{*}(\Omega_{34} \times Z) & \xrightarrow{\Upsilon_{12,34Z}} \widetilde{H}_{*}(\Omega_{32} \times \Omega_{14} \times Z) \\ & \swarrow & \swarrow & \swarrow \\ H_{*}(\Omega_{12}) \otimes H_{*}(\Omega_{34} \times Z) - \xrightarrow{\Upsilon_{12,34Z}} H_{*}(\Omega_{32} \times \Omega_{14} \times Z). \end{split}$$

The proof of Lemma 4.4.1 extends to this setting and gives the commutative diagram

(5.2.2)
$$\begin{split} \widetilde{H}_{*}(\Omega_{12}) \otimes \widetilde{H}_{*}(\Omega_{34} \times Z) & \xrightarrow{\Upsilon_{12,34Z}} \widetilde{H}_{*}(\Omega_{32} \times \Omega_{14} \times Z) \\ & & \downarrow^{[-]\times[-]} \\ & & \downarrow^{[-]} \\ & & H_{*}(\Omega_{12}) \otimes H_{*}(\Omega_{34} \times Z) & \xrightarrow{\Upsilon_{12,34Z}} H_{*}(\Omega_{32} \times \Omega_{14} \times Z). \end{split}$$

The following two lemmas will help us to compute $\tilde{\Upsilon}_{12,34Z}$ and $\Upsilon_{12,34Z}$.

LEMMA 5.2.3. — For any $a \in \widetilde{H}_*(\Omega_{12})$, any $b \in \widetilde{H}_*(\Omega_{34})$ and any homogeneous $c \in \widetilde{H}_*(Z)$, we have

$$\widetilde{\Upsilon}_{12,34Z}(a\otimes (b\times c)) = (-1)^{|c|} \widetilde{\Upsilon}_{12,34}(a\otimes b) \times c$$

Proof. — It suffices to consider the case where *a* and *b* are homogeneous. Let

$$\mathcal{K} = (K, \varphi, u, \kappa)$$
 and $\mathcal{L} = (L, \psi, v, \lambda)$

be smooth polycycles in Ω_{12}° and Ω_{34}° representing *a* and *b* respectively, and such that \mathcal{K} is transversal to \mathcal{L} . Let $\mathcal{N} = (N, \chi, z, \eta)$ be a polycycle in *Z* representing *c*. Then

$$\widetilde{\Upsilon}_{12,34}(a \otimes b) \times c = \big\langle \mathcal{D}(\mathcal{K}, \mathcal{L}) \times \mathcal{N} \big\rangle.$$

By the definition of $\widetilde{\Upsilon}_{12,34Z}$,

$$\widetilde{\Upsilon}_{12,34Z}(a\otimes(b\times c)) = \langle \mathcal{D}^Z(\mathcal{K},\mathcal{L}\times\mathcal{N}) \rangle.$$

Therefore, it is enough to show that

(5.2.3)
$$\mathcal{D}(\mathcal{K},\mathcal{L}) \times \mathcal{N} = (-1)^{|c|} \mathcal{D}^{Z}(\mathcal{K},\mathcal{L} \times \mathcal{N}).$$

We set $\mathcal{D}(\mathcal{K}, \mathcal{L}) = (D, \theta, w, \kappa \triangleleft \lambda)$ so that

$$\mathcal{D}(\mathcal{K},\mathcal{L})\times\mathcal{N}=\big(D\times N,\theta\times\chi,w\times z,(\kappa \rightsquigarrow \lambda)\times\eta\big).$$

We also set

$$\mathcal{D}^{Z}(\mathcal{K},\mathcal{L}\times\mathcal{N}) = \left(D^{Z},\theta^{Z},w^{Z},\delta^{Z}\right) \quad \text{with } \delta^{Z} = \left(\kappa \sphericalangle (\lambda \operatorname{pr}_{L}),\eta \operatorname{pr}_{N \mid D^{Z}}\right)$$

where $pr_L : L \times N \to L$ and $pr_N : K \times I \times (L \times N) \times I \to N$ are the cartesian projections. The map

$$(K \times I \times L \times I) \times N \longrightarrow K \times I \times (L \times N) \times I, \quad (k, s, l, t, n) \longmapsto (k, s, l, n, t)$$

restricts to a diffeomorphism $f : D \times N \to D^Z$ of degree $(-1)^{\dim(N)} = (-1)^{|c|}$. For any point (k, s, l, t, n) in $D \times N$, we have

$$\begin{split} \delta^{Z}f(k,s,l,t,n) &= \delta^{Z}(k,s,l,n,t) = \left((\kappa \rightsquigarrow \lambda)(k,s,l,t), \eta(n) \right) \\ &= \left((\kappa \rightsquigarrow \lambda) \times \eta \right)(k,s,l,t,n) \end{split}$$

Furthermore, the diffeomorphism f carries the partition $\theta \times \chi$ to θ^Z and the weight $w \times z$ to the weight w^Z . Hence, f is a diffeomorphism of polychains (5.2.3).

Lemma 5.2.4. — For any
$$a \in H_*(\Omega_{12})$$
, $b \in H_*(\Omega_{34})$ and $c \in H_*(Z)$, we have
 $\Upsilon_{12,34Z}(a \otimes (b \times c)) = \Upsilon_{12,34}(a \otimes b) \times c$

Proof. — It suffices to consider homogeneous *a*, *b*, *c*. By Lemma 3.3.5, we have

$$b \times c = [\langle b \rangle] \times [\langle c \rangle] = [\langle b \rangle \times \langle c \rangle].$$

We deduce that

$$\begin{split} \Upsilon_{12,34Z} \big(a \otimes (b \times c) \big) &= \Upsilon_{12,34Z} \big([\langle a \rangle] \otimes [\langle b \rangle \times \langle c \rangle] \big) \\ &= \big[\check{\Upsilon}_{12,34Z} (\langle a \rangle \otimes (\langle b \rangle \times \langle c \rangle)) \big] \\ &= (-1)^{|b|+|c|+n|a|} \big[\widetilde{\Upsilon}_{12,34Z} (\langle a \rangle \otimes (\langle b \rangle \times \langle c \rangle)) \big] \\ &= (-1)^{|b|+n|a|} \big[\widetilde{\Upsilon}_{12,34} (\langle a \rangle \otimes \langle b \rangle) \times \langle c \rangle \big] \\ &= (-1)^{|b|+n|a|} \big[\widetilde{\Upsilon}_{12,34} (\langle a \rangle \otimes \langle b \rangle) \big] \times \big[\langle c \rangle \big] \\ &= \Upsilon_{12,34} \big(a \otimes b \big) \times c \end{split}$$

where the second, fourth and fifth equalities follow from (5.2.2), Lemma 5.2.3 and Lemma 3.3.5 respectively.

Given two topological spaces *Y* and *Z*, a straightforward generalization of the constructions above and of Lemma 5.2.2 yields a bilinear map

$$\widetilde{\Upsilon}_{Y12,34Z}: \widetilde{H}_*(Y \times \Omega_{12}) \otimes \widetilde{H}_*(\Omega_{34} \times Z) \longrightarrow \widetilde{H}_*(Y \times \Omega_{32} \times \Omega_{14} \times Z).$$

A normalized version $\check{\Upsilon}_{Y12,34Z}$ of this map is defined by

$$\check{\Upsilon}_{Y12,34Z}(a\otimes b) = (-1)^{|b|+n|a|} \widetilde{\Upsilon}_{Y12,34Z}(a\otimes b)$$

MÉMOIRES DE LA SMF 154

100

for any homogeneous $a \in \widetilde{H}_*(Y \times \Omega_{12})$ and $b \in \widetilde{H}_*(\Omega_{34} \times Z)$. The corresponding map in singular homology is defined by the commutative diagram

(5.2.4)
$$\begin{split} \widetilde{H}_{*}(Y \times \Omega_{12}) \otimes \widetilde{H}_{*}(\Omega_{34} \times Z) & \xrightarrow{\Upsilon_{Y12,34Z}} \widetilde{H}_{*}(Y \times \Omega_{32} \times \Omega_{14} \times Z) \\ & \swarrow & \swarrow & \swarrow & \swarrow \\ & & & \downarrow [-] \\ & & H_{*}(Y \times \Omega_{12}) \otimes H_{*}(\Omega_{34} \times Z) - \xrightarrow{\Upsilon_{Y12,34Z}} \to H_{*}(Y \times \Omega_{32} \times \Omega_{14} \times Z). \end{split}$$

Then, again, we have the commutative diagram

(5.2.5)
$$\begin{split} \widetilde{H}_{*}(Y \times \Omega_{12}) \otimes \widetilde{H}_{*}(\Omega_{34} \times Z) & \xrightarrow{\Upsilon_{Y12,34Z}} \widetilde{H}_{*}(Y \times \Omega_{32} \times \Omega_{14} \times Z) \\ & & & \downarrow \\ & & \downarrow \\ & & \downarrow \\ & & H_{*}(Y \times \Omega_{12}) \otimes H_{*}(\Omega_{34} \times Z) \xrightarrow{\Upsilon_{Y12,34Z}} H_{*}(Y \times \Omega_{32} \times \Omega_{14} \times Z). \end{split}$$

Finally, Lemma 5.2.4 generalizes to the identity

(5.2.6)
$$\Upsilon_{Y12,34Z}((c \times a) \otimes (b \times d)) = c \times \Upsilon_{12,34}(a \otimes b) \times d$$

for any $a \in H_*(\Omega_{12})$, $b \in H_*(\Omega_{34})$ and $c \in H_*(Y)$, $d \in H_*(Z)$.

5.2.2. *Half-smooth polychains.* — We compute the intersection operations of Section 5.2.1 via so-called "half-smooth" polychains. Let Z be a topological space. A q-polychain

$$\mathcal{L} = (L, \psi, \nu, (\lambda', \lambda'') : L \longrightarrow \Omega_{34}^{\circ} \times Z)$$

is *half-smooth* if the restrictions of the map $\tilde{\lambda}' : L \times I \to M$ (adjoint to λ') to the manifolds with faces $L \times [0, 1/2]$ and $L \times [1/2, 1]$ are smooth. Furthermore, \mathcal{L} is *half-transversal* to a smooth *p*-polychain $\mathcal{K} = (K, \varphi, u, \kappa)$ in Ω_{12}° if for any face *E* of *K*, any face *F* of *L*, and any of the three sets $J = [0, 1/2], [1/2, 1], \{1/2\}$ the map

$$\tilde{\kappa} \times \tilde{\lambda}' : E \times I \times F \times J \longrightarrow M \times M$$

is weakly transversal to diag_M in the sense of Section 4.1.1. Then the set

$$D(J) = \{(k, s, l, t) \in K \times I \times L \times J : \tilde{\kappa}(k, s) = \tilde{\lambda}(l, t)\}$$

inherits from $K \times I \times L \times J$ a structure of a manifold with faces, and we have

$$(5.2.7) D(J) \subset K \times \operatorname{Int}(I) \times L \times (J \cap \operatorname{Int}(I)).$$

Set

$$D^{-} = D([0, 1/2]), \quad D^{+} = D([1/2, 1]), \quad D^{1/2} = D(\{1/2\}).$$

It is clear that $D^{1/2} = D^- \cap D^+ = \partial D^- \cap \partial D^+$ and

$$\dim D^- = \dim D^+ = p + q + 2 - n, \quad \dim D^{1/2} = p + q + 1 - n$$

Since \mathcal{L} may be non-smooth, we cannot consider the intersection polychain $\mathcal{D}^{\mathbb{Z}}(\mathcal{K},\mathcal{L})$. (A priori, the set $D^{-} \cup D^{+}$ does not have a structure of a manifold with faces.) Instead, we turn the disjoint union $D^{-} \sqcup D^{+}$ into a polychain which will serve as a substitute for $\mathcal{D}^{\mathbb{Z}}(\mathcal{K},\mathcal{L})$. The inclusion (5.2.7) allows us to use the same

construction as in Section 4.2.1 in order to upgrade D^- , D^+ , and $D^{1/2}$ to polychains in $\Omega_{32}^{\circ} \times \Omega_{14}^{\circ} \times Z$ denoted, respectively,

$$\mathcal{D}^- = \mathcal{D}^-(\mathcal{K}, \mathcal{L}), \quad \mathcal{D}^+ = \mathcal{D}^+(\mathcal{K}, \mathcal{L}), \quad \mathcal{D}^{1/2} = \mathcal{D}^{1/2}(\mathcal{K}, \mathcal{L}).$$

As can be checked from our conventions, the oriented manifold $D^{1/2}$ has the orientation inherited from $(-1)^{p+q+1+n}\partial D^-$ or, equivalently, the orientation inherited from $(-1)^{p+q+n}\partial D^+$. The inclusions $D^{1/2} \subset D^{\pm}$ are compatible with the polychain structures (except for the orientations): they map faces of $D^{1/2}$ diffeomorphically onto faces of D^{\pm} , map faces of the same type onto faces of the same type, commute with the identification diffeomorphisms of the faces, commute with the maps to $\Omega_{32}^{\circ} \times \Omega_{14}^{\circ} \times Z$, and the induced maps in π_0 commute with the weights. Also, a face of D^{\pm} having the same type as the image of a face F of $D^{1/2}$ must be the image of a face of $D^{1/2}$ of the same type as F. These facts allow us to form a (p+q+2-n)-polychain $\mathcal{D}^h = \mathcal{D}^h(\mathcal{K}, \mathcal{L})$ in $\Omega_{32}^{\circ} \times \Omega_{14}^{\circ} \times Z$ by taking the disjoint union $\mathcal{D}^- \sqcup \mathcal{D}^+$ and declaring that the images of any face of $D^{1/2}$ in D^- and D^+ have the same type and the identification diffeomorphism between them is the identity map. We shall sometimes write

$$\mathbb{D}^- \cup \mathbb{D}^+_{1/2}$$

for this polychain \mathcal{D}^h .

LEMMA 5.2.5. — Let \mathcal{K} be a smooth *p*-polycycle in Ω_{12}° and let \mathcal{L} be a half-smooth *q*-polycycle in $\Omega_{34}^{\circ} \times Z$ half-transversal to \mathcal{K} . Then $\mathcal{D}^{h}(\mathcal{K}, \mathcal{L})$ is a polycycle in $\Omega_{32}^{\circ} \times \Omega_{14}^{\circ} \times Z$ and

$$\left[\widetilde{\Upsilon}_{12,34Z}(\langle \mathcal{K} \rangle, \langle \mathcal{L} \rangle)\right] = \left[\mathcal{D}^{h}(\mathcal{K}, \mathcal{L})\right] \in H_{p+q+2-n}(\Omega_{32} \times \Omega_{14} \times Z).$$

Proof. — Lemma 4.2.2 directly extends to smooth polychains \mathcal{K} , \mathcal{K}' in Ω_{12}° and half-smooth polychains \mathcal{L} , \mathcal{L}' in $\Omega_{34}^{\circ} \times Z$ half-transversal to \mathcal{K} , \mathcal{K}' ; one should only replace \mathcal{D} by \mathcal{D}^h . This implies the first claim of Lemma 5.2.5.

There is an arbitrarily small deformation $\{\mathcal{L}^t = (L, \psi, v, ((\lambda')^t, \lambda''))\}_{t \in I}$ of $\mathcal{L}^0 = \mathcal{L}$ into a smooth polycycle \mathcal{L}^1 . We can assume that the restrictions of the maps

$$(\widetilde{\lambda}')^t : L \times [0,1] \longrightarrow M$$

to $L \times [0, 1/2]$ and $L \times [1/2, 1]$ are smooth maps smoothly depending on $t \in I$. As in the proof of Lemma 3.2.3, we derive from the deformation $\{\mathcal{L}^t\}_{t \in I}$ a (q + 1)-polychain \mathcal{R} in $\Omega_{34}^{\circ} \times Z$ such that $\partial^r \mathcal{R} = \operatorname{red}(\mathcal{L}^1) \sqcup \operatorname{red}(-\mathcal{L})$. The assumptions on the deformation imply that \mathcal{R} is half-smooth. Taking the deformation small enough, we can ensure that \mathcal{R} is half-transversal to \mathcal{K} . By the assumption $\partial^r \mathcal{K} = \emptyset$ and the generalized version of Lemma 4.2.2,

$$(-1)^{n+p+1}\partial^{r}\mathcal{D}^{h}(\mathcal{K},\mathcal{R}) = \operatorname{red}\mathcal{D}^{h}(\operatorname{red}\mathcal{K},\operatorname{red}(\mathcal{L}^{1})\sqcup\operatorname{red}(-\mathcal{L}))$$
$$= \operatorname{red}\mathcal{D}^{h}(\operatorname{red}\mathcal{K},\operatorname{red}\mathcal{L}^{1})\sqcup(-\operatorname{red}\mathcal{D}^{h}(\operatorname{red}\mathcal{K},\operatorname{red}\mathcal{L}))$$
$$= \operatorname{red}\mathcal{D}^{h}(\mathcal{K},\mathcal{L}^{1})\sqcup(-\operatorname{red}\mathcal{D}^{h}(\mathcal{K},\mathcal{L})).$$

Therefore

$$\left\langle \mathcal{D}^{h}(\mathcal{K},\mathcal{L})\right\rangle = \left\langle \mathcal{D}^{h}(\mathcal{K},\mathcal{L}^{1})\right\rangle \in \widetilde{H}_{p+q+2-n}(\Omega_{32} \times \Omega_{14} \times Z)$$

Projecting to singular homology, we obtain the equality

$$\left[\mathcal{D}^{h}(\mathcal{K},\mathcal{L})\right] = \left[\mathcal{D}^{h}(\mathcal{K},\mathcal{L}^{1})\right] \in H_{p+q+2-n}(\Omega_{32} \times \Omega_{14} \times Z).$$

Since the polycycle \mathcal{L}^1 is smooth, the manifold with faces underlying $\mathcal{D}^h(\mathcal{K}, \mathcal{L}^1)$ is obtained by cutting out the manifold with faces underlying $\mathcal{D}^Z(\mathcal{K}, \mathcal{L}^1)$ along a smooth compact oriented proper submanifold of codimension 1. This easily implies the equality $[\mathcal{D}^h(\mathcal{K}, \mathcal{L}^1)] = [\mathcal{D}^Z(\mathcal{K}, \mathcal{L}^1)]$. Thus,

$$\left[\widetilde{\Upsilon}_{12,34Z}(\langle \mathcal{K} \rangle, \langle \mathcal{L} \rangle)\right] = \left[\widetilde{\Upsilon}_{12,34Z}(\langle \mathcal{K} \rangle, \langle \mathcal{L}^1 \rangle)\right] = \left[\mathcal{D}^Z(\mathcal{K}, \mathcal{L}^1)\right] = \left[\mathcal{D}^h(\mathcal{K}, \mathcal{L})\right].$$

5.2.3. A Jacobi-type identity for Υ . — As in Section 4.3.5, the operations

$$\Upsilon_{12,34Z}$$
, $\check{\Upsilon}_{12,34Z}$, $\Upsilon_{12,34Z}$

generalize to all tuples $\star_1, \star_2, \star_3, \star_4 \in \partial M$. We pick two extra points $\star_5, \star_6 \in \partial M$. For $Z = \Omega_{56}$, the maps $\widetilde{\Upsilon}_{12,34Z}, \check{\Upsilon}_{12,34Z}, \Upsilon_{12,34Z}$ will be denoted respectively by

 $\widetilde{\Upsilon}_{12,3456}, \quad \check{\Upsilon}_{12,3456}, \quad \Upsilon_{12,3456}.$

Given a permutation (i, j, k, l, m, n) of (1, 2, 3, 4, 5, 6), we can accordingly renumber the points \star_1, \ldots, \star_6 and consider the corresponding maps $\widetilde{\Upsilon}_{ij,klmn}, \check{\Upsilon}_{ij,klmn}, \Upsilon_{ij,klmn}$.

We now establish a Jacobi-type identity for $\Upsilon_{ij,klmn}$.

LEMMA 5.2.6. — Consider the permutation maps

$$p_{231}: \Omega_{36} \times \Omega_{52} \times \Omega_{14} \longrightarrow \Omega_{52} \times \Omega_{14} \times \Omega_{36}, \quad (x, y, z) \longmapsto (y, z, x),$$

$$p_{312}: \Omega_{14} \times \Omega_{36} \times \Omega_{52} \longrightarrow \Omega_{52} \times \Omega_{14} \times \Omega_{36}, \quad (x, y, z) \longmapsto (z, x, y).$$

For any $a \in H_p(\Omega_{12})$, $b \in H_q(\Omega_{34})$ and $c \in H_r(\Omega_{56})$ with $p, q, r \ge 0$, we have the following equality in $H_{p+q+r+4-2n}(\Omega_{52} \times \Omega_{14} \times \Omega_{36})$:

$$\begin{split} \Upsilon_{12,5436} \big(a \otimes \Upsilon_{34,56}(b \otimes c) \big) \\ &+ (-1)^{(p+n)(q+r)} (\mathbf{p}_{312})_* \Upsilon_{34,1652} \big(b \otimes \Upsilon_{56,12}(c \otimes a) \big) \\ &+ (-1)^{(p+q)(r+n)} (\mathbf{p}_{231})_* \Upsilon_{56,3214} \big(c \otimes \Upsilon_{12,34}(a \otimes b) \big) = 0 \end{split}$$

Proof. — Set $\varepsilon = (-1)^{n(q+1)+pr}$. The definition of $\Upsilon_{34,56}$ and (5.2.2) imply that

$$\begin{split} &\Upsilon_{12,5436} \big(a \otimes \Upsilon_{34,56}(b \otimes c) \big) \\ &= \Upsilon_{12,5436} \big([\langle a \rangle] \otimes [\check{\Upsilon}_{34,56}(\langle b \rangle \otimes \langle c \rangle)] \big) \\ &= \big[\check{\Upsilon}_{12,5436}(\langle a \rangle \otimes \check{\Upsilon}_{34,56}(\langle b \rangle \otimes \langle c \rangle)) \big] \\ &= (-1)^{(r+nq)+(q+r+n+np)} \big[\widetilde{\Upsilon}_{12,5436}(\langle a \rangle \otimes \widetilde{\Upsilon}_{34,56}(\langle b \rangle \otimes \langle c \rangle)) \big] \\ &= \varepsilon (-1)^{q+p(n+r)} \big[\widetilde{\Upsilon}_{12,5436}(\langle a \rangle \otimes \widetilde{\Upsilon}_{34,56}(\langle b \rangle \otimes \langle c \rangle)) \big]. \end{split}$$

Using the naturality of the transformation [–], we also obtain that

$$(-1)^{(p+n)(q+r)}(\mathbf{p}_{312})_* \Upsilon_{34,1652}(b \otimes \Upsilon_{56,12}(c \otimes a)) = (-1)^{(p+n)(q+r)} [(\mathbf{p}_{312})_* \check{\Upsilon}_{34,1652}(\langle b \rangle \otimes \check{\Upsilon}_{56,12}(\langle c \rangle \otimes \langle a \rangle))]$$

$$= (-1)^{(p+n)(q+r)+(p+nr)+(r+p+n+nq)} [(p_{312})_* \widetilde{\Upsilon}_{34,1652}(\langle b \rangle \otimes \widetilde{\Upsilon}_{56,12}(\langle c \rangle \otimes \langle a \rangle))]$$

$$= \varepsilon (-1)^{r+q(n+p)} [(p_{312})_* \widetilde{\Upsilon}_{34,1652}(\langle b \rangle \otimes \widetilde{\Upsilon}_{56,12}(\langle c \rangle \otimes \langle a \rangle))]$$

and

$$\begin{aligned} (-1)^{(p+q)(r+n)}(\mathbf{p}_{231})_* \Upsilon_{56,3214}(c \otimes \Upsilon_{12,34}(a \otimes b)) \\ &= (-1)^{(p+q)(r+n)} \big[(\mathbf{p}_{231})_* \check{\Upsilon}_{56,3214}(\langle c \rangle \otimes \check{\Upsilon}_{12,34}(\langle a \rangle \otimes \langle b \rangle)) \big] \\ &= (-1)^{(p+q)(r+n)+(q+np)+(p+q+n+nr)} \big[(\mathbf{p}_{231})_* \widetilde{\Upsilon}_{56,3214}(\langle c \rangle \otimes \widetilde{\Upsilon}_{12,34}(\langle a \rangle \otimes \langle b \rangle)) \big] \\ &= \varepsilon (-1)^{p+r(n+q)} \big[(\mathbf{p}_{231})_* \widetilde{\Upsilon}_{56,3214}(\langle c \rangle \otimes \widetilde{\Upsilon}_{12,34}(\langle a \rangle \otimes \langle b \rangle)) \big]. \end{aligned}$$

Thus, it is enough to prove the following identity in $H_*(\Omega_{52} \times \Omega_{14} \times \Omega_{36})$, where $a \in \widetilde{H}_p(\Omega_{12})$, $b \in \widetilde{H}_q(\Omega_{34})$ and $c \in \widetilde{H}_r(\Omega_{56})$ are now any face homology classes:

$$(5.2.8) \qquad (-1)^{q+p(n+r)} \left[\widetilde{\Upsilon}_{12,5436}(a \otimes \widetilde{\Upsilon}_{34,56}(b \otimes c)) \right] \\ + (-1)^{r+q(n+p)} \left[(p_{312})_* \widetilde{\Upsilon}_{34,1652}(b \otimes \widetilde{\Upsilon}_{56,12}(c \otimes a)) \right] \\ + (-1)^{p+r(n+q)} \left[(p_{231})_* \widetilde{\Upsilon}_{56,3214}(c \otimes \widetilde{\Upsilon}_{12,34}(a \otimes b)) \right] = 0.$$

Slightly moving the points \star_1, \ldots, \star_6 in ∂M , we can assume that they are pairwise distinct. Let $\mathcal{K} = (K, \varphi, u, \kappa)$ be a smooth *p*-polycycle in Ω_{12}° representing *a*, let $\mathcal{L} = (L, \psi, v, \lambda)$ be a smooth *q*-polycycle in Ω_{34}° representing *b*, and let $\mathcal{N} = (N, \chi, z, \eta)$ be a smooth *r*-polycycle in Ω_{56}° representing *c*. We will assume that $\mathcal{K}, \mathcal{L}, \mathcal{N}$ are pairwise transversal in the sense of Section 4.1.1. This assumption and other transversality conditions imposed below in the course of the proof are always achieved by a small deformation of $\mathcal{K}, \mathcal{L}, \mathcal{N}$.

Let $\mathcal{D}_{bc} = \mathcal{D}(\mathcal{L}, \mathcal{N})$ be the intersection polycycle as defined in Section 4.2.1. Recall that its underlying manifold with faces, D_{bc} , consists of all tuples $(l, h, n, i) \in L \times I \times N \times I$ such that $\tilde{\lambda}(l, h) = \tilde{\eta}(n, i)$. Let (cb, bc) stand for the underlying continuous map

$$\lambda \triangleleft \eta : D_{bc} \longrightarrow \Omega_{54}^{\circ} \times \Omega_{36}^{\circ}$$

of \mathcal{D}_{bc} . The map $cb = \lambda \triangleleft \eta : D_{bc} \longrightarrow \Omega_{54}^{\circ}$ carries a point (l, h, n, i) to the path $I \to M$ which runs from \star_5 to $\tilde{\eta}(n, i)$ along $\tilde{\eta}(n, -)$ in the first half-time and then runs from $\tilde{\lambda}(l, h)$ to \star_4 along $\tilde{\lambda}(l, -)$ in the second half-time. (Here and below, the time parameter of paths always increases along subintervals of I with constant speed.) The map $bc = \lambda \triangleright \eta : D_{bc} \to \Omega_{36}^{\circ}$ carries (l, h, n, i) to the path $I \to M$ which runs from \star_3 to $\tilde{\lambda}(l, h)$ along $\tilde{\lambda}(l, -)$ in the first half-time and then runs from $\tilde{\eta}(n, i)$ to \star_6 along $\tilde{\eta}(n, -)$ in the second half-time. Thus the paths cb(l, h, n, i) and bc(l, h, n, i) are obtained from the paths $\tilde{\eta}(n, -)$ and $\tilde{\lambda}(l, -)$ by switching direction at the intersection point $\tilde{\lambda}(l, h) = \tilde{\eta}(n, i)$, see Figure 5.2.1 next page.

We set $I^{\circ} = \text{Int}(I) = (0, 1)$, $Z = \Omega_{36}^{\circ}$ and view \mathcal{D}_{bc} as a polycycle in $\Omega_{54}^{\circ} \times Z$. It is halfsmooth in the sense of Section 5.2.2. Slightly deforming the map $\tilde{\kappa} : K \times I \to M$ adjoint to κ , we can assume \mathcal{D}_{bc} to be half-transversal to \mathcal{K} in the sense of Section 5.2.2. In the sequel, we consider the associated (p + q + r + 4 - 2n)-polychains $\mathcal{D}_{abc}^{-} = \mathcal{D}^{-}(\mathcal{K}, \mathcal{D}_{bc})$ and $\mathcal{D}_{abc}^{+} = \mathcal{D}^{+}(\mathcal{K}, \mathcal{D}_{bc})$ in $\Omega_{52}^{\circ} \times \Omega_{14}^{\circ} \times Z$.

MÉMOIRES DE LA SMF 154

104

5.2. THE JACOBI IDENTITY



FIGURE 5.2.1. The polycycle \mathcal{D}_{bc} in $\Omega_{54}^{\circ} \times \Omega_{36}^{\circ}$.

On the one hand, the manifold with faces D_{abc}^- underlying the polychain \mathcal{D}_{abc}^- consists of all tuples

$$(5.2.9) (k, s, l, h, n, i, t) \in K \times I^{\circ} \times L \times I^{\circ} \times N \times I^{\circ} \times (0, 1/2]$$

such that $\tilde{\lambda}(l, h) = \tilde{\eta}(n, i)$ and $\tilde{\kappa}(k, s) = \tilde{\eta}(n, i * t)$. The map

$$(5.2.10) \qquad (ca, a(cb), bc): D^-_{abc} \longrightarrow \Omega^\circ_{52} \times \Omega^\circ_{14} \times Z = \Omega^\circ_{52} \times \Omega^\circ_{14} \times \Omega^\circ_{36}$$

underlying \mathcal{D}^-_{abc} is schematically shown in Figure 5.2.2 next page where one switches direction at the dotted intersections. The first coordinate

$$ca: D^-_{abc} \longrightarrow \Omega^\circ_{52}$$

sends any point (5.2.9) to the path $I \to M$ which goes from \star_5 to $\tilde{\eta}(n, i * t)$ along $\tilde{\eta}(n, -)$ in half-time and, next, goes from $\tilde{\kappa}(k, s)$ to \star_2 along $\tilde{\kappa}(k, -)$ in half-time. The map

$$a(cb): D^-_{abc} \longrightarrow \Omega^\circ_{14}$$

carries a point (5.2.9) to the path $I \to M$ which goes from \star_1 to $\tilde{\kappa}(k, s)$ along $\tilde{\kappa}(k, -)$ in half-time, next, goes from $\tilde{\eta}(n, i * t)$ to $\tilde{\eta}(n, i)$ along $\tilde{\eta}(n, -)$ in time $\left[\frac{1}{2}, 1 - \frac{1}{4(1-t)}\right]$ and, finally, goes from $\tilde{\lambda}(l, h)$ to \star_4 along $\tilde{\lambda}(l, -)$ in time $\left[1 - \frac{1}{4(1-t)}, 1\right]$. The map

$$bc: D^-_{abc} \longrightarrow \Omega^\circ_{36}$$

sends a point (5.2.9) to the path $I \to M$ which goes from \star_3 to $\tilde{\lambda}(l, h) = \tilde{\eta}(n, i)$ along $\tilde{\lambda}(l, -)$ in half-time and, next, goes from $\tilde{\eta}(n, i)$ to \star_6 along $\tilde{\eta}(n, -)$ in half-time.

On the other hand, the manifold with faces D^+_{abc} underlying the polychain \mathcal{D}^+_{abc} consists of all tuples

$$(5.2.11) (k, s, l, h, n, i, t) \in K \times I^{\circ} \times L \times I^{\circ} \times N \times I^{\circ} \times [1/2, 1)$$

such that $\tilde{\lambda}(l, h) = \tilde{\eta}(n, i)$ and $\tilde{\kappa}(k, s) = \tilde{\lambda}(l, h * t)$. The map

(5.2.12)
$$((cb)a, ab, bc) : D^+_{abc} \longrightarrow \Omega^\circ_{52} \times \Omega^\circ_{14} \times \Omega^\circ_{36}$$

is computed similarly to (5.2.10) and is schematically shown in Figure 5.2.2. We only note that the map $(cb)a : D^+_{abc} \to \Omega^\circ_{52}$ carries a point (5.2.11) to the path $I \to M$ which goes from \star_5 to $\tilde{\eta}(n, i)$ along $\tilde{\eta}(n, -)$ in time $[0, \frac{1}{4t}]$, next, goes from $\tilde{\lambda}(l, h)$ to $\tilde{\lambda}(l, h * t)$ along $\tilde{\lambda}(l, -)$ in time $[\frac{1}{4t}, \frac{1}{2}]$ and, finally, goes from $\tilde{\kappa}(k, s)$ to \star_2 along $\tilde{\kappa}(k, -)$ in the remaining half-time.

Consider also the polychain $\mathcal{D}_{abc}^{1/2} = \mathcal{D}^{1/2}(\mathcal{K}, \mathcal{D}_{bc})$ in $\Omega_{52}^{\circ} \times \Omega_{14}^{\circ} \times Z$. Its underlying (p+q+r+3-2n)-manifold with faces

$$D_{abc}^{1/2} = D_{abc}^- \cap D_{abc}^+$$

consists of the tuples (k, s, l, h, n, i, 1/2) such that $\tilde{\kappa}(k, s) = \tilde{\lambda}(l, h) = \tilde{\eta}(n, i)$. The underlying map

$$(ca, ab, cb): D^{1/2}_{abc} \longrightarrow \Omega^{\circ}_{52} \times \Omega^{\circ}_{14} \times \Omega^{\circ}_{36}$$

is the restriction of the maps (5.2.10) and (5.2.12), see Figure 5.2.2.



Figure 5.2.2. The polychains \mathcal{D}_{abc}^- , $\mathcal{D}_{abc}^{1/2}$ and \mathcal{D}_{abc}^+ .

Cyclically permuting *a*, *b*, *c*, we similarly obtain polychains \mathcal{D}_{bca}^{-} , $\mathcal{D}_{bca}^{1/2}$, \mathcal{D}_{bca}^{+} and \mathcal{D}_{cab}^{-} , $\mathcal{D}_{cab}^{1/2}$, \mathcal{D}_{cab}^{+} . Lemma 5.2.5 allows us to rewrite (5.2.8) as the identity

(5.2.13)
$$(-1)^{q+p(n+r)} \left[\mathcal{D}_{abc}^{-} \bigcup_{1/2} \mathcal{D}_{abc}^{+} \right] + (-1)^{r+q(n+p)} (\mathbf{p}_{312})_{*} \left[\mathcal{D}_{bca}^{-} \bigcup_{1/2} \mathcal{D}_{bca}^{+} \right]$$
$$+ (-1)^{p+r(n+q)} (\mathbf{p}_{231})_{*} \left[\mathcal{D}_{cab}^{-} \bigcup_{1/2} \mathcal{D}_{cab}^{+} \right] = 0$$

in $H_*(\Omega_{52} \times \Omega_{14} \times \Omega_{36})$. The idea of the proof is to show that the six polychains on the left-hand side of (5.2.13) cancel each other pairwise.

We first explain how to relate the polychains \mathcal{D}^-_{abc} and \mathcal{D}^+_{bca} . Observe that the manifold with faces D^+_{bca} underlying \mathcal{D}^+_{bca} consists of all tuples

$$(5.2.14) \qquad (l, h, n, i', k, s, t') \in L \times I^{\circ} \times N \times I^{\circ} \times K \times I^{\circ} \times [1/2, 1]$$

such that $\tilde{\eta}(n, i') = \tilde{\kappa}(k, s)$ and $\tilde{\lambda}(l, h) = \tilde{\eta}(n, i' * t')$. We define a smooth map

$$F: K \times I^{\circ} \times L \times I^{\circ} \times N \times I^{\circ} \times (0, 1/2] \longrightarrow L \times I^{\circ} \times N \times I^{\circ} \times K \times I^{\circ} \times [1/2, 1)$$

by the formula

$$F(k, s, l, h, n, i, t) = (l, h, n, i'(i, t), k, s, t'(i, t))$$
where $i': I^{\circ} \times (0, 1/2] \rightarrow I^{\circ}$ and $t': I^{\circ} \times (0, 1/2] \rightarrow [1/2, 1)$ are given by

(5.2.15)
$$i'(i,t) = 2it \text{ and } t'(i,t) = 1 - \frac{1-i}{2-4it}$$

Observe that the functions i', t' satisfy the equations i' = i * t and i = i' * t'. It easily follows that the transformation $(i', t') : I^{\circ} \times (0, 1/2] \rightarrow I^{\circ} \times [1/2, 1)$ is a diffeomorphism, so that F is a diffeomorphism carrying D_{abc}^{-} onto D_{bca}^{+} . The resulting diffeomorphism

$$F_{abc}: D^-_{abc} \longrightarrow D^+_{bca}$$

is compatible with the partitions and the weights of the polychains \mathcal{D}_{abc}^- , \mathcal{D}_{bca}^+ . Moreover,

$$(5.2.16) \qquad (ca, a(cb), bc) = p_{312}((ac)b, bc, ca)F_{abc} : D^-_{abc} \longrightarrow \Omega^\circ_{52} \times \Omega^\circ_{14} \times \Omega^\circ_{36}$$

up to homotopy of the second coordinate map compatible with the partitions. The map F_{abc} carries $D_{abc}^{1/2} \subset D_{abc}^-$ diffeomorphically onto $D_{bca}^{1/2} \subset D_{bca}^+$ via the permutation

 $(k, s, l, h, n, i, 1/2) \longmapsto (l, h, n, i, k, s, 1/2)$

and (5.2.16) holds on $D_{abc}^{1/2}$ as an equality of maps (no homotopy needed). One easily constructs a homotopy of the map

$$a(cb): D^-_{abc} \longrightarrow \Omega^\circ_{14}$$

into $((ac)b) \circ F_{abc}$ constant on $D_{abc}^{1/2}$. Since the left-hand side of (5.2.13) is preserved under such a homotopy of a(cb), we can assume that (5.2.16) is an equality of maps.

We prove now that

(5.2.17)
$$\deg F_{abc} = (-1)^{1+pn+qn+(p+1)(q+r)}.$$

The diffeomorphism F_{abc} carries the open subset $R^- = D^-_{abc} \setminus D^{1/2}_{abc}$ of D^-_{abc} onto the open subset $R^+ = D^+_{bca} \setminus D^{1/2}_{bca}$ of $D^+_{bca'}$ and deg F_{abc} is equal to the degree of the restricted diffeomorphism $R^- \to R^+$. Clearly,

$$\begin{aligned} R^- &= D^-_{abc} \cap X^-_{abc} \quad \text{where} \quad X^-_{abc} = K \times I^\circ \times L \times I^\circ \times N \times I^\circ \times (0, 1/2), \\ R^+ &= D^+_{bca} \cap X^+_{bca} \quad \text{where} \quad X^+_{bca} = L \times I^\circ \times N \times I^\circ \times K \times I^\circ \times (1/2, 1). \end{aligned}$$

Consider the maps

$$\begin{split} X^-_{abc} & \xrightarrow{G^-} M^4, \quad (k, s, l, h, n, i, t) \longmapsto \big(\tilde{\kappa}(k, s), \tilde{\eta}(n, i * t), \tilde{\lambda}(l, h), \tilde{\eta}(n, i)\big), \\ X^+_{bca} & \xrightarrow{G^+} M^4, \quad (l, h, n, i', k, s, t') \longmapsto \big(\tilde{\kappa}(k, s), \tilde{\eta}(n, i'), \tilde{\lambda}(l, h), \tilde{\eta}(n, i' * t')\big). \end{split}$$

Since \mathbb{N} is transversal to both \mathcal{K} and \mathcal{L} , the map G^- is transversal to $\operatorname{diag}_M \times \operatorname{diag}_M$ in the following sense: for any faces A, B, C of K, L, N respectively, the restriction of G^- to the interior of $A \times I^\circ \times B \times I^\circ \times C \times I^\circ \times (0, 1/2)$ is transversal to the interior of $\operatorname{diag}_M \times \operatorname{diag}_M$ (in the usual sense of differential topology). Similarly, the map G^+ is transversal to $\operatorname{diag}_M \times \operatorname{diag}_M$. Observe that $G^- = G^+ F|_{X_{abc}}$ and that the inverse

images of diag_{*M*} × diag_{*M*} under the maps G^- , G^+ are, respectively, the sets R^- , R^+ . We identify

(5.2.18) $\mathcal{V}_{M^4}(\operatorname{diag}_M \times \operatorname{diag}_M) = \operatorname{pr}_{12}^* \mathcal{V}_{M^2}(\operatorname{diag}_M) \oplus \operatorname{pr}_{34}^* \mathcal{V}_{M^2}(\operatorname{diag}_M)$

where $pr_{ii}: M^4 \to M^2$ is the cartesian projection defined by

$$\operatorname{pr}_{ii}(m_1, m_2, m_3, m_4) = (m_i, m_i).$$

As above, $\mathcal{V}_{M^2}(\operatorname{diag}_M)$ carries the orientation induced by that of $\operatorname{diag}_M \approx M$ using our orientation convention, and we give to (5.2.18) the product orientation. Pulling back the latter orientation along G^- , we obtain an orientation on the normal bundle of R^- in X^-_{abc} ; this oriented normal vector bundle is denoted by \mathcal{V}^- . The normal bundle of R^+ in X^+_{bca} is oriented similarly and denoted by \mathcal{V}^+ . Let T^- be the tangent bundle of R^- with the orientation induced by that of \mathcal{V}^- . Similarly, let T^+ be the tangent bundle of R^+ with the orientation induced by that of \mathcal{V}^+ . Clearly, the diffeomorphism

$$(i',t'): I^{\circ} \times (0,1/2] \longrightarrow I^{\circ} \times [1/2,1)$$

defined by (5.2.15) is orientation-reversing. Hence deg $F = (-1)^{1+(p+1)(q+r)}$, and since F carries R^- onto R^+ and induces an orientation-preserving map $\mathcal{V}^- \to \mathcal{V}^+$, we have

(5.2.19)
$$F_{abc}^*(T^+) = (-1)^{1+(p+1)(q+r)}T^-$$

Next, consider the following isomorphisms of oriented vector bundles over R^- , where *T* stands for the tangent bundle, ν stands for the normal bundle, and pr denotes the appropriate cartesian projection:

$$\begin{split} T(K \times I \times L \times I \times N \times I \times I)_{|R^{-}} & \cong \operatorname{pr}^{*} T(K \times I)_{|R^{-}} \oplus \operatorname{pr}^{*} T(L \times I \times N \times I)_{|R^{-}} \oplus \operatorname{pr}^{*} T(I)_{|R^{-}} \\ & \cong \operatorname{pr}^{*} T(K \times I)_{|R^{-}} \oplus \operatorname{pr}^{*} (\mathcal{V}_{L \times I \times N \times I}(D_{bc}) \oplus T(D_{bc}))_{|R^{-}} \oplus \operatorname{pr}^{*} T(I)_{|R} \\ & \cong (-1)^{n(p+1)} \operatorname{pr}^{*} \mathcal{V}_{L \times I \times N \times I}(D_{bc})_{|R^{-}} \oplus \operatorname{pr}^{*} T(K \times I \times D_{bc} \times I)_{|R^{-}} \\ & \cong (-1)^{n(p+1)} \operatorname{pr}^{*} \mathcal{V}_{L \times I \times N \times I}(D_{bc})_{|R^{-}} \oplus \mathcal{V}_{K \times I \times D_{bc} \times I}(R^{-}) \oplus T(R^{-}) \\ & \cong (-1)^{np} \underbrace{\mathcal{V}_{K \times I \times D_{bc} \times I}(R^{-}) \oplus \operatorname{pr}^{*} \mathcal{V}_{L \times I \times N \times I}(D_{bc})_{|R^{-}}}_{\mathcal{V}^{-}} \oplus T(R^{-}). \end{split}$$

It follows that $T^- = (-1)^{np} T(R^-)$. Similarly,

$$T(L \times I \times N \times I \times K \times I \times I)|_{R^+}$$

$$\cong \operatorname{pr}^* T(L \times I)|_{R^+} \oplus \operatorname{pr}^* T(N \times I \times K \times I)|_{R^+} \oplus \operatorname{pr}^* T(I)|_{R^+}$$

$$\cong \operatorname{pr}^* T(L \times I)|_{R^+} \oplus \operatorname{pr}^* (\mathcal{V}_{N \times I \times K \times I}(D_{ca}) \oplus T(D_{ca}))|_{R^+} \oplus \operatorname{pr}^* T(I)|_{R^+}$$

$$\cong (-1)^{n(q+1)} \operatorname{pr}^* \mathcal{V}_{N \times I \times K \times I}(D_{ca})|_{R^+} \oplus \operatorname{pr}^* T(L \times I \times D_{ca} \times I)|_{R^+}$$

$$\cong (-1)^{n(q+1)} \underbrace{\operatorname{pr}^* \mathcal{V}_{N \times I \times K \times I}(D_{ca})_{|R^+} \oplus \mathcal{V}_{L \times I \times D_{ca} \times I}(R^+)}_{(-1)^n \mathcal{V}^+} \oplus T(R^+).$$

Here the sign $(-1)^n$ accompanying \mathcal{V}^+ is the degree of the permutation map

 $M^2 \longrightarrow M^2$, $(m_1, m_2) \longmapsto (m_2, m_1)$.

It follows that $T^+ = (-1)^{nq} T(R^+)$. Formula (5.2.19) and the computations of T^+, T^- imply (5.2.17).

Cyclically permuting *a*, *b*, *c*, we obtain diffeomorphisms

$$F_{bca}: D^{-}_{bca} \longrightarrow D^{+}_{cab}$$
$$F_{cab}: D^{-}_{cab} \longrightarrow D^{+}_{abc}$$

such that

(5.2.20) $\deg F_{bca} = (-1)^{1+qn+rn+(q+1)(r+p)},$

(5.2.21)
$$\deg F_{cab} = (-1)^{1+rn+pn+(r+1)(p+q)}$$

To conclude the proof, we set

$$D^{\pm} = D^{\pm}_{abc} \sqcup D^{\pm}_{bca} \sqcup D^{\pm}_{cab}$$
 and $D^{1/2} = D^{+} \cap D^{-} = D^{1/2}_{abc} \sqcup D^{1/2}_{bca} \sqcup D^{1/2}_{cab}$

Clearly, $F_{cab}F_{bca}F_{abc} = \text{id on } D_{abc}^{1/2}$. Therefore any triangulation of $D_{abc}^{1/2}$ extends uniquely to a triangulation, $T^{1/2}$, of $D^{1/2}$ invariant under $F_{abc} \sqcup F_{bca} \sqcup F_{cab}$. (All triangulations in this argument are supposed to be locally ordered and to fit the given partitions, *cf.* Sections 3.1.2 and 3.3.2.) Subdividing, if necessary, $T^{1/2}$ we can assume that it extends to a triangulation, T^- , of D^- . Transferring T^- along the diffeomorphism

$$F_{abc} \sqcup F_{bca} \sqcup F_{cab} : D^- \longrightarrow D^+$$

we obtain a triangulation, T^+ , of D^+ also extending $T^{1/2}$. We use the triangulations T^- and T^+ to represent the left-hand side of (5.2.13) by a (p+q+r+4-2n)-dimensional singular chain. According to (5.2.17), (5.2.20) and (5.2.21), every singular simplex contributed by a top-dimensional simplex of T^- cancels with the corresponding singular simplex in T^+ . Therefore the singular chain in question is equal to zero and so is the left-hand side of (5.2.13).

5.2.4. *Proof of Theorem* **5.1.1** (*the end*). — Let $\{\{-, -, -\}\} \in \text{End}(A^{\otimes 3})$ be the tribracket induced by the intersection bibracket $\{\{-, -\}\}$ in $A = A(\mathcal{C})$. Pick any points \star_1, \ldots, \star_6 in ∂M and any homology classes $a \in H_p(\Omega_{12}), b \in H_q(\Omega_{34})$ and $c \in H_r(\Omega_{56})$. We need to show that the tensor

$$(5.2.22) \quad \{\!\{a, b, c\}\!\} = \{\!\{a, \{\!\{b, c\}\!\}'\}\!\} \otimes \{\!\{b, c\}\!\}'' \\ + (-1)^{(p+n)(q+r)} P_{312}(\{\!\{b, \{\!\{c, a\}\!\}'\}\!\} \otimes \{\!\{c, a\}\!\}'') \\ + (-1)^{(p+q)(r+n)} P_{231}(\{\!\{c, \{\!\{a, b\}\!\}'\}\!\} \otimes \{\!\{a, b\}\!\}'')$$

vanishes, where $P_{312}, P_{231} \in End(A^{\otimes 3})$ are the graded permutations defined in Section 1.2.1. For any *i*, *j*, *k*, *l*, *u*, *v* \in {1, . . . , 6}, let

$$\begin{aligned} & \varpi_{ij,kl} : H_*(\Omega_{ij}) \otimes H_*(\Omega_{kl}) \longrightarrow H_*(\Omega_{ij} \times \Omega_{kl}), \\ & \varpi_{ij,kl,uv} : H_*(\Omega_{ij}) \otimes H_*(\Omega_{kl}) \otimes H_*(\Omega_{uv}) \longrightarrow H_*(\Omega_{ij} \times \Omega_{kl} \times \Omega_{uv}) \end{aligned}$$

be the linear maps induced by the cross product. By definition of the intersection bibracket and Lemma 5.2.4,

$$\begin{split} \varpi_{52,14,36}\big(\{\!\{a,\{\!\{b,c\}\!\}'\}\} \otimes \{\!\{b,c\}\!\}''\big) &= \varpi_{52,14}\big(\{\!\{a,\{\!\{b,c\}\!\}'\}\} \times \{\!\{b,c\}\!\}'' \\ &= \Upsilon_{12,54}\big(a \otimes \{\!\{b,c\}\!\}') \times \{\!\{b,c\}\!\}'' \\ &= \Upsilon_{12,5436}\big(a \otimes (\{\!\{b,c\}\!\}' \times \{\!\{b,c\}\!\}'')\big) \\ &= \Upsilon_{12,5436}\big(a \otimes \varpi_{54,36}(\{\!\{b,c\}\!\})\big) \\ &= \Upsilon_{12,5436}\big(a \otimes \Upsilon_{34,56}(b \otimes c)\big) \,. \end{split}$$

Cyclically permuting *a*, *b*, *c*, we also obtain

$$\begin{split} \varpi_{52,14,36} \mathcal{P}_{312} \big(\{\!\{b, \{\!\{c,a\}\!\}'\} \otimes \{\!\{c,a\}\!\}'' \big) \\ &= (\mathcal{P}_{312})_* \varpi_{14,36,52} \big(\{\!\{b, \{\!\{c,a\}\!\}'\} \otimes \{\!\{c,a\}\!\}'' \big) \\ &= (\mathcal{P}_{312})_* \Upsilon_{34,1652} \big(b \otimes \Upsilon_{56,12} (c \otimes a) \big), \\ \varpi_{52,14,36} \mathcal{P}_{231} \big(\{\!\{c, \{\!\{a,b\}\!\}'\} \otimes \{\!\{a,b\}\!\}'' \big) \\ &= (\mathcal{P}_{231})_* \varpi_{36,52,14} \big(\{\!\{c, \{\!\{a,b\}\!\}'\} \otimes \{\!\{a,b\}\!\}'' \big) \\ &= (\mathcal{P}_{231})_* \Upsilon_{56,3214} \big(c \otimes \Upsilon_{12,34} (a \otimes b) \big). \end{split}$$

Combining the last three identities, formula (5.2.22) and Lemma 5.2.6, we obtain that $\varpi_{52,14,36}(\{\!\{a, b, c\}\!\}) = 0$. We conclude that $\{\!\{a, b, c\}\!\} = 0$.

5.3. Computations and examples

We compute Υ for spherical homology classes of complementary dimensions and for 0-dimensional homology classes. We use these results to determine the intersection bibracket in two examples.

5.3.1. *Intersection of spheres.* — Assume that $n = \dim(M) \ge 4$. We compute the operation Υ on the loop homology classes arising from spheres of complementary dimensions. Let us fix a base point s_k in the *k*-sphere S^k for every $k \ge 1$. For $x \in \partial M$, we let

$$\pi_k(M, x) = \left[(S^k, s_k), (M, x) \right]$$

be the *k*-th homotopy group of *M* at *x*. For $x, y \in \partial M$, we set

$$\pi_1(M, x, y) = \pi_0(\Omega(M, x, y)).$$

Consider base points \star, \star' in ∂M and integers $p, q \ge 2$ such that $p+q = n = \dim(M)$. Let $\Upsilon^{\pi}_{\star,\star'}$ be the following composition:

$$\begin{array}{cccc} \pi_p(M, \star) \times \pi_q(M, \star') & - & - & - & - & \Upsilon^{\pi}_{\star, \star'} \\ & & & & \\ & & & & \\ \hline & & & & \\ H_{p-1}(\Omega_{\star}) \otimes H_{q-1}(\Omega_{\star'}) \xrightarrow{\Upsilon} H_0(\Omega_{\star'\star} \times \Omega_{\star\star'}) \simeq \mathbb{K}[\pi_1(M, \star', \star)] \otimes \mathbb{K}[\pi_1(M, \star, \star')]. \end{array}$$

Here $\Omega_{\star} = \Omega(M, \star, \star)$, $\Omega_{\star\star'} = \Omega(M, \star, \star')$, $\bar{\partial}_* : \pi_*(M, \star) \to H_{*-1}(\Omega_{\star})$ is the connecting homomorphism of Section 5.1.3, and similar notation applies with \star and \star' exchanged. The following lemma computes $\Upsilon^{\pi}_{\star,\star'}$ when $\star \neq \star'$.

LEMMA 5.3.1. — Assume $\star \neq \star'$. Let

$$\alpha: (S^p, s_p) \longrightarrow (M, \star) \quad and \quad \beta: (S^q, s_q) \longrightarrow (M, \star')$$

be continuous maps such that

$$\alpha^{-1}(\partial M) = \{s_p\}, \quad \beta^{-1}(\partial M) = \{s_q\}$$

and $\alpha_{|S^{P} \setminus \{s_{p}\}}, \beta_{|S^{q} \setminus \{s_{q}\}}$ are transversal smooth maps. Then

(5.3.1)
$$\Upsilon^{\pi}_{\star,\star'}([\alpha],[\beta]) = (-1)^{n(p+1)+1} \sum_{(x,y)} \varepsilon(x,y) [\beta_y \alpha_x^{-1}] \otimes [\alpha_x \beta_y^{-1}].$$

Here: the sum runs over all $(x, y) \in S^p \times S^q$ such that $\alpha(x) = \beta(y)$; $\varepsilon(x, y)$ is the sign of the product orientation in $\alpha_*(T_xS^p) \oplus \beta_*(T_yS^q) = T_{\alpha(x)}M$ with respect to the orientation of M; α_x is the composition of α with a path from s_p to x in S^p and β_y is the composition of β with a path from s_q to y in S^q .

Proof. — For $k \ge 1$, let $h_k : I^k \to S^k$ be a continuous map such that $h_k(\partial I^k) = \{s_k\}$, $h_k|_{\operatorname{Int}(I^k)}$ is smooth and the quotient map $\overline{h}_k : I^k/\partial I^k \to S^k$ is a degree 1 homeomorphism. Then $\alpha h_p : I^p = I^{p-1} \times I \to M$ is adjoint to a continuous map $\omega_\alpha : I^{p-1} \to \Omega_*$ which carries ∂I^{p-1} to the constant path e_* . Let

$$\bar{\omega}_{\alpha}: I^{p-1}/\partial I^{p-1} \longrightarrow \Omega_{\star}$$

be the quotient map. Then

$$\bar{\partial}_p([\alpha]) = (\bar{\omega}_{\alpha})_* ([I^{p-1}/\partial I^{p-1}]) = [\mathcal{K}]$$

where $\mathcal{K} = (I^{p-1}, \theta_{p-1}, 1, \omega_{\alpha})$ is the polycycle in Ω_{\star} with weight 1 and with partition θ_{p-1} defined as the product of p-1 copies of the partition of I identifying {0} to {1}. Similarly, $\bar{\partial}_q([\beta]) = [\mathcal{L}]$ for $\mathcal{L} = (I^{q-1}, \theta_{q-1}, 1, \omega_{\beta})$.

The polycycles \mathcal{K} and \mathcal{L} are admissible in the sense of Section 4.3.2 where $\star_1 = \star_2 = \star, \star_3 = \star_4 = \star', U = \operatorname{Int}(I^{p-1}) \times \operatorname{Int}(I)$ and $V = \operatorname{Int}(I^{q-1}) \times \operatorname{Int}(I)$. We can therefore consider the intersection polychain $\mathcal{D}(\mathcal{K}, \mathcal{L})$ and by Lemma 4.3.3, it represents $\widetilde{\Upsilon}(\langle \mathcal{K} \rangle \otimes \langle \mathcal{L} \rangle)$. Then, using Lemma 4.4.1, we get

$$\Upsilon^{\pi}_{\star,\star'}([\alpha],[\beta]) = \Upsilon([\mathcal{K}],[\mathcal{L}]) = (-1)^{(q-1)+n(p-1)} \big[\widetilde{\Upsilon}(\langle \mathcal{K} \rangle \otimes \langle \mathcal{L} \rangle) \big]$$
$$= (-1)^{q+n(p+1)+1} \big[\mathcal{D}(\mathcal{K},\mathcal{L}) \big].$$

The intersection polycycle $\mathcal{D}(\mathcal{K}, \mathcal{L})$ is o-dimensional, and its points bijectively correspond to the pairs $(x, y) \in S^p \times S^q$ such that $\alpha(x) = \beta(y)$. Such a pair (x, y) contributes

 $\tilde{\varepsilon}(x, y) \left(\beta_{y} \bar{\alpha}_{x}, \alpha_{x} \bar{\beta}_{y} \right) \in \Omega_{\star \star} \times \Omega_{\star \star}$

to $\mathcal{D}(\mathcal{K}, \mathcal{L})$ where α_x, β_y are paths as in the statement of the lemma, $\bar{\alpha}_x$ is the composition of α with a path from x to s_p in S^p , and $\bar{\beta}_y$ is the composition of β with a path from y to s_q in S^q . Here $\tilde{\varepsilon}(x, y)$ is the sign of the linear isomorphism

$$T_{(x,y)}(S^p \times S^q) \xrightarrow{(\alpha \times \beta)_*} T_{(z,z)}(M \times M) \longrightarrow \frac{T_{(z,z)}(M \times M)}{T_{(z,z)} \operatorname{diag}_M} = \nu_{M \times M}(\operatorname{diag}_M)_{(z,z)},$$

where $z = \alpha(x) = \beta(y)$, $T_{(x,y)}(S^p \times S^q) = T_x S^p \oplus T_y S^q$ has the product orientation and $\nu_{M \times M}(\operatorname{diag}_M)$ has the orientation induced from that of $\operatorname{diag}_M \approx M$. The linear map $T_{(z,z)}(M \times M) = T_z M \oplus T_z M \to T_z M$ defined by $(u, v) \mapsto u - v$ induces an orientation-preserving isomorphism $\nu_{M \times M}(\operatorname{diag}_M)_{(z,z)} \to T_z M$. Composing with the linear isomorphism above, we obtain the map

$$\alpha_* \oplus (-\beta_*) : T_x S^p \oplus T_y S^q \longrightarrow T_z M$$

whose degree is $(-1)^q \varepsilon(x, y)$. Therefore $\tilde{\varepsilon}(x, y) = (-1)^q \varepsilon(x, y)$. Thus,

$$\begin{split} \Upsilon^{\pi}_{\star,\star'}([\alpha],[\beta]) &= (-1)^{n(p+1)+1} \sum_{(x,y)} \varepsilon(x,y) \Big[(\beta_y \bar{\alpha}_x, \alpha_x \bar{\beta}_y) \Big] \\ &= (-1)^{n(p+1)+1} \sum_{(x,y)} \varepsilon(x,y) \Big[\beta_y \bar{\alpha}_x \Big] \otimes \Big[\alpha_x \bar{\beta}_y \Big]. \end{split}$$

Since $p \ge 2$, the path α_x is well defined up to homotopy rel ∂I and $\bar{\alpha}_x$ is homotopic to α_x^{-1} . Similar claims hold for β since $q \ge 2$. This yields (5.3.1).

If we consider a single point \star in the boundary of M, then we can similarly compute the linear map

$$\Upsilon^{\pi} = \Upsilon^{\pi}_{\star,\star} : \pi_p(M,\star) \times \pi_q(M,\star) \longrightarrow \mathbb{K} \left[\pi_1(M,\star) \right] \otimes \mathbb{K} \left[\pi_1(M,\star) \right].$$

Fix a path $\varsigma : I \to \partial M$ from \star to a different point $\star' \in \partial M$, and consider maps

$$\alpha: (S^p, s_p) \longrightarrow (M, \star) \text{ and } \beta: (S^q, s_q) \longrightarrow (M, \star')$$

satisfying the conditions of Lemma 5.3.1. Transporting β along ς^{-1} , we obtain a map

$$\varsigma^{-1}\beta:(S^q,s_q)\longrightarrow (M,\star).$$

Applying Lemmas 4.4.2 and 5.3.1, we obtain that

(5.3.2)
$$\Upsilon^{\pi}([\alpha], [\varsigma^{-1}\beta]) = \Upsilon(\bar{\partial}_{p}[\alpha], \varsigma(\bar{\partial}_{q}[\beta])\varsigma^{-1})$$
$$= \varsigma \Upsilon(\bar{\partial}_{p}[\alpha], \bar{\partial}_{q}[\beta])\varsigma^{-1}$$
$$= \varsigma \Upsilon^{\pi}_{\star,\star'}([\alpha], [\beta])\varsigma^{-1}$$
$$= (-1)^{n(p+1)+1} \sum_{(x,y)} \varepsilon(x, y)[\varsigma \beta_{y} \alpha_{x}^{-1}] \otimes [\alpha_{x} \beta_{y}^{-1} \varsigma^{-1}].$$

This computation implies that the map Υ^{π} is determined by the pairing

$$(\operatorname{aug} \otimes \operatorname{id})\Upsilon^{\pi} : \pi_p(M, \star) \times \pi_q(M, \star) \longrightarrow \mathbb{K}[\pi_1(M, \star)],$$

where aug : $\mathbb{K}[\pi_1(M, \star)] \to \mathbb{K}$ is the addition of coefficients. Note that

(5.3.3)
$$(\operatorname{aug} \otimes \operatorname{id})\Upsilon^{\pi}([\alpha], [\varsigma^{-1}\beta]) = (-1)^{n(p+1)+1} \sum_{(x,y)} \varepsilon(x, y) [\alpha_x \beta_y^{-1} \varsigma^{-1}]$$

The pairing on the right-hand side is the well known "geometric intersection" of spherical cycles or the "Reidemeister pairing", see [**30**] or [**43**, Section 5].

5.3.2. *Intersection of arcs with spheres.* — Assume that $n = \dim(M) \ge 3$. We fix three points $\star_1, \star_2, \star_3 \in \partial M$ and consider the map $\Upsilon_{12,3}^{\pi}$ defined by the following composition:

$$\begin{array}{c} \pi_1(M, \star_1, \star_2) \times \pi_{n-1}(M, \star_3) & - & - & \gamma_{12,3} \\ \hline \\ \bar{\partial}_1 \times \bar{\partial}_{n-1} \\ \downarrow \\ H_0(\Omega_{12}) \otimes H_{n-2}(\Omega_{33}) & \xrightarrow{\Upsilon} H_0(\Omega_{32} \times \Omega_{13}) \simeq \mathbb{K} \left[\pi_1(M, \star_3, \star_2) \right] \otimes \mathbb{K} \left[\pi_1(M, \star_1, \star_3) \right].$$

As in the previous sections, $\Omega_{ij} = \Omega(M, \star_i, \star_j)$ for any $i, j \in \{1, 2, 3\}$. Lemma 5.3.1 easily adapts to this setting and yields the following computation of $\Upsilon_{12.3}^{\pi}$.

LEMMA 5.3.2. — Let $\alpha \in \Omega_{12}^{\circ}$ and let

$$\beta: (S^{n-1}, s_{n-1}) \longrightarrow (M, \star_3)$$

be a continuous map such that $\beta^{-1}(\partial M) = \{s_{n-1}\}$. Assume that $\star_1 \neq \star_3, \star_2 \neq \star_3$ and that $\alpha_{\mid (0,1)}, \beta_{\mid S^{n-1} \setminus \{s_{n-1}\}}$ are transversal smooth maps. Then

$$\Upsilon^{\pi}_{12,3}([\alpha],[\beta]) = -\sum_{(x,y)} \varepsilon(x,y) [\beta_y \alpha_{x1}] \otimes [\alpha_{0x} \beta_y^{-1}].$$

Here: the sum runs over all $(x, y) \in [0, 1] \times S^{n-1}$ such that $\alpha(x) = \beta(y)$; $\varepsilon(x, y)$ is the sign of the product orientation in $\alpha_*(T_x[0, 1]) \oplus \beta_*(T_yS^{n-1}) = T_{\alpha(x)}M$ with respect to the orientation of M; α_{0x} (respectively α_{x1}) is the path running along α from \star_1 to $\alpha(x)$ (respectively from $\alpha(x)$ to \star_2) in the positive direction and β_y is the composition of β with a path from s_{n-1} to y in S^{n-1} .

Lemma 5.3.2 can be adapted to the cases where $\star_1 = \star_3$ and/or $\star_2 = \star_3$. Besides, we can similarly define an operation

$$\Upsilon^{\pi}_{1,23}:\pi_{n-2}(M,\star_1)\times\pi_1(M,\star_2,\star_3)\longrightarrow \mathbb{K}\left[\pi_1(M,\star_2,\star_1)\right]\otimes\mathbb{K}\left[\pi_1(M,\star_1,\star_3)\right]$$

and compute it as in Lemma 5.3.2.

5.3.3. A simply connected example. — Fix 2g integers $p_1, q_1, \ldots, p_g, q_g \ge 2$ such that $p_1 + q_1 = \cdots = p_g + q_g = n$. Consider the closed smooth *n*-manifold

(5.3.4)
$$W = (S^{p_1} \times S^{q_1}) \ddagger \cdots \ddagger (S^{p_g} \times S^{q_g})$$

with the product orientation on each summand, and assume that $M = W \setminus \text{Int}(D)$ where *D* is a closed *n*-ball smoothly embedded in *W*. Fix a point $\star \in \partial M$ and consider the Pontryagin algebra $A_{\star} = H_*(\Omega_{\star})$ where $\Omega_{\star} = \Omega(M, \star, \star)$. We now compute the intersection bibracket in A_{\star} .

For an appropriate choice of *D*, of the base points $\{s_k \in S^k\}_k$, and of the balls along which the connected sums are performed in (5.3.4), the sets

$$X_i = S^{p_i} \times \{s_{q_i}\} \subset S^{p_i} \times S^{q_i}$$
 and $Y_i = \{s_{p_i}\} \times S^{q_i} \subset S^{p_i} \times S^{q_i}$

are embedded spheres in M, for all i = 1, ..., g. Since M is simply connected, these spheres define certain elements $x_i^{\pi} \in \pi_{p_i}(M, \star)$ and $y_i^{\pi} \in \pi_{q_i}(M, \star)$, respectively. Consider the corresponding elements of the Pontryagin algebra

$$x_i = \bar{\partial}_{p_i}(x_i^{\pi}) \in A_{\star}^{p_i - 1}, \quad y_i = \bar{\partial}_{q_i}(y_i^{\pi}) \in A_{\star}^{q_i - 1}.$$

Since *M* deformation retracts to a wedge of 2g spheres isotopic to $X_1, Y_1, \ldots, X_g, Y_g$, it follows from [5, III.1.B] (or, alternatively, from [1, Corollary 2.2]) that $x_1, y_1, \ldots, x_g, y_g$ freely generate the unital graded algebra A_{\star} .

In particular, if $\mathbb{K} = \mathbb{Z}$, then A_{\star} is a free abelian group. Therefore condition (5.1.2) is satisfied for any ground ring \mathbb{K} . Hence the intersection bibracket {{-, -}} in A_{\star} is defined for any \mathbb{K} , and is fully determined by its values on the generators. These values can be computed from the formula (5.3.2): for any *i*, *j* = 1, ..., *g*,

(5.3.5)
$$\{\!\{x_i, y_j\}\!\} = \delta_{ij}(-1)^{q_i(p_i+1)+1} 1 \otimes 1, \quad \{\!\{y_j, x_i\}\!\} = \delta_{ij}(-1)^{p_i+1} 1 \otimes 1,$$

(5.3.6)
$$\{\{x_i, x_j\}\} = 0, \quad \{\{y_i, y_j\}\} = 0.$$

Here we use the assumption that the spheres X_i , Y_j have codimension ≥ 2 in M and so can be made disjoint from the interiors of arcs connecting them to \star . As a consequence, we observe that the bibracket $\{\{-, -\}\}$ is a graded version of the bibracket associated by Van den Bergh [42] with the (double of the) quiver Q_g having a single vertex and g edges.

The graded module $\check{A}_{\star} = A_{\star}/[A_{\star}, A_{\star}]$ is freely generated by words in the letters $x_1, y_1, \ldots, x_g, y_g$, subject to the *cyclic relations*

$$w_1 w_2 = (-1)^{|w_1| \cdot |w_2|} w_2 w_1$$

for any words w_1, w_2 where $|w_i|$ is the sum of the degrees of the letters appearing in w_i . The (2 - n)-graded Lie bracket $\langle -, - \rangle$ in \check{A}_{\star} induced by $\{\{-, -\}\}$ is a graded version of the necklace Lie bracket associated to Q_g , see [4], [19].

For any integer $N \ge 1$, the Gerstenhaber bracket $\{-, -\}$ in $(A_{\star})_N^+$ induced by $\{\{-, -\}\}$ can be computed from (5.3.5), (5.3.6). In particular, $(A_{\star})_1^+ = \text{Com}(A_{\star})$ is the unital commutative graded algebra with free generators $x_1, y_1, \ldots, x_g, y_g$ in degrees $|x_i| = p_i - 1$, $|y_i| = q_i - 1$, and for any $i, j = 1, \ldots, g$,

$$\{x_i, y_j\} = (-1)^{q_i(p_i+1)+1} \delta_{ij}, \quad \{y_j, x_i\} = (-1)^{p_i+1} \delta_{ij}, \quad \{x_i, x_j\} = 0, \ \{y_i, y_j\} = 0.$$

The bracket $\{-, -\}$ is a graded version of the standard Poisson bracket in the symmetric algebra of a free module of rank 2g equipped with a symplectic form.

5.3.4. *A non-simply connected example.* — Let $n \ge 3$. We compute the intersection bibracket in the Pontryagin algebra of the exterior of a ball in $W = S^1 \times S^{n-1}$. We endow *W* with the product orientation and set

$$X = S^1 \times \{s_{n-1}\} \subset W$$
 and $Y = \{s_1\} \times S^{n-1} \subset W$

where $s_1 \in S^1$ and $s_{n-1} \in S^{n-1}$ are the base points. As above, assume that $M = W \setminus \text{Int}(D)$ where D is a closed n-ball smoothly embedded in $W \setminus (X \cup Y)$. Pick a point $\star \in \partial M = \partial D$ and connect it to the point $s = (s_1, s_{n-1}) \in \text{Int}(M)$ by a path $\gamma : I \to M$ such that $\gamma^{-1}(X \cup Y) = \{1\}$. Up to homotopy relative to the endpoints, there are two such paths; we take the path γ such that a positive tangent vector of γ followed by a positively oriented basis of $T_{s_{n-1}}S^{n-1}$ yields a positively oriented basis of T_sM , see Figure 5.3.1 below. Transporting X and Y along γ , we obtain certain homotopy classes $x^{\pi} \in \pi_1(M, \star)$ and $y^{\pi} \in \pi_{n-1}(M, \star)$. Consider the corresponding elements

$$x = \overline{\partial}_1(x^{\pi}) \in A^0_{\star}$$
 and $y = \overline{\partial}_{n-1}(y^{\pi}) \in A^{n-2}_{\star}$

of the algebra $A_{\star} = H_*(\Omega_{\star})$. Note that *x* is invertible in $A^0_{\star} \simeq \mathbb{K}[\pi_1(M, \star)]$. We claim that the unital graded algebra A_{\star} is generated by $x^{\pm 1}$ and *y* subject to the only relation $xx^{-1} = 1$. Indeed,

$$A_{\star} = \bigoplus_{i \in \mathbb{Z}} x^i H_*(\Omega^{\mathrm{null}}_{\star})$$

where $\Omega^{\text{null}}_{\star}$ is the connected component of Ω_{\star} consisting of null-homotopic loops. The space $\Omega^{\text{null}}_{\star}$ can be identified with the loop space of the universal cover of M. This cover has the homotopy type of a wedge of countably many copies of S^{n-1} since M deformation retracts to $X \cup Y \cong S^1 \vee S^{n-1}$. Therefore, the unital graded algebra $H_*(\Omega^{\text{null}}_{\star})$ is freely generated by the elements $\{x^iyx^{-i}\}_{i\in\mathbb{Z}}$, and the claim above easily follows.



FIGURE 5.3.1. The manifold $M = (S^1 \times S^{n-1}) \setminus \text{Int}(D)$.

In particular, if $\mathbb{K} = \mathbb{Z}$, then A_{\star} is a free abelian group. Therefore the intersection bibracket $\{\{-,-\}\}$ in A_{\star} is defined for any ground ring \mathbb{K} . To determine $\{\{-,-\}\}$, it

suffices to compute its values on the generators *x*, *y*. For degree reasons,

$$(5.3.7) \qquad \{\!\{x, x\}\!\} = 0$$

Let ς be an arc in ∂M connecting \star to another point \star' . By Lemma 5.3.2, we obtain

$$\{\!\{x, \varsigma^{-1}y\varsigma\}\!\} = -\varsigma^{-1}x \otimes \varsigma, \quad \{\!\{\varsigma^{-1}y\varsigma, x\}\!\} = \varsigma \otimes \varsigma^{-1}x.$$

This implies the equalities

$$(5.3.8) \quad \{\{x, y\}\} = -x \otimes 1, \quad \{\{y, x\}\} = 1 \otimes x.$$

Observe next that $x^{-1}yx$ and $\varsigma^{-1}y\varsigma$ are images under the connecting homomorphism of certain elements of $\pi_{n-1}(M, \star)$ and $\pi_{n-1}(M, \star')$ that can be represented by disjoint embedded spheres. It follows that $\{\{x^{-1}yx, \varsigma^{-1}y\varsigma\}\} = 0$ which implies that $\{\{x^{-1}yx, y\}\} = 0$. Using the Leibniz rules and (5.3.8), we deduce that

$$(5.3.9) {\{y, y\}} = 1 \otimes y - y \otimes 1$$

Using (5.3.7)–(5.3.9), one can also compute the graded Lie bracket $\langle -, - \rangle$ in \check{A}_{\star} and the Gersthenhaber bracket $\{-, -\}$ in $(A^+_{\star})_N$ for any integer $N \ge 1$.

CHAPTER 6

PROPERTIES OF THE INTERSECTION BIBRACKET

In this chapter, *M* is a smooth oriented connected manifold of dimension $n \ge 2$ such that $\partial M \neq \emptyset$ and the condition (5.1.2) is satisfied.

6.1. The scalar intersection form

We derive from the intersection bibracket of M a scalar intersection form and compute it in terms of usual homology intersections. We begin with algebraic preliminaries.

6.1.1. *The scalar form induced by a bibracket.* — Consider an arbitrary graded category C and the associated graded algebra A = A(C), see Section 2.2.1. Given an augmentation $\varepsilon : A \to \mathbb{K}$ and a *d*-graded bibracket {{-,-}} in C with $d \in \mathbb{Z}$, we define the *induced scalar form*

• :
$$A \times A \longrightarrow \mathbb{K}$$
 by $a \bullet b = (\varepsilon \otimes \varepsilon)(\{\{a, b\}\})$

for any $a, b \in A$. Observe that

$$a \bullet (bc) = (a \bullet b)\varepsilon(c) + \varepsilon(b)(a \bullet c)$$

for any $a \in A$, $b \in \text{Hom}_{\mathcal{C}}(X, Y)$, $c \in \text{Hom}_{\mathcal{C}}(Y, Z)$ with $X, Y, Z \in Ob(\mathcal{C})$; similarly,

$$(ab) \bullet c = \varepsilon(a)(b \bullet c) + (a \bullet c)\varepsilon(b)$$

for any $c \in A$, $a \in \text{Hom}_{\mathbb{C}}(X, Y)$, $b \in \text{Hom}_{\mathbb{C}}(Y, Z)$ with $X, Y, Z \in \text{Ob}(\mathbb{C})$. Furthermore, if the bibracket {{-, -}} is *d*-antisymmetric, then

 $a \bullet b = -(-1)^{|a|_d \cdot |b|_d} b \bullet a$ for any homogeneous $a, b \in A$.

6.1.2. The scalar form induced by the intersection bibracket. — The path homology category $\mathcal{C} = \mathcal{C}(M)$ of the manifold M has a canonical augmentation $\varepsilon : A(\mathcal{C}) \to \mathbb{K}$ obtained as the direct sum over all $\star, \star' \in \partial M$ of the compositions

$$H_*(\Omega(M, \star, \star')) \longrightarrow H_0(\Omega(M, \star, \star')) \longrightarrow \mathbb{K},$$

where the left arrow is the obvious projection and the right arrow carries the homology classes of all points to 1. By the previous subsection, this augmentation together with the intersection bibracket induce a bilinear form $\bullet : A(\mathcal{C}) \times A(\mathcal{C}) \rightarrow \mathbb{K}$.

We compute • in terms of standard homological intersections in M. For simplicity, we assume in the rest of this section that $n \ge 3$, though the case n = 2 may be considered similarly. For any points $\star_1, \star_2, \star_3, \star_4 \in \partial M$, we define a linear map

It suffices to define the restriction of • to $H_k \otimes H_l$ for any $k, l \ge 0$. If $k + l \ne n$, then this restriction is equal to zero. Suppose now that k + l = n. When $\{\star_1, \star_2\} \cap \{\star_3, \star_4\} = \emptyset$, the form • is the standard homological intersection, see, for example, **[6]**. When $\{\star_1, \star_2\} \cap \{\star_3, \star_4\} \ne \emptyset$, we separate two cases. If $k \ge 2$, then $H_k(M, \{\star_1, \star_2\})$ is canonically isomorphic to $H_k(M)$ and the pairing • is induced by the homological intersection $H_k(M) \otimes H_l(M, \{\star_3, \star_4\}) \rightarrow \mathbb{K}$. The case $l \ge 2$ is treated similarly using that $H_l(M, \{\star_3, \star_4\})$ is canonically isomorphic to $H_k(M)$. Note that the assumption $k + l = n \ge 3$ guarantees that $k \ge 2$ or $l \ge 2$. If both these inequalities hold true, then the two definitions above give the same pairing.

The next lemma yields a version of the homological suspension homomorphism due to Serre [40, §IV.5].

LEMMA 6.1.1. — Let $\Omega = \Omega(M, \star, \star')$ with $\star, \star' \in \partial M$. There is a unique homomorphism

$$\Sigma: H_*(\Omega) \longrightarrow H_{*+1}(M, \{\star, \star'\})$$

such that for every polycycle $\mathcal{K} = (K, \varphi, u, \kappa)$ in Ω , we have

(6.1.2)
$$\Sigma([\mathcal{K}]) = \left[(K \times I, \varphi \times \tau, u \times 1, \tilde{\kappa}) \right]$$

where τ is the trivial partition on I = [0, 1].

Proof. — The uniqueness of Σ is a direct consequence of Theorem 3.3.4. To prove the existence, define a continuous map

$$\operatorname{ev}: \Omega \longrightarrow M$$
 by $\operatorname{ev}(\alpha) = \alpha(1/2)$

and set $\Omega^{\partial} = ev^{-1}(\{\star, \star'\})$. The formula

$$d(\alpha, s)(t) = \begin{cases} \star & \text{if } t \in [0, 1/2 - s/2], \\ \alpha(s + 2t - 1) & \text{if } t \in [1/2 - s/2, 1 - s/2], \\ \star' & \text{if } t \in [1 - s/2, 1] \end{cases}$$

defines a continuous map $d : (\Omega \times I, \Omega \times \partial I) \to (\Omega, \Omega^{\partial})$. Let

$$\Delta: H_*(\Omega) \longrightarrow H_{*+1}\left(\Omega, \Omega^{\partial}\right)$$

be the linear map sending any $x \in H_*(\Omega)$ to $d_*(x \times [I, \partial I])$ (the definition of Δ is inspired by [9, §5] and [28, Remark 3.2.3]). Finally, we set $\Sigma = ev_* \Delta$. To check (6.1.2), observe that the fundamental class $[I, \partial I] \in H_1(I, \partial I)$ is represented by the 1-dimensional

MÉMOIRES DE LA SMF 154

polycycle $\mathcal{I} = (I, \tau, 1, \text{id} : I \to I)$ relative to ∂I . Lemma 3.3.5 implies that for any polycycle $\mathcal{K} = (K, \varphi, u, \kappa)$ in Ω ,

$$\begin{split} \Sigma([\mathcal{K}]) &= \operatorname{ev}_* d_* \big([\mathcal{K}] \times [\mathcal{I}] \big) \\ &= (\operatorname{ev} d)_* [\mathcal{K} \times \mathcal{I}] = \big[(K \times I, \varphi \times \tau, u \times 1, \tilde{\kappa}) \big]. \end{split}$$

We can now state the main result of this section.

THEOREM 6.1.2. — For any $\star_1, \star_2, \star_3, \star_4 \in \partial M$, the following diagram commutes:

$$H_*(\Omega(M, \star_1, \star_2)) \otimes H_*(\Omega(M, \star_3, \star_4)) \xrightarrow{\bullet} \mathbb{K}$$

$$\xrightarrow{-(-1)^{n|-|}\Sigma \otimes \Sigma} \downarrow$$

$$H_*(M, \{\star_1, \star_2\}) \otimes H_*(M, \{\star_3, \star_4\})$$

The proof of Theorem 6.1.2 proceeds in three steps. First we consider arbitrary disjoint subsets ∂_-M , ∂_+M of ∂M and the standard homology intersection form

$$H_*(M, \partial_-M) \otimes H_*(M, \partial_+M) \longrightarrow H_*(M).$$

We denote this form by \odot and compute it in terms of polycycles. Secondly, we relate \odot to the operation Υ . Finally, we deduce Theorem 6.1.2.

LEMMA 6.1.3. — Let $\partial_- M$, $\partial_+ M$ be disjoint subsets of ∂M . Let $\mathcal{K} = (K, \varphi, u, \kappa)$ be a smooth *p*-polycycle in *M* relative to $\partial_- M$, let $\mathcal{L} = (L, \psi, v, \lambda)$ be a smooth *q*-polycycle in *M* relative to $\partial_+ M$ such that the map $\kappa \times \lambda : K \times L \to M \times M$ is transversal to diag_M in the sense of Section 4.1.1. Let $D = (\kappa \times \lambda)^{-1}(\operatorname{diag}_M)$ and let $\operatorname{pr}_K : K \times L \to K$ be the cartesian projection. Then *D* is a manifold with faces and, for some orientation, partition θ , and weight *w* on *D*, the polychain $\mathcal{D} = (D, \theta, w, \kappa \operatorname{pr}_K \mid_D)$ is a polycycle such that

(6.1.3)
$$[\mathcal{K}] \odot [\mathcal{L}] = (-1)^{q(p+n)} [\mathcal{D}] \in H_{p+q-n}(M).$$

Proof. — The transversality assumption ensures that *D* inherits from $K \times L$ a structure of a manifold with corners, see [35]. The same argument as at the beginning of Section 4.2.1 shows that *D* is a manifold with faces. We orient *D* so that the induced orientation of its normal bundle in $K \times L$ is the pull-back of the orientation of the normal bundle of diag_{*M*} \approx *M* in $M \times M$ via $(\kappa \times \lambda)|_D$. The partition θ of *D* is defined as follows: the faces of *D* are the connected components of the intersections $(F \times G) \cap D$ where *F* and *G* range over faces of *K* and *L* respectively; two such faces $C \subset (F \times G) \cap D$ and $C' \subset (F' \times G') \cap D$ are of the same type if *F*, *F'* are of the same type, *G*, *G'* are of the same type, and $(\varphi_{F,F'} \times \psi_{G,G'})(C) = C'$. Then

$$\theta_{C,C'} = (\varphi_{F,F'} \times \psi_{G,G'})|_C.$$

The weight *w* of *D* carries a connected component *Z* of *D* to u(X)v(Y) where *X*, *Y* are connected components of *K*, *L* respectively, such that *Z* is contained in $X \times Y$. Then $\mathcal{D} = (D, \theta, w, \kappa \operatorname{pr}_{K \mid D})$ is a polycycle satisfying (6.1.3).

We leave the general case of this claim to the reader and prove it only under the following assumptions: *K* and *L* are transversal compact oriented smooth submanifolds of *M* such that $\partial K = \partial M \cap K \subset \partial_{-}M$ and $\partial L = \partial M \cap L \subset \partial_{+}M$; the partitions φ

of *K* and ψ of *L* are trivial; the weights $u : \pi_0(K) \to \mathbb{K}$ and $v : \pi_0(L) \to \mathbb{K}$ send all connected components to $1 \in \mathbb{K}$; the maps $\kappa : K \to M$ and $\lambda : L \to M$ are the inclusions. Under these assumptions, we have

$$[\mathcal{K}] \odot [\mathcal{L}] = [K] \odot [L] = [K \cap L]$$

where $[K] \in H_*(M, \partial_-M)$, $[L] \in H_*(M, \partial_+M)$ and $[K \cap L] \in H_*(M)$ are the fundamental classes, and $K \cap L$ is oriented so that

(6.1.4)
$$\nu_M(K \cap L) = \nu_M(K)|_{K \cap L} \oplus \nu_M(L)|_{K \cap L}$$

(this agrees with the orientation rule in [6, p. 375]). Since $D = (K \times L) \cap \text{diag}_M$ corresponds to $K \cap L \subset M$ under the standard identification $\text{diag}_M \approx M$, we need only to compare the orientation of D with that of $K \cap L$. Note the following orientation-preserving isomorphisms of oriented vector bundles:

$$\begin{split} T(M^2)_{|K \times L} &= \operatorname{pr}_K^* \left(T(M)_{|K} \right) \oplus \operatorname{pr}_L^* \left(T(M)_{|L} \right) \\ &\cong \operatorname{pr}_K^* \nu_M(K) \oplus \operatorname{pr}_K^* T(K) \oplus \operatorname{pr}_L^* \nu_M(L) \oplus \operatorname{pr}_L^* T(L) \\ &\cong (-1)^{p(q+n)} \operatorname{pr}_K^* \nu_M(K) \oplus \operatorname{pr}_L^* \nu_M(L) \oplus \operatorname{pr}_K^* T(K) \oplus \operatorname{pr}_L^* T(L) \\ &\cong (-1)^{p(q+n)} \operatorname{pr}_K^* \nu_M(K) \oplus \operatorname{pr}_L^* \nu_M(L) \oplus T(K \times L) \end{split}$$

where $pr_K : K \times L \to K$ and $pr_L : K \times L \to L$ are the cartesian projections. Restricting to $D \subset K \times L$, we obtain

$$T(M^{2})|_{D} \cong (-1)^{p(q+n)} (\operatorname{pr}_{K}^{*} \nu_{M}(K))|_{D} \oplus (\operatorname{pr}_{L}^{*} \nu_{M}(L))|_{D} \oplus T(K \times L)|_{D}$$
$$\cong (-1)^{p(q+n)} p^{*} (\nu_{M}(K)|_{K \cap L}) \oplus p^{*} (\nu_{M}(L)|_{K \cap L}) \oplus T(K \times L)|_{D}$$
$$\cong (-1)^{p(q+n)} p^{*} \nu_{M}(K \cap L) \oplus \nu_{K \times L}(D) \oplus T(D)$$

where *p* is the identification diffeomorphism $D \rightarrow K \cap L$. On the other hand,

$$\begin{split} T(M^2)|_D &\cong \nu_{M \times M}(\operatorname{diag}_M)|_D \oplus T(\operatorname{diag}_M)|_D \\ &\cong \nu_{M \times M}(\operatorname{diag}_M)|_D \oplus p^*\left(T(M)|_{K \cap L}\right) \\ &\cong \nu_{M \times M}(\operatorname{diag}_M)|_D \oplus p^*\nu_M(K \cap L) \oplus p^*T(K \cap L) \\ &\cong (-1)^{(p+q)n}p^*\nu_M(K \cap L) \oplus \nu_{M \times M}(\operatorname{diag}_M)|_D \oplus p^*T(K \cap L). \end{split}$$

Since $v_{K \times L}(D) = v_{M \times M}(\operatorname{diag}_M)|_D$ as oriented vector bundles, we deduce that

$$T(D) = (-1)^{p(q+n)} \cdot (-1)^{(p+q)n} p^* T(K \cap L) = (-1)^{q(p+n)} p^* T(K \cap L)$$

and (6.1.3) follows.

LEMMA 6.1.4. — Let, under the assumptions of Lemma 6.1.3, $\star_1, \star_2 \in \partial_-M$ and $\star_3, \star_4 \in \partial_+M$. Let ε_1 be the composition of the augmentation $\varepsilon : H_*(\Omega_{32} \times \Omega_{14}) \to \mathbb{K}$ with the linear map $\mathbb{K} \to H_*(M)$ sending $1 \in \mathbb{K}$ to $[\star_1] \in H_0(M)$. Then the following diagram

MÉMOIRES DE LA SMF 154

commutes:

 \sim

Proof. — Let $pr_{32} : \Omega_{32} \times \Omega_{14} \to \Omega_{32}$ be the cartesian projection. Clearly, the map

$$ev: \Omega_{32} \longrightarrow M$$

is homotopic to the constant map $\alpha \mapsto \star_3$ so that, in homology, $(\text{ev} \text{ pr}_{32})_* = \varepsilon_1$. Pick now any $a \in H_p(\Omega_{12})$ and $b \in H_q(\Omega_{34})$ with $p, q \ge 0$. Let $\mathcal{K} = (K, \varphi, u, \kappa)$ be a smooth reduced *p*-polycycle in Ω_{12}° and let $\mathcal{L} = (L, \psi, v, \lambda)$ be a smooth reduced *q*-polycycle in Ω_{34}° transversely representing the pair of face homology classes $(\langle a \rangle, \langle b \rangle)$. Set

$$\mathcal{D}(\mathcal{K},\mathcal{L})=(D,\theta,w,\kappa \triangleleft \lambda).$$

Then

$$\begin{split} \varepsilon_{1} \Upsilon_{12,34}(a \otimes b) &= (-1)^{q+np} (\operatorname{ev} \operatorname{pr}_{32})_{*} \left([\mathcal{D}(\mathcal{K}, \mathcal{L})] \right) \\ &= (-1)^{q+np} \operatorname{ev}_{*} \left[(D, \theta, w, \kappa \triangleleft \lambda) \right] \\ &= (-1)^{q+np} \left[(D, \theta, w, \tilde{\kappa} \circ \operatorname{pr}_{\mid D}) \right] \end{split}$$

where pr : $K \times I \times L \times I \rightarrow K \times I$ is the cartesian projection. We deduce from Lemmas 6.1.1 and 6.1.3 that

$$\begin{aligned} &-(-1)^{np} \Sigma(a) \odot \Sigma(b) \\ &= -(-1)^{np} \left[(K \times I, \varphi \times \tau, u \times 1, \tilde{\kappa}) \right] \odot \left[(L \times I, \psi \times \tau, v \times 1, \tilde{\lambda}) \right] \\ &= (-1)^{1+np+(q+1)(p+1+n)} \left[(D, \theta, w, \tilde{\kappa} \circ \operatorname{pr}_{\mid D}) \right] \\ &= (-1)^{1+q+(q+1)(p+1+n)} \varepsilon_1 \Upsilon_{12,34}(a \otimes b) = (-1)^{(q+1)(p+n)} \varepsilon_1 \Upsilon_{12,34}(a \otimes b). \end{aligned}$$

Since (q + 1)(p + n) is even if p + q = n - 2 and $\varepsilon_1 \Upsilon_{12,34}(a \otimes b) = 0$ otherwise, we obtain the claim of the lemma.

We can now complete the proof of Theorem 6.1.2. Set

$$\partial_{-}M = \{\star_1, \star_2\}$$
 and $\partial_{+}M = \{\star_3, \star_4\}.$

Suppose first that $\partial_- M \cap \partial_+ M = \emptyset$. The desired claim is obtained by combining the diagram in Lemma 6.1.4 with the obvious diagram

$$\begin{array}{c} H_*(\Omega_{32} \times \Omega_{14}) \xrightarrow{\varpi_{32,14}} H_*(\Omega_{32}) \otimes H_*(\Omega_{14}) \\ & \varepsilon_1 \\ & \downarrow \\ H_*(M) \xrightarrow{\varepsilon_1} \\ & & \downarrow \varepsilon \otimes \varepsilon \\ \end{array}$$

where $\varpi_{32,14}$ denotes the inverse of the cross product isomorphism as before, and the bottom horizontal arrow is the standard augmentation. To handle the case $\{\star_1, \star_2\} \cap \{\star_3, \star_4\} \neq \emptyset$, consider a smooth isotopy $\{\phi^t : M \to M\}_{t \in I}$ of $\phi^0 = \operatorname{id}_M$

which is constant outside of a small neighborhood of the points \star_1, \star_2 and such that the point $\star'_i = \phi^1(\star_i)$ lies in $\partial M \setminus \{\star_3, \star_4\}$ for i = 1, 2. The diffeomorphism $\phi^1 : (M, \star_1, \star_2) \to (M, \star'_1, \star'_2)$ induces horizontal isomorphisms in the commutative diagram

$$\begin{array}{c} H_*\big(\Omega(M, \star_1, \star_2)\big)\big) & \longrightarrow \\ & \longrightarrow \\ -(-1)^{n|-|\Sigma} \\ & & \downarrow \\ &$$

Note that the upper horizontal arrow coincides with the isomorphism $(\varsigma_1, \varsigma_2)_{\#}$ defined in Section 4.4.1 where $\varsigma_i : I \to \partial M$ is the path $t \mapsto \phi^t(\star_i)$. Tensoring this diagram by the obvious commutative diagram

$$\begin{array}{c} H_*(\Omega(M, \star_3, \star_4))) & \xrightarrow{\mathrm{id}} & H_*(\Omega(M, \star_3, \star_4)) \\ & \Sigma \\ & & \downarrow \\ & & \downarrow \\ H_*(M, \{\star_3, \star_4\}) & \xrightarrow{\mathrm{id}} & H_*(M, \{\star_3, \star_4\}) \end{array}$$

we obtain a commutative diagram

By the first part of the proof, we have the diagram in Theorem 6.1.2 for the points $\star'_1, \star'_2, \star_3, \star_4$. Combining it with the diagram above we obtain the required diagram. Indeed, according to (4.4.3), the upper line represents the scalar form $\bullet: H_*(\Omega(M, \star_1, \star_2)) \otimes H_*(\Omega(M, \star_3, \star_4)) \to \mathbb{K}$. In the bottom line we obviously get $\bullet: H_*(M, \{\star_1, \star_2\}) \otimes H_*(M, \{\star_3, \star_4\}) \to \mathbb{K}$.

6.2. The reducibility

The path homology category $\mathcal{C} = \mathcal{C}(M)$ has a natural structure of a graded Hopf category, which generalizes the usual Hopf algebra structure on the Pontryagin algebra. The comultiplication Δ in \mathcal{C} is the direct sum over all $\star, \star' \in \partial M$ of the linear maps

$$H_*(\Omega(M, \star, \star')) \longrightarrow H_*(\Omega(M, \star, \star')) \otimes H_*(\Omega(M, \star, \star'))$$

induced by the diagonal maps $\Omega(M, \star, \star') \to \Omega(M, \star, \star') \times \Omega(M, \star, \star')$. (Note that we use here the condition (5.1.2).) The counit ε in \mathcal{C} is the augmentation defined in Section 6.1.1. For $\star, \star' \in \partial M$, the inversion of paths induces a homeomorphism

$$\Omega(M, \star, \star') \longrightarrow \Omega(M, \star', \star)$$

which in its turn induces a graded linear isomorphism

 $H_*(\Omega(M, \star, \star')) \longrightarrow H_*(\Omega(M, \star', \star));$

the direct sum of these isomorphisms over all $\star, \star' \in \partial M$ defines an antipode *s* in \mathbb{C} . It is well-known that the path homology category \mathbb{C} with this data is a cocommutative Hopf category.

LEMMA 6.2.1. — The intersection bibracket in $\mathcal{C} = \mathcal{C}(M)$ is reducible.

Proof. — Let $A = A(\mathcal{C})$ be the graded algebra associated with \mathcal{C} and let $\Lambda = \Lambda(\{\{-, -\}\})$ be the map (2.3.1) associated with the intersection bibracket $\{\{-, -\}\}$ in \mathcal{C} . We must show that $\Lambda(a, b) \in \Delta(A)$ for any $a, b \in A$. Since Λ is bilinear, it suffices to consider the case where $a \in H_p(\Omega_{12})$ and $b \in H_q(\Omega_{34})$ for some $p, q \ge 0$. Here $\Omega_{ij} = \Omega(M, \star_i, \star_j)$, and $\star_1, \star_2, \star_3, \star_4$ are four points in ∂M . Observe that a path in ∂M starting from \star_2 represents a certain element $v \in A^0$ and, by Lemma 2.3.2,

$$\Lambda(av, b) = \Lambda(a, b)\varepsilon(v) + \Delta(a)\Lambda(v, c) = \Lambda(a, b).$$

Similarly, a path in ∂M ending at \star_1 represents a certain $u \in A^0$ and

$$\Lambda(ua,b) = \Lambda(u,b)\varepsilon(a) + \Delta(u)\Lambda(a,b) = \Delta(u)\Lambda(a,b).$$

Thus it suffices to consider the case where $\{\star_1, \star_2\} \cap \{\star_3, \star_4\} = \emptyset$. Pick transversal smooth polycycles

$$\mathcal{K} = (K, \varphi, u, \kappa)$$
 in Ω_{12}° and $\mathcal{L} = (L, \psi, v, \lambda)$ in Ω_{34}°

representing respectively $\langle a \rangle \in \widetilde{H}_p(\Omega_{12})$ and $\langle b \rangle \in \widetilde{H}_q(\Omega_{34})$. We form the intersection polycycle $\mathcal{D} = (D, \theta, w, \kappa \diamond \lambda)$ in $\Omega_{32} \times \Omega_{14}$ as in Section 4.2.1. By definition, $\{\!\{a, b\}\!\}$ in $H_*(\Omega_{32}) \otimes H_*(\Omega_{14})$ corresponds to the homology class

$$(-1)^{q+np}[\mathcal{D}] \in H_{p+q+2-n}(\Omega_{32} \times \Omega_{14})$$

under the isomorphism

$$\varpi_{32,14}: H_*(\Omega_{32}) \otimes H_*(\Omega_{14}) \longrightarrow H_*(\Omega_{32} \times \Omega_{14})$$

induced by the cross product in homology. Consider the tensor

$$T = a^{(1)} \otimes \{\!\!\{a^{(2)}, b^{(1)}\}\!\!\} \otimes b^{(2)} \in H_*(\Omega_{12}) \otimes H_*(\Omega_{32}) \otimes H_*(\Omega_{14}) \otimes H_*(\Omega_{34}).$$

Applying (5.2.6) with \star_1 , \star_3 exchanged and with $Y = \Omega_{12}$, $Z = \Omega_{34}$, we obtain

$$\begin{aligned} \varpi_{12,32,14,34}(T) &= a^{(1)} \times \Upsilon_{12,34}(a^{(2)} \otimes b^{(1)}) \times b^{(2)} \\ &= \Upsilon_{Y12,34Z}(\operatorname{diag}_*(a), \operatorname{diag}_*(b)) \in H_*(\Omega_{12} \times \Omega_{32} \times \Omega_{14} \times \Omega_{34}) \end{aligned}$$

where $\varpi_{12,32,14,34}$ is the isomorphism induced by the cross product in homology and diag_{*} : $H_*(\Omega_{ij}) \rightarrow H_*(\Omega_{ij} \times \Omega_{ij})$ is induced by the diagonal map $M \rightarrow M \times M$. The homology class $\Upsilon_{Y12,34Z}(\text{diag}_*(a), \text{diag}_*(b))$ is represented by the polycycle

$$(-1)^{q+np} \left(D, \theta, w, \kappa' \times (\kappa \diamond \lambda) \times \lambda' : D \longrightarrow \Omega_{12} \times \Omega_{32} \times \Omega_{14} \times \Omega_{34} \right)$$

where $\kappa' : D \to \Omega_{12}$ is obtained by projecting $D \subset K \times I \times L \times I$ onto K and applying κ , whereas $\lambda' : D \to \Omega_{34}$ is obtained by projecting onto L and applying λ . Consider

now the homeomorphisms $\{J_i : \Omega_{3i} \to \Omega_{i3}\}_{i=2,4}$ induced by the inversion of paths, the concatenation maps $\{c_i : \Omega_{1i} \times \Omega_{i3} \to \Omega_{13}\}_{i=2,4}$, and the map

 $\mu = (\mathbf{c}_2 \times \mathbf{c}_4)(\mathrm{id}_{\Omega_{12}} \times J_2 \times \mathrm{id}_{\Omega_{14}} \times J_4)(\kappa' \times (\kappa \rightsquigarrow \lambda) \times \lambda') : D \longrightarrow \Omega_{13} \times \Omega_{13}.$

It follows that the image of the homology class

$$\Lambda(a,b) = a^{(1)}s\big(\{\!\{a^{(2)}, b^{(1)}\}\!\}'\big) \otimes \{\!\{a^{(2)}, b^{(1)}\}\!\}''s(b^{(2)})$$

under the cross product isomorphism $\varpi_{13,13}$: $H_*(\Omega_{13}) \otimes H_*(\Omega_{13}) \rightarrow H_*(\Omega_{13} \times \Omega_{13})$ is represented by the polycycle

$$(-1)^{q+np}(D,\theta,w,\mu)$$

To analyze this polycycle, let $\mu_1, \mu_2 : D \to \Omega_{13}$ be the first and the second coordinates of μ . For any point $(k, s, l, t) \in D$, the path $\mu_1(k, s, l, t)$ is obtained by concatenation of the following three paths:

- (i) the path $\kappa(k)$ from \star_1 to \star_2 ;
- (ii) the initial segment of the path $(\kappa(k))^{-1}$ from \star_2 to the point $\kappa(k)(s) = \lambda(l)(t)$;
- (iii) the terminal segment of the path $(\lambda(l))^{-1}$ from the latter point to \star_3 .

This concatenated path goes along a terminal segment of the path $\kappa(k)$ twice in opposite directions. Therefore the path $\mu_1(k, s, l, t)$ is homotopic to a path $\nu(k, s, l, t)$ obtained by concatenation of just two paths: the initial segment of the path $\kappa(k)$ from \star_1 to the point $\kappa(k)(s) = \lambda(l)(t)$ and the terminal segment of the path $(\lambda(l))^{-1}$ from the latter point to \star_3 . The homotopy in question may be defined by an explicit formula which applies to all points $(k, s, l, t) \in D$. Therefore, it determines a homotopy of the polycycle (D, θ, w, μ_1) into the polycycle (D, θ, w, ν) . A similar argument applies to the path $\mu_2(k, s, l, t)$ and yields a homotopy of the polycycle (D, θ, w, μ_2) into (D, θ, w, ν) . Applying these two homotopies coordinatewise we obtain a homotopy of the polycycles (D, θ, w, μ) and $(D, \theta, w, (\nu, \nu))$. It is obvious that the homology class represented by the latter polycycle belongs to the image of the map

$$\operatorname{diag}_*: H_*(\Omega_{13}) \longrightarrow H_*(\Omega_{13} \times \Omega_{13}).$$

We conclude that $\Lambda(a, b) \in \Delta(A)$.

Lemma 6.2.1 and the (2-n)-antisymmetry of the intersection bibracket of *M* implies that it shares all the properties established in Lemma 2.3.3. Note that the associated pairing λ generalizes the Reidemeister pairing (5.3.3).

П

The results of this section are analogues of the known properties of the intersection bibracket in dimension two, see [36]. In dimension two, the role of λ is played by the homotopy intersection form introduced in [41].

6.3. The string bracket

In this section, we relate the intersection bibracket of *M* to the Chas-Sullivan string bracket in loop homology. By a *loop* in *M* we mean a continuous map

 $S^1 \longrightarrow M$ where $S^1 = \mathbb{R}/\mathbb{Z}$.

MÉMOIRES DE LA SMF 154

Let $\mathbb{L} = \mathbb{L}(M)$ be the space of loops in *M* with compact-open topology.

▶ The loop homology of *M* is $\mathbb{H} = H_*(\mathbb{L})$.

▶ The *string homology*, \mathcal{H} , of *M* is the *S*¹-equivariant homology of \mathbb{L} where *S*¹ acts on \mathbb{L} by $(s\gamma)(t) = \gamma(s+t)$ for any *s*, $t \in S^1$ and $\gamma \in \mathbb{L}$.

Thus, $\mathcal{H} = H_*(\mathcal{E} \times_{S^1} \mathbb{L})$ where \mathcal{E} is the total space of the universal S^1 -principal fiber bundle and $\mathcal{E} \times_{S^1} \mathbb{L}$ is the quotient of $\mathcal{E} \times \mathbb{L}$ by the diagonal action of S^1 . Since \mathcal{E} is contractible, the projection $\mathcal{E} \times \mathbb{L} \to \mathcal{E} \times_{S^1} \mathbb{L}$ induces a linear map $\mathbb{E} : \mathbb{H} \to \mathcal{H}$.

Chas and Sullivan [9] defined a degree 2 - n Lie bracket in \mathcal{H} called the *string bracket*. For n = 2, this is the Goldman bracket discussed in Section 5.1.6. We assume that $n \ge 3$ and relate the string bracket to the Lie bracket $\langle -, - \rangle$ in

$$\check{A}_{\star} = A_{\star} / [A_{\star}, A_{\star}]$$

defined in Section 5.1.4.

LEMMA 6.3.1. — Let $\star \in \partial M$, $\Omega_{\star} = \Omega(M, \star, \star)$, and let $r : \Omega_{\star} \hookrightarrow \mathbb{L}$ be the inclusion map. The induced homology homomorphism

$$r_*: A_{\star} = H_*(\Omega_{\star}) \longrightarrow \mathbb{H} = H_*(\mathbb{L})$$

annihilates $[A_{\star}, A_{\star}]$ and induces a linear map $\mathbb{R} : \check{A}_{\star} \to \mathbb{H}$. The composition

 $(-1)^n \to \mathbb{R} : \check{A}_{\star} \longrightarrow \mathcal{H}$

is a graded Lie algebra homomorphism.

Proof. — Let $c: \Omega_{\star} \times \Omega_{\star} \to \Omega_{\star}$ be the concatenation of loops. For $a \in A^{p}_{\star}, b \in A^{q}_{\star}$

$$ab - (-1)^{pq}ba = c_*(a \times b) - (-1)^{pq}c_*(b \times a) = c_*(a \times b) - c_*p_*(a \times b)$$

where $p:\Omega_\star\times\Omega_\star\to\Omega_\star\times\Omega_\star$ is the transposition. Therefore, to show that

$$r_*(ab - (-1)^{pq}ba) = 0,$$

it suffices to prove that $r_*c_* = r_*c_*p_*$. Clearly,

$$rcp = (\frac{1}{2} \cdot)rc$$

where $(\frac{1}{2} \cdot) : \mathbb{L} \to \mathbb{L}$ stands for the action of $1/2 \in \mathbb{R}/\mathbb{Z} = S^1$. Since $(\frac{1}{2} \cdot)$ is homotopic to the identity, $r_{\rm CP}$ is homotopic to $r_{\rm C}$. We deduce that $r_*([A_\star, A_\star]) = 0$.

Recall the definition of the string bracket [-, -] in \mathcal{H} . Let $M : \mathcal{H} \to \mathbb{H}$ be the degree 1 lift map in the Gysin sequence of the S^1 -bundle $\mathcal{E} \times \mathbb{L} \to \mathcal{E} \times_{S^1} \mathbb{L}$. Chas and Sullivan define a linear map

$$\bullet_{\mathrm{CS}}:\mathbb{H}\otimes\mathbb{H}\longrightarrow\mathbb{H}$$

of degree -n called the *loop product* (and denoted by • in [9]). For a detailed exposition, the reader is referred for instance to Cieliebak [11]. For a homogoneous $x \in \mathcal{H}$ and any $y \in \mathcal{H}$,

(6.3.1)
$$[x, y] = (-1)^{|x|+n} \operatorname{E} (\operatorname{M}(x) \bullet_{\operatorname{CS}} \operatorname{M}(y)).$$

This formula implies that to prove the second claim of the lemma, it is enough to show the commutativity of the diagram



Let $a \in H_p(\Omega_{\star})$ and $b \in H_q(\Omega_{\star})$ with $p, q \ge 0$. Let $h : A_{\star} \to A_{\star}$ be the canonical projection. To compute $\langle h(a), h(b) \rangle$, we pick a path ς in ∂M connecting \star to another point \star' . By definition,

$$\begin{split} \left\langle h(a), h(b) \right\rangle &= (-1)^{q+np} h c_* \left([\widetilde{\Upsilon}(\langle a \rangle \otimes \langle b \rangle)] \right) \\ &= (-1)^{q+np} h c_* \left([((\varsigma^{-1}, e_\star)_{\sharp} \times (e_\star, \varsigma^{-1})_{\sharp}) \widetilde{\Upsilon}(\langle a \rangle \otimes (\varsigma, \varsigma)_{\sharp} \langle b \rangle)] \right). \end{split}$$

Let $\mathcal{K} = (K, \varphi, u, \kappa)$ be a reduced smooth polycycle in $\Omega^{\circ}_{\star} = \Omega^{\circ}(M, \star, \star)$ and let $\mathcal{L} = (L, \psi, v, \lambda)$ be a reduced smooth polycycle in $\Omega^{\circ}_{\star'} = \Omega^{\circ}(M, \star', \star')$ such that $(\mathcal{K}, \mathcal{L})$ transversely represents the pair $(\langle a \rangle, (\varsigma, \varsigma)_{\sharp} \langle b \rangle)$. Consider the intersection polychain

Then

$$\mathcal{D}(\mathcal{K},\mathcal{L}) = (D,\theta,w,\kappa \triangleleft \lambda)$$

where, in the last two lines, c is the concatenation of paths

 $\Omega(M, \star', \star) \times \Omega(M, \star, \star') \longrightarrow \Omega(M, \star', \star') = \Omega_{\star'}$

and $r': \Omega_{\star'} \hookrightarrow \mathbb{L}$ is the inclusion. On the other hand,

$$\operatorname{MER} h(a) = \operatorname{ME} r_*(a) = \operatorname{ME} \left[r_*(a) \right] = \operatorname{ME} \left[(K, \varphi, u, r\kappa) \right].$$

Using the computation of the map $ME : \mathbb{H} \to \mathbb{H}$ in [9], [11] (where this map is denoted by Δ), we obtain

where we use the following notation: $\bar{\varphi}$ is the partition on $K \times S^1$ induced by φ (by identifying $F \times S^1$ to $G \times S^1$ via $\varphi_{F,G} \times \operatorname{id}_{S^1}$ for any faces F, G of the same type in K); \bar{u} is the weight on $K \times S^1$ induced by u via the equality $\pi_0(K \times S^1) = \pi_0(K)$; the map $\bar{\kappa} : K \times S^1 \to \mathbb{L}$ is defined using the action of S^1 on \mathbb{L} by $(k, s) \mapsto s(r\kappa(k))$ for $k \in K$ and $s \in S^1$. The sign $(-1)^p$ in (6.3.4) is caused by a permutation of the two factors of $K \times S^1$ with respect to [**9**], [**11**].

Similarly,

$$\operatorname{MER} h(b) = \operatorname{ME} \left[r_{\ast}(b) \right] = \operatorname{ME} \left[r_{\ast}'(\varsigma, \varsigma)_{\sharp}(b) \right] = (-1)^{q} \left[(L \times S^{1}, \bar{\psi}, \bar{\nu}, \bar{\lambda}) \right]$$

where the map $\overline{\lambda} : L \times S^1 \to \mathbb{L}$ is defined by $(l, s) \mapsto s(r'\lambda(l))$ for $l \in L$ and $s \in S^1$. The loop product \bullet_{CS} can be computed in terms of face homology. This gives

(6.3.5)
$$\operatorname{MER} h(a) \bullet_{\mathrm{CS}} \operatorname{MER} h(b) = (-1)^{p+q} \left[(K \times S^1, \bar{\varphi}, \bar{u}, \bar{\kappa}) \right] \bullet_{\mathrm{CS}} \left[(L \times S^1, \bar{\psi}, \bar{\nu}, \bar{\lambda}) \right] = (-1)^{p+q+np} \left[(\overline{D}, \bar{\theta}, \bar{w}, \bar{\kappa} \infty \bar{\lambda}) \right].$$

Here \overline{D} is the inverse image of diag_{*M*} under the map

$$K \times S^1 \times L \times S^1 \longrightarrow M \times M, \quad (k, s, l, t) \longmapsto \big(r \kappa(k)(s), r' \lambda(l)(t) \big).$$

Note that \overline{D} has a structure of a manifold with faces inherited from $K \times S^1 \times L \times S^1$. The orientation, the partition $\overline{\theta}$ and the weight \overline{w} of \overline{D} are as in the definition of the intersection operation \mathcal{D} in Section 4.2.1. The map

$$\bar{\kappa} \propto \bar{\lambda} : \bar{D} \longrightarrow \mathbb{L}$$

sends a point (k, s, l, t) to the loop that first goes along the loop $\bar{\kappa}(k, s)$ and then along the loop $\bar{\lambda}(l, t)$. Note the sign $(-1)^{np}$ in (6.3.5), which arises from the difference between our orientation conventions and those in [11].

The map $K \times I \times L \times I \to K \times S^1 \times L \times S^1$ determined by the canonical projection $I \to S^1$ induces an orientation-preserving diffeomorphism $D \cong \overline{D}$ which carries the partition θ into $\overline{\theta}$ and the weight *w* into \overline{w} . Using the action of $1/4 \in S^1$ on \mathbb{L} , one easily constructs a homotopy between the maps $r'c(\kappa \Rightarrow \lambda)$ and $\overline{\kappa} \propto \overline{\lambda}$ from $D \cong \overline{D}$ to \mathbb{L} . It follows that

$$\left[(D, \theta, w, r' c(\kappa \rightsquigarrow \lambda)) \right] = \left[(\overline{D}, \overline{\theta}, \overline{w}, \overline{\kappa} \infty \overline{\lambda}) \right] \in \mathbb{H},$$

and we deduce (6.3.2) from (6.3.3) and (6.3.5).

We show that a spherical boundary component of the manifold M determines a moment map for the intersection bibracket. This allows us to define an H_0 -Poisson structure on the Pontryagin algebras of certain manifolds without boundary.

6.4.1. The moment map. — Assume that $n = \dim(M) \ge 3$ and that *S* is a component of ∂M homeomorphic to the sphere S^{n-1} . Fix a point $\star \in S$ and set $A_{\star} = H_{*}(\Omega_{\star})$ where $\Omega_{\star} = \Omega(M, \star, \star)$. The orientation-preserving homeomorphisms $S^{n-1} \cong S$ represent an element $\mu^{\pi} = \mu_{S}^{\pi}$ of $\pi_{n-1}(M, \star)$. Recall the connecting homeomorphism

$$\overline{\partial}_{n-1}: \pi_{n-1}(M, \star) \longrightarrow A^{n-2}_{\star} = H_{n-2}(\Omega_{\star})$$

of Section 5.1.3 and set

$$\mu = \mu_S = \bar{\partial}_{n-1}(\mu^{\pi}) \in A^{n-2}_{\star}.$$

LEMMA 6.4.1. — The element μ is a moment map for the intersection bibracket $\{\{-, -\}\}$ in A_{\star} in the sense of Section 2.4.2.

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

Proof. — Consider the path homology category $\mathcal{C} = \mathcal{C}(M)$ of M and the intersection bibracket $\{\{-, -\}\}$ in the associated graded algebra $A = A(\mathcal{C})$. Pick a smooth closed n-ball $D \subset \text{Int}(M)$ and consider the smooth manifold $P = M \setminus \text{Int}(D)$. As above, we can consider the path homology category of P and the intersection bibracket $\{\{-, -\}\}_P$ in the associated graded algebra. Consider the restriction of $\{\{-, -\}\}_P$ to the algebra

$$B = \bigoplus_{\star_1, \star_2 \in S} H_* \big(\Omega(P, \star_1, \star_2) \big).$$

The inclusion $P \hookrightarrow M$ induces a graded algebra homomorphism $\iota : B \to A$. The definition of the intersection bibracket implies that the following diagram commutes:



FIGURE 6.4.1. The manifold $P = M \setminus \text{Int}(D)$.

We must prove that

$$\{\{\mu, a\}\} = a \otimes 1 - 1 \otimes a \text{ for any } a \in A_{\star} \subset A.$$

To this end, fix a path α in *S* leading from \star to a distinct point $\star' \in S$: see Figure 6.4.1. The path α represents an element in $H_0(\Omega, \star, \star') \subset A^0$ denoted also by α . This element is invertible, and its inverse $\alpha^{-1} \in A^0$ is represented by the inverse path. Set

$$a' = \alpha^{-1} a \alpha \in H_*(\Omega(M, \star', \star')).$$

(In the notation of Section 4.3.4, $a' = (\alpha, \alpha)_{\#}(a)$.) Clearly, $\{\{\mu, a'\}\} = \alpha^{-1}\{\{\mu, a\}\}\alpha$. Hence it suffices to prove that

(6.4.2)
$$\{\{\mu, a'\}\} = a' \alpha^{-1} \otimes \alpha - \alpha^{-1} \otimes \alpha a'.$$

The homology class a' can be represented by a polycycle \mathcal{K} in $\Omega^{\circ}(M, \star', \star')$. Choosing the ball D close enough to ∂M , we can ensure that it does not meet the image of \mathcal{K} . Then $a' = \iota(b)$ for the homology class $b \in H_*(\Omega(P, \star', \star'))$ represented by \mathcal{K} . Similarly, $\mu = \iota(\tau)$ for some $\tau \in H_{n-2}(\Omega(P, \star, \star))$. We deduce from (6.4.1) that

$$\{\!\{\mu, a'\}\!\} = (\iota \otimes \iota)(\{\!\{\tau, b\}\!\}_P).$$

To proceed, we pick an embedded path in *P* leading from a point $\star'' \in \partial D$ to \star' and meeting ∂P only in the endpoints. This path defines an invertible element β in $H_0(\Omega(P, \star'', \star'))$. Set

$$c = \beta b \beta^{-1} \in H_*(\Omega(P, \star'', \star'')).$$

Note that $\{\{\tau, c\}\}_P = 0$ since *c* can be represented by a polycycle whose image does not meet *S*. Therefore

$$\begin{split} \{\!\{\mu, a'\}\!\} &= (\iota \otimes \iota) \big(\{\!\{\tau, b\}\!\}_P\big) = (\iota \otimes \iota) \big(\{\!\{\tau, \beta^{-1} c\beta\}\!\}_P\big) \\ &= (\iota \otimes \iota) \big(\{\!\{\tau, \beta^{-1}\}\!\}_P c\beta + \beta^{-1} c\{\!\{\tau, \beta\}\!\}_P\big) \\ &= (\iota \otimes \iota) \big(-\beta^{-1} \{\!\{\tau, \beta\}\!\}_P \beta^{-1} c\beta + \beta^{-1} c\{\!\{\tau, \beta\}\!\}_P\big). \end{split}$$

By Lemma 5.3.2, we obtain $\{\!\{\beta,\tau\}\!\}_P = -\alpha \otimes \beta \alpha^{-1}$. Therefore $\{\!\{\tau,\beta\}\!\}_P = \beta \alpha^{-1} \otimes \alpha$. We conclude that

$$\{\!\{\mu, a'\}\!\} = (\iota \otimes \iota)(-\alpha^{-1} \otimes \alpha b + b\alpha^{-1} \otimes \alpha),$$

which proves (6.4.2).

We deduce from Lemma 6.4.1 that the following three conditions are equivalent:

(i) $A_{\star} = \mathbb{K};$

(ii) $\mu = 0;$

(iii) the intersection bibracket in A_{\star} is zero.

6.4.2. *Intersections in manifolds without boundary.* — Let *W* be a smooth connected oriented manifold of dimension $n \ge 3$ without boundary. Our construction of a bibracket in the Pontryagin algebra of a manifold requires the base point to lie in the boundary, so that it does not apply to *W*. However, under certain assumptions on *W* we can use the Hamiltonian reduction of Section 2.4 to define an H_0 -Poisson structure on the Pontryagin algebra of *W*. To this end, pick a base point $\star \in W$ and a smooth closed *n*-ball $D \subset W$ with $\star \in \partial D$. Consider the smooth manifold $M = W \setminus \text{Int}(D)$ with $\partial M = \partial D = S^{n-1}$; as everywhere in this chapter, we assume that the condition (5.1.2) is satisfied. Let

$$A_{\star} = H_*(\Omega(M, \star, \star))$$
 and $B_{\star} = H_*(\Omega(W, \star, \star))$

be the Pontryagin algebras of *M* and *W*, respectively. The inclusion $M \hookrightarrow W$ induces a graded algebra homomorphism $p : A_{\star} \to B_{\star}$, Clearly, $p(\mu) = 0$ where $\mu = \mu_{\partial M}$ belongs to A_{\star}^{n-2} . Therefore, Ker $p \supset A_{\star}\mu A_{\star}$.

THEOREM 6.4.2. *Assume that the homomorphism* $p : A_{\star} \rightarrow B_{\star}$ *is onto and* Ker $p = A_{\star}\mu A_{\star}$. Then the intersection bibracket $\{\{-, -\}\}$ in A_{\star} induces an H_0 -Poisson structure of degree 2 - n on B_{\star} . This structure does not depend on the choice of the ball D.

Proof. — The first claim follows from Lemma 2.4.3. The independence of the choice of the ball is a consequence of the naturality of the intersection bibracket under diffeomorphisms, and the fact that for any balls $D_1, D_2 \subset W$ with $\star \in \partial D_1 \cap \partial D_2$ there is a diffeomorphism $f : W \to W$ such that $f(D_1) = D_2$, $f(\star) = \star$, and f is isotopic to id_W in the class of diffeomorphisms $W \to W$ fixing \star . Such an f acts on B_{\star} as the identity, and the result follows.

SOCIÉTÉ MATHÉMATIQUE DE FRANCE 2017

Recall from Theorem 2.4.2 that an H_0 -Poisson structure on B_{\star} induces Gerstenhaber brackets on the trace algebras of B_{\star} . Hence Theorem 6.4.2 allows us to associate Gerstenhaber algebras with W.

Some manifolds do not satisfy the assumptions of Theorem 6.4.2, for example, $W = S^n$ (in this case $A_{\star} = \mathbb{K}$ and $B_{\star} = \mathbb{K}[x]$ where the generator x has degree n - 1, *cf.* [5]). Nonetheless, according to [24] and [16], these assumptions are satisfied if W is a closed simply connected manifold whose cohomology algebra $H^*(W) = H^*(W; \mathbb{K})$ is not generated by a single element and \mathbb{K} is a field whose characteristic is equal to zero or is sufficiently large.

6.4.3. *Example.* — We consider the example of Section 5.3.3 and keep the same notation. Thus

$$W = (S^{p_1} \times S^{q_1}) \sharp \cdots \sharp (S^{p_g} \times S^{q_g})$$
 and $M = W \setminus (\text{an open ball}).$

The element $\mu = \mu_{\partial M} \in A_{\star} = H_*(\Omega(M, \star, \star))$ can be computed as follows. As a topological manifold, *M* is the boundary-connected sum of the manifolds

$$M_j = (S^{p_j} \times S^{q_j}) \setminus (\text{an open ball})$$

where $j = 1, \ldots, g$. Hence

$$\mu^{\pi} = \mu_{\partial M}^{\pi} = \sum_{j=1}^{g} \operatorname{in}_{j} \left(\mu_{\partial M_{j}}^{\pi} \right) \in \pi_{n-1}(M)$$

where in_{*j*} : $\pi_{n-1}(M_j) \rightarrow \pi_{n-1}(M)$ is the inclusion homomorphism (we can ignore the base point because M_j and M are simply-connected). By the definition of the Whitehead bracket $[-, -]_{Wh}$ in $\pi_*(M)$, we have

$$\operatorname{in}_{j}(\mu_{\partial M_{i}}^{\pi}) = [x_{i}^{\pi}, y_{i}^{\pi}]_{\mathrm{Wh}}$$

where $x_i^{\pi} \in \pi_{p_j}(M)$ and $y_i^{\pi} \in \pi_{q_j}(M)$ are represented by the two factors of M_j . Thus,

$$u^{\pi} = [x_1^{\pi}, y_1^{\pi}]_{\mathrm{Wh}} + \dots + [x_g^{\pi}, y_g^{\pi}]_{\mathrm{Wh}} \in \pi_{n-1}(M)$$

Recall that the bracket [-, -] in A_{\star} induced by the Pontryagin multiplication is related to the Whitehead bracket in $\pi_{\star}(M)$ by the formula

$$[x_i, y_i] = \left[\bar{\partial}_{p_i}(x_i^{\pi}), \bar{\partial}_{q_i}(y_i^{\pi})\right] = (-1)^{p_i} \bar{\partial}_{p_i+q_i-1}\left([x_i^{\pi}, y_i^{\pi}]_{\mathrm{Wh}}\right) \in A_{\star}.$$

Therefore

$$\mu = (-1)^{p_1} [x_1, y_1] + \dots + (-1)^{p_g} [x_g, y_g] \in A_{\star}.$$

A direct computation on the generators $x_1, y_1, ..., x_g, y_g$ of A_{\star} using (5.3.5)–(5.3.6) confirms that μ is a moment map of the intersection bibracket {{-, -}} of M, as claimed by Lemma 6.4.1.

Consider in more detail the case g = 1 and set

$$p = p_1, \quad q = q_1, \quad x = x_1 \in A^{p-1}_{\star}, \quad y = y_1 \in A^{q-1}_{\star}.$$

The loop space of $W = S^p \times S^q$ based at \star is the product of the loop spaces of S^p and S^q . By the Künneth theorem, the Pontryagin algebra $B_{\star} = H_*(\Omega(W, \star, \star))$ is (as a graded algebra) the tensor product of the Pontryagin algebras of S^p and S^q . Since the graded algebra A_{\star} is freely generated by x, y, the quotient $A_{\star}/A_{\star}\mu A_{\star}$ is the

MÉMOIRES DE LA SMF 154

commutative graded algebra freely generated by *x*, *y*. It is clear that the assumptions of Theorem 6.4.2 are satisfied here for any ground ring K. Theorem 6.4.2 yields an H_0 -Poisson structure $\langle -, - \rangle$ of degree 2 - n on the (commutative) graded algebra B_{\star} . The bracket $\langle -, - \rangle$ in $\check{B}_{\star} = B_{\star}$ is then a Gerstenhaber bracket of degree 2 - n. It coincides with the Gerstenhaber bracket $\{-, -\}$ in $Com(A_{\star})$ computed in Section 5.3.3.

6.4.4. *Example.* — We consider the example of Section 5.3.4 and keep the same notation. Thus, $W = S^1 \times S^{n-1}$ and $M = W \setminus (an \text{ open ball})$. The element

$$\mu = \mu_{\partial M} \in A_{\star} = H_*(\Omega(M, \star, \star))$$

can be computed as follows. Consider the cylinder $I \times S^{n-1}$ with the product orientation and pick a closed *n*-ball *D* in its interior. Then

$$[\partial D] = \left[\{1\} \times S^{n-1}\right] - \left[\{0\} \times S^{n-1}\right] \in \pi_{n-1}\left((I \times S^{n-1}) \setminus \operatorname{Int}(D)\right).$$

(Here *D* carries the orientation induced by *M* and ∂D carries the orientation inherited from *D*.) It follows that $\mu^{\pi}(M) = (x^{\pi})^{-1} \cdot y^{\pi} - y^{\pi}$ where the dot denotes the action of $\pi_1(M, \star)$ on $\pi_{n-1}(M, \star)$. We deduce that

$$\mu = x^{-1}yx - y \in A_{\star}.$$

A direct computation using (5.3.7)–(5.3.9) confirms that μ is a moment map of the intersection bibracket $\{\{-, -\}\}$ of M, as claimed by Lemma 6.4.1.

By the Künneth theorem, the Pontryagin algebra B_{\star} of W is (as a graded algebra) the tensor product of the Pontryagin algebras of S^1 and S^{n-1} . Thus, B_{\star} is the commutative graded algebra freely generated by $x^{\pm 1} \in B^0_{\star}$ and $y \in B^{n-2}_{\star}$. As a consequence, the assumptions of Theorem 6.4.2 are satisfied (for any ground ring \mathbb{K}) so that the intersection bibracket {{-, -}} of M induces an H_0 -Poisson structure $\langle -, - \rangle$ of degree 2 - n on B_{\star} . Since B_{\star} is commutative, this structure is a Gerstenhaber bracket of degree 2 - n. According to (5.3.7)–(5.3.9), it is given by

$$\langle x, x \rangle = 0, \quad \langle x, y \rangle = -x, \quad \langle y, y \rangle = 0.$$

6.4.5. *Remark.* — The results of this section are high-dimensional analogues of the well-known properties of surfaces. The Pontryagin algebra of a closed connected oriented surface is the group algebra $B = \mathbb{K}[\pi]$ where π is the fundamental group of the surface. Then $\check{B} = B/[B, B] = \mathbb{K}[\check{\pi}]$ is the module freely generated by the set $\check{\pi}$ of conjugacy classes in π . The Goldman Lie bracket in \check{B} is the canonical H_0 -Poisson structure on B, see [**36**, Section 9].

BIBLIOGRAPHY

- J. F. ADAMS & P. J. HILTON "On the chain algebra of a loop space", Comment. Math. Helv. 30 (1956), pp. 305–330.
- [2] A. ALEKSEEV, Y. KOSMANN-SCHWARZBACH & E. MEINRENKEN "Quasi-Poisson manifolds", Canad. J. Math. 54 (2002), pp. 3–29.
- [3] Y. BEREST, X. CHEN, F. ESHMATOV & A. RAMADOSS Noncommutative Poisson structures, derived representation schemes and Calabi-Yau algebras, pp. 219–246 in *Mathematical aspects of quantization*, Contemp. Math., vol. 583, Amer. Math. Soc., Providence, RI, 2012.
- [4] R. BOCKLANDT & L. L. BRUYN "Necklace Lie algebras and noncommutative symplectic geometry", *Math. Z.* **240** (2002), pp. 141–167.
- [5] R. BOTT & H. SAMELSON "On the Pontryagin product in spaces of paths", Comment. Math. Helv. 27 (1953), pp. 320–337.
- [6] G. E. BREDON *Topology and Geometry*, Graduate Texts in Mathematics, vol. 139, Springer-Verlag, New York, 1993.
- [7] E. CASTILLO & R. DÍAZ "Homology and manifolds with corners", Afr. Diaspora J. Math. (N.S.) 8 (2009), pp. 100–113.
- [8] J. CERF "Topologie de certains espaces de plongements", Bull. Soc. Math. France 89 (1961), pp. 227–380.
- [9] M. CHAS & D. SULLIVAN "String topology", Preprint (1999), arXiv:math/ 9911159, http://arxiv.org/abs/math/9911159.
- [10] _____, String topology in dimensions two and three, Algebraic topology, Abel Symp., pp. 33–37, vol. 4, Springer, Berlin, 2009.
- [11] K. CIELIEBAK "Lectures on String Topology", Draft (January 2013) available at www.math.uni-augsburg.de/prof/geo/Dokumente/string.pdf.

BIBLIOGRAPHY

- [12] R. COHEN & J. D. JONES "A homotopy theoretic realization of string topology", *Math. Ann.* 324 (2002), pp. 773–798.
- [13] W. CRAWLEY-BOEVEY "Poisson structures on moduli spaces of representations", J. Algebra 325 (2011), pp. 205–215.
- [14] A. DOUADY "Variétés à bord anguleux et voisinages tubulaires", *Séminaire Henri Cartan* **14** (1961-2, exp. 1), pp. 1–11.
- [15] Y. FÉLIX, S. HALPERIN & J.-C. THOMAS Rational homotopy theory, Graduate Texts in Mathematics, vol. 205, Springer-Verlag, New York, 2001.
- [16] Y. FÉLIX & D. TANRÉ "Sur l'homologie de l'espace des lacets d'une variété compacte", Ann. Sci. École Norm. Sup. (4) 25 (1992), pp. 617–627.
- [17] V. V. FOCK & A. A. ROSLY "Poisson structure on moduli of flat connections on Riemann surfaces and the *r*-matrix (Russian), Moscow Seminar in Math. Physics", English translation: *American Mathematical Society Transl. Ser.* 2, 191 (1999), pp. 67–86.
- [18] D. FUKS & V. ROKHLIN Beginner's course in topology. Geometric chapters, Springer Series in Soviet Math., Springer-Verlag, Berlin, 1984.
- [19] V. GINZBURG "Non-commutative symplectic geometry, quiver varieties, and operads", Math. Res. Lett. 8 (2001), pp. 377–400.
- [20] W. M. GOLDMAN "The symplectic nature of fundamental groups of surfaces", Adv. Math. 54 (1984), pp. 200–225.
- [21] _____, "Invariant functions on Lie groups and Hamiltonian flows of surface group representations", *Invent. Math.* **85** (1986), pp. 263–302.
- [22] V. GUILLEMIN & A. POLLACK Differential topology, Prentice-Hall, Inc., Englewood Cliffs, N.J., 1974.
- [23] K. GURUPRASAD, J. HUEBSCHMANN, L. JEFFREY & A. WEINSTEIN "Group systems, groupoids, and moduli spaces of parabolic bundles", *Duke Math. J.* 89 (1997), pp. 377–412.
- [24] S. HALPERIN & J. LEMAIRE "Suites inertes dans les algèbres de Lie graduées ("Autopsie d'un meurtre. II")", *Math. Scand.* **61** (1987), pp. 39–67.
- [25] K. JÄNICH "On the classification of O(n)-manifolds", Math. Ann. 176 (1968), pp. 53–76.
- [26] D. JOYCE *On manifolds with corners*, Advances in geometric analysis, pp. 225–258 in Adv. Lect. Math., vol. 21, Int. Press, Somerville, MA, 2012.
- [27] C. KASSEL Quantum groups, Graduate Texts in Mathematics, vol. 155, Springer-Verlag, New York, 1995.

134

BIBLIOGRAPHY

- [28] N. KAWAZUMI & Y. KUNO "The logarithms of Dehn twists", Quantum Topol. 5 (2014), pp. 347–423.
- [29] _____, "Intersections of curves on surfaces and their applications to mapping class groups", *Ann. Inst. Fourier (Grenoble)* **65** (2015), pp. 2711–2762.
- [30] M. KERVAIRE "Geometric and algebraic intersection numbers", *Comment. Math. Helv.* **39** (1965), pp. 271–280.
- [31] P. LAMBRECHTS & D. STANLEY "Poincaré duality and commutative differential graded algebras", *Ann. Sci. École Norm. Supér.* (4) **41** (2008), pp. 495–509.
- [32] F. LAUDENBACH "A note on the Chas-Sullivan product", *Enseign. Math.* (2) 57 (2011), pp. 3–21.
- [33] L. LE BRUYN & C. PROCESI "Semisimple representations of quivers", *Trans. Amer. Math. Soc.* **317** (1990), pp. 585–598.
- [34] L. LE BRUYN & G. V. DE WEYER "Formal structures and representation spaces", *J. Algebra* **247** (2002), pp. 616–635.
- [35] J. MARGALEF-ROIG & E. O. DOMÍNGUEZ *Differential topology*, North-Holland Math. Studies, vol. 173, North-Holland Publishing Co., Amsterdam, 1992.
- [36] G. MASSUYEAU & V. TURAEV "Quasi-Poisson structures on representation spaces of surfaces", Int. Math. Res. Not. (2014), pp. 1–64.
- [37] _____, "Brackets in representation algebras of Hopf algebras", *Preprint arXiv*:1508.07566 (2015), to appear in J. Noncommut. Geom.
- [38] J. R. MUNKRES *Elementary differential topology*, Annals of Math. Studies, vol. 54, Princeton University Press, Princeton N.J., 1963.
- [39] C. PROCESI "A formal inverse to the Cayley-Hamilton theorem", J. Algebra 107 (1987), pp. 63–74.
- [40] J.-P. SERRE "Homologie singulière des espaces fibrés. Applications", Ann. of Math. (2) 54 (1951), pp. 425–505.
- [41] V. G. TURAEV "Intersections of loops in two-dimensional manifolds (Russian)", Mat. Sb. 106(148 (1978), pp. 566–588, English translation: Math. USSR, Sb. 35 (1979), pp. 229–250.
- [42] M. VAN DEN BERGH "Double Poisson algebras", Trans. Amer. Math. Soc. 360 (2008), pp. 5711–5769.
- [43] C. T. C. WALL "Surgery on compact manifolds. Second edition", Math. Surveys and Monographs 69 (1999).

INDEX

 (G, \mathfrak{g}) -algebra, 22 H_0 -Poisson structure, 39 d-graded Jacobi identity, 16 *d*-graded Leibniz rules, 17, 18 d-graded Lie algebra, 16 *d*-graded Lie bracket, 16 d-graded permutation, 18 d-transpose, 19 associated pairing, 24 augmentation, 31 bibracket, 8, 18, 30 *d*-antisymmetric, 20 Gerstenhaber, 8, 22 of degree d, 18 reducible, 36 bracket, 16 *d*-antisymmetric, 16 degree of, 16 Gerstenhaber, 17 commutator, 17 concatenation pairing, 81 connecting homomorphism, 95 convolution product, 32, 33 cyclic relations, 114 derivation, 16 diagonal, 69 dimension function, 43 double Gerstenhaber algebra, 22 double Gerstenhaber category, 30 double Poisson bracket, 8 Eilenberg-Zilber chain map, 61 face combinatorial, 53 principal, 43

proper, 44 type of, 44 face chain complex, 48 face homology, 9, 48, 50 relative, 53 smooth, 66, 67, 98 transversely represented, 73 graded algebra, 13 commutative, 13 Gerstenhaber, 17 homomorphism of, 14 Hopf, 32 symmetric, 39 unital, 27 graded category, 29 graded module, 13 element of degree of, 13 homogeneous, 13 graded permutation, 18 homotopy compatible with the partition, 65 Hopf category, 32 antipode of, 31 cocommutative, 32 comultiplication of, 31 counit of, 31 involutive, 32 intersection bibracket, 8, 94 scalar form of, 118 Lie pair, 22 loop homology, 125 loop product, 125 manifold with corners, 43

INDEX

manifold with faces, 43 diffeomorphism of, 44 face of, 43 oriented, 44 partition of, 44 smooth map of, 44 map compatible with a partition, 46 good, 70 smooth, 63, 67, 70 transversal, 69, 70 transversal to the diagonal, 69 weakly transversal, 69 moment map, 40 oriented bordism group, 52 partition compatible with weight, 54 identification map of, 44 trivial, 44 weight of, 46 path homology category, 10, 93 path space, 67 proper, 67 polychain, 9, 46 boundary of, 47 cylinder, 49 deformation of, 50 diffeomorphism of, 46 empty, 46 half-smooth, 101 half-transversal, 101 opposite, 47 reduced, 47 reduced boundary of, 48 reduction of, 9 smooth, 65, 67, 97 polychains

cross product of, 51 diffeomorphic, 46 disjoint union of, 48 homologous, 48 module of, 48 reduced cross product of, 51 transversal, 72, 98 transversely represented, 98 polycycle, 9, 49 admissible, 80 relative, 53 Pontryagin algebra, 8, 95 Pontryagin product, 95 representation algebra, 8, 16, 95 unital, 27 scalar form, 117 shuffle, 60 singular chain, 53 fundamental, 56 singular chain complex, 53 singular homology, 53 cross product in, 61 singular manifold, 47 singular simplex, 53 boundary of, 53 string bracket, 125 string homology, 125 trace, 26 trace algebra, 26 triangulation fitting the partition, 55 local order on, 55 locally ordered, 55 tribracket induced, 20 universal enveloping algebra, 28

MÉMOIRES DE LA SOCIÉTÉ MATHÉMATIQUE DE FRANCE Nouvelle série

2017

- 153. M.P. GUALDANI, S. MISCHLER, C. MOUHOT Factorization of Non-Symmetric Operators and Exponential H-Theorem
- 152. M. MACULAN Diophantine applications of geometric invariant theory
- 151. T. SCHOENEBERG Semisimple Lie Algebras and their classification over p-adic Fields
- 150. P.G. LEFLOCH, YUE MA The Mathematical Validity of the f(R) Theory of Modified Gravity

2016

- 149. BEUZART-PLESSIS R. La conjecture locale de Gross-Prasad pour les représentations tempérées des groupes unitaires
- 148. MOKHTAR-KHARROUBI M. Compactness properties of perturbed sub-stochastic C_0 -semigroups on $L^1(\mu)$ with applications to discreteness and spectral gaps
- 147. CHITOUR Y., KOKKONEN P. Rolling of manifolds and controllability in dimension three
- 146. KARALIOLIOS, N. Global aspects of the reducibility of quasiperiodic cocycles in compact Lie groups
- 145. V. BONNAILLIE-NOËL, M. DAUGE, N. Ground state energy of the magnetic Laplacian on corner domains
- 144. P. AUSCHER, S. STAHLHUT A priori estimates for boundary value elliptic problems via first order systems & A priori estimates for boundary value elliptic problems via first order systems

2015

- 143. R. DANCHIN, P.B. MUCHA Critical functional framework and maximal regularity in action on systems of incompressible flows
- 142. Joseph AYOUB Motifs des variétés analytiques rigides
- 140/141. Yong LU, Benjamin TEXIER A Stability Criterion for High-Frequency Oscillations

2014

- 138/139. Takuro MOCHIZUKI Holonomic D-Modules with Betti Structures
 - 137. Paul SEIDEL Abstract Analogues of Flux as Symplectic Invariants
 - 136. Johannes SJÖSTRAND Weyl law for semi-classical resonances with randomly perturbed potentials

2013

- 135. L. PRELLI Microlocalization of subanalytic sheaves
- 134. P. BERGER Persistence of Stratification of Normally Expanded Laminations
- 133. L. DESIDERI Problème de Plateau, équations fuchsiennes et problème de Riemann Hilbert
- 132. X. BRESSAUD, N. FOURNIER One Dimensional General Forest Fire Processes

2012

- 130-131. Y. NAKKAJIMA Weight filtration and slope filtration on the rigid cohomology of a variety in characteristic p > 0
 - 129. W. A STEINMETZ-ZIKESCH Algèbres de Lie de dimension infinie et théorie de la descente
 - 128. D. DOLGOPYAT Repulsion from resonances

2011

127. B. LE STUM – The overconvergent site

- 125/126. J. BERTIN, M. ROMAGNY Champs de Hurwitz
 - 124. G. HENNIART, B. LEMAIRE Changement de base et induction automorphe pour GL_n en caractéristique non nulle

2010

- 123. C.-H. HSIAO Projections in several complex variables
- 122. H. DE THÉLIN, G. VIGNY Entropy of meromorphic maps and dynamics of birational maps
- 121. M. REES A Fundamental Domain for V₃
- 120. H. CHEN Convergence des polygones de Harder-Narasimhan

2009

- 119. B. DEMANGE Uncertainty principles associated to non-degenerate quadratic forms
- 118. A. SIEGEL, J. M. THUSWALDNER Topological properties of Rauzy fractals
- 117. D. HÄFNER Creation of fermions by rotating charged black holes
- 116. P. BOYER Faisceaux pervers des cycles évanescents des variétés de Drinfeld et groupes de cohomologie du modèle de Deligne-Carayol

2008

- 115. R. ZHAO, K. ZHU Theory of Bergman Spaces in the Unit Ball of \mathbb{C}^n
- 114. M. ENOCK Measured quantum groupoids in action
- 113. J. FASEL Groupes de Chow orientés
- 112. O. BRINON Représentations p-adiques cristallines et de de Rham dans le cas relatif

2007

- 111. A. DJAMENT Foncteurs en grassmanniennes, filtration de Krull et cohomologie des foncteurs
- 110. S. SZABÓ Nahm transform for integrable connections on the Riemann sphere
- 109. F. LESIEUR Measured quantum groupoids
- 108. J. GASQUI, H. GOLDSCHMIDT Infinitesimal isospectral deformations of the Grassmannian of 3-planes in \mathbb{R}^6

- 107. I. GALLAGHER, L. SAINT-RAYMOND Mathematical study of the betaplane model: Equatorial waves and convergence results
- 106. N. BERGERON Propriétés de Lefschetz automorphes pour les groupes unitaires et orthogonaux
- 105. B. HELFFER, F. NIER Quantitative analysis of metastability in reversible diffusion processes via a Witten complex approach: the case with boundary
- 104. A. FEDOTOV, F. KLOPP Weakly resonant tunneling interactions for adiabatic quasi-periodic Schrödinger operators

Instructions aux auteurs

Les Mémoires de la Société Mathématique de France publient, en français ou en anglais, des articles longs de recherche ou des monographies de la plus grande qualité, qui font au moins 80 pages.

Les *Mémoires* sont le supplément du *Bulletin de la Société Mathématique de France* et couvrent l'ensemble des mathématiques. Son comité de rédaction est commun à celui du *Bulletin*. Le manuscrit doit être envoyé au format pdf au comité de rédaction, à l'adresse électronique memoires@smf.emath.fr.

Les articles acceptés doivent être composés en LaTeX avec la classe *smfbook*, disponible sur le site de la SMF *smf.emath.fr*, ou avec toute classe standard.

Instructions to Authors

In the *Mémoires de la Société Mathématique de France* are published, in French or in English, long research articles or monographs of the highest mathematical quality, that are at least 80 pages.

The *Mémoires* are the supplement of the *Bulletin de la Société Mathématique de France*. Articles in all areas of mathematics are considered. The *Mémoires* have the same editorial board as that of the *Bulletin de la SMF*.

The manuscript must be sent in pdf format to the editorial board to the email address *memoires@smf.emath.fr.*

The accepted articles must be composed in LaTeX with the *smfbook* class, available on the SMF website *smf.emath.fr*, or with any standard class.

A fundamental geometric object derived from an arbitrary topological space M with a marked point \star is the space of loops in M based at \star . The Pontryagin algebra A of (M, \star) is the singular homology of this loop space with the graded algebra structure induced by the standard multiplication of loops. When M is a smooth oriented manifold with boundary and \star is chosen on ∂M , we define an "intersection" operation $A \otimes A \to A \otimes A$. We prove that this operation is a double bracket in the sense of Michel Van den Bergh satisfying a version of the Jacobi identity. We show that our double bracket induces Gerstenhaber brackets in the representation algebras of A. These results extend our previous work on surfaces, where A is the group algebra of the fundamental group of a surface and the Gerstenhaber brackets in question are the usual Poisson brackets on the moduli spaces of representations of such a group. The present work is inspired by the results of William Goldman on surfaces and by the ideas of string topology due to Moira Chas and Dennis Sullivan.

Un objet géométrique fondamental qu'on associe à tout espace topologique M avec un point marqué \star est son espace de lacets basés en \star . L'algèbre de Pontryagin A de (M, \star) est l'homologie singulière de cet espace de lacets, avec sa structure d'algèbre graduée induite par la multiplication usuelle des lacets. Lorsque M est une variété orientée lisse à bord et \star est choisi sur ∂M , nous définissons une opération "d'intersection" $A \otimes A \to A \otimes A$. Nous prouvons que cette opération est un crochet double au sens de Michel Van den Bergh satisfaisant une variante de l'identité de Jacobi. Nous montrons que ce crochet double induit des crochets de Grstenhaber sur les algèbres de représentations de A. Ceci étend notre précédent travail sur les surfaces, où A est l'algèbre de groupe du groupe fondamental d'une surface et les crochets de Grstenhaber en question sont les crochets de Poisson habituels sur les espaces de modules de représentations d'un tel groupe. Le présent travail est inspiré des résultats de William Goldman sur les surfaces, et des idées de la topologie des cordes due à Moira Chas et Dennis Sullivan.